# CASE NUMBER: 99-070 Filed 7:30.99

# COMMONWEALTH OF KENTUCKY BEFORE THE PUBLIC SERVICE COMMISSION

JUL S CIEEE

0.

#### **IN THE MATTER OF:**

## RATE APPLICATION OF WESTERN KENTUCKY GAS COMPANY

Case No. 99-070

# PETITION FOR CONFIDENTIALITY OF CERTAIN INFORMATION PROVIDED IN RESPONSE TO THE STAFF'S INITIAL REQUEST FOR INFORMATION

Comes now Western Kentucky Gas Company ("Western"), pursuant to 807 KAR 5:001, Section 7, and all other applicable law, and for its Petition for Confidentiality, states as follows:

In Item No. 5 of the Staff's initial Request for Information, Western was requested to provide the calculations supporting the determination that Western has experienced an \$800,000 decline in lost industrial margins. These calculations are set forth in the attached "Response to PSC DR 1, DR Item 5", which is marked as Exhibit A.

The information contained in Exhibit A reveals volume and discount levels for each special contract industrial customer for whom a discount has been negotiated, disclosure of all of which is necessary in order to provide the calculations requested by the Commission.

Pursuant to KRS 61.878(1)(c) the following documents are eligible for confidential

#### treatment:

4

"Upon and after July 15, 1992, records confidentially disclosed to an agency or required by an agency to be disclosed to it, generally recognized as confidential or proprietary which if openly disclosed would permit an unfair commercial advantage to competitors of the entity that disclosed the records ----".

This is the same standard adopted by the Commission pursuant to 807 KAR 5:0001, Section 7. Company specific details concerning volumes and confidentially negotiated discounts with private enterprises are generally recognized as confidential and proprietary. Disclosure of details pertaining to a particular customer's volume and discount, are likely to cause substantial competitive harm to Western. Knowledge of these facts will provide Western's competitors with a substantial advantage in future business negotiations with Western's customers. Western's competitors would have a clear advantage in competing for these customers since knowledge of a existing Western discounts would enable them to slightly undercut Western's charges. On the other hand, Western's unregulated competitors are not required to make public similar information.

Accordingly, the value of the information is derived by not being readily ascertainable by Western's competitors who would have a clear economic advantage upon disclosure. Negotiations concerning the discounts were maintained with strict confidentiality. None of this information is posted or otherwise generally made available within the company or without. Only those employees of Western who have a legitimate need to know have knowledge of the information contained in Exhibit A.

Additionally, disclosure of this information would put Western at a disadvantage in future negotiations of special contracts with other industrial customers. There would be little room for

room for bargaining when a potential customer knows exactly what discounts Western has negotiated with other industrial customers. This likewise would put Western at an unfair commercial disadvantage.

WHEREFORE, Western respectfully request that the attached be treated a confidential. One copy of the attached response has been submitted with the confidential portions highlighted for review and consideration by the Commission. Redacted copies of these documents have been submitted with Western's filing.

Respectfully submitted this  $\underline{\mathcal{3}}$  day of July, 1999.

Douglas Walther Atmos Energy Corporation P.O. Box 650205 Dallas, TX 75265

SHEFFER - HUTCHINSON - KINNEY Mark R. Hutchinson 115 E. Second St. Owensboro, KY 42303

John N. Hughes 124 West Todd Street Frankfort, KY 40601

Attorneys for Western Kentucky Gas Company

By: Neural Mut

#### VERIFICATION

I, Gary Smith, being duly sworn under oath, state that I am Vice President of Marketing of Western Kentucky Gas Company, and that the foregoing statements are true of my own

knowledge except as to those matters therein stated on information and belief, and as to those matters I believe them to be true.

<u>Inech</u>

## STATE OF KENTUCKY COUNTY OF DAVIESS

SUBSCRIBED AND SWORN to before me by Gary Smith on this the  $27^{H}$  day of July, 1999.

Notary Public -My Commission:

## **CERTIFICATE OF SERVICE**

I hereby certify that on the 30th day of July, 1999, the original of this Petition, together with fifteen (15) copies, was filed with the Kentucky Public Service Commission, 730 Schenkel Lane, Frankfort, Kentucky 40602, and a true copy thereof mailed by first class mail to the following named person:

Hon. David Spenard Assistant Attorney General Office of Rate Intervention 1024 Capitol Center Drive Frankfort, Kentucky 40601

Mark R. Hutchinson

O:\USERS\BKK\WKG\PSC\RATE\PET3.

JOHN N. HUGHES Attorney at Law Professional Service Corporation 124 WEST TODD STREET FRANKFORT, KENTUCKY 40601

Telephone: (502) 227-7270

July 30, 1999

Telecopier: (502) 875-7059

\* CENED

JUL 3 0 1999

FUELIC BERVICE COMMISSION

Ms. Helen Helton Executive Director Kentucky Public Service Commission 730 Schenkel Lane Frankfort, KY 40602

Re: Case No. 99-070

Dear Ms. Helton:

Please file the Response of Western Kentucky Gas Company to the Commission's First Request for Information and Petition for Confidentiality. Item 47f is not completed and is expected to be filed the week of August 9.

Thank you for your assistance, and if there are any questions about this matter or if additional information is needed, please contact me.

ncerely Yours, I. Hughes/

(Attorney for Western Kentucky Gas Company

cc: Attorney General

## COMMONWEALTH OF KENTUCKY

#### BEFORE THE PUBLIC SERVICE COMMISSION

In the Matter of:

THE APPLICATION OF WESTERN KENTUCKY GAS COMPANY FOR AN ADJUSTMENT OF RATES CASE NO. 99-070

# ORDER

IT IS ORDERED that Western Kentucky Gas Company ("Western") shall file with the Commission the original and 15 copies of the following information, with a copy to all parties of record. The information requested herein is due no later than July 30, 1999. Each copy of the data requested should be placed in a bound volume with each item tabbed. When a number of sheets are required for an item, each sheet should be appropriately indexed, for example, Item 1(a), Sheet 2 of 6. Include with each response the name of the witness who will be responsible for responding to questions relating to the information provided. Careful attention should be given to copied material to ensure that it is legible. Where information requested herein has been provided along with the original application, in the format requested herein, reference may be made to the specific location of said information in responding to this information request. When applicable, the information requested herein should be provided for total company operations and jurisdictional operations, separately.

1. Provide an organization chart for Atmos Energy Corporation ("Atmos") as of June 1, 1999. The chart should show all regulated utility divisions and non-regulated businesses.

2. With the exception of Western, provide the following information for each of the regulated utility divisions:

a. The state jurisdictions in which the division operates.

b. The year of the division's last completed general rate case proceeding. If the division is currently involved in a general rate case, indicate when that proceeding is expected to be completed.

c. The authorized rate of return on common equity awarded or authorized in the last completed general rate case proceeding. If the division is currently involved in a general rate case, indicate the rate of return on common equity requested by the division.

3. Western's last general rate case was Case No. 95-010.<sup>1</sup> In the August 10, 1995 Order in that case, the Commission expressed its concern about Western's practices of funding only a portion of its other post-retirement employee benefits ("OPEBs") and of funding internally rather than using a protected fund administered by a third party. Describe Western's current practice with regard to these two issues. In addition, provide justification for Western's current level of OPEB funding.

4. In the October 20, 1995 Order, which adopted the October 9, 1995 Settlement Agreement, the Commission acknowledged Western's agreement to perform a new depreciation study no later than its next general rate application. However, the

<sup>1</sup> Case No. 95-010, An Adjustment of Rates of Western Kentucky Gas Company, Orders dated August 10, 1995 and October 20, 1995.

-2-

October 20, 1995 Order reminded Western that the Commission's concerns about its depreciation study, expressed in the August 10, 1995 Order, would remain pertinent for the review of the new depreciation study.

a. Explain in detail how the concerns expressed in the August 10, 1995 and October 20, 1995 Orders have been addressed by the September 30, 1997 depreciation study filed with Western's application.

b. Refer to the September 30, 1997 depreciation study filed with the application. Provide the calculations for the net salvage rates with details and supporting documentation which support the change, to a negative 150 percent, in the salvage rate for Account 380, Services.

5. Refer to the Application, Volume 1 of 10, Tab 1, Item 5. Provide the calculations that support Western's determination that it has experienced an \$800,000 decline in lost industrial margins and \$1,600,000 in lost margins due to energy conservation and declining customer usage. Include a detailed explanation as to how Western identified these specific causes.

6. Refer to the Application, Volume 2 of 10, Tab 1, the Testimony of Conrad E. Gruber. On pages 12 through 18 of the testimony is a discussion of Western's efficiency and productivity improvements. Explain in detail how the benefits of these efficiency and productivity improvements have been incorporated into Western's forecasted period. Include a dollar quantification of the improvements and identify where this quantification has been shown in the forecasted period.

7. Refer to the Application, Volume 2 of 10, Tab 4, the Testimony of Betty L. Adams. On page 5 of the testimony it is stated that Western's O&M budget was

-3-

converted into National Association of Regulatory Utility Commissioners ("NARUC") accounts. Provide a detailed explanation as to why Western utilized the NARUC accounts rather than following the Uniform System of Accounts prescribed by the Federal Energy Regulatory Commission.

8. On page 10 of the Adams Testimony is a discussion of changes in Western's operational budget due to the implementation of the Shared Services approach. For each change listed on page 10, compare in dollar amounts the changes in Western's O&M budget with the changes in expenses from the Shared Services program. If the increase in Shared Services expense exceeded the reduction in Western's O&M budget for a particular change, explain in detail why it was cost effective to make the change.

9. Refer to the Application, Volume 2 of 10, Tab 5, the Testimony of David H. Doggette, concerning Western's capital budgets.

a. Is it correct that none of the capital budgets submitted in Western's application reflect Western's normal "bottom up" development approach?

b. Has Western prepared its capital budgets for FY2001 through FY2003, following the "bottom up" approach, utilizing the new Oracle accounting system? If yes, provide the FY 2001 through FY 2003 capital budgets, using the "bottom up" approach, in a format similar to that shown in Exhibit DHD-1.

c. Has Western performed any analysis or review to determine if using a "FY 1999 baseline" approach for its capital budgeting produces a similar result as the "bottom up" approach? If yes, provide copies of that analysis or review. If no, explain why such an analysis or review has not been performed.

-4-

d. Have the FY 2000 and FY 2001 capital budgets filed in this proceeding been entered into the Atmos Capital Budget Gathering System? If no, explain in detail why this has not occurred.

e. Provide copies of the approved FY 2000 and FY 2001 capital budgets within 10 days of their approval by Atmos's Board of Directors. Explain in detail all differences between the approved FY 2000 and FY 2001 capital budgets and those submitted with the application.

f. Provide the capital budgets for the Information Technology strategy and other Shared Services projects, for the FY 2000 through FY 2003 periods. This information should show the amounts for the total budget, as well as Western's forecasted investment, for each fiscal year. Explain in detail how Western's forecasted investment was determined.

10. Refer to the Application, Volume 2 of 10, Tab 5, Exhibit DHD-1, page 2 of 6. Provide the workpapers and assumptions used to determine that the projected overhead rate should be 50 percent and the projected increase in maintenance and improvements should be 36.25 percent for the FY 2000 capital budget. Include an explanation as to how these determinations were made.

11. Refer to the Application, Volume 2 of 10, Tab 6, the Testimony of Donald P. Burman. On pages 6 and 7 of the testimony is a discussion of Western's accounting for pension expense.

a. Provide the balances of Western's pension assets and obligations shown on its balance sheet as of March 31, 1999, as well as for the base period and

-5-

forecasted period. Include the account numbers and titles used to record these assets and obligations.

b. Has Western included its pension assets and obligations in its calculation of its rate base? If so, provide the reference to the appropriate schedules and workpapers that disclose this inclusion.

12. Provide the actual capital ratios for Atmos as of June 30, 1999.

13. Refer to the Application, Volume 2 of 10, Tab 7, the Testimony of John P. Reddy. On page 4 of the testimony, it is stated that Atmos's objective is to maintain a capital structure comprised of approximately 50 percent equity and 50 percent debt. However, on page 6 of the Reddy Testimony, the summary of Atmos's five-year financial plan for FY 2001 through FY 2003 shows a capital structure comprised of equity ranging between 53 percent and 61.9 percent.

a. Explain why Atmos's five-year financial plan presents capital structures that are significantly different than Atmos's expressed capital structure objectives.

b. The equity portion of Atmos's capital structure as of March 31, 1999 was 44.19 percent. The equity portion shown on page 6 of the Reddy Testimony shows a forecasted equity component of 49.8 percent for FY 2000. Explain in detail why it is reasonable to expect such an improvement in the equity in less than a two-year time frame.

14. Refer to pages 6 and 7 of the Reddy Testimony. For each of the assumptions listed below, indicate how reasonable that assumption is for Atmos. Include any analyses, studies, or other documentation that support these assumptions.

-6-

a. A return to normal long-term weather patterns for the other Atmos utility divisions beginning in FY 2000.

b. The issuance of \$26 million of new equity in November 1999.

c. Raising \$20 million of new equity annually through stock plans.

d. No significant acquisitions. Explain how this assumption is not in conflict with the statements contained on page 5 of the 1998 Atmos Annual Report to shareholders.

e. Cash flow from depreciation will fund ongoing capital spending requirements.

f. No new long-term debt issues.

15. Refer to the Reddy Testimony, the attachment titled "FR10(9)(h)11," Sheets 1 and 2 of 3. Explain why each of the assumptions listed on these schedules is reasonable. Include any analyses, studies, or other documentation that support these assumptions.

16. Refer to the Application, Volume 2 of 10, Tab 12, the Testimony of Michael Marks. On page 4 of the testimony it is stated "Central to the program are two important provisions of the agreement between Western and the Commission." In the August 10, 1995 Order in Case No. 95-010, the Commission stated:

Thus, the Settlement merely establishes a framework for developing a program which will qualify for rate recovery under KRS 278.285. No specific programs or related cost recovery mechanism have been included. Therefore, the Commission makes no decision or findings of fact related to any portion of the DSM provisions included in the Settlement.<sup>2</sup>

<sup>&</sup>lt;sup>2</sup> Case No. 95-010, August 10, 1995 Order, at 4.

In light of the Commission's statement in Case No. 95-010, explain in detail what agreement is being referenced in Mr. Marks Testimony on page 4.

17. Refer to the Marks Testimony, page 4. Concerning the Western Demand Side Management ("DSM") Collaborative:

a. Identify the members of the Western DSM Collaborative at its inception and as of June 30, 1999.

b. Provide copies of the bylaws or other documents that govern the operation of the Western DSM Collaborative.

c. Identify the chairperson of the Western DSM Collaborative as of June 30, 1999.

18. Refer to the Marks Testimony, page 8. For each of the measures ultimately selected for inclusion in the WKG CARES program, provide the results of the following DSM benefit cost test:

a. Total Resource Cost ("TRC").

- b. Utility Cost Test ("UCT").
- c. Ratepayer Impact Measure ("RIM").
- d. Participant Test ("PT").

19. Refer to the Marks Testimony, page 8. It is stated that the TRC test measures the costs and benefits of a conservation measure from the broadest perspective as it represents the net benefit to society. Isn't it correct that the TRC calculates the net impact on a utility and its customer base as a whole, instead of as a net benefit to society? If no, provide documentation to support the contention that the focus of the TRC is the net benefit to society.

-8-

20. Concerning the process and impact evaluations of WKG CARES:

a. Provide copies of the written process and impact evaluation reports. If no written report was provided, explain in detail why not.

b. Explain why the Applied Energy Group, Inc. ("AEG") was retained to perform these evaluations, given Mr. Marks association with AEG.

c. Explain in detail why actual customer savings were not developed as part of the impact evaluation.

d. Explain how the impacts of the weather experienced during the evaluation period were taken into consideration when pre- and post-treatment consumption values were determined.

e. Provide the workpapers showing the calculation of the TRC, UCT, RIM, and PT values for the impact evaluation.

f. Explain why the Societal Test is referenced on page 13 of the Marks Testimony instead of the TRC.

g. Provide the calculations and assumptions used to determine the environmental externality adder reflected in the Societal Test. Include an explanation as to why it was believed an environmental externality adder should have been included in the analysis.

h. Did the calculation of the TRC during the impact evaluation include a determination of Western's lost revenues associated with WKG CARES? If no, explain why the exclusion of lost revenues would not skew the results of the TRC.

21. On page 13 of the Marks Testimony is the statement that Western is not attempting to recover the revenue erosion caused by WKG CARES. However,

-9-

throughout its application, Western has cited one of the reasons for the proposed revenue increase has been the impact of residential energy conservation efforts. Is it correct that Western is attempting to recover these lost revenues through its general rate increase, rather than through the DSM tariff? Explain the response.

22. On page 14 of the Marks Testimony is a discussion of the benefit to Western's uncollectible accounts as a result of WKG CARES. If the actual impact on uncollectible accounts was not determined as part of the impact evaluation, explain in detail the basis for the assumption that uncollectible accounts were impacted favorably by WKG CARES.

23. Provide copies of all program materials associated with WKG CARES, including the program parameters established regarding allowable program measures.

24. Provide the following information concerning WKG CARES:

a. Describe the specific changes in customers' consumption patterns that Western is attempting to influence through WKG CARES.

b. Indicate whether WKG CARES is consistent with Western's most recent long-range integrated resource plan ("IRP"). Include copies of Western's most recent IRP.

c. Does WKG CARES result in any unreasonable prejudice or disadvantage to any class of customers? Explain the response.

d. Describe the extent to which WKG CARES programs are available, affordable, and useful to all customers.

25. Given the provisions of KRS 278.285, explain in detail why Western waited until the filing of this rate proceeding to recover the costs for WKG CARES, which began in 1996.

26. On page 18 of the Marks Testimony is the statement that Western is seeking to recover the costs associated with the 3-year pilot program which was approved by the Commission in the October 20, 1995 Order in Case No. 95-010. The October 20, 1995 Order approving the October 9, 1995 Settlement in Case No. 95-010 specifically stated that:

In all other respects this proposal mirrors the July 18, 1995 Settlement. Those provisions not addressed herein which were previously addressed and accepted in the Commission's Order of August 10, 1995 are approved without discussion.<sup>3</sup>

The pilot DSM program is not discussed in the October 20, 1995 Order, and the Commission stated in the August 10, 1995 Order that it made no decision or findings of fact related to the DSM provisions included in the Settlement. Explain in detail how Western has concluded that the Commission approved the 3-year pilot program.

27. Provide the following information concerning the costs associated with WKG CARES:

A schedule of the costs incurred during the 3-year pilot. These amounts should be identified using Western's account numbers, with account titles.
 The amounts should be presented in total and fiscal year amounts.

<sup>&</sup>lt;sup>3</sup> Case No. 95-010, October 20, 1995 Order, at 3.

b. Were the pilot costs expensed or deferred by Western? If deferred, provide the Western detailed account numbers utilized and the basis for Western assuming deferral was appropriate.

c. A schedule of the costs proposed for the 1999 – 2002 period. These amounts should be identified using Western's account numbers, with account titles. The amounts should be presented in total and fiscal year amounts.

28. Provide an analysis that examines the completion percentage for capital budget projects. The analysis should be by individual capital project, by year, for fiscal years 1994 through 1998. The analysis should show the completion percentage (actual expenditure to budget amount) for each project in each fiscal year.

29. Refer to the Application, Volume 3 of 10, Tab 1, titled "FR 10(9)(b)," page 3 of 6, the forecasted test year capital budget.

a. Does Western assume that all the capital projects included on this schedule will be completed and included in rate base by the end of the forecasted test year?

b. If yes to part (a), explain in detail the basis for this assumption and why it is reasonable.

c. If no to part (a), provide a schedule showing the amounts from the capital budget that Western included in the rate base by the end of the forecasted test year.

30. Refer to the Application, Volume 3 of 10, Tab 7, titled "FR 10(9)(h)2." Provide all the assumptions used by Western to determine the amounts shown for the

-12-

base year, the forecasted year, and the 2000 through 2003 fiscal years. Include an explanation as to why each assumption is reasonable.

31. Refer to the Application, Volume 3 of 10, Tab 7, titled "FR 10(9)(h)4." Provide a revision of this schedule that fixes the forecasted required rate of return at 9.97 percent for each year shown.

32. Concerning Western Kentucky Gas Resources, Inc. ("WKG Resources"):

a. Describe the nature of WKG Resources' business operations.

b. Indicate when WKG Resources was established.

c. Indicate whether any of Western's assets, liabilities, capital, or personnel were transferred to WKG Resources.

33. Refer to the Application, Volume 3 of 10. Explain why the operating income amounts shown for the test year (January 2000 – December 2000) on the following schedules are not in agreement:

a. Tab 3, titled "FR 10(9)(d)."

b. Tab 7, titled "FR 10(9)(h)1 and FR 10(10)(i)1."

c. Tab 7, titled "FR 10(9)(h)4."

34. Refer to the Application, Volume 9 of 10, Tab 2, titled "FR 10(9)(u)," Schedules 1 and 2. Provide the following information for Schedule 1 and Schedule 2:

a. An explanation of how the amounts shown under the column titled "WKG \$" were determined. Include any supporting workpapers, assumptions, and other documentation.

b. The determination of the percentages shown under the column titled "WKG %." Include any supporting workpapers, assumptions, and other documentation.

c. For each Shared Service Unit listed on the schedule, an explanation of why the particular allocation methodology is reasonable for the allocation of the particular unit costs.

35. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-2.2.

a. For both the base and forecasted periods, provide a schedule that links the plant additions shown on Schedule B-2.2 with the capital budget projects shown in Volume 3 of 10, Tab 1, Exhibit DHD-1.

b. For the base period, provide the workpapers, analysis, assumptions, and other documentation that support the amounts shown for "Retirements" and "Transfers/Reclassifications." Include a description for each retirement or transfer/reclassification shown on the schedule.

c. For the forecasted period, explain why Western assumed there would be no retirements. Include in the explanation a discussion of why such an assumption is reasonable.

36. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-2.3, for both the base and forecasted periods.

a. Identify the business unit of Atmos referred to in the schedule as "Division 02 General Office." b. Indicate whether the assignment of the Division 02 General Office plant is the result of a general allocation of plant by Atmos or whether Western contracted for these assets.

37. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-3, for both the base and forecasted periods. For each of the plant accounts listed below, provide a detailed explanation as to why the accumulated depreciation exceeds the plant investment. Also explain why accumulated depreciation in excess of plant investment should be included in the calculation of Western's rate base.

a. Account No. 331.00 – Production Gas Wells Equipment.

- b. Account No. 332.10 Field Lines.
- c. Account No. 332.20 Tributary Lines.
- d. Account No. 334.00 Field Meas. & Reg. Station Equipment.
- e. Account No. 351.20 Compression Station Equipment.
- f. Account No. 366.30 Other Structures (Transmission Plant).
- g. Account No. 375.10 Meas. & Reg. Station Equipment General.
- h. Account No. 399.00 Other Tangible Property (base period only).

i. Account No. 399.89 – Other Tangible Property – System Software (forecasted period only).

38. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-3.2. Each of the accounts listed below is shown on Schedule B-3.2 as being fully depreciated. Explain why Western has included a 12-month depreciation expense for each of these accounts, and why it is reasonable to include depreciation expense for a plant account that is fully depreciated per the company's books. a. Account No. 331.00 – Production Gas Wells Equipment (base period only).

b. Account No. 332.10 – Field Lines (base period only).

c. Account No. 332.20 – Tributary Lines (base period only).

d. Account No. 334.00 – Field Meas. & Reg. Station Equipment (base period only).

e. Account No. 351.20 – Compression Station Equipment (base and forecasted periods).

f. Account No. 366.30 – Other Structures (base and forecasted periods).

g. Account No. 375.10 – Meas. & Reg. Station Equipment General (base and forecasted periods).

39. Refer to the Application, Volume 10 of 10, Tab 6, Schedule F-6. Provide the following information concerning the costs for the preparation of this case:

a. A detailed schedule of costs incurred to date. Include the date of the transaction, check number or other document reference, the vendor, amount, a description of the services performed, and the account number in which the expenditure was recorded. Indicate any costs incurred for this case during the base year. Include copies of invoices received from the vendors.

b. A detailed explanation of how the estimate shown on Schedule F-6 was determined, with all supporting workpapers and calculations.

c. Monthly updates of the actual costs incurred during the course of this proceeding, in the manner prescribed above. Updates will be due on September 3,

-16-

1999 (Western's due date for responses to additional data requests), November 1, 1999 (due date for requests to Intervenors), and 30 days after the date of the public hearing or the due date for final briefs, whichever date is later.

40. Refer to the Application, Volume 10 of 10, Tab 8, Schedule H.

a. Explain how the Uncollectible Accounts Expense percentage was determined. Include all supporting workpapers, assumptions, and calculations.

b. Provide a schedule showing Western's actual Uncollectible Accounts Expense percentage for the base period and the five previous fiscal years. Include all supporting workpapers, assumptions, and calculations.

c. Explain how the PSC Fees percentage was determined. Include all supporting workpapers, assumptions, and calculations.

41. Refer to the Application, Volume 10 of 10, Tab 8, Schedule I-1. In light of the revenue increase requested by Western, explain why Schedule I-1 shows Western is forecasted to experience net losses beginning in fiscal year 2001.

42. Refer to the Application, Volume 10 of 10, Tab 15, Summary of Factors schedule, referenced as "WP Factors." Concerning the calculation of the Residual Factor:

a. Explain why the Residual Factor is based on calendar year 1998 data.

b. Provide the Residual Factors for all other Atmos utility and nonutility business divisions. c. Explain why it is reasonable to base the Residual Factor on the Western to Atmos ratios for Gross Direct PP&E, Average Number of Customers, and Total O&M Expense.

d. Explain the reference to the "Gray Book."

43. Refer to the Application, Volume 10 of 10, Tab 15, WP B-2, base period for Division 09, pages 3 and 4 of 4. For each of the column headings listed below, identify the source of information for the items contained in the column and provide the cross-reference to where that item can be found in the Application. If the item has not been provided in the Application, provide the source of information for the item in question.

a. "WKG Direct Additions FY 99."

b. "service prog. WKG adds April 1999."

- c. "WKG OH 98 carryover."
- d. "WKG OH FY 99."
- e. "02 OH FY 99."

44. Refer to the Application, Volume 10 of 10, Tab 15, WP B-2, forecasted period for Division 09, pages 3 and 4 of 4. For each of the column headings listed below, identify the source of information for the items contained in the column and provide the cross-reference to where that item can be found in the Application. If the item has not been provided in the Application, provide the source of information for the item information for the item source of information for the item source of information.

- a. "WKG Direct Additions FY2000."
- b. "WKG OH FY2000."

c. "02 OH FY2000."

d. "WKG Additions FY2001."

45. Refer to the Application, Volume 10 of 10, Tab 15, WP B-3.2, for both the base and forecasted periods, Division 09.

a. For the base period, explain the reason for the columns titled "Total Company Adjusted Jurisdiction – Reserve" and "Reserve Computation."

b. For the forecasted period, explain the reason for the columns titled "Division 09 13 Month Avg. – Reserve" and "Reserve Computation."

c. For both periods, explain why the "12 Month Expense" column includes a reference to 95.45 percent.

d. Provide the calculations used to determine the 95.45 percent.

e. For the forecasted period, explain the reference to "ELG" in the annual accrual rate column.

46. Refer to the Application, Volume 10 of 10, Tab 15, WP B-4.1, Working Capital Components, for both base and forecasted periods. Provide a breakdown showing all accounts and subaccounts contained in the Prepayments for each period. This breakdown should use the same accounting system as was used to show the detail for Materials and Supplies.

47. Refer to the Application, Volume 2 of 10, Tab 10, the Testimony of Thomas H. Petersen and Volume 9 of 10, Tab 3, the class cost-of-service study.

a. Provide the name of the model used for the cost-of-service study and specify whether this is an industry model or an in-house model. Provide a narrative description of the study and written operating procedures for running the model.

-19-

b. Do any of Western's affiliates use this model? If so, identify any concerns that have been expressed or modifications that have been made by other state regulatory commissions in cases in which this model has been presented in support of those affiliates' revenue allocation or rate design proposals.

c. Identify any other models that were considered by Western prior to selecting this model and whether the other models provided similar results and guidance. If other models were reviewed, why was this model considered superior? If no other models were reviewed, how can the Commission be assured that the guidance represented by this model is the most reasonable?

d. Class load factor is defined by Mr. Petersen as the average daily use divided by design day use or maximum daily contract level. Expand on this definition and explain how this factor is used in the model.

e. On page 3 of his testimony Mr. Petersen states that the rate classes selected use available data that captures the class differences in load characteristics. Can the available data be readily subdivided into groupings other than the five rate classes used in the study?

f. Mr. Petersen also states that the study was performed using fiscal year 1998 data and that results using the forecasted test period would follow a pattern similar to that of the historic cost-of-service analysis. Provide the results for the forecasted period and the appropriate workpapers in the same form as provided in the original filing (i.e., nineteen pages of model results and nine pages of supporting workpapers).

-20-

g. What are the results of the model for the historic test period normalized for weather using the proposed rates and the resulting revenue levels? What are the results of the model using the forecasted test year with the proposed rates and resulting revenue levels? Provide these results and the appropriate workpapers in the same form as provided in the original filing.

h. An adjustment for customer accounts was incorporated into the model. Explain why this was the only adjustment other than the adjustment to normalize weather incorporated into the model.

i. Notes are included in many pages of the study describing rules, actions or assumptions applied to the particular worksheet. Provide a narrative description of these rules and actions and the source of the assumptions.

48. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Rate G-1 and G-2. Why is there no page reference for: (1) the Weather Normalization Adjustment; (2) the Gas Cost Adjustment Rider; and (3) the Margin Loss Recovery Rider? Does Western agree that including page references for each of these items would enable the tariff reader to better follow the tariffs without being required to constantly refer to the tariff index?

49. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 26, Weather Normalization Adjustment Rider ("WNA").

a. The tariff shows an effective date of July 24, 1999, while page 37 of the Testimony of Gary L. Smith indicates the WNA would go into effect November 1, 2000. What is the correct effective date for the WNA?

-21-

b. The tariff states "Base loads and heating sensitivity factors will be determined by class and computed annually." Provide a detailed description of how base loads and heating sensitivity factors will be determined. Include example calculations if necessary.

c. Page 38 of the Testimony of Gary L. Smith sets out a proposed schedule for filing periodic reports with the Commission. Should the schedule and a description of these reports be included in the tariff? Why or why not?

d. At page 37 of the Testimony of Gary L. Smith he states that the proposed WNA mirrors that of Western's affiliate, United Cities Gas Company. Provide for the last three years (fiscal or calendar), a comparison of United Cities' residential revenues both with and without the impacts of its WNA.

50. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 27, Gas Cost Adjustment Rider ("GCA").

a. Since the proposed GCA is zero-based, is it still necessary that Western file its GCA on a monthly basis?

b. Explain in detail why quarterly GCA filings, as submitted by Kentucky's other major LDCs, would not be sufficient to meet Western's gas cost recovery needs.

51. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 27I, the Margin Loss Recovery Rider ("MLR").

a. The tariff does not specify this, but the Testimony of Gary L. Smith, at page 29, indicates that the proposal will shift lost revenues to sales customers. Why

-22-

is the proposed shift to sales customers only? Explain how the proposed 90 percent / 10 percent sharing between customers and the company was developed.

b. The MLR tariff does not specify the rate schedules to which it would be applied. Was this an oversight or was it done intentionally? Explain why an "Applicable" provision designating the appropriate rate schedules should not be included in the tariff.

c. How was it determined that the MLR should be adjusted on a semiannual basis, as opposed to monthly, quarterly, or annually?

52. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 30d, Gas Research Institute ("GRI") R & D Rider.

a. The unit charge in the tariff is proposed to be billed "according to the transition schedule outlined in the pipeline's tariffs." Provide the transition schedules for each of the pipelines serving Western.

b. What was Western's "level of contribution per Mcf" as of December31, 1998?

c. Why is the proposed tariff rider to be "applicable to all gas transported by the Company other than Rate T-3 and T-4 Carriage Service"?

d. Does "all gas transported" mean sales and transportation volumes or transportation volumes only? Explain why it should be one or the other.

e. Identify any other methods of GRI cost recovery that Western considered and explain why those methods were not selected.

f. Identify the benefits that accrue to Western's ratepayers from Western's funding of GRI's R & D.activities.

-23-

53. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 49, Alternative Receipt Point Service. Provide cost support for the proposed distribution charge of \$.10 per Mcf.

54. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 51, Special Charges.

a. Even though rate schedules G-1, G-2, LVS-1, LVS-2, T-3, and T-4 all include sections headed "Late Payment Charge," Sheet No. 51 and the Testimony of Gary L. Smith, indicate the proposed Late Payment Charge of 5 percent will be applied only to Rate G-1 sales service. Explain the reasoning for applying the 5 percent charge to only one rate schedule.

b. What other local gas distribution companies is Western aware of that have a late payment charge which is applicable to only one of several rate schedules?

c. What is the purpose of the Late Payment Charge section in the tariffs, other than Rate G-1, identified in part (a) above?

d. Provide the amount of annual revenue that Western expects the Late Payment Charge to generate. Include supporting calculations and sufficient narrative explanation to explain the calculations.

55. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 30a, Demand-Side Management Cost Recovery Mechanism, and the Testimony of Gary L. Smith and Michael Marks on the same subject.

a. Explain why the WKG CARES program is proposed for another three years, as opposed to one or two years, or on a permanent basis.

-24-

b. What consideration, if any, was given to implementing the program on a permanent basis?

c. Are the non-permanent nature of the proposed three-year program and the proposal to recover costs for the three-year pilot program the only reasons for proposing a surcharge mechanism rather than including the prospective costs for recovery through base rates? If there are other reasons for using a surcharge mechanism, explain them in detail.

d. The tariff itself does not specifically mention the annual filing with the Commission discussed in the Testimony of Michael Marks. Was this an oversight or intentional? Provide any reasons why Western would be opposed to including a statement in the tariff identifying and describing the annual filings proposed by Mr. Marks.

56. Refer to Volume 1 of 10 of the Application, Tab 6, Proposed Tariffs, at Sheet No. 67, Rules and Regulations, Part (I), Premises charge and the Testimony of Daniel Ives.

a. The proposed Premises Charge is only for the residential customer class and Mr. Ives discusses this on page 11 of his testimony. Even though 84 percent of customer growth is in the residential class, explain why Western would choose not to address the same problem of incremental versus embedded costs for the remaining one-sixth of its customer growth occurring in other customer classes.

b. On page 10 of his testimony Mr. Ives discusses the proposed fifteen-year recovery period for the Premises Charge. Explain why a shorter life, based

-25-

on the Internal Revenue Service's MACRS system, is appropriate for per books accounting by a regulated entity.

c. Provide support for Mr. Ives statement that "a fifteen-year recovery period is consistent with what is being used elsewhere in the industry."

57. Refer to Volume 2 of 10 of the Application, the Testimony of David H. Doggette, pages 12-14, and Exhibit DHD-2.

a. Some of the service charge studies discussed by Mr. Doggette and included in Exhibit DHD-2 covered Western's fiscal year 1998. Identify any studies, other than the April 1999 survey of banks, that cover a period other than fiscal year 1998.

b. Does the summary analysis on Exhibit DHD-2, page 1 of 8, at Column 3, represent the actual number of orders charged for fiscal year 1998, or does it represent the actual number of orders for any period? Explain what Column 3 represents.

c. Explain why Exhibit DHD-2, page 1 of 8, does not include all the special charges included in Western's proposed tariffs at Sheet No. 51.

d. What impact, if any, do the proposed revenues in Exhibit DHD-2, page 1 of 8, Column 15, or the increase in revenues in Column 16 have on the increase in Other Revenues derived from comparing Exhibits GLS-7 and GLS-1 of the Testimony of Gary L. Smith?

58. Refer to Volume 2 of 10 of the Application, Tab 11, the Testimony of Gary L. Smith at pages 4-12.

-26-

a. Provide the historical data for the three-year period referred to by Mr. Smith on page 6 of his testimony including: (1) the number of customers by customer class; (2) sales volumes, by customer class, adjusted for normal weather; (3) annual changes in volumes for industrial sales and transportation deliveries; and (4) the level of volume migration from sales to transportation volumes.

b. Provide detailed calculations showing the derivation of the adjustment for industrial sales and transportation deliveries referred to by Mr. Smith on page 8, lines 25-28 of his testimony.

c. On page 7, lines 5-6 of his testimony, Mr. Smith refers to "historical growth rates averaging slightly less than 2,000 for the three prior years." To what three years does Mr. Smith refer? How does this statement reconcile with the table on page 12 of his testimony that reflects an average of at least 2,156 for any three-year period included therein?

d. For each year in the five-year period covered in the table on page 12 of Mr. Smith's testimony, provide a breakdown of growth in residential customers between "new construction" and "on-main conversions."

59. Refer to Volume 2 of 10 of the Application, Tab 11, the Testimony of Gary L. Smith, and Exhibits GLS-1 through GLS-7.

a. Exhibit GLS-3 summarizes the impact of industrial contract adjustments and volume changes. Provide supporting workpapers and narrative descriptions of these changes, by customer (the actual identity of the customers may be omitted and reference made by numbers and/or letters, i.e., – customer 1a).

b. Exhibit GLS-6 summarizes the volume adjustment for declining customer usage. Provide supporting workpapers and narrative descriptions of the calculations made to derive the adjustment.

c. Exhibit GLS-1 shows revenues at current rates reflecting all adjustments to derive test year volumes while Exhibit GLS-7 shows revenues at proposed rates reflecting the same adjustments. Are there any differences in the two exhibits other than: (1) different rates / margins; (2) Alternative Receipt Point volumes and revenues; (3) the amounts shown for Additional Contract Reformation; and (4) the amounts shown for Other Revenue? If yes, identify and explain those differences.

60. Refer to Volume 2 of 10 of the Application, Tab 8, the Testimony of Dr. Donald A. Murry.

a. Page 6, lines 20 through 22, indicates that the Commission should make allowances for the added risk of the inclusion of short-term debt in the capital structure. In what way should the Commission make such an allowance?

b. Page 5, beginning on line 17, indicates that Atmos raises capital for Western's operations. Is this beneficial to Western? If so, should the Commission make allowances for Western's ability to access this capital source? Would it be more risky for Western if it had to raise capital itself for its operations?

c. Provide an explanation of why each company in the group of comparative companies is considered to be a viable comparison to Atmos.

d. Provide the most recently approved return on equity for each of the comparable companies, along with the date each was approved.

-28-

e. Do any of the comparable companies use a weather normalization adjustment, a premises charge, or a margin loss recovery mechanism to stabilize their earnings? If so, which ones?

f. What effect would the implementation of a weather normalization adjustment have on Western's financial risk?

g. What effect would the implementation of a premises charge have on Western's financial risk?

h. What effect would a margin loss recovery mechanism have on Western's financial risk?

61. Refer to Volume 10 of 10 of the Application, Tab 4, Summary of Jurisdictional Adjustments by Major Accounts, Schedule D-1, Sheet 1 of 4 and Schedule D-2.1, Sheets 1 and 2.

a. Provide supporting workpapers for the revenue and gas purchases adjustments on these schedules, or reference where provided if already included in the application or in response to other requests contained in this Order.

b. Provide narrative descriptions of the workpapers provided in response to part (a) above.

c. Explain in detail the reasons for the proposed reductions to Service Revenues and Other Gas Service Revenues.

62. Refer to Volume 2 of 10 of the Application, Tab 1, the Testimony of Conrad E. Gruber, specifically, the table on page 11 which denotes Western's operating and maintenance cost efficiencies in comparison to industry averages.

-29-

a. Provide copies of the A.G. Edwards study cited and a detailed schedule of Western's operating and maintenance expenses, identified by fiscal year ("FY"), used to determine Western's Costs per Meter as shown in the table.

b. Provide a detailed schedule of Western's gas utility employees by
job classification for the period used to determine Western's number of employees per
1,000 customers as shown in the table.

c. Provide documentation used for external reporting purposes to substantiate the number of meters in service and thousands of customers served for the period represented by the table.

63. Refer to Volume 2 of 10 of the Application, Tab 1, the Testimony of Conrad E. Gruber, specifically page 16, which indicates that installation of the Oracle system was expected to be completed by July 1999, and refers to the series of IT projects that are essential for Y2K readiness.

a. Has the Oracle system been implemented and tested for Y2K readiness?

b. Provide a description of the other projects scheduled for completion prior to the end of 1999 to assure Western's customers of Y2K readiness.

c. Identify any costs associated with Western's Y2K readiness that are included in the base year or the forecasted year.

64. Refer to Volume 2 of 10 of the Application, Tab 6, the Testimony of Donald P. Burman, and Volume 4 of 10, Tab 4, Filing Requirement 10(9)(m).

-30-
a. Explain whether FR 10(9)(m) is a complete conversion table to NARUC accounts as presented for operating revenue and expenses in Volume 10 of 10, FR 10(10)(c), Schedule C-2.1 and Schedule C-2.2.

b. Are the references in the "detail" sections of this exhibit to NARUC accounts the same as the accounts used to file the annual FERC Form No. 2 with the Commission?

65. Refer to Volume 2 of 10 of the Application, Tab 4, the Testimony of Betty L. Adams. Does Western have a conversion table that converts its current chart of accounts for general ledger purposes to the NARUC accounts as presented for operating revenue and expenses in Volume 10, FR 10(10)(c), Schedule C-2.1 and Schedule C-2.2? If yes, provide the conversion table.

66. Are the NARUC account numbers referenced in Ms. Adams' testimony at page 5, lines 5 through 23, the same as the account numbers used to determine the account balances for the annual FERC Form No. 2 filed with the Commission? If no, is there a conversion table that converts from NARUC accounts, to FERC accounts, to Western's general ledger chart of accounts?

67. Refer to Ms. Adams' testimony.

a. Are the operating revenue and expenses in Volume 10, FR 10(10)(c), Schedule C-2.1 and Schedule C-2.2 according to NARUC accounts available according to Western's current general ledger chart of accounts? Resubmit these schedules according to the current chart of accounts.

b. Does Western have operating revenue and expenses in the detailed manner described above according to its current chart of accounts which

-31-

compare budgeted amounts to actual year-to-date totals for the FY 1998, 1997, 1996, 1995 and 1994?

c. If yes to part (a) provide the budget to actual comparison for those years. Provide a brief explanation for accounts with a budget to actual variance of 5 percent or greater.

d. Resubmit Volume 10, FR 10(10)(d), the summary of jurisdictional adjustments, according to Western's general ledger chart of accounts.

e. Resubmit Volume 9, FR 10(9)(d) for the base year and test year according to Western's general ledger chart of accounts.

f. Are the jurisdictional adjustments in Volume 10, FR 10(10)(d) and FR 10(9)(d) for the base year and test year by account as Western would submit to the Commission in a FERC Form No. 2 annual report? If not, resubmit these schedules according to the FERC accounts.

g. Are the operating revenue and expenses in Volume 10, FR 10(10)(c), Schedule C-2.1 and Schedule C-2.2 according to FERC Form No. 2 as filed annually with the Commission available? If no, resubmit these schedules according to the FERC accounts.

68. Refer to Volume 2 of 10 of the Application, Tab 4, the Testimony of Betty L. Adams. On page 6, line 24, a table is presented to point out Western's overall operating and maintenance ("O & M") budgeting effectiveness for FY 1994 through 1998. Provide the source documents from which this table was created, with amounts detailed according to Western's general ledger chart of accounts.

-32-

69. On page 8, line 22 of Ms. Adams' testimony there is a discussion of the increase in the forecasted test period labor costs that is attributed to "the planned filling of a number of vacant employee positions and... a four percent wage increase."

a. Provide the number of vacant employee positions, by job classification, that Western intends to fill.

b. Provide the actual employee positions, by job classification, and provide the actual period used to determine the number of vacant positions necessary.

c. Does Western have both union and non-union employees? If there are union employees, provide the job classifications and a copy of the union contract.

d. Provide a schedule showing the derivation of the proposed 4 percent wage increase along with an explanation of how wage increases are determined for management, union and non-union employees.

e. Provide a list of the planned positions being filled that were previously held by contractors, by job classification, and break down the list further by identifying contractors performing construction activities, not operational duties.

f. Identify how many years the contractors have performed construction activities and whether these activities are now being considered in the planned construction budget.

g. If the planned positions are replacing contractors that have been performing construction services, and construction services with contractors are included in the planned construction budget, provide a detailed description of the expected benefits from the addition of the planned operating and maintenance employees that Western's customers will receive that they have not been receiving.

-33-

70. Resubmit Volume 9, FR 10(9)(h)9 for the years 2000 through 2003 with employees separated by job classifications. Also, provide a comparison of budgeted to actual number of employees for FY 1994 through 1998. Provide references in each of these schedules to the employee numbers by Western's chart of account number, NARUC account, and FERC account. Explain any increases or decreases of 5 percent or more in employee numbers from year to year.

71. On page 9, lines 10 through 17 of Ms. Adams' testimony is a discussion of the increase in the forecasted test period communications expense that is attributed to increased use of mobile data terminals ("MDTs") and higher cellular usage. Provide a schedule showing the cost amounts, MDT units acquired and plant accounts charged since the project inception, showing the years in which the investments in MDTs were made.

a. Does Western have contracts for communications expenses, such as long distance and cellular usage?

b. If yes, provide the old and new contracts and an explanation of why the usage was deemed to increase in the forecasted period considering any contractual changes or changes in services used.

c. Give a quantified determination of how this increase was estimated and whether the costs are under contract or not.

72. Refer to Ms. Adams' testimony. Provide an explanation, complete with a quantified determination, of how the increase in uncollectible write-offs was calculated for the forecasted test period.

-34-

a. Provide the accounts receivable aging schedules for the last two fiscal years.

b. Explain why under-budgeted write-offs for the six-month period in FY 1999 provided sufficient reason to adjust the forecasted period.

c. Provide an accounts receivable aging schedule for the last month of actual results in the test period.

d. Provide an account analysis of Western's reserve for uncollectibles comparing the actual charge-offs with the year to date provision for uncollectibles (expense) for the six-month period of actual results in the FY 1999 that comprises part of the base year. If different, schedule Western's monthly provision for uncollectibles (expense) in comparison to the year-to-date ("YTD") budget.

e. Provide a comparison of the YTD budget to actual provision for uncollectibles (expense) for the last two fiscal years.

f. Provide a comparison of the reserve for uncollectibles to accounts receivable for FY 1997, FY 1998 and the end of the six-month actual period included in the base year.

73. In Volume 10 of 10 of the Application, Schedule D2.2, Sheet 2 of 2, "ADJ 7" includes the "transfer of Human Resources expenses from Shared Services of \$67,700."

a. Provide a list of the job(s) transferred, an explanation of the previous job(s) function with Shared Services, and an explanation of the job(s) function with Western.

-35-

b. Did similar job reclassifications occur with Western's gas distribution affiliates? If yes, give an explanation of the reasoning. If no, why not?

74. In Volume 10 of 10 of the Application, Schedule D2.3, Sheet 1 of 1, "ADJ 2" is described as an adjustment to "reflect the amortization of the PSC Assessment for 1997 paid in 1999." Provide a detailed explanation and calculations to support the determination of this adjustment.

75. Refer to Ms. Adams' testimony at page 10, line 6, where adjustments to Western's FY 1999 budget decreasing O & M expenses and increasing Shared Services expenses are discussed in regard to the utilization of the new "Customer Support Center in Amarillo, Texas." Provide quantified schedules, referenced to Western's chart of accounts, with explanations of the cost shifts discussed, i.e., decreased number of employees by job classification to Western, new charges by Shared Services.

76. Refer to Ms. Adams' testimony at page 10, line 17, where adjustments to Western's FY 1999 budget decreasing O & M expenses and increasing Shared Services expenses are discussed in regard to the United Cities Gas merger. Provide quantified schedules, referenced to Western's chart of accounts, with explanations of the cost shifts discussed, i.e., decreased number of employees by job classification to Western, new charges by Shared Services.

77. Refer to Ms. Adams' testimony at page 10, line 22, where adjustments to Western's FY 1999 budget from non-labor savings in the "proposed Gas Meter Performance Control Program" are referenced. Provide a detailed schedule with a calculation showing how these savings were determined.

-36-

78. Refer to Ms. Adams' testimony at page 10, line 22, where adjustments to Western's FY 1999 budget from transferring the rates and regulatory vice-president position from the Shared Services staff is mentioned. Did similar job reclassifications occur with Western's gas distribution affiliates? If yes, give an explanation of the reasoning. If no, why not?

79. Refer to Ms. Adams' testimony at page 13, lines 15 through 18, where the adjustments to Western's "Shared Services" forecasted budget includes an adjustment for a "decrease in the labor portion of Atmos' administrative and general overheads, which is a capital expense."

a. Provide a description of these costs and a schedule of these costs, with reference to the accounts charged in the Shared Services forecasted budget.

b. Explain how these costs represent a capital expense as Atmos administrative and general overhead, but upon reclassification as a Shared Service expense become an operating and maintenance expense subject to recovery through customers' rates.

c. Were these costs similarly reclassified for Shared Services charges to Western's gas distribution affiliates? If yes, give an explanation of the reasoning. If no, why not?

80. Provide the basis for the beginning of Western's FY 1999 budgeting process in quantitative form, i.e., prior year's budget, prior year's actual results, by Western's current chart of accounts.

-37-

a. Provide a listing of the known adjustments made at the beginning of Western's budgetary process, i.e., increase or decrease in employee numbers, reductions for expenses non-recurring in nature.

b. Provide a listing of the known adjustments made at the major decision-points of Western's budgetary process, i.e., increase or decrease in employee numbers, reductions for expenses non-recurring in nature, deferring or accelerating maintenance projects.

81. Provide the basis for the beginning of the Western's "Shared Services" FY 1999 budgeting process in quantitative form, i.e., prior year's budget, prior year's actual results, prior year service adjusted by changes in affiliated charges, by Western's current chart of accounts.

a. Provide a listing of the known adjustments made at the beginning of Shared Services' budgetary process, i.e., increase or decrease in employee numbers, reductions for expenses non-recurring in nature.

b. Provide a listing of the known adjustments made at the major decision-points of Shared Services' budgetary process, i.e., increase or decrease in employee numbers, reductions for expenses non-recurring in nature, deferring or accelerating projects.

82. Refer to Volume 2 of 10 of the Application, Tab 4, the Testimony of Betty M. Adams, page 3, which shows her sponsoring FR 10(10)(d) and FR 10(10)(f) and Volume 10 of 10, Tabs 4 and 6, which include FR 10(10)(d) and FR 10(10)(f). Provide a schedule of the rate-making adjustments for country club dues, promotional advertising and sales expenses, employee party and gift expenses and pension expense in

-38-

reference to the "Detailed Adjustments" in Volume 10, FR 10(10)(d)2.1, FR 10(10)(d)2.2, and FR 10(10)(d)2.3, as applicable.

a. Provide the location of the above adjustments in FR 10(10)(d)1, with specific account number references.

b. Provide the location of the above adjustments in FR 10(10)(d)1, with specific account number references to the same schedule as previously requested to be resubmitted in Western's general ledger account number form.

83. Refer to Volume 2 of 10 of the Application, Tabs 2 and 4, the testimony of R. Earl Fischer and Betty L. Adams. To some extent, both witnesses address the issue of direct billed intercompany services and allocated service costs from Atmos's Shared Services Business Unit ("Shared Services") to Western and other Atmos business units.

a. Provide a detailed operating statement for Shared Services for FY 1997, 1998, and year to date FY 1999 actual, with detailed intercompany revenue accounts to reflect similar services provided by Shared Services for the Atmos operating divisions. Specifically, reference Shared Services' revenue accounts to Western's expense accounts by current chart of accounts. Provide the FY 1999 Shared Services budget and provide updates of FY 1999 actual data as it becomes available.

b. Provide contractual agreements between Western and Shared Services since 1997, with a schedule of expected service cost increases that are included in the determination of the base year or the forecasted year. Reference these service costs to Western's expense accounts by current chart of accounts.

-39-

c. Provide Shared Services' cost allocation manual, if available. If none is available, describe the cost allocation process for the portion of the cost not directly billed to Western and provide copies of all internal policies and procedures relating to the allocation of costs from Shared Services to Western.

Done at Frankfort, Kentucky, this

## PUBLIC SERVICE COMMISSION

N.

For the Commission

ATTEST:

**Executive Director** 

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 1 Witness: Gruber

## Data Request:

Provide an organization chart for Atmos Energy Corporation ("Atmos") as of June 1, 1999. The chart should show all regulated utility divisions and non-regulated businesses.

## Response:

See attached Atmos organization chart.



# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item Question 2 Witness: Thomas H. Petersen

## Data Request:

- 2. With the exception of Western, provide the following information for each of the regulated utility divisions:
  - a. The state jurisdictions in which the division operates.

b. The year of the division's last completed general rate case proceeding. If the division is currently involved in a general rate case, indicate when that proceeding is expected to be completed.

c. The authorized rate of return on common equity awarded or authorized in the last completed general rate case proceeding. If the division is currently involved in a general rate case, indicate the rate of return on common equity requested by the division.

### Response:

a) Atmos Energy Corporation provides regulated utility service in twelve states. It does business as Energas Company in Texas; as Greeley Gas Company in Colorado, Kansas and Missouri; as Trans Louisiana Gas Company in Louisiana; as United Cities Gas Company in Georgia, Illinois, Iowa, Missouri, South Carolina, Tennessee and Virginia; and as Western Kentucky Gas Company in Kentucky.

٦

b&c) Energas Company's last general rate proceeding was completed in 1996. The case was settled with no rate of return specified.

Greeley Gas Company's last general rate proceeding in Colorado was completed in 1998. The case was settled with no rate of return specified. The company's last general rate proceeding in Kansas was completed in 1995. The case was settled with no rate of return specified.

Trans Louisiana Gas Company's last general rate proceeding was completed in 1995. The approved range for rate of return on common equity was 11.75 to 12.25 percent. The Company is currently involved in a proceeding that is expected to be completed later in 1999. The company has proposed a rate of return on common equity of 12.25% in the current proceeding.

United Cities Gas Company's last general rate proceeding in Georgia was completed in 1996 with an approved rate of return on common equity of 11.5 percent. The company's last general rate proceeding in Illinois was completed in 1997 with an approved rate of return on common equity of 10.94 percent. The company's last general rate proceeding in Iowa was completed in 1996 with an approved rate of return on common equity of 11.0 percent. The company's last general rate proceeding in Missouri was completed in 1995 with an approved rate of return on common equity of 12.15 percent. The company's last general rate proceeding in South Carolina was completed in 1995 with an approved rate of return on common equity of 11.75 percent. The company's last general rate proceeding in Tennessee was completed in 1995. The case was settled with no rate of return specified. The company last general rate proceeding in Virginia was completed in 1998 with an approved rate of return on common equity of 11.0 percent.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 3 Witness: Donald P. Burman

## Data Request:

Western's last general rate case was Case No. 95-010.<sup>1</sup> In the August 10, 1995 Order in that case, the Commission expressed its concern about Western's practices of funding only a portion of its other post-retirement employee benefits ("OPEBs") and of funding internally rather than using a protected fund administered by a third party. Describe Western's current practice with regard to these two issues. In addition, provide justification for Western's current practice with regard to these two issues. In addition, provide justification for Western's current level of OPEB funding.

#### Response:

Although the Commission expressed its concern that Western fund its OPEB obligation to the extent the expenditures were tax deductible, Western has continued to only fund the amount of claims incurred, which has benefits for Western's customers, and has recorded a liability for the difference between the actuarially determined amounts and the amounts paid. Western is continuing to consult with its tax accountants to determine the maximum tax deductibility.

<sup>&</sup>lt;sup>1</sup> Case No. 95-101, An Adjustment of Rates of Western Kentucky Gas Company, Orders dated August 10, 1995 and October 20, 1995.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 4 Witness: Donald P. Burman

#### Data Request:

In the October 20, 1995 Order, which adopted the October 9, 1995 Settlement Agreement, the Commission acknowledged Western's agreement to perform a new depreciation study no later than its next general rate application. However, the October 20, 1995 Order reminded Western that the Commission's concerns about its depreciation study, expressed in the August 10, 1995 Order, would remain pertinent for the review of the new depreciation study.

- (a) Explain in detail how the concerns expressed in the August 10, 1995 and October 20, 1995 Orders have been addressed by the September 30, 1997 depreciation study filed with Western's application.
- (b) Refer to the September 30, 1997 depreciation study filed with the application. Provide the calculations for the new salvage rates with details and supporting documentation, which support the change to a negative 150 percent, in the salvage rte for Account 380, Services.

Response:

(a)

- The Company acknowledges the Commission's concerns expressed in the earlier orders. However, the Company believes that those concerns were misplaced. While it is true that the prior depreciation study (as of September 30, 1992) utilized only six years of history for certain analyses, other analyses were conducted which encompassed an entire period of history using a different life analysis tool. The current study also utilized both analysis methodologies, not to mention, the additional 5 years of actual aged historical retirement experience. In addition, both studies gave consideration to the future plans and expectations of the Company. Therefore, the Company is confident that the current study methodology and conclusions adequately address those concerns, and the results of the study reflected in this application are fair and reasonable, and do not reward nor penalize any generation of ratepayers.
- (b) The attached spreadsheet provides the quantitative supporting documentation for Account 380. There is a distinct trend toward higher cost of removal. Greater reliance was placed on 1997 due to the reflection of a change in accounting. Therefore, the Company believes that 1997 is the best expectation of what the future will be also recognizing the cost of removal trend and the reduced salvage. A conservative approach was taken due to the significant change and only 50% of the actual experienced percentages were utilized.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Response 4 (b) Account 380 - Services Net Salvage Analysis

Year	Retirements	Salvage (*)	Cost of Removal (*)	Salvage %	COR %	NS %
1993	110,087	696,702	176,179	633%	160%	473%
1994	156,110	573,554	280,944	367%	180%	187%
1995	204,293	354,291	426,682	173%	209%	-35%
1996	176,565	507,706	442,262	288%	250%	37%
1997	215,379	68,465	750,510	32%	348%	-317%
	862,434	2,200,718	2,076,577			

(\*)Note: These amounts are for the total distribution plant function.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 5 Witness: Smith

#### Data Request:

Refer to the Application, Volume 1 of 10, Tab 1, Item 5. Provide the calculations that support Western's determination that it has experienced an \$800,000 decline in lost industrial margins and \$1,600,000 in lost margins due to energy conservation and declining customer usage. Include a detailed explanation as to how Western identified these specific causes.

### Response:

Western's calculations supporting margin losses since our last rate case in 1995, associated with discounts to retain bypass-vulnerable industrial margins customers and those attributable to declining Residential and Commercial customer usage, is attached as Schedule PSC DR NO. 1 - DR Item 5. The information relating to industrial discounts is filed under a petition for confidentiality due to the necessity of revealing the affected volume and discount level for purposes of this computation.

Western identified these specific causes during the process of developing the Fiscal Year 2000 Budget (and the calendar year 2000 budget for the Test Period in this Case). Margin reductions necessary to retain bypass-vulnerable accounts and their accumulated affect on Western's financial performance has long been recognized, and monitored, in the preparation of forward-looking budgets. The calculation of the \$800,000 margin losses was performed to record the impact of this phenomenon since Western's last rate case. The calculation is shown on Schedule DR-5, Sheets 1 and 2.

Prior to Western's preparation of the FY 2000 budget, however, the Company had not monitored the cumulative magnitude of Residential and Commercial energy conservation, or its impact on our financial performance since the last rate case. Refer to pages 12 (lines 28-31) through 13 (lines 1-16) of my testimony for more information on our discovery of this matter. To quantify the impact on Western's margin, we computed the current annual margin reduction from the test period in the last rate case. The calculation is shown on Schedule DR-5, Sheet 3. PSC DR NO. 1 DR Item 5 Sheet 1 of 3

CONFIDENTIAL

KPSC Data Request Dated July 16, 1999 DR Item 5 Western Kentucky Gas Company Case No. 99-070 Witness: Smith



PSC DR NO. 1 DR Item 5 Sheet 2 of 3

> Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 5 Witness: Smith

CONFIDENTIAL





	Weste KPSC Da	rn Kentucky Gas Com Case No. 99-070 ta Request Dated July DR Item 5 Witness: Smith	pany DR Ite Sheet 3 16, 1999
o. ne	Description	Calculations	Source
	(a)	(9)	(c)
	Residential Conservation Effect:		
~	FY 1994 Number of Residential Customers:	145,689	Financial Statistics, 9/94
<del></del>	Average Decline in Mcf/year/customer:	(1.73)	See Derivation in PSC DR 59(b)
	Total decline for five years to FY 1999, Mcf	(1,260,210)	Line 3 times Line 4 times Five Years
5	Average Commodity Margin/Mcf	\$1.0615	Avg. Commodity Margin, Per Books 9,
~ ~	Margin Loss Attributable to Conservation -	- (\$1,337,713)	Line 5 times Line 6
~ c	Commercial Conservation Effect:		
,	FY 1994 Number of Com/PA Customers:	18,148	Financial Statistics, 9/94
2	Average Decline in Mcf/year/customer:	(3.00)	See Derivation in PSC DR 59(b)
ŝ	Total decline for five years to FY 1999, Mcf	(272,220)	Line 11 times Line 12 times Five Year
4	Average Commodity Margin/Mcf	\$0.9810	Avg. Commodity Margin, Per Books 9
5 9	Margin Loss Attributable to Conservation -	(\$267,048)	Line 13 times Line 14
5	Total Conservation Effect Since FY 1994 -	(\$1,604,761)	Line 7 plus Line 15

.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 6 Witness: Gruber

#### Data Request:

Refer to the Application, Volume 2 of 10, Tab 1, the Testimony of Conrad E. Gruber. On pages 12 through 18 of the testimony is a discussion of Western's efficiency and productivity improvements. Explain in detail how the benefits of these efficiency and productivity improvements have been incorporated into Western's forecasted period. Include a dollar quantification of the improvements and identify where this quantification has been shown in the forecasted period.

#### Response:

The benefits of Western's efficiency and productivity improvements have been discussed at length in Mr. Gruber's testimony. See Western's response to DR 8 for a discussion and of how the benefits of many of these and related programs are reflected in Western's total costs. Although the quantification of such benefits are discussed in DR 8 as a component of FY1999, these same benefits will continue and expand into FY2000 as the various programs become more integrated into our overall service delivery process. The quantification of these benefits for FY2000 is hindered because these programs are so inter-related that quantification of the benefits or savings must be viewed from the perspective of the total operations

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 7 Witness: Betty L. Adams

## Data Request:

Refer to the Application, Volume 2 of 10, Tab 4, the testimony of Betty L. Adams. On page 5 of the testimony it is stated that Western's O&M budget was converted into National Association of Regulatory Utility Commissioners ("NARUC") accounts. Provide a detailed explanation as to why Western utilized the NARUC accounts rather than following the Uniform System of Accounts prescribed by the Federal Energy Regulatory Commission.

#### Response:

Western has traditionally used the NARUC chart of accounts and converted data as needed for our FERC Form 2. Starting in June of this year with our new financial system we have converted to using the FERC account coding and will no longer use the NARUC account coding.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 8 Witness: Betty L. Adams

## Data Request:

On page 10 of the Adams Testimony is a discussion of changes in Western's operational budget due to the implementation of the Shared Services approach. For each change listed on page 10, compare in dollar amounts the changes in Western's O&M budget with the changes in expenses from the Shared Services program. If the increase in Shared Services expense exceeded the reduction in Western's O&M budget for a particular change, explain in detail why it was cost effective to make the change.

#### Response:

In 1998, Western Kentucky Gas began to experience the effects of significant changes in its operations. These changes included the beginning of a number of key interrelated service delivery process changes, including the establishment of the Customer Support Center and various other business process changes. (See Gruber testimony, pages 12-18.) FY 1998 was truly a year of transition which makes it difficult to compare with either future or prior years.

The appropriate comparison is to compare FY97 to FY99. The decrease in WKG's O&M costs from FY97 to FY99 was \$5.088M while the increase in Shared Services costs from FY97 to FY99 was \$4.281M. Taking into account inflationary increases and the increases in salaries from FY97 to FY99, the reduction is actually greater.

To address the question, however, WKG's O&M costs will decrease from FY98 to FY99 by \$1.253M (See Volume 10, Tab 9, FR 10(9)(h)1 and 10(9)(i)-1.) FY99 is expressed as 6 months actual and 6 months budget. The \$1.253M decrease is itemized below, as is the corresponding increase in Shared Services costs of \$3.075M for the same period.

Total O&M expense for FY98 was \$22.189M. Total FY99 (6 plus 6) is projected at \$24.011M, or an increase of \$1.822M. However, \$2.2M of FY98 capitalized start-up costs for the various service improvements offset the additional labor, benefits and other related costs of transferred responsibilities to the Customer Support Center and credit and collection function during the FY98 transition from the old to new service delivery processes. The result is a net reduction in total cost from FY98 to FY99

	WKG (millions)	S. Svcs (millions)
Office personnel reduction due to closing of offices (amount for 5.5 months)	\$(.750)	\$
Additional labor, benefits and other related costs for above	e	1.186 (1)
Credit & Collections		1.158 (1)
Gas Control function (12 mos.)	(.174)	.110
Accounting functions	(.110)	. <b>.045</b>
Measurement (Meter Shop)	(.105)	
VP Rates (partial year)	.076	(.076)
Vacancies/Pay increases etc.	(.190)	
Various		652
Totals	\$(1.253)	\$3.075
Capitalized start-up cost of various service improvements	<u></u>	<u>(2.200) (1)</u>
Net Increase in Shared Service O&M expenses	\$(1.253)	\$.875

J

1. Additional labor, benefits and other related costs plus Credit & Collections expense offset by FY98 capitalization start up cost of various service improvements.



## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 9 a through f Witness: David H. Doggette

### Data Request:

Refer to the Application, Volume 2 of 10, Tab 5, the Testimony of David H. Doggette, concerning Western's capital budgets.

a. Is it correct that none of the capital budgets submitted in Western's application reflect Western's normal "bottom up" development approach?

#### **Response:**

Yes. The submitted budgets are forecasted. Please see the testimony of David H. Doggette, page 9, lines 7 through 14.

b. Has Western prepared its capital budgets for FY2001 through FY2003, following the "bottom up" approach, utilizing the new Oracle accounting system? If yes, provide the FY 2001 through FY 2003 capital budgets, using the "bottom up" approach, in a format similar to that shown in Exhibit DHD-1.

#### Response:

No.

c. Has Western performed any analysis or review to determine if using a "FY 1999 baseline" approach for its capital budgeting produces a similar result as the "bottom up" approach? If yes, provide copies of that analysis or review. If no, explain why such an analysis or review has not been performed.

## **Response:**

No. Although the "bottoms up" approach was not directly utilized for the submitted budgets, appropriate employees were consulted for input and guidance.

d. Have the FY 2000 and FY 2001 capital budgets filed in this proceeding been entered into the Atmos Capital Budget Gathering System? If no, explain in detail why this has not occurred.

## Response:

No. Capital budgets are entered into the Atmos Capital Budget Gathering System on an annual basis after approval by the Atmos Board of Directors.

e. Provide copies of the approved FY 2000 and FY 2001 capital budgets within 10 days of their approval by Atmos' Board of Directors. Explain in detail all differences between the approved FY 2000 and FY 2001 capital budgets and those submitted with the application.

## Response:

WKG will submit the requested information upon its approval.

f. Provide the capital budgets for the Information Technology strategy and other Shared Services projects, for the FY 2000 through FY 2003 periods. This information should show the amounts for the total budget, as well as Western's forecasted investment, for each fiscal year. Explain in detail how Western's forecasted investment was determined.

### **Response:**

The capital budgets for the Information Technology strategy and other Shared Services projects for FY 2000 through FY 2003 is attached as DR 9, Schedules 1 & 2. Western's forecasted investment was determined by using Western's percentage of customers to the total Atmos customer count, approximately 17%.

# DR 9, Schedule 1

# TOTAL ATMOS BUDGET

IT Strategy	\$ <u>FY 2000</u> 11,428,000	\$ <u>FY 2001</u> 8,470,000	\$ <u>FY 2002</u> 10,124,000	\$ <u>FY 2003</u> 2,015,000
Shared Service	\$ 451,000	\$ 443,000	\$ 442,000	\$ 450,000
Total Capital	\$ 11,879,000	\$ 8,913,000	\$ 10,566,000	\$ 2,465,000

# DR 9, Schedule 2

.

# WESTERN'S FORECASTED INVESTMENT

	<u>FY 2000</u>	<u>FY 2001</u>	<u>EY 2002</u>	<u>FY 2003</u>
IT Strategy	\$ 1,980,000	\$ 1,468,000	\$ 1,754,000	\$ 349,000
Shared Service	\$ 76,000	\$ 74,000	\$ 74,000	\$ 75,000
Total Capital	\$ 2,056,000	\$ 1,542,000	\$ 1,828,000	\$ 424,000
			Y,	

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 10 Witness: David H. Doggette

## Data Request:

Refer to the Application, Volume 2 of 10, Tab 5, Exhibit DHD-1, page 2 of 6. Provide the workpapers and assumptions used to determine that the projected overhead rate should be 50 percent and the projected increase in maintenance and improvements should be 36.25 percent for the FY 2000 capital budget. Include an explanation as to how these determinations were made.

## **Response:**

The 50 percent overhead rate was based on a review of Atmos' Administrative & General (A&G) overhead dollars and Western's business unit overhead dollars for the FY 1999. The Atmos A & G allocation was \$1,106,000 and the Western business unit allocation was \$1,840,000. Which when added to Western's direct capital expenditures for FY 1999 results in a total capital budget of \$8,408,000. The FY 1999 overhead rate is therefore 53%, hence a similar rate of 50% was applied for overheads in FY 2000. The projected 36.25% increase in maintenance and improvements for FY 2000 is a result of prudent management practices during abnormal weather periods, such as deferring certain non-safety related projects.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 11 Witness: Donald P. Burman

## Data Request:

Refer to the Application, Volume 2 of 10, Tab 6, the Testimony of Donald P. Burman. On pages 6 and 7 of the testimony is a discussion of Western's accounting for pension expense.

- a. Provide the balances of Western's pension assets and obligations shown on its balance sheet as of March 31, 1999, as well as for the base period and forecasted period. Include the account numbers and titles used to record these assets and obligations.
- b. Has Western included its pension assets and obligations in its calculation of its rate base? If so, provide the reference to the appropriate schedules and workpapers that disclose this inclusion.

#### Response:

a. The balance of Western's pension assets shown on it's balance sheet as of March 31, 1999, as well as for the base period and forecasted period was \$13,331,467. This amount was included in account number 1860740490, Pension Assets Noncur WKG.

b. No.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 12 Witness: John P. Reddy

# Data Request:

Provide the actual capital ratios for Atmos as of June 30, 1999.

### Response:

Balance sheet information for the month of June 30, 1999 is not yet available as the quarter closing process is still underway. Atmos will release third quarter earnings on August 11, 1999 and balance sheet data for June will be available at that time.

1

Atmos actual capital ratios as of May 31, 1999 are as follows:

	Amount	Percent
Long-term Debt*	\$402,797,957	44.39%
Short-term Debt	\$102,956,149	11.35%
Shareholders' Equity	\$401,556,013	44.26%
Total	\$907,310,119	<u>100.00%</u>

\*Includes current maturities portion of long-term debt.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 13 a and b Witness: John P. Reddy

## Data Request:

Refer to the Application, Volume 2 of 10, Tab 7, the Testimony of John P. Reddy. On page 4 of the testimony, it is stated that Atmos' objective is to maintain a capital structure comprised of approximately 50 percent equity and 50 percent debt. However, on page 6 of the Reddy Testimony, the summary of Atmos' five-year financial plan for FY 2001 through FY 2003 shows a capital structure comprised of equity ranging between 53 percent and 61.9 percent.

- a. Explain why Atmos' five-year financial plan presents capital structures that are significantly different than Atmos' expressed capital structure objectives.
- b. The equity portion of Atmos' capital structure as of March 31, 1999 was 44.19 percent. The equity portion shown on page 6 of the Reddy Testimony shows a forecasted equity component of 49.8 percent for FY 2000. Explain in detail why it is reasonable to expect such an improvement in the equity in less than a two-year time frame.

#### Response:

a. The reasons for the improvement in the capital ratios are explained beginning on Line 18, page 6 of the Reddy Testimony. The improvement reflects a number of assumptions including successfully reducing the impact of weather on earnings and cash flows, issuing new equity through benefits plans and through a public offering, and a return to normal capital spending levels. Weather and capital spending patterns are shown below.

	1996	1997	1998	1999E	2000E
Degree Days - % of normal	101%	98%	95%	TBD*	100%
Capital Spending (Millions)	\$118	\$122	\$135	\$102	<b>\$80</b>

\*For the first six months of FY1999, weather conditions were 16% warmer than normal and 13% warmer than 1998.

It is reasonable to assume a reduced negative impact of weather through a combination of a return to more normal weather patterns (experienced as recently as 1997 and 1996) and approval of WNA's as discussed in Mr. Reddy's testimony. It is also reasonable to recognize the reduced capital spending levels made possible by completion of the major service improvement initiatives discussed in Mr. Gruber's testimony. Finally, it is reasonable to recognize the effects of ongoing and anticipated equity

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 13 a and b Witness: John P. Reddy

issuances. However, it is not possible to know with certainty that all of the Company's assumptions will be realized. For that reason, the five-year financial plan is updated annually. If warmer-than-normal weather should continue, or reappear in future plan years, the improvement in Atmos' capital ratios will be affected. Also, the Company's success in making future acquisitions will affect capital structure and provide opportunities to make adjustments to keep overall debt/equity ratios in line with stated objectives.

b. The reasons for this improvement in the equity component of capital structure are explained in Mr. Reddy's testimony, beginning on page 6, line 18.
Elsewhere in his testimony, Mr. Reddy explains why the Company's debt balances have exceeded target levels. See especially page 4, line 8 through page 5, line12 of Mr. Reddy's testimony. To recap the reasons for improvement in the equity ratio, they include:

Reducing earnings sensitivity to weather patterns through WNA's, weather hedging/insurance, rate design in other Atmos' LDC jurisdictions that shifts a greater portion of fixed cost recovery to the monthly charge, and similar initiatives.

Reduced levels of capital spending as major service improvement initiatives are completed and placed in service.

Additional equity issuances through Company benefits plans and public offerings. The Atmos Board of Directors will be asked to approve the filing of a Universal Shelf Offering with the SEC at the Company's August 1999 Board meeting. This will enable the Company to issue equity as needed in FY2000 to meet its capital structure objectives, to refinance existing long-term debt, to finance acquisitions and otherwise create financial flexibility.

It is reasonable to expect that the Company can achieve its capital structure objectives through these initiatives in the anticipated time frame.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 14 a through f Witness: John P. Reddy

### Data Request:

Refer to pages 6 and 7 of the Reddy Testimony. For each of the assumptions listed below, indicate how reasonable that assumption is for Atmos. Include any analyses, studies, or other documentation that support these assumptions.

- a. A return to normal long-term weather patterns for the other Atmos utility divisions beginning in FY 2000.
- b. The issuance of \$26 million of new equity in November 1999.
- c. Raising \$20 million of new equity annually through stock plans.
- d. No significant acquisitions. Explain how this assumption is not in conflict with the statements contained on page 5 of the 1998 Atmos Annual Report to shareholders.
- e. Cash flow from depreciation will fund ongoing capital spending requirements.
- f. No new long-term debt issues.

#### Response:

- a. This is a reasonable assumption as explained in the response to DR Item 13a. The Atmos utilities experienced normal weather conditions in 1996 and 1997 and for planning purposes, it is reasonable to assume a return to normal long-term weather patterns. Further, Atmos is taking other steps to reduce the effects of weather on earnings, cash flow and capital structure as discussed in Mr. Reddy's Testimony and summarized in the response to DR Item 13. This is, therefore, a reasonable assumption.
- b. As explained in the response to DR 13b, the Atmos Board of Directors will be asked to approve the filing of a Universal Shelf Offering with the SEC at the Company's August 1999 Board meeting. This will enable the Company to issue equity as needed in FY2000 to meet its capital structure objectives, to refinance existing long-term debt, to finance acquisitions and otherwise create financial flexibility. This is, therefore, a reasonable assumption.
- c. As explained in the response to DR 15, Assumption 1, the Company's various benefit plans and the Direct Stock Purchase Plan generate new equity issuances of approximately \$20 million annually. This is, therefore, a reasonable assumption.
- d. As explained in the response to DR 15, Assumption 8, Atmos expects to continue to make acquisitions but cannot accurately predict the timing, size



# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 14 a through f Witness: John P. Reddy

or financial implications of future acquisitions. It is, therefore, reasonable to build a base financial plan that excludes the impact of future acquisitions.

- e. Cash flow from depreciation will be sufficient to fund ongoing capital spending requirements if the other assumptions reflected in the financial plan are reasonable. These other assumptions include reducing the effects of weather (explained in the response to DR 13a and b) and reducing capital spending to historical levels (explained in the response to DR 13a). This is, therefore, a reasonable assumption.
- f. No new long-term debt issues will be required if the other assumptions reflected in the financial plan are reasonable. These other assumptions include reducing the effects of weather (explained in the response to DR 13a and b), reducing capital spending to historical levels (explained in the response to DR 13a), and ongoing equity issuances through Company benefit plans and public offerings (explained in the response to DR 15, Assumption 2, and DR 13b).
#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 15 Witness: John P. Reddy

#### Data Request:

Refer to the Reddy Testimony, the attachment titled "FR10(9)(h)11," Sheets 1 and 2 of 3. Explain why each of the assumptions listed on these schedules is reasonable. Include any analyses, studies, or other documentation that support these assumptions.

#### Response:

The referenced assumptions are:

- 1. "All numbers are actual through March 1999." This refers to the numbers on the schedule for debt and equity balances for the Base Year Capital Structure schedules. Actual debt and equity balances are reflected for the months of September 1998 through March 1999 (latest month available before the rate case filing date) with financial plan estimates reflected thereafter.
- 2. "Assumes base stock capital growth rate of \$20 million per year via various plans." Atmos has an Employee Stock Ownership Plan (ESOP) for employees pursuant to which Atmos may make a matching contribution for the account of the participant in an amount determined each year by the Board of Directors. (Prior to 1999, the Company sponsored a 401(k) savings plan for the United Cities Division employees (UCG Plan) pursuant to which the Company made certain matching contributions.) Atmos also has a Direct Stock Purchase Plan ("DSPP") with a Dividend Reinvestment feature for DSPP participants. These plans produced increases in the Company's equity balances of over \$14 million in FY1998 and nearly \$10 million in the first six months of FY1999.

	1998 Shares Issued*
ESOP	52,473
DSPP	531,353
UCG Plan	55,500
	639,326

\*As shown on page 37 of the 1998 Annual Report to Shareholders.

It is reasonable to assume that these plans will continue to add roughly \$20 million annually to Atmos' equity base.

3. "1999 monthly Net Income is per the monthly budget." As part of the fiveyear financial plan, detailed monthly budgets are prepared for the initial year

#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 15 Witness: John P. Reddy

of the plan. 1999 is the first year of the latest plan and it is reasonable to use the monthly net income figures from the plan to forecast net income for April through September, the end of Atmos' fiscal year.

- 4. "2000 monthly net income is patterned after the 1999 budget." It is reasonable to start with 1999's monthly net income and make adjustments for anticipated differences in FY2000, the most significant of which are discussed in Assumption 5, below.
- 5. "Net income is adjusted monthly for: 1) additional short-term interest based on a 6% annual rate on \$22,000,000 overall Net Income shortfall due to warm weather; 2) reduced short-term interest due to any issuance of equity (assumed to pay down ST debt); and 3) additional depreciation related to various service initiatives."

The current five-year financial plan was completed in September of 1998. Since then, certain assumptions that were relied upon in forecasting 1999 results have not borne out, requiring that appropriate adjustments be made to 1999 figures and subsequent balances. With regard to assumption 1) above, plan numbers for 1999 monthly net income have been adjusted to reflect the effects of lower net income (estimated at \$22 million below plan) due to the impact of significantly warmer than normal weather. One impact of reduced net income is the need to increase short-term borrowings in 1999 above plan levels, creating higher short-term interest expense (6% interest rate applied to \$22 million increased short-term debt balance to offset decline in net income). With regard to assumption 2), new equity issuances, whether public issues or through benefits plans, are assumed to reduce short-term borrowing requirements and, hence, short-term debt balances. With regard to assumption 3), additional annual depreciation expense has been reflected associated with the in-service dates of the service improvement initiatives discussed in Mr. Gruber's testimony, but not fully budgeted in the September five-year financial plan.

- 6. "Dividends are adjusted to reflect any new issuance of equity; they are calculated on the monthly ending shares outstanding." This is a straightforward assumption since quarterly dividends are paid on all shares outstanding, including those issued to the public and those issued through various Company benefits plans.
- "Assumed dividend rate is \$1.10 for FY1999, \$1.14 for 2000, and \$1.18 for 2001." \$1.10 is the actual indicated dividend rate for 1999 (\$0.275 per quarter). Although it is up to the Atmos Board to establish future dividend

Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 15 Witness: John P. Reddy

rates, as indicated on page 52 of the 1998 Annual Report to Shareholders, Atmos has increased the dividend for 11 consecutive years, and by 4 cents annually in each of 1996, 1997, 1998 and 1999. It is therefore reasonable to budget 4 cent annual increases in the dividend for financial plan purposes.

- 8. "Long-term debt balances are per financial plan, with no additional issuances for acquisitions." Atmos has completed five major acquisitions over the past twelve years and expects to continue to make acquisitions in the future. However, as discussed on page 5 of the 1998 Annual Report to Shareholders, "We continually evaluate opportunities. However, the decision is not solely in our hands." Because it is impossible to predict with any degree of accuracy the timing, size and financial implications of future acquisitions, Atmos has not included any acquisitions in its financial plan. However, should Atmos complete a merger transaction within the plan period, that would likely present the opportunity to accelerate or maintain the Company's objective of achieving a 50/50 debt/equity capital structure. For example, many utility acquisitions are accomplished by the acquiring company issuing stock to be exchanged for shares of the acquired company, thereby raising the percentage of equity in the capital structure of the acquiring company.
- 9. "Short-term debt is per the 5-year financial plan, reduced for equity issuances, increased for Net Income shortfall due to warm weather in 1999, increased for additional dividends related to stock issuances, and increased for tax-effected change in short-term interest (one iteration only)." These assumptions and their reasonableness have been explained in items 1-8, above. For ease of modeling, the after-tax amount of interest on the additional short-term debt was added to the month of April 1999; hence the reference to "one iteration only."

3

#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 16 Witness: Marks

#### Data Request:

Refer to the Application, Volume 2 of 10, Tab 12, the Testimony of Michael Marks. On page 4 of the testimony it is stated "Central to the program are two important provisions of the agreement between Western and the Commission." In the August 10, 1995 Order in Case No. 95-010, the Commission stated:

Thus, the Settlement merely establishes a framework for developing a program which will qualify for rate recovery under KRS 278.285. No specific programs or related cost recovery mechanism have been included. Therefore, the Commission makes no decision or findings of fact related to any portion of the DSM provisions included in the Settlement. (Footnote – Case No. 95010, August 10, 1995 Order, at 4.)

In light of the Commission's statement in Case No. 95-010, explain in detail what agreement is being referenced in Mr. Marks testimony on page 4.

#### Response:

Correctly stated, Mr. Marks was referring to the Commission's understanding of the process Western had agreed to undertake in implementing the program. The agreement itself was between the various parties to Case No. 95-010, as a result of the settlement of that case. The Commission approved the overall settlement and the program was implemented within the framework of that settlement.

#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 17.a. Witness: Marks

#### Data Request:

17. Refer to the Marks Testimony, page 4. Concerning the Western Demand Side Management ("DSM") Collaborative:

a. Identify the members of the Western DSM Collaborative at its inception and as of June 30, 1999.

#### Response:

Inception:

J. F. Carnahan, Senior V. P., Technical Services, WKG

Hon. Ann Louise Cheuvront, Assistant Attorney General, Office of the Attorney General, Commonwealth of Kentucky

Scott Crocker, Executive Director, Cumberland Trace Legal Services

Kip Bowmar, Executive Director, Kentucky Association for Community Action, Inc.

As of June 30, 1999:

W. J. Senter, V. P., Rates and Regulatory Affairs, WKG

Hon. Ann Louise Cheuvront, Assistant Attorney General, Office of the Attorney General, Commonwealth of Kentucky

Scott Crocker, Executive Director, Cumberland Trace Legal Services

Kip Bowmar, Executive Director, Kentucky Association for Community Action, Inc.

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 17. b. Witness: Marks

### Data Request:

17. Refer to the Marks Testimony, page 4. Concerning the Western Demand Side Management ("DSM") Collaborative:

b. Provide copies of the bylaws or other documents that govern the operation of the Western DSM Collaborative.

### Response:

A copy of the bylaws is attached.

ATTACHMENT = DR 17. 6.

WESTERN KENTUCKY GAS COMPANY BY-LAWS OF THE DSM COLLABORATIVE

**PURPOSE:** The intent of these by-laws is to govern the process for operational procedures of the Western Kentucky Gas Company Demand-Side Management Collaborative.

#### ARTICLE I MEMBERSHIP

Membership in the Collaborative is to be representative of Residential customers of Western Kentucky Gas Company. There shall be no individual members.

Section 1 -- Executive Committee

The founding members of the Collaborative are:

(1) Community Action Agencies

Kip Bowman		Primary
Staff Member	KAC	Alternate
Staff Member	KAC	Alternate

(2) Cumberland Trace Legal Services

Scott Crocker	Primary
Norma Perry	Alternate
Margaret Ryan	Alternate

(3) Office of the Attorney General, Kentucky

Ann Louise Cheuvront Primary Staff Member/Att'y Gen. Alternate Staff Member/Att'y Gen. Alternate

(4) Western Kentucky Gas Company

Jay Carnahan	Primary
Larry Moore	Alternate

(5) Kentucky Division of Energy (Non-Voting)

Primary Alternate

(6) Public Service Commission of Kentucky (Non-Voting)

> Primary Alternate

Section 2 -- Technical Committee Membership

(1) Audubon Area Community Services

Sheila White

Primary Alternate

(2) Blue Grass Community Action Agency

Houston Jones

Primary Alternate

(3) Community Action of Southern Kentucky

Betty Vincent

Primary Alternate

(4) Pennyrile Allied Community Services

Martha McGill Primary Alternate

(5) West Kentucky Allied Services

Thelma Sayers Primary Alternate

(6) Western Kentucky Gas Company

Jim Allison Primary Alternate

(7) Western Kentucky Gas Company

Garry Milligan Primary Gary Smith Alternate

(8) Western Kentucky Gas Company

Ben Boyd Primary Alternate

(9) State Energy Assistance Branch

Gary Brown Primary Alternate

Section 3 -- New Members

A petition for membership may be made by filing a written request with the Chairperson of the Executive Committee. Application for membership will be considered by the entire membership and may be added by unanimous approval of the Collaborative.

Section 4 -- Duties of Membership

Members will be responsible for the following:

- >> Designate in writing one (1) voting representative and up to two (2) alternates to represent a customer member
- >> Attend collaborative meetings in person, by proxy, or by telephone
- >> Stay informed on issues relating to DSM
  programs and activities
- >> Review, recommend, and endorse DSM programs for Western Kentucky Gas.

Section 5 -- Termination of Membership

Any member groups may voluntarily terminate their membership at any time via filing written notice with the Chairperson.

> Automatic termination shall occur if a member fails to attend personally, by telephone, or by proxy, three (3) consecutive meetings of the Collaborative. Consideration for returning to the Collaborative may be given by means of unanimous vote.

#### ARTICLE II VOTING RIGHTS

- Section 1 -- All votes of the Executive Committee require a unanimous agreement for passage. Abstaining votes shall be considered agreement. Passage or recommendations of the Technical Committee requires a majority vote. Abstaining votes of the Technical Committee shall be considered non-votes.
- Section 2 -- Each member shall have one (1) vote to be cast by the authorized representative or designated alternate.
- Section 3 -- Proxy votes may be made in writing. Only the authorized representative may submit proxy votes.

#### ARTICLE III DECISION MAKING

Section 1 -- General Decision-Making

All voting members of the Executive Committee will vote on issues affecting Western Kentucky Gas DSM programs, including

> collaborative membership and rule-making. If a member is absent from a meeting and all other members present agree upon an item, the issue will be outlined in the meeting minutes, and the absent member will have five (5) working days upon receipt of the minutes to present an objection to the issue, if necessary.

Section 2 -- Sub-Committee Decision Making

The Technical Committee will develop, consider, and recommend DSM programs to the Executive Committee after deciding by majority vote.

Section 3 -- Changes in Programs

Once a program is established, the terms of the program may be changed only by unanimous consent.

#### ARTICLE IV DISPUTE RESOLUTION

- Section 1 -- In the event a unanimous decision cannot be reached by the utility members of the Executive Committee, the members shall determine whether the impasse is informational or policy-based.
- Section 2 -- In the event an informational dispute exists, the Collaborative may agree on an expert to analyze the disputed information and advise them as to what is accurate.

- Section 3 -- If a policy dispute should occur, the Collaborative or sub-group may secure an outside facilitator to assist the group with reaching an acceptable agreement.
- Section 4 -- No action will be taken without a unanimous agreement. Abstaining votes shall be considered agreement.

#### <u>ARTICLE V LEADERSHIP</u>

- Section 1 -- A Chairperson may convene and preside at collaborative meetings. Additionally, the sub-group may have a Chairperson.
- Section 2 -- The Collaborative may select the Chairperson by unanimous vote of the members and may rotate among the members every year. The sub-group may select a Chairperson by a unanimous vote on an annual basis. The Chairperson may serve for a twelve-month period with the election of a new Chairperson to be done in the month of December.

ARTICLE VI MEETINGS

- Section 1 -- Meetings of the Collaborative shall be held at least once per quarter.
  - Section 2 -- The Technical Committee sub-group may meet monthly (or more frequently at the call of

> the Chairperson of the Collaborative or that sub-group or at least two [2] members of the Collaborative or sub-group). However, the sub-group must meet once per quarter.

Section 3 -- Reasonable written notice shall be given to all members in advance of meeting, providing the date, time, and location of the meetings.

Section 4 -- Minutes shall be kept of all meetings and circulated for approval to the membership.

#### ARTICLE VII CONTRACTS, CONSULTANTS, AND ADMINISTRATION

- Section 1 -- Costs for consultants and other outside entities retained by the sub-group shall be recovered only from the funds available from the WKG Care's DSM program.
- Section 2 -- Contracts between the Collaborative, consultants, and outside entities shall be authorized by the Collaborative and signed by the Chairperson.
- Section 3 -- All contracts with consultants and outside entities will specify:
  - (A) The name of a contact person from the Collaborative or sub-group;

> (B) That all parties to the contract shall have access to information submitted by the consultant or outside entity.

#### ARTICLE VIII AMENDMENT

- Section 1 -- Amendment of Western Kentucky Gas' Demand-Side Management By-Laws shall only be approved by unanimous consent of all Collaborative members.
- Section 2 -- Any amendment proposal to these by-laws shall be made by a member in writing; full text of the amendment shall be provided to all members of the Collaborative at least ten (10) days prior to the meeting at which the amendment is to be voted upon.

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 17. c. Witness: Marks

Data Request:

17. Refer to the Marks Testimony, page 4. Concerning the Western Demand Side Management ("DSM") Collaborative:

c. Identify the chairperson of the Western DSM Collaborative as of June 30, 1999

Response:

W. J. Senter, V. P., Rates and Regulatory Affairs, WKG

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 18 Witness: Marks

#### Data Request:

- 18. Refer to the Marks Testimony, page 8. For each of the measures ultimately selected for inclusion in the WKG CARES program, provide the results of the following DSM benefit cost test:
- a. Total Resource Cost ("TRC").
- b. Utility Cost Test ("UCT").
- c. Ratepayer Impact Measure ("RIM").
- d. Participant Test ("PT").

#### Response:

See attachment entitled "Pre-Implementation Benefit/Cost Screening Results" dated October 25, 1996, Page 7, Table 1.

HACHMENT



# WESTERN KENTUCKY GAS COMPANY COMMUNITY ACTION RESIDENTIAL ENERGY SERVICES

# PRE-IMPLEMENTATION BENEFIT/COST SCREENING RESULTS

Prepared by



HAMPTON STRATEGIES, INC. 898 Veterans Memorial Highway Hauppauge, New York 11788 (516) 234-2400

October 25, 1996

## INTRODUCTION

The Western Kentucky Gas Company Demand-Side Management Program Collaborative (Collaborative) is seeking to offer a number of energy conservation/efficiency measures to low-income customers of Western Kentucky Gas (WKG) through the Western Kentucky Gas Company Community Action Residential Energy Services (WKG CARES) Program. This document describes the results of preliminary benefit/cost analyses performed by Hampton Strategies, Inc. (Hampton) for the specific measures considered by the Collaborative.

As part of the WKG CARES Program design, the Collaborative prepared a preliminary list of demand-side management (DSM) measures for consideration. These measures were developed with input from the five community action agencies through which the WKG programs will be implemented. These five agencies are currently providing energy-related services to the low-income community. The preliminary measures are:

- 1) Attic Insulation
- 2) Clean and Tune Furnace
- 3) Water Heater Replacement
- 4) Furnace Replacement
- 5) Wall Insulation
- 6) Floor Insulation
- 7) Duct Repair
- 8) Duct Insulation
- 9) Water Heater Wraps
- 10)Infiltration Reduction

Hampton was retained by the Collaborative to provide consulting services related to the design and evaluation of the WKG CARES Program. As part of this effort, Hampton performed a series of benefit/cost analyses for these measures based on estimates ofmeasure costs, avoided gas and electricity costs, measure impacts, and other necessary input assumptions.

It should be recognized that, at this pre-implementation/design stage, there are no WKG-specific data regarding program costs and impacts. Specific assumptions are made based on the experience of the Weatherization Directors from the five community action agencies and the State Weatherization Office, as well as Hampton's experience with similar DSM programs offered elsewhere. These assumptions allow for the assessment of estimated benefits and costs of these measures, and hence, provide program design guidance. Actual results may vary and an ongoing program evaluation

[m[252]

Page 1



effort will serve to measure these results and allow for subsequent program redesign and optimization.

This document describes the DSM measures analyzed, the benefit/cost tests conducted, and the results of those tests. Hampton's recommendations regarding measures that can be included in the WKG CARES are located at the end of this document.

## **DESCRIPTION OF MEASURES**

This section describes the measures considered as part of the WKG CARES Program design. Each of the measures is described along with the cost and impact assumptions used in the benefit/cost analyses. More detailed information regarding the assumptions used in the analyses are provided in the DSM Screening and Assessment Assumption Worksheets provided in Attachment A. Note that all cost estimates described in each measure and submeasure include materials, labor, and an overhead factor of 20%.

- 1) Attic Insulation: Install attic insulation in homes where such insulation is either nonexistent or insufficient. This measure will reduce thermal loss through the attic and thereby increase the home's thermal efficiency. The Weatherization Directors and the State Weatherization Office estimate that attic insulation can save 15-18% on heating bills in homes that have no existing attic insulation and 10-15% in homes that have some attic insulation. These estimates are reasonably consistent with estimates used elsewhere. An estimate of 15% energy savings was used in the benefit/cost analyses, based on the assumption that homes taking this measure would be those with minimal or no attic insulation. Based on experience, the Weatherization Directors and the State Weatherization Office estimate that the cost of this measure is approximately \$400 per home.
- 2) Clean and Tune Furnace: Repair, clean, and tune existing furnaces to increase furnace efficiency. The State Weatherization Office estimates that this measure could result in 8-10% energy savings. Based on these estimates, Hampton assumed an 8% energy savings in the benefit/cost analyses. Based on experience, the State Weatherization Office estimates that the costs of this measure are approximately \$150 per home.
- 3) Water Heater Replacement: Replace inefficient gas water heaters and older electric water heaters with new gas water heaters. In order to differentiate between these two cases, the following submeasures were analyzed:
  - a) Water Heater Replacement (Gas-to-Gas): Replace older, inefficient gas water heaters with new gas water heaters. The following assumptions were made: 1) the replacement water heater has an efficiency factor of .54 (current NAECA minimum); 2) the older water heater has an efficiency factor of .50, which is probably conservative since many older water heaters often have lower efficiency factors; 3) the installation of low-flow shower heads are included as part of this measure. Consistent with other programs offering low-flow shower heads, it is estimated that this will



result in a further savings of 5%. Based on experience, the Weatherization Directors estimate that the cost of a new gas water heater is approximately \$700 per home. The low-flow shower heads are estimated to cost \$20 per home. The costs are consistent with similar programs conducted elsewhere.

- b) Water Heater Replacement (Electric-to-Gas): Replace older electric water heaters with new gas water heaters where gas service to the house already exists and the electric water heater is at least 5 years old. The following is assumed: 1) the replacement water heater has an efficiency factor of .54 (current NAECA minimum); 2) the older electric water heater has an efficiency factor of .70; 3) low-flow shower heads would be installed as part of this measure to result in a further savings of 5%. Based on experience, the Weatherization Directors estimate that the costs of a new water heater are approximately \$700 per home while the low-flow shower heads are estimated to cost an additional \$20 per home.
- 4) *Furnace Replacement*: Replace inefficient gas furnaces with either standard or high-efficiency gas furnaces. In order to differentiate between these two cases, the following submeasures were analyzed:
  - a) *Furnace Replacement (Standard Efficiency)*: Replace existing inefficient gas furnaces with standard efficiency gas furnaces. It is assumed that: 1) the existing furnace has an AFUE of 60% (which is conservative, since it is the maximum efficiency for which this measure will be implemented); 2) the replacement furnace has an AFUE of 82%, close to the NAECA minimum). Based on experience, the Weatherization Directors estimate that the cost of a new standard efficient furnace is approximately \$1200 per home.
  - b) Furnace Replacement (High Efficiency): Replace existing inefficient gas furnaces with high-efficiency gas furnaces (assuming technical feasibility given the need for sidewall venting). These assumptions were used: 1) the existing furnace has an AFUE of 60% (which is conservative, since it is the maximum efficiency for which this measure will be implemented) and 2) the replacement furnace has an AFUE of 90%. Based on experience, the Weatherization Directors estimate that the cost of a new high efficient furnace is approximately \$1450 per home.
- 5) Wall Insulation: Install wall insulation in homes where such insulation is either nonexistent or insufficient. This measure will reduce thermal loss through the walls and increase the home's thermal efficiency. The Weatherization Directors and the State Weatherization Office estimate that attic insulation can save up to 15% on heating bills. Hampton assumed an 8% energy savings in the benefit/cost analyses, which is also consistent with the estimate used for attic insulation (in that an average home would save more energy from attic insulation than wall insulation). Based on experience, the Weatherization



Directors and the State Weatherization Office estimate that the cost of this measure is approximately \$500 per home.

- 6) *Floor Insulation*: Install floor insulation in mobile homes where such insulation is either nonexistent or insufficient in order to reduce thermal loss through the floor and increase the mobile home's thermal efficiency. The Weatherization Directors and the State Weatherization Office estimate that floor insulation can save up to 15% on heating bills in mobile homes. An estimate of 5% energy savings was used in the benefit/cost analyses. Based on experience, the Weatherization Directors and the State Weatherization Office estimate that the cost of this measure is approximately \$400 per home.
- 7) Duct Repair: Repair ducts in nomes where such ducts leak into unconditioned spaces in order to improve the efficiency of the space heating system. The State Weatherization Office and the Weatherization Directors estimate that this measure can save 15% on heating bills, and this estimate is reasonably consistent with estimates used elsewhere. Hampton assumed a 10% energy savings in the benefit/cost analyses. Based on experience, the Weatherization Directors and the State Weatherization Office estimate that the cost of this measure is approximately \$150 per home.
- 8) **Duct Insulation**: Install insulation on ducts in unconditioned portions of the home in order to improve the efficiency of the space heating system. The State Weatherization Office estimates that this measure can save 10% on heating bills. An estimate of 5% energy savings was used in the benefit/cost analyses. Based on experience, the Weatherization Directors and the State Weatherization Office estimate that the cost of this measure is approximately \$200 per home.
- 9) Water Heater Wraps: Install insulating wraps on gas water heaters to reduce thermal losses. The State Weatherization Office estimates that this measure would save 1-2% of the water heating bills. An estimate of 1% reduction in water heating energy use was assumed in the benefit/cost analyses. The State Weatherization Office estimates the cost of this measure at approximately \$20 per home.
- 10) Infiltration Reduction: Reduce air infiltration through the installation of caulking, weather stripping, etc. Based on experience elsewhere, Hampton . assumed an estimate of 10% savings of space heating energy use in the benefit/cost analyses. The State Weatherization Office estimates the cost of this measure at approximately \$250 per home.



## **DESCRIPTION OF BENEFIT/COST TESTS**

The benefit/cost tests utilized in the analyses of the proposed DSM measures for the WKG CARES program are those described in the *Standard Practice Manual: Economic Analysis of Demand-Side Management Programs* (Standard Practice Manual), jointly developed by the California Public Utilities Commission and the California Energy Commission. A brief description of each of these tests follows:

- The Participant Test compares the bill savings plus incentives provided to customers with the costs of the measure. This test addresses the net benefits of the measure from the perspective of those customers who participate in the program.
- The Utility Test compares the reductions in the utility's supply costs to the costs to the utility to implement the program (i.e., administrative and incentive costs). This test addresses the net costs of providing a resource option from the perspective of the utility.
- The Ratepayer Impact Measure (RIM) Test addresses what could happen to customer rates as a result of changes in utility revenues and operating costs associated with the measure. Sometimes referred to as the "Non-Participant Test," this test addresses the perspective of the utility customer who does not participate in the program since it reflects changes in rates without the net benefits of direct participation.
- The Total Resource Cost (TRC) Test compares the reduction in the utility's supply costs to the incremental costs of the measure (i.e., administrative and customer costs). This test is the broadest perspective in that it represents the net benefits to society, including both participants and non-participants. (A variant of the TRC Test is the "Societal Cost Test" which also includes the net benefits of externalities. The Societal Perspective was not considered in this analysis).

The specific benefit and cost components of each of these tests is shown in Figure 1.

In this document, the results of the benefit/cost analysis are expressed as a benefit/cost ratio. A result of 1.0 indicates benefits are equal to costs, a result less than 1.0 indicates costs are greater than benefits, and a result greater than 1.0 indicates benefits are greater than costs.





Figure 1 STANDARD COST-BENEFIT TESTS

It should be noted that the benefit/cost tests were conducted independently for each measure. In other words, it was assumed that the specific measure would be implemented absent other measures. While this provides an indication of the attractiveness of the measure from various perspectives, it does not take into account the interactive effects of many of these measures. For example, a furnace replacement measure might reduce energy consumption by "X" if no other changes are made to the house. Similarly, attic insulation might reduce energy consumption by "Y" if no other changes are made to the house. However, if both of these measures are implemented, the net reduction in energy use will be less than "X+Y."



Given the large number of permutations that arise from the various pairing of measures, the lack of WKG program-specific data, the lack of information relative to the likelihood of each pairing, and the lack of data relative to the "interactive effects" of each pairing, accounting for this interactive effect was not considered to be necessary at the design phase of the program. Instead, conservative estimates were used for energy impacts for each measure in this pre-implementation benefit/cost analysis. Measures that pass this analysis are expected to have a reasonable probability of being cost-effective, despite interactive effects.

## RESULTS

The results of the benefit/cost analyses are presented in Table 1.

	Table NEFITS/COST ANA	1 Lyses Results.		
Measure	Participant Test	Utility Test	RIM Test	TRCTest
1) Attic insulation	2.89	1.17	0.47	1.27
2) Clean & Tune Furnace	2.71	0.98	0.42	1.03
3) Water Heater Replacement a. Gas-to-Gas	1.30	0.15	0.12	0.16
b. Electric-to-Gas	1.73	0.62	0.52	0.64
4) Furnace Replacement a. Standard Efficiency	2.06	0.65	0.36	0.70
b. High Efficiency	2.09	0.67	0.36	0.72
5) Wall Insulation	1.81	0.50	0.31	0.55
6) Floor Insulation	1.62	0.39	0.26	0.42
7) Duct Repair	3.14	1.23	0.46	1.28
8) Duct insulation	1.79	0.46	0.28	0.48
9) Water Heater Wraps	1.46	0.21	0.16	0.22
10) Infiltration Reduction	3.02	1.25	0.49	1.37
Aggregate Results:	2.20	0.77	0.41	0.82



As indicated on Table 1, each of the measures pass the Participant Test (i.e., the results are 1.0), which is to be expected since participants receive the benefits of the measure and the costs of the measure are all paid for by the WKG CARES program. Each of the measures also fail the RIM Test (i.e., the results are 1.0), which is typical of measures that reduce overall energy consumption. The most relevant test for these types of programs is the TRC Test since it offers the broadest perspective.<sup>1</sup> As indicated in Table 1, only four of the measures pass the TRC Test.

In order to provide some options regarding the mixture of measures to be included in the WKG CARES Program, the following three scenarios were assessed:

- 1. Implement Only Those Measures That Independently Pass the TRC Test
- 2. Implement The Maximum Number of Measures That Pass TRC In Aggregate
- 3. Develop Cost-Sharing Arrangements to Allow For Full Spectrum of Measures to be implemented

Each of these three scenarios are discussed as follows:

FIRST SCENARIO:

### Implement Only Those Measures That independently Pass the TRC Test

The most stringent criteria for measure selection is for each measure to pass the TRC Test. In this case the only measures to be implemented would be:

- Attic Insulation
- Clean and Tune Furnace
- Duct Repair
- > Infiltration Reduction

In this scenario, the three-year direct measure expenditures by WKG CARES would be approximately \$500,000, given the number of participants per measure estimated by the State Weatherization Office. This scenario assumes that all program and measure costs are borne by the WKG CARES Program.

SECOND SCENARIO:

### Implement The Maximum Number of Measures That Pass TRC In Aggregate

This scenario is intended to maximize the number of measures implemented and recognize the importance of the TRC Test, given full funding by WKG CARES. The procedure used was to drop measures beginning with the lowest TRC Test results until

In the case of measures, such as these where the full measure cost is assumed by the sponsoring utility, the Utility Test is essentially the same as the TRC Test. The small difference in test results is due to different discount rates being applied to each test.



----

----

----

WESTERN KENTUCKY GAS COMPANY COMMUNITY ACTION RESIDENTIAL ENERGY SERVICES

the aggregate TRC Test result reached 1.0 or better. As a result, the following measures can be implemented with an aggregate TRC Test result of 1.02:

- > Attic Insulation
- > Clean & Tune Furnace
- Water Heater Replacement (Electric-to-Gas)
- Furnace Replacement (Standard Efficiency)
- Furnace Replacement (High Efficiency)
- > Duct Repair
- Infiltration Reduction

In this scenario, the three-year direct measure expenditures by WKG CARES would be approximately \$784,000, given the number of participants per measure estimated by the State Weatherization Office.

### THIRD SCENARIO: Develop Cost-Sharing Arrangements to Allow For Full Spectrum of Measures to be Implemented

In this scenario, we examined the level of cost sharing needed between WKG CARES and the CAP Agencies in order to make each measure pass the TRC Test. For those measures which do not pass the TRC Test (see Table 1) the amount of WKG CARES funding was modified to the point where the TRC Test would equal 1.0. Those specific measures and the amount of funding to be picked up by the CAP Agencies are listed in Table 2.

Under this scenario, all of the measures could be implemented.<sup>2</sup> Full measure costs would be paid by WKG CARES for the following programs:

- > Attic Insulation
- Clean and Tune Furnace
- Duct Repair
- > Infiltration Reduction

Costs for the measures in Table 2 would be shared, as indicated, by WKG CARES and the CAP Agencies.

<sup>2</sup> The exception is water heater wraps. Hampton recommends that the WKG CARES Program does not offer this measure regardless of its economics. Improperly installed, a water heater wrap on a gas water heater can pose a safety risk. In addition, some manufacturers will void warranties if such wraps are installed.





Western Kentucky Gas Company Community Action Residential Energy Services <u>Pre-Implementation Benefit/Cost Screening Results</u>

Cost Sharing Scenaric						
Measure.	Total Funding	WKG CARES	CAP Agencies			
Water Heater Replacement (Gas-to-Gas)	\$720	\$116	\$604			
Water Heater Replacement (Electric-to-Gas)	\$720	\$244	\$476			
Furnace Replacement (Standard Efficiency)	\$1,200	\$836	\$364			
Furnace Replacement (High Efficiency)	\$1,450	\$1,039	<b>\$</b> 411			
Wall Insulation	\$500	\$270	<b>\$</b> 230			
Floor Insulation	\$400	\$167	<b>\$</b> 233			
Duct Insulation	\$200	\$94	\$106			

In this scenario, the three-year direct measure expenditures by WKG CARES would be approximately \$830,000, given the number of participants per measure estimated by the State Weatherization Office.

<sup>3</sup> Hampton suggests that the acceptance of this measure not be based solely on the TRC test result, since its purpose is not energy savings, but energy cost reduction for low-income customers.



## RECOMMENDATIONS

Based on the results of the benefit/cost analyses, and other considerations as discussed, the following recommendations are offered regarding the design of the WKG CARES Program:

- 1. The Third Scenario should be considered. This scenario will allow for the maximum contribution by WKG CARES to take cost-effective DSM measures and will also provide the greatest benefit to low-income customers, maximize the financial contribution by WKG CARES to the ongoing efforts of the CAP Agencies, and protect the interests of all WKG customers. In addition, this scenario allows each measure to pass the TRC test independently, as well as in aggregate.
- 2. The Water Heater Wrap measure should not be implemented due to the safety and equipment warranty issues previously mentioned.
- 3. The Attic Insulation measure should only be implemented in homes where there is no, or very minimal, existing attic insulation.
- 4. Care should be taken to limit the number of measures which WKG funds in the same home since the interactive effects can reduce the cost-effectiveness of each measure. This is particularly true in homes where gas is not the sole means of space heating (e.g., homes that have wood-burning stoves).
- 5. Consideration should be given to the implementation of a Furnace Upgrade measure in place of the two furnace replacement measures. This measure assumes that the CAP Agency would have installed a standard efficiency (82% AFUE) furnace absent WKG CARES and provides solely for an upgrade in furnace efficiency to 90% AFUE. The test results of this measure are as follows:
  - Participant Test: 2.24
  - > Utility Test: 0.76
  - > RIM Test: 0.39
  - > TRC Test: 0.82

If this measure were adopted and applied under the Third Scenario, the cost sharing would be as follows:

- > WKG Cares: \$204 per participant
- > CAP Agencies: \$46 per participant





Constant Constant States and Constant State	GENER	AL INFOR	MATION	a de la servició de la constante de la servició de la constante de la servició de la constante de la constante La constante de la constante de
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS
1	Attic Insulation	3	Residential Low-Income	1 150 2 200 3 200
and a second and a s	MEASURE-S	PECIFIC IN	FORMATIO	Ň
	INPUT	BAS	ECASE	DSM CASE
MEASURE	/DEVICE DATA: Description	Pre-C	Condition	Post-Condition
ļ	Measure/Device Lifetime (Yrs.)		20	20
	Equipment Cost (Per/Unit)		0	\$320
	Annual Incremental O&M Costs		\$0	\$0
LOAD IMPA	ACT ASSUMPTIONS: Efficiency Rating Annual Energy Consumption: Natural Gas (Therms) Electricity (kWh)	. 1	V/A 750	N/A 638
PROGRAM	REBATE/INCENTIVES:		••• •	
	Customer Rebate		-	\$320
	Dealer/Other Rebate		-	•
PROGRAM	DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs Recurring Per Part. Costs One-Time Per Part. Costs		-	\$80
RATES:		Gas Re	sidential	Gas Residential
	COSTS:	Residentia	Space Heat	Residential Space Heat
LOAD SHAI	PES:	Residentia	Space Heat	Residential Space Heat

GENERAL INFORMATION				
		IMPLENT. PERIOD	END USE/	NUMBER OF TARGETED
CODE #	DSM PROGRAM TITLE	(YEARS)	CUST. TYPE	PARTICIPANTS
2	Clean & Tune Furnace	3	Residential Low-Income	Year Rate 1 15 2 20
	MEASURE-S	PECIFIC II	FORMATIO	<u> </u>
i i se ne konstruktivni se state state N	INPUT	BAS	ECASE	DSM CASE
MEASURE	/DEVICE DATA: Description	Pre-C	Condition	Post-Condition
	Measure/Device Lifetime (Yrs.) Equipment Cost (Per/Unit)		10	10 \$120
	Annual Incremental O&M Costs			•
	ACT ASSUMPTIONS: Efficiency Rating	. 6	35%	70%
	Natural Gas (Therms) Electricity (kWh)		750	690
PROGRAM	REBATE/INCENTIVES:	••••		
,	Customer Rebate Dealer/Other Rebate	. <b></b>	-	\$120
PROGRAM	DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs	····• ··	-	······································
	Recurring Per Part. Costs One-Time Per Part. Costs		-	\$30
RATES:		Gas Re	sidential	Gas Residential
	COSTS:	Residentia	Space Heat	Residential Space Heat
LOAD SHA	PES:	Residentia	Space Heat	Residential Space Heat

	GENER	AL INFOR	MATION		<u> </u>
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
ЗA	Water Heater Replacement (Gas-to-Gas)	3	Residential Low-Income	Year Rate 1 2 3	30 50 50
	MEASURE-S	PECIFIC II	NFORMATIO	N	· · · ·
	INPUT	BAS	E CASE	DSM CASE	
MEASURE	/DEVICE DATA: Description	Old Gas V	Nater Heater	New Gas Water Heate	:r
	Measure/Device Lifetime (Yrs.)	• • ····	13	13	
	Equipment Cost (Per/Unit)		•	\$580	
	Annual Incremental OaM Costs	<u> </u>		<b>.</b> -	
LOAD IMP	ACT ASSUMPTIONS:		50	.54	
	Annual Energy Consumption:				
	Natural Gas (Therms) Electricity (kWh)		-	- 208	
PROGRAM	REBATE/INCENTIVES			· · · · · · · · · · · · · · · · · · ·	
	Customer Rebate		-	\$580	_
	Dealer/Other Rebate		•	-	
PROGRAM	DIRECT COSTS:	<b>.</b>			
	One-Time Fixed Costs		-	-	
	Recurring Per Part. Costs		-		
	One-Time Per Part. Costs	••••••	-	\$140	• •
RATES:		Gas R	esidential	Gas Residential	-
	COSTS:	Residentia	al Water Heat	Residential Water Hea	t
LOAD SHA	PES:	Residentia	al Water Heat	Residential Water Hea	t

a ang pang ang pang pang pang bang pang pang pang pang pang pang pang p	GENER	AL INFOR	MATION		•
CODF #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/	NUMBER OF TARGETED PARTICIPANTS	
0002 #		() criter		Year Rate	
3B	Water Heater Replacement (Electric-to-Gas)	3	Residential Low-Income	1 2 2	30 50
en language. In language anna fheirichteach Anna fheirige anna chuide fheirichteach Anna fheirige anna chuide anna fheirige fheir	MEASHDES		SOPMATIO	<b>N</b>	<u></u>
n an an an an Arran an Arran an Arra a Arra an Arra an	INPUT	BAS	ECASE	DSM CASE	<u> </u>
MEASURE	/DEVICE DATA: Description	Electric V	Vater Heater	Gas Water Heater	
	Measure/Device Lifetime (Yrs.) Equipment Cost (Per/Unit)	•	13	<u>13</u> \$580	
	Annual Incremental O&M Costs		•		
	ACT ASSUMPTIONS		•••••	-	
	Efficiency Rating		.70	.54	
-	Annual Energy Consumption: Natural Gas (Therms) Electricity (KWh)	4	.950	208	
DDOOD AN		·····	i	 	
PROGRAW	Customer Rebate		-	\$580	
PROGRAM	DIRECT COSTS:	· . ···	· ·····		
	Recurring Fixed Costs		-	•	
	One-Time Fixed Costs		-	40 - 1	
	Recurring Per Part. Costs				
-	One-Time Per Part. Costs		-	\$140	
RATES:		Gas Re	esidential	Gas Residential	
	COSTS:	Residentia	I Water Heat	Residential Water Heat	
LOAD SHA	PES:	Residentia	I Water Heat	Residential Water Heat	

Service Same services

. . . .

. . .

. . . .

4

\_\_\_ .\_...

- -

. \_ - - -

al a construction of the second s a construction of the second sec	GENER	AL INFOR	MATION	(a) A second s second second s second second secon second second sec	
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
4A	Furnace Replacement (Standard Efficiency)	3	Residential Low-Income	Year Rate 1 2 3	20 30 30
a in a substantia and a substantia definition and a substantia definition definition and a substantia definition and a substantia definition and a substantia definition and a substantia definition and a substantia definition	MEASURE-S	PECIFIC IN	FORMATIO	N	
	INPUT	BAS	E CASE	DSM CASE	
MEASURE	/DEVICE DATA: Description	Old	Furnace	New Furnace (Standard Efficiency)	
	Measure/Device Lifetime (Yrs.)		18	18	
<b>}</b> .	Equipment Cost (Per/Unit)		-	\$960	
	Annual Incremental O&M Costs	<u></u>	a		
LOAD IMP	ACT ASSUMPTIONS: Efficiency Rating Annual Energy Consumption:	60%	AFUE	82% AFUE	-
	Natural Gas ( <u>Therms</u> ) Electricity (kWh)		750	549	
PROGRAM	REBATE/INCENTIVES: Customer Rebate Dealer/Other Rebate	·····	•	\$960	
PROGRAN	DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs		-	-	· • • • • •
	One-Time Per Part. Costs	······································		\$240	
RATES:		Gas R	esidential	Gas Residential	
AVOIDED	COSTS:	Residentia	I Space Heat	Residential Space Hea	at
LOAD SHA	PES:	Residentia	I Space Heat	Residential Space Hea	at

. . . . .

\_ \_\_ \_ \_ \_ \_ \_ \_ \_ . . . . . . .

. . . . .

- ---

<ul> <li>All and the second secon</li></ul>	GENER	RAL INFOR	MATION	Construction and Const Construction and Construction	
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	••••
4B	Furnace Replacement (High Efficiency)	3	Residential Low-Income	1 2	20 30
n de la companya de la companya de la comp	MEASURE.S	PECIFIC II	FORMATIO	<u> </u>	30
ار بار بار بار بار بار بار بار بار بار ب	INPUT	BAS	ECASE	DSM CASE	
MEASURE	DEVICE DATA: Description	Old I	Furnace	New Fumace (High Efficiency)	
	Measure/Device Lifetime (Yrs.)		18	18	
-	Annual Incremental O&M Costs		•	<u>41,100</u>	
LOAD IMPA	ACT ASSUMPTIONS: Efficiency Rating Annual Energy Consumption: Natural Gas (Therms) Electricity (kWh)	60%	• AFUE 750 •	90% AFUE 500	
PROGRAM	REBATE/INCENTIVES:				
	Customer Rebate Dealer/Other Rebate		- 	\$1,160	
PROGRAM	DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs Recurring Per Part. Costs One-Time Per Part. Costs		-	\$290	
RATES:		Gas Re	sidential	Gas Residential	
	COSTS:	Residentia	Space Heat	Residential Space Heat	t
LOAD SHAF	PES:	Residentia	I Space Heat	Residential Space Hea	t

...
GENERALINFORMATION					
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
				Year Rate	•
5	Wall Insulation	3	Residential Low-Income	1	75 100
				3	100
y and a second standard and a second second second standard and the second second second second second second second	MEASURE-S	PECIFIC II	FORMATIO	N	
ygeven er syn rennen sin ter bereiten. Der bereiten er sin	INDIT	BAS	FCASE	DSM CASE	
		0/10			
MEASURE	/DEVICE DATA: Description	Pre-C	Condition	Post-Condition	
-	Measure/Device Lifetime (Yrs.)	• •••	20	20	
	Equipment Cost (Per/Unit)		-	\$400	
-	Annual Incremental O&M Costs		• ··· · ·	•	
LOAD IMP	ACT ASSUMPTIONS:				
	Efficiency Rating	<u>N/A</u>		<u>N/A</u>	
	Annual Energy Consumption:			000	
	Natural Gas (Therms)	750		690	
	Electricity (KVVh)	-		-	
PROGRAM	REBATE/INCENTIVES:	<u></u>			
	Customer Rebate		-	\$400	
	Dealer/Other Rebate		•	-	
70000		· • • • • • • • • • • • • • • • • • • •			<b></b>
PROGRAM	DIRECT COSTS:		_	-	
-	Checulting Fixed Costs		- -	• • • • •	
•	Recurring Per Part Costs			-	
	One-Time Per Part. Costs	······································		\$100	
-					
RATES:	• · · • • • · · · · · · · · · · · · · ·	Gas R	esidential	Gas Residential	
	COSTS:	Residentia	I Space Heat	Residential Space He	eat
LOAD SHA	PES:	Residentia	I Space Heat	Residential Space He	eat

GENERAL INFORMATION					
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
6	Floor Insulation	3	Residentiai Low-Income	Year Rate 1 75 2 100 3 100	
	MEASURE-S	PECIFIC IN	FORMATIO	N	
	INPUT	BAS	ECASE	DSM CASE	
MEASURE	/DEVICE DATA: Description	Pre-C	Conditon	Post-Condition	
	Measure/Device Lifetime (Yrs.)	····	20	20	
	Equipment Cost (Per/Unit)		-	\$320	
	Annual Incremental O&M Costs	·····	-	•	
LOAD IMP/ -	LOAD IMPACT ASSUMPTIONS: Efficiency Rating		۹/A	N/A	
_	Natural Gas (Therms) Electricity (kWh)	7	′50 -	713	
PROGRAM	REBATE/INCENTIVES:				
_	Customer Rebate	•		\$320	
-	Dealer/Other Rebate		-	•	
PROGRAM	DIRECT COSTS: Recurring Fixed Costs	· · · · · · · · · · · · · · · · · · ·	•	•	
	Recurring Per Part. Costs		•	·····	
	One-Time Per Part. Costs	· · · · · · · · · · · · · · · · · · ·	•	\$80	
RATES:	- <u></u>	Gas Re	sidential	Gas Residential	
AVOIDED COSTS:		Residential Space Heat		Residential Space Heat	
LOAD SHAP	PES:	Residential	Space Heat	Residential Space Heat	

\_\_\_\_

GENERAL INFORMATION					
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
7	Duct Repair	3	Residential Low-Income	Year Rate	150 200 200
	MEASURE-S	PECIFIC	FORMATIO	N	
	INPUT	BAS	ECASE	DSM CASE	
MEASURE	DEVICE DATA: Description	Pre-C	condition	Post-Condition	
	Measure/Device Lifetime (Yrs.)		10	10	
	Equipment Cost (Per/Unit)		•	\$120	
	Annual Incremental O&M Costs		•	•	
LOAD IMP	ACT ASSUMPTIONS: Efficiency Rating	N/A		N/A	
	Annual Energy Consumption:	750		875	
	Electricity (kWh)		-	-	
PROGRAM	REBATE/INCENTIVES:		· · · · · · · · · · · · · · · · · · ·	1	
	Customer Rebate	•		\$120	
-	Dealer/Other Rebate		•	-	
PROGRAM DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs					
	Recurring Per Part. Costs	• •	•	¢20	
-	One-Time Per Part. Costs		• • • • • • • • • • • • • • • • • • • •	300	
RATES:	·	Gas Re	sidential	Gas Residential	
	COSTS:	Residentia	Space Heat	Residential Space Hea	at
LOAD SHA	PES:	Residentia	Space Heat	Residential Space Hea	at

	GENERAL INFORMATION				
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
8	Duct Insulation	Э	Residential Low-Income	Year Rate	100 150 150
	MEASURE-SI	PECIFIC IN	FORMATIO	N	
	INPUT	BAS	E CASE	DSM CASE	
MEASURE	E/DEVICE DATA: Description	Pre-C	condition	Post-Condition	
	Measure/Device Lifetime (Yrs.)		10	10	
1	Equipment Cost (Per/Unit)			\$160	
	Annual Incremental O&M Costs		•		·
LOAD IMP	LOAD IMPACT ASSUMPTIONS: Efficiency Rating Annual Energy Consumption: Natural Gas (Therms)		V/A	N/A 713	- <b></b>
PROGRAM		· · · · · · · · · · · · · · · · · · ·	-		
	Customer Rebate	-		\$160	
ļ.	Dealer/Other Rebate			-	-
PROGRAM DIRECT COSTS: <u>Recurring Fixed Costs</u> <u>One-Time Fixed Costs</u> <u>Recurring Per Part. Costs</u> <u>One-Time Per Part. Costs</u>		- · · · · · · · · · · · · · · · · · · ·		- <u>·</u>	··· •
RATES:		Gas Re	sidential	Gas Residential	
AVOIDED	COSTS:	Residentia	Space Heat	Residential Space Hea	at I
LOAD SHA	PES:	Residential	Space Heat	Residential Space Hea	at

GENERAL INFORMATION					
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
9	Water Heater Wraps	3	Residential Low-Income	Year Rate	150 200
	MEASURE-S	PECIFIC IN	FORMATIO	N	200
	INPUT	BAS	ECASE	DSM CASE	
MEASURE/	DEVICE DATA: Description	Pre-C	Condition	Post-Condition	
	Measure/Device Lifetime (Yrs.)		10	10	
	Equipment Cost (Per/Unit)		. <del>.</del>	\$15	
	Annual Incremental O&M Costs		•	•	
LOAD IMPACT ASSUMPTIONS: Efficiency Rating Annual Energy Consumption: Natural Gas (Therms) Electricity (kWh)		N/A 25 <u>0</u>		N/A 248	
PROGRAM REBATE/INCENTIVES: Customer Rebate Dealer/Other Rebate		-		\$15	
PROGRAM DIRECT COSTS: <u>Recurring Fixed Costs</u> <u>One-Time Fixed Costs</u> <u>Recurring Per Part. Costs</u> <u>One-Time Per Part. Costs</u>					
RATES:	······································	Gas Re	sidential	Gas Residential	-
AVOIDED C	OSTS:	Residentia	Water Heat	Residential Water Heal	t
LOAD SHAF	PES:	Residential	Water Heat	Residential Water Heat	t

Bernstein (Editoria), die Konstand Stein die Statistikanistik is statistik einer die Statistikanistik is statistik einer die Statistikanistik die Statistik einer die Statistikanistik die Statistik einer die Statistik die Statistik einer die Statis	GENERAL INFORMATION				
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS	
10	Infiltration Measures	3	Residential Low-Income	Year ARate	150 200 200
(a) A second se Second second seco	MEASURE-S	PECIFIC IN	FORMATIO		
	INPUT	BAS	ECASE	DSM CASE	
MEASURE	/DEVICE DATA: Description	Pre-C	condition	Post-Condition	
	Measure/Device Lifetime (Yrs.) Equipment Cost (Per/Unit)		20	20 \$200	
	Annual Incremental O&M Costs		•	•	
	ACT ASSUMPTIONS: Efficiency Rating	N/A		N/A	
Annual Energy Consumption: Natural Gas (Therms) Electricity (kWh)		750 -		<u>675</u> -	
PROGRAM	REBATE/INCENTIVES:	<i></i>		·	••••••
	Customer Rebate Dealer/Other Rebate	- -		\$200 -	
PROGRAM	DIRECT COSTS: Recurring Fixed Costs One-Time Fixed Costs Recurring Per Part. Costs One-Time Per Part. Costs		-	\$50	
RATES:	· · · · · · · · · · · · · · · · · · ·	Gas Re	esidential	Gas Residential	
	OSTS:	Residentia	Space Heat	Residential Space Hea	at
LOAD SHAF	PES:	Residentia	Space Heat	Residential Space Hea	at

GENERAL INFORMATION				
CODE #	DSM PROGRAM TITLE	IMPLENT. PERIOD (YEARS)	END USE/ CUST. TYPE	NUMBER OF TARGETED PARTICIPANTS
11	Furnace Upgrade	3	Residential Low-Income	Year Rate 1 20 2 30 3 30
	MEASURE-S	PECIFIC IN	IFORMATIO	N
	INPUT	BASI	E CASE	DSM CASE
MEASURE	DEVICE DATA: Description	Old F	urnace	New Furnace (High Efficiency)
	Measure/Device Lifetime (Yrs.)		18	
-	Annual Incremental O&M Costs	•	-	\$200
-		· ••••••••••••••••••••••••••••		
	ACT ASSUMPTIONS: Efficiency Rating	82%	AFUE	90% AFUE
-	Annual Energy Consumption: <u>Natural Gas (Therms)</u> Electricity (kWh)		49	500
PROGRAM	REBATE/INCENTIVES:		. <u></u>	
	Design/Other Repate		•	\$200
PROGRAM		<u></u>		-
F	Recurring Fixed Costs Dne-Time Fixed Costs		• • • • •	•
	Recurring Per Part. Costs Dne-Time Per Part. Costs		•	\$50
RATES:	· · ·	Gas Re	sidential	Gas Residential
AVOIDED C	OSTS:	Residential	Space Heat	Residential Space Heat
LOAD SHAP	PES:	Residential	Space Heat	Residential Space Heat

#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 19 Witness: Marks

#### Data Request:

19. Refer to the Marks Testimony, page 8. It is stated that the TRC test measures the costs and benefits of a conservation measure from the broadest perspective as it represents the net benefit to society. Isn't it correct that the TRC calculates the net impact on a utility and its customer base as a whole, instead of as a net benefit to society? If no, provide documentation to support the contention that the focus of the TRC is the net benefit to society.

#### Response:

The TRC Test is interpreted as the Societal Test without externalities. In fact, we chose in the evaluation report for the Collaborative to call the TRC test the "Societal Test excluding externality adder". I regret any confusion that this may have caused.

#### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20. a. Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

a. Provide copies of the written process and impact evaluation reports. If no written report was provided, explain in detail why not.

#### Response:

See attachment entitled, "Process and Impact Evaluation of Western Kentucky Gas WKG CARES Program".

TACHMENT

P.R. 20, A, D.R. 20, C, D.R. 20, d, Evaluation of Western Kentucky Gas WKG CARES Program

Prepared by

Applied Energy Group, Inc. Hauppauge, New York

Prepared for

Western Kentucky Gas Company Demand-Side Management Program Collaborative

**July 1998** 

# **Table of Contents**

I.	Executive Summary
II.	Overview6A.ObjectivesB.Methodology1.Process Evaluation2.Impact Evaluation8C.C.Report Organization
III.	WKG CARES Program Process Overview and Findings9A.Program Description9B.Program Design and the Role of CAP Agencies11C.Program Delivery by the CAP Agencies13D.Program Objectives and Goals17E.Program Results19F.Program Promotion201.Program Promotion by WKG202.Program Promotion by CAP Agencies213.Education Component21G.Data Tracking and Reporting221.Tracking222.Program Performance Reporting by WKG23H.Program Communication Issues24I.Barriers to Program Participation25J.CAP Agency Reactions to the WKG CARES Program26
IV.	Customer Reaction to the Program - Telephone Survey Results and Analysis29A.Description of the Data-Collection ProcessB.Sample Design and Confidence Levels1.Sample Design292.Confidence Levels30C.Survey Tabulations and Findings
V.	Impact Evaluation Overview and Findings42A.Objectives42B.Methodology421.General Approach422.PRISM Methodology433.Analysis Stages45C.Data Sources47

	D.	1.WKG Cares Program Tracking System472.Customer Consumption Data473.Weather Data484.WKG Corporate Data48Impact Evaluation Findings491.Sample Design492.Comparison of Key Statistics (Sample and Study Population)513.Gross and Net Change-Score Results534.BENCOST Model Results55
	<b>E</b>	Participant Bill Savings 57
VI.	Overa A. B.	all Conclusions and Recommendations58General Conclusions58Recommendations60

- Appendices
- Appendix A Bill Stuffer
- Appendix B Monthly Summary Report
- Appendix C Program Manager Protocol, CAP Agency Protocol & Participant Telephone Survey Instrument
- Appendix D PRISM Modeling Output
- Appendix E SAS Statistical Output
- Appendix F BENCOST Cost-Effectiveness Output

## I. Executive Summary

This evaluation of Western Kentucky Gas' WKG CARES program examined the program from both impact and process perspectives. The process evaluation found that the program is a good example of the way in which a DSM program can be designed that capitalizes on the strengths of a number of partners in the implementation process. The program was developed as one element of a rate proceeding decided by the Kentucky Public Service Commission (PSC) in 1995 (Case No. 95-010). Central to the program are two important provisions of the agreement between WKG and the Kentucky PSC. These are, first, that the delivery of program services should be coordinated through local CAP agencies and secondly, that program oversight and guidance should be provided through a collaborative process. The resultant WKG CARES program design achieves the efficiencies and benefits accruing from piggy-back types of programs, while at the same time, maximizing the benefits that program participants could receive from both the CAP agency weatherization efforts, as well as WKG's program expenditures. The program manifests a high degree of inherent efficiency as its infrastructure is already in place via the five CAP agencies through which it operates. Recommendations stemming from the process evaluation are offered to refine an already effective program and include the following:

- Secure permission from the Collaborative to prudently fund paid promotional efforts. The effectiveness of a promotional strategy relying solely on free advertising time/space was limited in this program. If a <u>small</u> portion of the program funding surplus can be used to buy appropriate advertising coverage, the investment may not only attract further participants but will also help WKG and the Collaborative in maximizing the benefits made available from the program over its three-year planning horizon.
- Hold more frequent status meetings with CAP agencies. Use these as an opportunity for cross-sharing of ideas as well as a vehicle for the transmission of goal achievement reports. Throughout the course of the evaluation, it became apparent that there are certain program elements where a common understanding among the CAPs and WKG is not present. One means of improving the situation is to re-instate the periodic meetings between all parties that were present at the beginning of the program. *While not an issue during the period under evaluation*,

the loss of oil overcharge funds as of July 1, 1998, will exercise a negative impact on the CAPs and potentially the WKG CARES Program itself. Meetings such as those recommended here will foster dialogue between all interested parties that may help reduce any negative impact of the loss in funding.

- Develop an educational supplement that can be uniformly distributed by all CAPs to all incoming participants that heightens the conservation message while clearly conveying the role played by WKG. There is no standard "educational component" offered by all CAPS. It is worthwhile for WKG to consider either preparing materials on its own or securing ready-made booklets, brochures, etc., that it can supply to the CAPs for subsequent distribution to participants.
- Contact the remaining CAP agencies that share a portion of WKG's territory and solicit their participation in the program. Three additional CAP agencies serve WKG customers. They are not included in the current roster of participating agencies, nor are they understood as having a significant WKG customer population. Nonetheless, WKG should contact these agencies for the purposes of assessing its interest in joining with the other five CAPs. Since there is no incremental expense on WKG's part to add additional CAP agencies, any contribution of additional participants from the additional agencies will only stand to benefit the program.
- Design and implement written contracts with each of the participating CAP agencies. Written contracts do not presently capture the relationship between the agencies and WKG. For the purposes of liability protection, as well as to exactly define the responsibilities of all parties, written contracts should be set in place.

The impact evaluation of the WKG CARES program sought to estimate the energy savings (Mcf) and associated bill reductions for those Western Kentucky Gas customers participating in the WKG CARES program. As a corollary, the evaluation produced estimates of the WKG CARES program's cost-effectiveness during this period from the societal, ratepayer, utility, and participant perspectives.



Gross and net energy savings experienced by participants are presented below in Tables I.1 through I.3. Table I.1 presents the results of the change-score analysis. Mean estimates of pre and post-treatment consumption were developed using PRISM for a sample of the program participants in order to calculate a gross energy savings. The gross per-participant energy savings is then simply the difference between the pre and post-treatment consumptions (i.e, *87.8 - 70.9*) or 16.9 Mcf.

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	87.8
Mean Post-Treatment Normalized Annual Consumption (Mcf)	70.9
Mean Gross Per-Participant Change in NAC (Mcf)	16.9
Program Participants (n)	16.1
Program-Wide Energy Savings (Mcf)	2,721

Table I.1 Participant Gross Energy Savings

Mean estimates of pre- and post-treatment consumption were also developed for a sample of program non-participants in order to control non-programmatic influences on consumption. These customers experienced a minimal reduction in consumption of one ccf (0.1 Mcf). This represents a 0.13% reduction. Table I.2 shows the results.

Table I.2 Non-Participant PRISM Results

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	78.8
Mean Post-Treatment Normalized Annual Consumption (Mcf)	78.7
Mean Per-Participant Change in NAC (Mcf)	0.1
% Change in Annual Consumption	0.13%
CAF (1-% Change in Annual Consumption)	0.9987



The final step in the development of net energy savings was to adjust the participant's mean pre-consumption to account for the "naturally- occurring" reduction in consumption experienced by the program non-participants. This adjustment is a relative adjustment and entails reducing the pre-treatment consumption by 0.13% (or multiplying by 0.9987, the *Control Adjustment Factor*). This adjustment yields a net energy savings of 16.8 Mcf. The total energy savings for the study population is then 2,704.8 Mcf. See Table I.3 for the results.

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	87.8
Control Adjustment Factor	0.9987
Mean Pre-Treatment Energy Consumption (Mcf) Adjusted for Control Adjustment Factor (CAF)	87.7
Mean Post-Treatment Normalized Annual Consumption (Mcf)	70.9
Mean Net Per-Participant Change in NAC (Mcf)	16.8
Program Participants (n)	161
Program-Wide Energy Savings	2,705

Table I.3 Participant Net Energy Savings

Tables I.4 and I.5 present the results of the benefit-cost analysis which employed the BENCOST benefit-cost software. Note that AEG performed the benefit-cost analysis with and without an environmental externality adder (which accounts for environmental benefits associated with reduced natural gas consumption).

#### Table I.4 Benefit-Cost Results (Excluding Environmental Adder)

Societal Test	1.12
Ratepayer Impact Measure	0.65
Utility Test	0.66
Participant Test	681.35



Societal Test	1.17
Ratepayer Impact Measure	0.65
Utility Test	0.66
Participant Test	681.35

## Table I.5 Benefit-Cost Results (Including Environmental Adder)

The results of the impact evaluation suggest the following conclusions:

• Substantial energy savings are associated with participation in the WKG CARES program.

Ý

- The net per-participant energy savings has been estimated as 16.8 Mcf for those WKG customers who participated in the program between November 1996 and September 1997 which represents approximately a 19% reduction in energy consumption.
- The annual bill reduction associated with this energy savings is \$82 or 16.7% of the average participant's annual natural gas bill.
- Based on the energy savings estimates, the societal benefit-cost ratio is 1.12 (without an environmental adder) and 1.17 (with an environmental adder) demonstrating the program's cost-effectiveness.

Based on these findings and conclusions from the impact evaluation, the following recommendations are offered:

- The WKG CARES program should continue as is with no modifications in delivery or measure specification suggested by the level of the achieved program impacts.
- WKG should consider performing an evaluation of year-one participants following year-three to if such savings degradation has occurred.

## II. Overview

This report represents the results of Applied Energy Group, Inc.'s (AEG) process and impact evaluations of Western Kentucky Gas' WKG CARES Program. This demand side management/low income customer assistance program commenced on November 1, 1996, and will continue in operation for at least three years.

### A. Objectives

The objectives of the process evaluation include the following:

- To provide and capture an institutional memory of the program, its operation, and its results
- To provide recommendations, as appropriate and where required, concerning program delivery, promotion, and communications
- To identify ways, if any, to improve the efficiency of the program design and implementation process
- To determine the adequacy of program staffing, both internal and secondarily through Community Assistance Program agencies (CAPs), program resources, and quality control
- To examine and comment upon the working relationships between all parties involved in designing and implementing the program
- To explore and critique the piggy-back relationship and value-added benefits provided by WKG CARES to the Department of Energy and State-funded services provided by CAP agencies



The objectives of the impact evaluation include the following:

- To estimate the energy savings (Mcf) attributable to the WKG CARES Program
- To develop estimates of the WKG CARES Program's cost-effectiveness from the societal, ratepayer, utility, and participant perspectives
- To develop an estimate of annual bill savings accruing to program participants

#### B. Methodology

#### 1. Process Evaluation

Process evaluations rely on both quantitative and qualitative data to provide a clear picture of program implementation and performance from a number of perspectives. Each research design requires careful consideration to determine which methods are most appropriate to the accomplishment of evaluation objectives. In the case of WKG CARES, AEG's evaluation focused on program sponsors, implementors, and recipients through a variety of research tools including:

- Interviews with program management
- Interview with the Executive Director of the Kentucky Association for Community Action
- Interviews with Community Assistance Program Agency directors
- Surveys of program participants
- Analysis of program data bases and reporting formats
- Reviews of relevant program documentation

Initial and follow-up interviews were conducted during April and May 1998. The surveys of program participants took place during May 1998. (See Appendix C for the protocols and the participant telephone survey instrument.)



AEG Applied Energy Group

### 2. Impact Evaluation

To determine and analyze the quantitative results of the program, i.e., its impacts, AEG conducted a statistical billing analysis to estimate the gross and net energy Mcf savings associated with participation in the WKG CARES Program. AEG used the Princeton Scorekeeping Method (PRISM) to develop estimates of pre- and post-treatment normalized annual consumption values for a sample of program participants and non-participants. The resulting gross and net savings estimates were then projected to the program population. In addition these energy savings estimates were used to develop estimated bill reductions for program participants. The net energy savings values were also integrated into a benefit-cost model<sup>1</sup> to provide benefit-cost ratios from the societal, ratepayer, utility, and participant perspectives.

#### C. Report Organization

Chapter III of this report presents a profile of the WKG CARES Program, incorporating the impetus for establishment of the program, a discussion of its configuration and design, a description of delivery mechanisms, and the importance of the role played by the CAP agencies. The findings yielded by the process evaluation are developed throughout this discussion of the program's design and operation.

Chapter IV contains the reaction of participants to the program in the form of survey results, analysis, and tabulation.

Chapter V presents the results of the impact evaluation.

Chapter VI presents the conclusions drawn from both the process and impact evaluations and offers recommendations to improve the functionality and effectiveness of the program.

<sup>&</sup>lt;sup>1</sup> AEG utilized a benefit-cost model (BENCOST) which was developed by the Minnesota Department of Public Service specifically for gas utility DSM programs. AEG is familiar with this model through work we have done for three different Minnesota gas utilities.



## III. WKG CARES Program Process Overview and Findings

This chapter presents an overview of the WKG CARES Program as well as key findings from the process evaluation.

#### A. Program Description

The WKG CARES Program was initiated as a pilot Demand Side Management/Low Income Customer Assistance Program. The program is directed at low income customers who, for the most part, own their own homes. The program focuses on the delivery of weatherization measures to the homes of qualifying low-income residents and the reduction of their gas utility bill. The program was developed as one element of a rate proceeding decided by the Kentucky Public Service Commission (PSC) in 1995 (Case No. 95-010). Central to the program are two important provisions of the agreement between WKG and the Kentucky PSC. These are, first, that the delivery of program services should be coordinated through local CAP agencies and secondly, that program oversight and guidance should be provided through a collaborative process.

"To enhance the success of the program, Western agrees to work with a collaborative work group made up an internal team and representatives of Kentucky Legal Services, Inc., the Attorney General's office and community action agencies having expertise at working with low income customers' utility problems. The Commission shall be invited to participate also, at its discretion. The purpose of the Collaborative will be to establish a practical, detailed plan for implementing the DSM program. Unless otherwise agreed to by Western, all programs will be designed to qualify for full or partial recovery pursuant to KRS 278.285.......Western agrees to work with the Kentucky Association for Community Action and the local Community Action agencies in its service area to help administer the program."<sup>2</sup>

<sup>&</sup>lt;sup>2</sup> Kentucky Public Service Commission Case 95-010 Decision, p.6.

A total of 300 low-income residences were targeted initially for treatment during each year of the three-year program. The letter of stipulation and agreement called for a maximum of \$1,500 to be expended per treated residence <sup>3</sup>, with a total program cost not to exceed \$450,000 per year. WKG agreed to commit to fund the pilot program for three years regardless of the cost recovery effectiveness, although all program elements were to be designed to qualify for full recovery.

The first year of the program spanned the period November 1, 1996, through October 31, 1997. For the purposes of this process evaluation however, the period from November 1, 1996, through January 31, 1998 will be addressed.

Program management, as it is customarily understood in DSM programs, is the responsibility of WKG. It is the responsibility of the WKG Program Manager to coordinate the day-to-day functioning of the program. The Collaborative (The Western Kentucky Gas Company Demand-Side Management Program Collaborative) mentioned above was formed to provide additional oversight in the design and implementation of the program and to ensure that the interests of all participants to the process were most effectively and most equitably served. Major policy decisions regarding the program are the responsibility of the Collaborative. The group comprising the Collaborative, as envisioned in the rate case, was to include representation from an internal team from WKG, as well as representation from Kentucky Legal Services, Inc., the Attorney General's office, and either the Community Action agencies themselves or someone representing them. The Commission declined the opportunity to participate. Additionally, a representative was not available from Kentucky Legal Services, Inc. As a result, a representative from Cumberland Trace Legal Services (a regional legal service) was included in the Collaborative. At the time of this evaluation, a total of four representatives (one from each of the above mentioned organizations) comprise the Collaborative.

<sup>&</sup>lt;sup>3</sup> As is further discussed later in this evaluation WKG subsequently modified this ceiling to an average of \$1,500 per home, with a maximum of \$2,000, provided that the *average* expense of all treated residences did not exceed \$1,500.

### B. Program Design and the Role of CAP Agencies

WKG CARES is a piggy-back type of DSM program as it overlays benefits or program measures supplied by the utility on top of benefits made available through other programs, such as those provided by CAP agencies. WKG CARES is similar to programs in other states that are directed at improving the comfort level, safety, and energy efficiency of residences owned (or in some cases, merely occupied) by low-income families through utility-funded programs.<sup>4</sup>

There are a total of 26 CAP agencies in Kentucky. WKG works in this program with five agencies in this group that cover approximately 90% of its service territory. The five participating agencies serve contiguous areas within WKG's territory. The Executive Director of the Kentucky Association for Community Action reported that there are three CAP agencies that could participate in the program that are not already involved. However, it is not known whether these agencies reflect geographic areas containing large eligible populations that could benefit from the program.

Finding: However, it is possible that these remaining agencies could provide at least some additional participants to the program. Additionally, their inclusion would make program benefits available to all eligible WKG residential customers.

Simply stated, WKG supplies funding to CAP agencies to augment CAP expenditures for weatherization services. WKG does not install measures with its own staff, nor does it employ sub-contractors independent of the CAPs. CAP agencies receive their normal funding generally from federal and state grant programs and provide a variety of services, one of which is weatherization. Examples of other services provided by CAP agencies would be the qualification of potential recipients of Home Energy Assistance Program (HEAP) benefits that are made available annually to low-income households through this federal block-grant program, Meals on Wheels, aid to seniors, etc.

<sup>&</sup>lt;sup>4</sup> An example of such a program is New York State's Utility Low-Income Energy Assistance Program (ULIEEP) which was established through a cooperative rather than a collaborative process, and was approved by the NYS PSC in June 1992. Another example can be found in Minnesota, where most of the gas utilities have piggy-back type weatherization programs with CAP agencies located in their service territories.

The selection of weatherization measures specifically authorized and funded by WKG is the result of a joint analysis by WKG and the Collaborative, including the five agencies that presently participate in the program. A series of measures were proposed by the group for consideration. These potential program offerings were subjected to a series of standard cost benefit tests to weigh their net costs and benefits from a number of different perspectives. The ultimate choice of measures was based on a cost sharing arrangement between the CAP agencies and WKG. This scenario enabled the installation of the widest range of measures, with each measure passing the Total Resource Cost Test. This test measures the costs and benefits of a conservation measure from the broadest perspective in that it represents the net benefits to society, including both participants and non-participants.<sup>5</sup> Table III.1, which follows, presents the measures currently being installed by the CAPS that are either fully funded by WKG or are funded on a shared-cost basis.

Efficiency Measures Implemented at WKG Expense or on a Shared-Cost Basis in the WKG CARES Program		
Attic insulation		
Wall insulation		
Floor insulation		
Infiltration reduction		
Water heater replacements/repairs		
Duct insulation		
Repair/replace furnace		
Clean/repair furnace		
Duct repair		

Table III.1

<sup>&</sup>lt;sup>5</sup> See <u>Standard Practice Manual: Economic Analysis of Demand-Side Management Programs</u>, developed by the California PUC and the California Energy Commission.

## C. Program Delivery by the CAP Agencies

The CAP agencies treat low-income homes to improve the home's "envelope integrity" by replacing or repairing specific equipment and structural components. The first step in the delivery process is the qualification of the customer for admission to the program. There are very stringent income qualifications for the program. In summary, in order for a residential customer to qualify, total household income cannot exceed 125% of the federal poverty level. Eligibility is determined by the CAP agencies. WKG plays no role in this process.

Only single family homes and mobile homes are eligible for the WKG funds. As reported by one of the CAP agencies, the State weatherization guidelines allow a maximum of between 10% and 15% of the eligible homes to be rentals (the application is filled out by the current tenant). The WKG CARES Program adopted and is governed by this guideline since it piggy-backs on the weatherization program. However, if a rental home is weatherized, it must adhere to certain specifications:

- The landlord is required to pay 75% of the cost of heating system improvements
- The landlord must sign a contract stating that he/she will not raise the rent or sell the home to make a profit on the weatherization improvements
- If the landlord is eligible based on his/her income, the home can be weatherized under the program without any stipulations

The program process begins with the "intake" at the local CAP offices within each county. Potential program participants (most commonly comprised of people walking into the local offices and asking for assistance in paying their energy bills or requesting weatherization services) are required to fill out applications. The information required includes the following:

- Name, address, telephone number
- Number of occupants living in the home
- Income level and source of income

Customers are required to supply documentation to substantiate the income level they claim. The level of funding available is calculated according to the federal standards using a point system.

The point system is based on the following factors: the total family members, the number of handicapped and elderly family members, income level, primary fuel type, and the percent of the fuel cost as a factor of total yearly income. The sum of the point values assigned to each of these scoring attributes is the basis for a residential dwelling being selected for weatherization. Because of limited funds availability, households with the highest point total are selected for weatherization measure installation. Customers who are handicapped and/or elderly receive additional points which gives them priority. The following table illustrates the point system process as reported by one of the CAP Agencies.

Table III.2

<u>THE</u>	POINT SYSTEM -	
1.	Total family members	
	<ul> <li>Each family member</li> </ul>	1 point
	<ul> <li>Each family member who is elderly</li> </ul>	5 points
	<ul> <li>Each family member who is handicapped</li> </ul>	5 points
	<ul> <li>If a family member is both elderly &amp; handicapped</li> </ul>	10 points
2.	Income level	
	<ul> <li>120-125% of federal income poverty guidelines</li> </ul>	1 point
	<ul> <li>75-99% of federal income poverty guidelines</li> </ul>	2 points
	<ul> <li>Less than 75% of federal income poverty guidelines</li> </ul>	3 points
3.	Primary fuel	
	<ul> <li>If primary fuel is electricity</li> </ul>	8 points
	<ul> <li>If primary fuel is propane</li> </ul>	6 points
	<ul> <li>If primary fuel is fuel oil</li> </ul>	4 points
	<ul> <li>If primary fuel is wood</li> </ul>	4 points
	<ul> <li>If primary fuel is natural gas</li> </ul>	3 points
	<ul> <li>If primary fuel is coal</li> </ul>	2 points
4.	The percentage of total income used to pay fuel bill	
	<ul> <li>If 0-6% of fuel cost is from total yearly income</li> </ul>	2 points
	<ul> <li>If 7-13% of fuel cost is from total yearly income</li> </ul>	4 points
	<ul> <li>If 14-20% of fuel cost is from total yearly income</li> </ul>	6 points
	<ul> <li>If 21-27% of fuel cost is from total yearly income</li> </ul>	8 points
	<ul> <li>If 28-33% of fuel cost is from total yearly income</li> </ul>	10 points
	<ul> <li>If 34% or more of fuel cost is from total yearly income</li> </ul>	15 points

Four of the five CAPs reported that during the intake process, WKG customers are informed of the additional funding available through the WKG CARES Program. One CAP stated that it did not have a standardized policy of informing the potential participants that partial funding and expanded benefits were provided by WKG CARES. Although the CAPs claim that they inform potential participants of WKG's role, only 29% of customers surveyed as part of this evaluation indicated an awareness of WKG's contribution. (See Chapter IV, Table IV.3.)

Finding: Regardless of the reason for this general lack of awareness, and while the gaining of recognition for program funding is not a significant motivation for WKG's participation, the Company nonetheless has the right to have program participants educated as to WKG's role and the additional weatherization services made possible as a result of WKG funding.

Once eligibility is determined, a pre-inspector from the CAP Agency is sent to the home to do a pre-inspection. As part of the process, a blower door test is performed to determine the level of air leaks and infiltration within the home. Additionally, each home is required to have a State-required National Energy Audit (NEAT) performed using a standardized State-supplied audit form. This audit determines the savings investment ratio, energy savings in dollars per year, and the total energy savings. All information gathered during the pre-inspection is entered by the CAP agency into its own database. The NEAT Audit information is entered into a NEAT software program (provided by the State Weatherization Dept.), and the output from this software determines the most cost-effective measures for that particular home. Based on this information, the CAPs will select the mix of energy efficient measures they can install with the funds they have available through the weatherization program. If the CAP client is a Western Kentucky Gas customer, then the CAPs are authorized by WKG to install additional approved weatherization improvements to the home through the WKG CARES Program. If gas leaks are found during the pre-inspection, they are fixed immediately.

Four of the five CAPS have internal installers who do the actual installations. One of the CAPs bids the work out to sub-contractors. Once the CAP determines the measures that will be installed, the materials are ordered and secured. The individual CAPs reported that the installer or installation team is sent out to do the work within a "few months" of the preinspection date. This apparent delay is the result of work load as was reported by the agencies. Once the work is completed, a post-inspection is performed by the CAP within 4 months (depending on the CAP agency, the post-inspection could take place as early as 1 week after installation) to verify that the measures were installed properly. At this time, the post-inspectors answer any questions or concerns the customer may have. All CAPs reported that they give additional education at this time.



WKG imposes its own quality control on the steps taken by CAPs. They inspect at least 10% of all homes that have been weatherized using WKG CARES funds. During the program evaluation period in which this report is based, the Program Manager stated that WKG had found only one home which WKG had post-inspected, in which the CAP agency's installer was required to return and do additional work. (The installer had not placed a safety mat under the furnace which was required by Federal Code Standards.) Additionally, the State audits a random selection of treated homes each year. The federal government also performs audits on a random basis.

Finding: Based on interviews of CAP management, it is apparent that the agencies are very cooperative with WKG and appreciative of WKG's injection of added weatherization funding. The agencies are positive in the attitude toward their clients and display a pattern of concern and interest in their dealings with payment troubled or otherwise disadvantaged low-income residents.

### D. Program Objectives and Goals

The program was designed as a three-year pilot program as mentioned previously. The original program goal was to fund a maximum of 300 low income homes per year (25 per month) with a target maximum of \$1,500 per household, not to exceed a total program cost of \$450,000 per year. In an effort to provide the CAPs with additional flexibility in spending the available WKG funds, WKG modified the program at the end of approximately 12 months of operation from a *maximum* funding limit of \$1,500 per home to an *average* funding of \$1,500 per home, with a *maximum* of \$2,000 per home. This design change afforded the CAPs the ability to do more inclusive weatherization installations in homes as needed.

The participating CAP agencies are required to use the WKG CARES Program funding for all WKG customers who are eligible for federal weatherization funding under the poverty guidelines.



Goals were originally communicated by kick-off meetings with the weatherization program directors. This was followed by written documents defining the measures that would be acceptable for the program. It appears that the only goal communicated called for an annual combined total of 300 homes to be weatherized by all agencies.

According to the Program Manager, formal written contracts were never established. Instead, during the initial meeting with the CAPs, WKG explained the program in detail, which included budget amounts per eligible home. The CAPs were very receptive to this concept and immediately began using the program.

While establishment of the original program goals was very specific, no individual agency goals were established. This was because it was difficult to predict goal attainment potential without in-depth study of each CAP's service area demographics and the condition of housing stock within the area.

Finding: However, while no individual agency goals were established, each CAP agency was instructed, as the program commenced, to treat as many homes as was possible. Thus, while there were no specific agency targets, there was the mandate to extend program benefits to as many homes as was possible.



### E. Program Results

In its first 15 months of operation, the program has resulted in the weatherization of a large number of residential dwellings. However, the originally intended goals have not been achieved. Table III.3, which follows, presents the agencies and the number of homes each has weatherized during the evaluation period with WKG CARES funding and participation.

Community Action Program Agency	Number of Homes Weatherized*
Audubon Area Community Services	46
Bluegrass Community Action Agency	8
Community Action of Southern Kentucky	57
Pennyrile Allied Community Action Agency	74
West Kentucky Allied Community Action Agency	45
Total	230

Table III.3

\*Homes weatherized from November 1, 1996 through January 31, 1998.

Prorating the original goal of 300 homes per year results in a target of 375 homes for the period being evaluated. Thus, the actual number of homes treated equates to 62% of the goal. Likewise, prorated program expenditure targets for the period total \$562,500. Actual expenditures were \$ 358,291 representing 64% of the goal. Recognizing that the first four months of the program were considered the "start-up" period, a more reasonable goal would be a monthly average of homes weatherized beginning in month 5 (March 1997) and continuing through month 17 (February 1998) of the program. This monthly average of 17 homes falls 8 homes short of the 25 homes per month goal (300 homes per year) mentioned above.



## F. Program Promotion

## 1. Program Promotion by WKG

The principal form of promotion offered by WKG is comprised of press releases that are prepared and disseminated to the public broadcast media and Public Service Broadcasting stations. WKG also provides print announcements to all newspaper media. *(See Appendix A for a sample of the newspaper ad.)* Bill stuffers are used once each year to promote the WKG CARES Program. None of the WKG CARES funding is used to pay for program promotion. This decision to forego paid advertising was made by the Collaborative.

Finding: This lack of WKG funded paid advertising may be one factor in the inability of the program to achieve its first-year participation goal.

During the course of this evaluation, it was reported that because of competition for free advertising announcements or notices, there was little if any guarantee that WKG CARES promotional materials would ever be aired or printed. This finding is supported by the fact that only one customer who was surveyed (2.2%) reported learning of the program through a newspaper ad. (See Chapter IV, Table IV.4.) The results of this evaluation indicate that the principal form of promotion for this program is a combination of recognition of need by the CAPs and simple word-of-mouth.

Finding: One other potential source of referrals to the program are WKG customer account representatives who interface with customers in-person and by telephone.

These account representatives represent an excellent referral/recognition source and provided with a brief, easy-to-understand program fact sheet, could be a valuable asset to the program.

Based on the responses of customers who were asked how they had first learned of the program, this avenue for the WKG CARES program promotion is not being capitalized upon.



### 2. Program Promotion by CAP Agencies

The CAP agencies use a variety of methods to promote weatherization. While not promotion in the strict sense of the word, the most common means of introducing customers to WKG CARES occurs in the CAP offices during the application process where WKG customers should be informed that, if they are eligible for the weatherization assistance, they are also eligible for the WKG CARES Program.

Since the CAP agencies provide a wide range of services to their communities in addition to the weatherization program, there are numerous internal need recognition and referral mechanisms. For example, if a customer comes into the agency for help in paying their utility bill and the agency feels their bills appear too high, they will be referred to the weatherization section of the agency. CAP agencies have placed ads in local newspapers, on the radio, and placed flyers at local food stamp offices promoting weatherization services. One CAP sponsored an "Open House" at its offices. Another CAP publishes a newsletter, "Senior Star", which has included an article about the WKG CARES Program. Additionally, the low-income community has learned about the WKG CARES Program through word-of-mouth promotion.

#### 3. Education Component

Finding: It was reported by the agencies that education is provided at various stages of the program process although there is no standardized education component, nor are brochures describing WKG's role in the program or its recommendations for energy-use reduction given to customers.

When customers come into the Agency to fill out applications, the Agency generally provided suggestions on how to save energy and lower energy bills. For example, clients might be asked if they keep windows open in cool weather because it is too hot in the home. If they give an affirmative reply, the client is then informed that if they lowered the thermostat setting the home wouldn't be as warm and windows could remain closed, thus conserving energy.

During the initial inspection of the home, the pre-inspectors explain what they are doing during the audits (NEAT) so customers can understand the reasons for the audit, together with its results. During the course of the weatherization, the installers will explain how the equipment/measures work and how to maintain the various measures that have been installed. As an example, if a new heating system is installed, the installer will explain and/or demonstrate how to change the filters and how to set the thermostat temperature as part of a periodical maintenance program. When the post-inspection is performed, the inspector will again explain how to use each piece of equipment, where appropriate. Most of the CAPs will provide phone numbers and contact names for the customer to use if they have questions relating to the weatherization measures or equipment that were just installed.

### G. Data Tracking and Reporting

### 1. Tracking

Each CAP Agency is responsible for collecting and tracking specific information for each qualifying customer who takes advantage of weatherization/WKG CARES services. The forms used by the CAPs are provided by the Kentucky State Cabinet for Human Resources - Department for Social Services for the Weatherization Assistance Program, and include the following:

Forms Required by the Federal Government:

- WX800 "The Application & Prioritization for Weatherization Services"
- WX900 "The Work Order"

Forms Required by the State:

WX710 Completed Dwelling Report

As was the case with the previously mentioned blower door test results, the information on these forms is entered into databases established by the individual CAP agencies. Each month, the CAP agencies are required to send the WKG Program Manager a hard copy of the above forms for each home (weatherization measures installed) completed within the month that utilized WKG CARES funding. Additionally, each CAP sends an invoice for



the past month, which includes a summary of the number of homes that have completed weatherization in the past month and the WKG portion of the costs.

These tracking forms are standardized Weatherization Program forms which contain detailed information concerning each customer. Because of the WKG CARES Program's piggy-back nature, the information that these forms provide represents all the relevant and necessary information required by the WKG CARES Program. There are no additional tracking formats required solely to meet the requirements of WKG CARES. The WKG Program Manager decided that WKG would be able to work with the forms that the CAPs were already required to use as part of the weatherization program. The program manager simply asked the CAPs to include with these forms one additional page detailing the actual WKG funding used, by qualifying measure, for each home which had been weatherized. The program manager felt that the CAPs should not be burdened with additional pagerwork.

Finding: WKG and the Program Manager have attempted wherever possible to minimize program administrative expenses so that the maximum possible amount of program funding can flow to actual weatherization measures.

## 2. Program Performance Reporting by WKG

On a monthly basis, the WKG Program Manager compiles a report summarizing the current status and accomplishments of the program. The report contains the following information:

- The reporting month
- The name of each participating CAP Agency
- The current program year fund status
- The average cost per home weatherized
- The number of homes weatherized to date
- The number of completed WKG post-inspection audits for the program year



This report is provided to each member of the Collaborative, WKG Senior management, and to each of the participating CAPs, thus providing updated information to all of the interested parties. A copy of this report is included in Appendix B.

### H. Program Communication Issues

All CAP agencies reported generally good communications with WKG. At the inception of the program, the Program Manager arranged to conduct the WKG CARES Program meetings in conjunction with the ongoing CAP quarterly meetings (which were established a number of years ago for the purpose of discussing DSM weatherization projects). After the CAP meeting was adjourned, the Program Manager would meet separately with the Weatherization Program Directors of the WKG participating CAPs. These meetings addressed specific issues and questions related to the WKG CARES Program and its operation. One of the principle reasons for these meetings was to ensure that the CAP agencies were reporting information in a uniform manner.

After the first complete program year, the Program Manager has reduced the frequency of status and update meetings due to the fact that the CAPs are now familiar with the WKG CARES Program and have generated a limited number of questions or concerns in recent months.

Finding: It was reported by some agencies that they would appreciate more frequent meetings if for no other reason than to provide a forum for sharing common experiences.

Although not in the setting of a formal meeting, the CAP Weatherization Program Directors reported that they communicate by telephone with the Program Manager, on average, once a month.

Finding: WKG offers an "open door" policy in which the CAP Directors can always call and have their questions and concerns addressed and answered. The agencies were quite supportive of the WKG Program Manager and were particularly appreciative of his availability and quick turnaround in providing needed information.


While not an event that occurred during the course of this evaluation, it was reported that oil overcharge funds, which have been provided to CAP agencies, are eliminated, effective July 1, 1998. This will stress the resources of the CAPs to treat as many homes as was done in the past, both with and without WKG CARES Program support. More frequent meetings/contacts with the CAPs will foster dialogue between all interested parties that may help reduce any negative impact of the loss in funding.

## I. Barriers to Program Participation

Finding: Low income residents can be prevented from participating in weatherization/WKG CARES for a number of reasons.

One such reason results from a lack of promotion reaching the targeted population. A number of potential customers may be unaware of the weatherization program and the WKG CARES Program simply because information concerning the availability of the program does not reach them. This is not a criticism of CAP efforts, but rather is a recognition that promotional expense allocations are limited in the case of the CAPs and non-existent in the case of WKG.

A second barrier to participation stems from the fact that there are strict limits to the number of rental dwellings allowed to be weatherized each year with program supplied funding. The number of rental homes that can be weatherized each year is limited to between 10% and 15% of the total number of homes treated. As a result, a number of low-income families cannot avail themselves of program benefits. Note that there is a lack of consistent interpretation/understanding of this limitation among the participating CAP agencies. Some CAP agencies believe the maximum saturation of rental homes can be 10% while others believe it to be 15%. Regardless of the correct proportion, it was reported that the State allows exceptions to the rule and is willing to "work with the CAPs" on this point when needed.

A third barrier arises when the customer does not meet the technical criteria for income eligibility and thus, is prevented from participating in the program even though he/she may effectively still be poor. Agencies reported that there is strict adherence to Federal income standards, with no exceptions, in the certification process.

Roof repair is a forth barrier to participation. As reported by one of the CAP Weatherization Directors, according to the Federal regulations, if as part of the weatherization treatment roof repair is required, the CAP must also install attic insulation, wall insulation, floor insulation and general heat waste prevention measures, *through its own funding allotment for the residence.* Since the CAP has limited funds and the roof repairs are generally quite costly (upwards of \$2,000 on average, as reported by the CAPs), a number of homes cannot be treated as the funds remaining after the roof repairs will not cover the cost of the other required measures. (See Section "J" for a further discussion of this issue.)

## J. CAP Agency Reactions to the WKG CARES Program

Finding: The WKG CARES Program owes its success to the efforts of the CAP agencies.

As such, the reactions of these individual organizations to the program and their suggestions for how it might be improved are valuable for program planning in the future. As mentioned previously, each of the five participating agencies was contacted and interviewed, some on multiple occasions.

Finding: There is a clear understanding among the agencies of the purpose of WKG CARES and the role it plays in helping low-income families reduce their gas utility bills. Without exception, the reaction of the CAPs to the program and to WKG is highly positive.

Clearly the program is seen as a means of improving the comfort level of participants and the level of safety within their homes through the ability of the agencies to provide additional measures paid for by WKG CARES. Interestingly, one agency indicated that a very visible result of the program was its ability to buy and install completely new outside gas heating units in a number of cases, a situation that would not have been possible without WKG funding.

Direct WKG CARES Program promotion by the agencies is somewhat informal. In other words, it is comprised of flyers at food stamp offices, open houses, newsletters, and some level of radio and newspaper advertising. The use of radio and newspapers is limited because of the associated expense. As is borne out by the comments of participants, the WKG CARES program promotion by the agencies, other than word-of-mouth, appears to be somewhat ineffective. Present WKG CARES Program promotional activities are a deficiency in the program that, if restructured, may be a motivator for additional participation.

The CAPs operate and provide weatherization under a group of state and federal regulations. While securing modifications to these regulations may be quite difficult, if not impossible, the regulations concerning homes with roof leaks is particularly troublesome to the CAPs. The requirement to install a series of potentially very costly insulation measures coincident with roof repairs effectively prevents the CAPs from treating a number of homes in the low-income housing stock. Further, the limitation cannot include piggy-back funding from WKG CARES. Repair of roof leaks does not meet the cost/benefit requirements of WKG CARES. A few months into the program, a recommendation was made from the program manager to the Collaborate requesting that WKG CARES money be allowed for roof repairs. The recommendation suggested that WKG pay for the roof repairs and then take the credit for any energy savings that resulted from whatever weatherization was done to the home with DOE funds, since nothing could have occurred without first repairing the roof. The Collaborate declined approval on this recommendation.

Finding: The net result of this situation is that the CAP agencies are precluded from treating a number of homes that otherwise could have been treated if their funding could be expended on roof repairs while WKG CARES funding could be dedicated to the follow-on insulation and associated inside improvements that are cost/benefit justified.

Again, the issue may not have a solution, but appeals by the CAPs to the State and/or the Federal government may result in a waiver if only for the time period during which WKG CARES is in effect.



As was discussed elsewhere, while the monthly program status reports received from the WKG Program Manager are useful to the agencies, three of the five agencies suggested that the quarterly meetings with the Program Manager, that took place in the early months of the program, be reinstated.

Ŷ,



# IV. Customer Reaction to the Program - Telephone Survey Results and Analysis

## A. Description of the Data-Collection Process

For the purpose of this report, 230 homes were weatherized during the program evaluation period. A telephone survey was conducted using a sample of 50 of the 230 WKG CARES participants (50 homes) to determine their perceptions of the program, satisfaction with the program and the measures installed, whether or not they were aware WKG supplied the CAPs with partial funding for each of the homes, and to gather general demographics.

All program data used for the process evaluation was taken directly from the information collected from the client applications and inspection forms prepared by the CAP agencies for the program. The sample of 50 participants includes homes weatherized from program inception (October 1996) through January 31, 1998.

#### B. Sample Design and Confidence Levels

## 1. Sample Design

In order to ensure non-bias survey results, a sample of 50 program participants was selected from the 230 completed projects. This sample was stratified by CAP agency. Table IV.1 presents the weatherization project populations and the number of telephone surveys completed within each stratum.

Sample Size and Error calculations for Proportions 1998 Western Kentucky Gas WKG CARES All CAP Agencies 230 Participant Survey Population Sample Size 50 Survey Sample 0.800 Estimate of Customer Satisfaction from Survey 0.4000 std dev 0.0566 std error 0.884652 finite pop correction 0.0500 adjusted std error 90 confidence band 1.645 # of std errs <u>8.2%</u> Margin of Error in Proportion +/-

0.800 +/- 0.082
0.718
0.882



## Figure 2

Sample Size and Error calculations for Proportions		
Western Kentucky Gas		
All CAP Agencies		
Population	230	Participant Survey
Sample Size	50	Survey Sample
		N
Actual Results of Customer Satisfaction from Survey	0.880	
std dev	0.3250	
std error	0.0460	
finite pop correction	0.884652	
adjusted std error	0.0407	
confidence band	90	
# of std errs	1.645	
	0 70/	
Margin of Error in Proportion +/-	<u>6.7%</u>	

Error Band	0.880 +/- 0.067
Lower Bound	0.813
Upper Bound	0.947



CAP Agency	Number of Homes Weatherized	Number of Completed Home Surveys
Audubon Area Community Services	46	11
Bluegrass Community Action Agency	8	5
Community Action of Southern Kentucky	57	` 11
Pennyrile Allied Community Action Agency	74	12
West Kentucky Allied Community Action Agency	45	11
Total	230	50

#### Table IV.1

#### 2. Confidence Levels

Figure 1, on the following page, illustrates the basis for the sample design, showing that out of a population of 230 participants, a survey sample of 50 participants would produce an *estimated* customer satisfaction level of 80% with a margin of error in proportion of +/- 8.2 percentage points.

Figure 2, shows that of the 230 participants, the survey sample of 50 participants, based on the *actual* response rate to the question, produced an overall satisfaction of 88% with a margin of error in proportion of +/- 6.7 percentage points. Overall satisfaction is defined for this illustration as respondents who indicated either "1" (extremely satisfied) or "2" (very satisfied) when asked their opinion on their overall satisfaction of WKG as their service provider.



Table IV.2 details the individual CAP agency margins of error based on a positive response to overall satisfaction of WKG as the customer's service provider.

Cap Agency	Total Number of Responses	Positive Satisfaction Level	Margin of Error
Audubon Area Community Services	11	100%	0%
Bluegrass Community Action Agency	5	100%	0%
Community Action of Southern Kentucky	11	91%	12.8%
Pennyrile Allied Community Action Agency	12	75%	18.8%
West Kentucky Allied Community Action Agency	11	82%	16.6%
OVERALL	50	88%	6.7%

Table IV.2

A statistical comparison between the lowest satisfaction level CAP agency and the overall satisfaction level showed no significant difference in response. Therefore, the survey concludes that the CAP Agency was not a determining factor of overall satisfaction.

Finding: It was also determined that the customers' overall satisfaction with WKG as their energy provider was not effected by whether customers knew WKG had paid for some of the energy savings measures.

Fourteen customers (out of the 50 customers surveyed) knew that WKG had partially paid for the measures that were installed. Of these, 86% of the customers (12 of the 14), indicated that they were "extremely satisfied" or "very satisfied" with WKG as their energy provider. Thirty-four of the 50 customers surveyed indicated that they did not know WKG partially paid for the measures that were installed. Of these, 88% were "extremely satisfied" or "very satisfied" with WKG as their energy provider. The two remaining customers indicated "don't know" as to whether they knew if WKG had paid for some or all of the measures installed.

## C. Survey Tabulations and Findings

All 50 participants surveyed were confirmed to be Western Kentucky Gas customers and recalled having energy savings measures installed in their homes within approximately the past year, free of charge.

The first issue explored concerned the understanding or awareness that program participants had of WKG's role in the program. Participants were asked whether they were aware that WKG had paid for some of the energy savings measures installed in the home. Although, as reported earlier, four of the five CAP agencies indicated that they informed potential participants of WKG's role during the qualification process, less than one-third of participants were aware or could recall WKG's part in the weatherization effort.

Awareness of WKG's Role	Number of Responses8	
Yes	14 (29.2%)	
No	34 (70.8%)	
Don't Know	2	

\* Percentage based on 48 responses.

Participants were then queried on the subject of how they had first come to learn of the program. As Table IV.4 indicates, word-of-mouth remains the largest information vehicle for this program. How had participants learned about the WKG CARES Program? Only one person had heard of the program through newspaper advertising. This may support the finding that the program may be missing opportunities as a result of the prohibition against paid WKG CARES Program promotional messages.

	Number of Responses*
Friend	11 (23.9%)
CAP agency	10 (21.7%)
Relative	6 (13.0%)
Had Participated in Weatherization Program Before	5 (10.9%)
Social Worker	3 (6.5%)
Neighbor	2 (4.3%)
Word of Mouth	2 (4.3%)
Newspaper Ad	1 (2.2%)
Office Where Bills are Paid	1 (2.2%)
Inquired on Own	1 (2.2%)
A Contractor	1 (2.2%)
Grandson's Medical Insurance	1 (2.2%)
State Agency	1 (2.2%)
Knew From HUD	1 (2.2%)
Don't Know	4

Table I	V.4
---------	-----

\*Percentages based on 46 responses.

As Table IV.5 shows, there is also the lack of a thorough understanding of who installed the measures. While no participant incorrectly credited WKG as the installer of the measures, only 78% could recall that it was a CAP agency who had installed the measures.

	Number of Responses*
Western Kentucky Gas	0
The CAP Agency	39 (78%)
The State of Kentucky	0
Don't know	11 (22%)

Tabl	e l	V.5
------	-----	-----

Customers were asked to rate how satisfied they were with a range of program attributes. Respondents provided ratings on a scale of 1 to 5, where 1 indicated "Extremely Satisfied" and 5 indicated "Extremely Dissatisfied". The responses to the question, "How satisfied were you with the following aspects of the program ...?" were as follows:

	Average Satisfaction Rating	Number in Average	Number of "Don't Know" Responses
The ease of scheduling the installation	1.23	48	2
The friendliness and helpfulness of the contractor who installed the measures	1.24	49	1
The information you received about the program	1.33	46	4
The quality of the measures installed	1.47	49	1
The amount of time it took to install the measures your home	1.48	50	0
The energy cost savings as a result from the equipment installed	1.50	44	6

Table IV.6

One factor that contributes to either a positive or negative impression of the program, in general, is the performance of the installation contractor. As Table IV.7 demonstrates, with the exception of six customers (12%) who indicated that there had been some form of damage to their homes associated with the weatherization program, respondents reported timely appointments, reasonably good after-project clean-up, and freedom from damage to the premises.

	Yes	No	Don't know
Arrive at the scheduled time (keep the appointment)*	47 (97.9%)	1 (2.1%)	2
Cause any damage during the installation	6 (12%)	44 (88.0%)	0
Clean up after the installation	46 (92%)	4 (8.0%)	0

\*Percentage based on 48 responses.

The following list of damages were reported by the participants:

- Installer drilled hole through main electrical wire in home (installer fixed this and re-paneled the wall.)
- Installer damaged the hardwood floor
- Customer noticed leak in roof after installer had weatherized home
- Customer noticed leak in pipe after home was weatherized (installer returned and fixed this).
- Customer was unhappy with the repair to a crack in the wall. The installer used red putty.

The goals of the WKG CARES Program include improving the overall comfort and safety of participants. While more than 90% of respondents indicated at least some improvement in comfort levels in the home, the ability to pay energy bills, home safety and the overall condition of the home, only 77% perceived that the value of their home had been improved by the installation of program measures. Table IV.8 provides answers to the question, "Did any of the following conditions improve since your participation in the WKG CARES Program?"

	A Great Deal	Somewhat	Not At All	Don't Know
The comfort of your home	37 (74%)	10 (20%)	3 (6%)	0
Your ability to pay your energy bills*	34 (72.3%)	9 (19.1%)	4 (8.5%)	3
The overall condition of your home*	32 (68.1%)	11 (23.4%)	4 (8.5%)	3
The safety of your home*	30 (65.2%)	13 (28.3%)	3 (6.5%)	4
Your ability to pay bills other than energy bills*	26 (56.5%)	15 (32.6%)	5 (10.9%)	4
Your overall health*	17 (47.2%)	8 (22.2%)	11 (30.6%)	14
Whether the value of your home has increased*	11 (31.4%)	16 (45.7%)	8 (22.9%)	15

Tab	le	IV	.8
Tab	le	IV	.8

\*Percentages do not include "Don't Know" responses.

Table IV.9 investigates the incidence of "Snap-Back", or, using increased amounts of energy once improvements in energy efficiency have been put into effect, i.e., trading bill savings for increased comfort. Twenty-five percent of respondents did indicate increasing the heating system thermostat setting, while 4% indicated increasing the temperature of their water heater. "Since the measures have been installed, have you increased your comfort level in your home by increasing the temperature settings on ..."

	Yes	No	Don't know
Raising the heating system thermostat	12 (25%)	36 (75%)	2
Raising the water heater thermostat*	2 (4.3%)	45 (95.7%)	2

\*Note that one respondent did not have a water heater.

Again using the satisfaction rating scale mentioned earlier, participants were asked to provide an, "Overall satisfaction rating with Western Kentucky Gas as your energy service provider". WKG received high marks for overall customer satisfaction, with the average response rating of 1.26. (Note that "Extremely Satisfied" =1, "Satisfied" =3, and "Extremely Dissatisfied" =5). Table IV.10 explores the issue of growth in positive opinion of WKG as a result of the program and their participation in it. The following responses were provided to the question, "Has your overall opinion of Western Kentucky Gas improved, fallen, or remained the same as a result of your participation in the program?"

Opinion of WKG	Number of Responses*	
Has Improved	22 (53.7%)	
Has Fallen	2 (2.4%)	
Has Remained the Same	17 (43.9%)	
Don't Know	9	

\*Percentages based on 41 responses.

The one participant who was dissatisfied stated that the local payment center lost her payment and she had to pay her WKG bill twice that month. It is interesting to note that 53.7% of respondents indicated an improvement of their opinion of WKG, while only 29% had reported knowing that WKG paid for the a portion of the measures that had been installed. This apparent dichotomy is probably best explained by recognizing that during the telephone survey conversation, respondents gained a better understanding of WKG's role and may have changed their opinion *as the survey was progressing*.

- Respondents were then asked if they had participated in any other energy conservation programs during the past year. The responses indicate that only one customer (2%) out of 50 respondents had participated in another program.
- When asked whether they had received any other weatherization services from any other state agencies or programs, again only one respondent indicated that they had received such services.

One important program guideline, as mentioned earlier, was that no more than 10% to 15% of treated homes could be rental dwellings. As Table IV.11 shows, the guideline is being followed for the overall program.

Table IV.11			
	Number of Responses		
Own	47 (94%)		
Rent	3 (6%)		

- The average number of years participants have been at their current address was 14.1 years, with 46 years being the longest and 10 months being the least amount of time.
- The program treated at least 9 homes, and potentially more, that were at least 58 years old, Table IV.12 presents responses to the question, " Approximately, in what year was the home built?"

	Number of Responses*
Between 1980 to 1984	3 (11.5%)
Between 1970 to 1979	2 (7.7%)
Between 1960 to 1969	6 (23.1%)
Between 1950 to 1959	4 (15.4%)
Between 1940 to 1949	2 (7.7%)
1939 or Before	9 (34.6%)
Don't Know	24

\*Percentages based on 26 responses.



Table IV.13 indicates that 12% of respondents make use of some form of supplementary heat in their homes. Customers were asked, "Do you use an additional or supplemental source of heat for your home?"

T-61- 11/49

Additional or Supplemental	Number of Responses	
Source of Heat?		`
Yes	6 (12%)	
No	44 (88%)	

All 6 positive respondents stated that they used electricity as the additional/supplemental source of heat.

When asked whether any major changes had been made in the home since the energy saving measures were installed, four respondents indicated that such changes had been undertaken.

Table	IV.14

Major Changes?	Number of Responses	
Yes	4 (8%)	
Νο	46 (92%) ·	

The major changes reported by these four program participants were as follows: *Participant 1*: Built porches onto the front and back of the home.

Participant 2: Purchased a new dishwasher and dryer.

Participant 3: Purchased a new refrigerator and 2 additional people living in the home.

Participant 4: Purchasing a used refrigerator.

On average, 2.4 people are currently living in each of the 50 surveyed households.

The age breakdown of the residents of these dwellings are described in Table IV.15 which follows:

	Average Number	Number in Average
Under the age of 5	1.50	8
5 - 13 years old	1.45	11
14 - 19 years old	1.09	11
20 - 24 years old	1.25	4
25 - 34 years old	1.17	6
35 - 44 years old	1.19	16
45 - 54 years old	1.09	11
55 - 64 years old	1.30	10
65 - 74 years old	1.21	14
75 or older	1.00	5

Table I	V.15
---------	------

Almost half of the respondents (52%) had head of households between 55 and 74 years of age. Table IV.16 illustrates all head of household ages.

Age of Head of Household	Number of Responses
20 - 24	1 (2%)
25 - 34	3 (6%)
35 - 44	9 (18%)
45 - 54	7 (14%)
55 - 64	11 (22%)
65 - 74	15 (30%)
75 or older	4 (8%)

Table IV.16

• Of survey respondents, 6 (12%) were males and 44 (88%) were females.

## V. Impact Evaluation Overview and Findings

This chapter presents the results of the evaluation for the WKG CARES Program from the impact perspective. AEG conducted a statistical billing analysis to estimate the change in annual Mcf consumption for a sample of program participants and non-participants. The resulting net savings estimates were then projected to the program population and used to develop estimates of participant bill savings (\$) and benefit-cost ratios by which the cost-effectiveness of the WKG CARES Program could be judged.

## A. Objectives

As mentioned earlier, the following were the primary objectives associated with this evaluation study:

- To estimate the energy savings (Mcf) attributable to the WKG CARES Program
- To develop estimates of the WKG CARES Program's cost-effectiveness from the societal, ratepayer, utility, and participant perspectives
- To develop an estimate of the annual bill savings accruing to program participants

## B. Methodology

## 1. General Approach

The general framework of this analysis is a comparison of the pre- and post-treatment Mcf consumption for program participants. These consumption values have been annualized and adjusted for weather effects in order to control two possible confounding influences (differences in the number of days and daily temperatures in the pre- and post-treatment periods). In addition, this study incorporated a comparison group of program non-participants in order to control other possible influences on energy consumption (e.g., economics).

Table V.1 describes the study periods established for this analysis.

Time Period	Designation	
November 1995 - October 1996	Pre-Treatment Period	
November 1996 - September 1997	Treatment Period	
April 1997 - March 1998	Post-Treatment Period	

Table V.1

Note that the overlap of the treatment and post-treatment periods (April - September 1997) should not be problematic since the most significant energy impacts would occur during the heating season (November - March) which is contained entirely within the post-treatment period.

## 2. PRISM Methodology

The primary analysis technique employed in this evaluation was the Princeton Scorekeeping Methodology (PRISM). The energy savings estimated through the PRISM billing analysis is based on the changes in weather-normalized annual energy consumption from the pre- to the post-treatment period on a customer-by-customer basis. Since impacts are measured on a customer-by-customer basis, the PRISM analysis does not segment savings by end-use or individual treatment measure.

PRISM uses an iterative estimation procedure to fit the following model to billing period data for each customer in the analysis sample (for both the pre and post-treatment analysis periods):

$$F_i = \alpha + B H_i$$
 (T) +  $\varepsilon_i$ 

where  $F_i$  is the average daily consumption for customer i within each billing period, calculated by dividing billing energy use by the number of days in the billing period. The intercept term ( $\alpha$ ) measures average daily non-weather sensitive usage. B is an estimate

AEG Applied Energy Group

of the heating slope, and  $H(\tau)$  measures heating degree-days per day, based on the reference temperature that optimizes the fit of the model to the data.<sup>6</sup>

The PRISM heating load model generates several model statistics by customer for each analysis period (i.e., pre- and post-treatment):

- Coefficient estimates and t-statistics for the intercept and heating slope
- Annualized baseload (i.e., non-weather-sensitive) energy consumption
- Weather-normalized estimates of annual heating consumption
- An overall estimate of normalized annual energy consumption (NAC)
- Goodness of fit statistics such as R<sup>2</sup> and the coefficient of variation (CV).

These statistics are employed to test the reliability of the PRISM model results, to screen unreliable data from the analysis, and to develop the energy savings estimates. AEG used its standard criteria to remove "unreliable" results from the sample: ( $R^2 > .7$ , CV(NAC) < 0.25). The key model statistics are defined in the following table.

PRISM Statistic	Description
NAC	Normalized Annual Consumption (Mcf); an estimate of the total annual household energy consumption in a year with typical weather; generated for both pre-treatment and post-treatment periods.
R²	The amount of variation in the daily energy consumption explained by the PRISM model; a standard goodness-of-fit measurement.
CV(NAC)	Coefficient of variation of the NAC; equal to the SE(NAC) / NAC.

Table - V.2

<sup>&</sup>lt;sup>6</sup> The reference temperature is used to calculate average heating degree days which is equal to the greater of zero, or the reference temperature minus average daily temperature.

## 3. Analysis Stages

There were four primary stages in this analysis:

a. Initial Data Processing and Screening

The participant and non-participant billing data sets were initially formatted to conform with the standard PRISM data structure. Estimated reads were combined with subsequent actual reads, missing periods were identified and associated with a usage of -1, and overlapping periods due to tenant turnover were combined to create a single billing history. The processed data sets were then analyzed to screen out those customers whose billing histories contained certain data anomalies which would adversely impact the reliability of the analysis. Specifically, it was required that customers had to have greater than 100 days of billing data in both the pre-treatment (November 95/ March 96) and post-treatment (November 97/March 98) winters. Furthermore, the PRISM software requires that there be a minimum of four actual reads.

b. Development of Gross Energy Savings Estimates

Normalized Annual Consumption (NAC) estimates were developed for each of the program participants for the pre-treatment and post-treatment periods using the PRISM software. The PRISM outputs were then screened using the associated model statistics to ensure that the results were reliable. For each participant in the final sample, a change-score was calculated as below:

$$\Delta NAC_{gross} = NAC_{pre} - NAC_{post}$$

The sample results were extrapolated to the program population by calculating a mean perparticipant energy savings which was multiplied by the total number of participants during the relevant time period (November 96 - September 97). This process yielded an estimate of the gross energy savings. Note that in the calculation of the mean per-participant savings AEG screened out influential observations by removing from the sample those customers whose  $\Delta$ NAC fell outside the range of the sample mean plus or minus one standard deviation.

## c. Development of Net Energy Savings Estimates

In order to control for the impact of non-programmatic influences on consumption (other than weather), AEG included in its analysis a control group of program non-participants. As with the program participants, pre- and post-treatment NAC values were estimated using PRISM. The mean % change in consumption from the pre-treatment to the post-treatment period was calculated as a control adjustment factor (CAF):

 $CAF = 1 - [\Sigma(PreNAC_{non-participant}) - \Sigma(PostNAC_{non-participant})] / \Sigma(PreNAC_{non-participant})$ 

The pre-treatment NAC for program participants was then adjusted to reflect what would have occurred in the absence of the WKG CARES Program and a net energy savings estimate was calculated:

$$\Delta NAC_{net} = (CAF^*NAC_{pre}) - NAC_{post}$$

The aggregate energy savings was calculated as above using a mean per-participant *net* energy savings. Again, a single screen was employed to remove influential observations (i.e., those whose change-score fell outside the range of the sample mean plus or minus one standard deviation).

## d. Development of Benefit-Cost Ratios

The net energy savings values estimated above were incorporated into a benefit-cost software model (BENCOST) which provided benefit-cost ratios from the societal, ratepayer, utility, and participant perspectives.



#### C. Data Sources

This evaluation required the integration of data from four primary sources: the WKG CARES Program database, WKG customer consumption data, regional weather data, and WKG corporate data :

## 1. WKG Cares Program Tracking System

The WKG CARES Program tracking system is a comprehensive database containing the relevant data associated with a program participant and the nature of their participation in the program. This includes:

- Customer information and demographic data such as customer name and address
- Structural information such as the size of the structure, existing insulation levels, and primary heating fuel
- Installation data such as the type of measures installed and actions taken through the program

#### 2. Customer Consumption Data

Consumption data was obtained through Western Kentucky Gas's customer information system. Data collected included gas consumption, read dates, read days, and bill codes. WKG initially provided billing data for a total of 230 participants for the time period November 1995 through February 1998. In order to enhance the reliability of the evaluation findings, WKG provided a supplemental billing data set which included an additional one to two months of billing data for 161 program participants. Note that the second set of billing data was limited to those program participants treated by the end of September 1997 whereas the initial data set included customers treated through January 1998. Customers treated after September 1997 were excluded from the final billing analysis due to insufficient post-treatment billing data.



WKG also provided customer billing data for 124 program non-participants for a similar time-frame. Each participating CAP agency had provided WKG with an abbreviated list of its clients who had been treated through a non-WKG program at least one year prior to the inception of the WKG CARES Program. This non-participant group was included in the analysis to control for non-programmatic factors other than weather which could have impacted energy consumption among the low-income population segment through this time-period (i.e., economic and pricing factors).

#### 3. Weather Data

A data set of historical daily mean temperatures was obtained from the National Climatological Data Center Internet site and supplemented with current data obtained from the Midwestern Climate Center. The data sets included weather data for three stations in WKG's service territory: Louisville, Bowling Green, and Paducah. Participants and non-participants were matched to the three sites based on a substring of their WKG account numbers.

#### 4. WKG Corporate Data

Additional data inputs were provided by WKG which were employed in the benefit-cost analyses which included:

- retail rates
- commodity and supply costs
- current and historical customer and sales figures
- escalation factors
- corporate income tax rates
- net operating revenues
- gross operating revenues
- corporate discount rates

## D. Impact Evaluation Findings

## 1. Sample Design

The following tables present the sample design analysis for the participant and non-participant samples, respectively. A total of 114 participants and 112 non-participants were used in the final change-score analysis. These values represent 71% of the study population and 90% of the initial non-participant sample.

As previously noted, the intent of the screening process is to maximize the reliability of the final savings estimates by eliminating those data points which fail certain reliability criteria or which are considered to be statistical outliers. Furthermore, in order to accurately model natural gas consumption through the winter heating season, it was necessary to require that the billing histories contain actual (*rather than estimated*) data for a minimum of 100 winter days (November - March) in the pre- and post-treatment periods. Table V.3 describes the Participant Sample Design and Table V.4 describes the Non-Participant Sample Design.

Screen	Sample Size (pre-screen)	Quantity Removed	Sample Size (post-screen)
Initial Sample			161
Less than 100 winter days in pre or post	161	1	160
Less than four actual periods in pre or post	160	5	155
Failed PRISM screens (R <sup>2</sup> >= .7, CV(NAC) <= .25)	155	29	126
Failed Outlier Screen: ΔNAC > x̄ + σ or ΔNAC < x̄ - σ	126	12	114
Final participant sample			114

## Table V.3 - Participant Sample Design

Table V.4 - Non-Participant Sample Design

Screen	Sample Size (pre-screen)	Quantity Removed	Sample Size (post-screen)
Initial Sample			124
Less than 100 winter days in pre or post	124	1	123
Less than four actual periods in pre or post	123	0	123
Failed PRISM screens (R <sup>2</sup> >= .7, CV(NAC) <= .25)	123	5	118
Failed Outlier Screen: ΔNAC > x̄ + σ or ΔNAC < x̄ - σ	118	6	112
Final non-participant sample		<b>·</b>	112



## 2. Comparison of Key Statistics (Sample and Study Population)

The following tables demonstrate the representativeness of the final participant sample through a comparison of *measures installed*, *structure size*, and *implementation agency* between the final analysis sample and the study population.

Table V.5 includes for each measure treatment the percentage of participants receiving the treatment in both the final study sample of 112 and study population of 161. As is evident by inspection, the treatment profile for the final sample closely approximates that of the study population. In all cases, the percentage of participants in the final sample who received a particular treatment differed by less than 3% from the percentage of the study population receiving that same treatment.



Measure	Percentage of Sample	Percentage of Population
Attic Insulation	94%	94%
Wall Insulation	46%	47%
Floor Insulation	71%	70%
Duct Repair	32%	¥ 29%
Ceiling Insulation	93%	94%
Hot Water Tank Wraps	63%	62%
Infiltration Measures	95%	96%
Water Heater Replacement	16%	16%
Water Heater Repair	44%	42%
Primary Heat Replacement	39%	40%
Primary Heat Repair	41%	41%
Primary Heat Cleaning	24%	22%
Secondary Heat Replacement	3%	5%
Secondary Heat Repair	7%	6%
Secondary Heat Cleaning	3%	3%

## Table V.5 - Comparison of Treatment Profiles (Sample and Study Population)

Table V.6 presents a comparison of structural characteristics between the study sample and the population. Energy consumption is a function of building size (among other factors) and, given similar energy conservation treatments, energy savings is a function of building size as well. Therefore, it is important in these impact studies to establish that the size of the homes does not differ between the study sample and population. It is particularly critical for analyses in which savings estimates are developed on a perparticipant basis. As the following table demonstrates, the distribution of these structural variables for the sample closely approximates that of the population.

	Sample Mean	Population Mean
House Volume (f³)	8,257 (sd =3,212 )	8,232 (sd= 3,271)
Living Space (f²)	1,065 (sd =361 )	1,077 (sd = 366)

 Table V.6 - Structure Size

Table V.7 below illustrates that the distribution of participants across CAP agencies closely matches that of the study population and is not significantly skewed. The most evident differences, which are statistically insignificant, are that the Southern KY CAP is under represented by 2.7% and the Pennyrile CAP over represented by 3.5%.

CAP Agency	Percentage of Sample	Percentage of Population
Audubon	16.7%	17.4%
Blue Grass	3.5%	3.7%
Southern KY	24.6%	27.3%
Pennyrile	37.7%	34.2%
West KY	17.5%	17.4%

Table V.7 - Distribution Across CAP Agencies

#### 3. Gross and Net Change-Score Results

Table V.8 presents the results of the change-score analysis. As explained above in the *Methodology* section, mean estimates of pre and post-treatment consumption were developed using PRISM for a sample of the program participants in order to calculate a gross energy savings. The gross per-participant energy savings is then simply the difference between the pre and post-treatment consumptions (i.e, *87.8 - 70.9*) or 16.9 Mcf.

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	87.8
Mean Post-Treatment Normalized Annual Consumption (Mcf)	70.9
Mean Gross Per-Participant Change in NAC (Mcf)	16.9
Program Participants (n)	161
Program-Wide Energy Savings (Mcf)	`2,721

 Table V.8 - Participant Gross Energy Savings

Mean estimates of pre- and post-treatment consumption were also developed for a sample of program non-participants in order to control non-programmatic influences on consumption. These customers experienced a minimal reduction in consumption of one ccf (0.1 Mcf). This represents a 0.13% reduction.

**Table V.9 - Non-Participant PRISM Results** 

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	78.8	
Mean Post-Treatment Normalized Annual Consumption (Mcf)	78.7	
Mean Per-Participant Change in NAC (Mcf)	0.1	
% Change in Annual Consumption	0.13%	
CAF (1-% Change in Annual Consumption)	0.9987	

The final step in the development of net energy savings was to adjust the participant's mean pre-consumption to account for the "naturally- occurring" reduction in consumption experienced by the program non-participants. This adjustment is a relative adjustment and entails reducing the pre-treatment consumption by 0.13% (or multiplying by 0.9987, the *Control Adjustment Factor*). This adjustment yields a net energy savings of 16.8 Mcf. The total energy savings for the study population is then 2,704.8 Mcf. The following table shows these results.

Mean Pre-Treatment Normalized Annual Consumption (Mcf)	87.8
Control Adjustment Factor	0.9987
Mean Pre-Treatment Energy Consumption (Mcf) Adjusted for Control Adjustment Factor (CAF)	87.7
Mean Post-Treatment Normalized Annual Consumption (Mcf)	70.9
Mean Net Per-Participant Change in NAC (Mcf)	16.8
Program Participants (n)	161
Program-Wide Energy Savings	2,705

Table V.10 - Participant Net Energy Savings

Appendix E contains the raw PRISM Output for program participants and non-participants. Append F contains the output generated by the SAS Proc Means procedure used in developing the change scores for participants and non-participants.

#### 4. BENCOST Model Results

Table V.11 and V.12 present the results of the benefit-cost analysis. For each of the four perspectives, the net present value (NPV) of the program's benefits and costs are calculated and compared. Note that the benefit-cost tests were run with and without the environmental externality adder. This adder captures the social benefit of reducing environmental damage associated with the consumption of natural gas and factors solely into the societal test benefit-cost ratio.



Societal Test	1.12	
Ratepayer Impact Measure	0.65	
Utility Test	0.66	
Participant Test	681.35	

Table V.11 - Benefit-Cost Results (Excluding Externality Adder)

Table V.12 - Benefit-Cost Results (Including Externality Adder)

Societal Test	1.17
Ratepayer Impact Measure	0.65
Utility Test	0.66
Participant Test	681.35

Table V.13 summarizes the inputs used in the modified BENCOST model. Retail rates, commodity and demand costs, tax rates and profit margins were provided by Western Kentucky Gas. Program costs were calculated by summing the treatment costs across the study population (\$216,517.42) and allocating the first-year Collaborative costs on a perparticipant basis (\$185.50 per treated home). The environmental damage factor for natural gas was obtained from the Minnesota Department of Public Service and is based on 1995 Environmental Protection Agency data.

The social discount rate of 3% is the current rate of inflation and is used to discount the benefits associated with the societal benefit-cost test. Since there is only a one-time program cost, there is no need to discount the future stream of social costs. The use of the inflation rate in the social test is justified given that it is the only neutral intergenerational risk factor which can be used to discount the future streams of program benefits. Costs and benefits for all other tests were discounted at WKG's corporate discount rate of 10%.

Benefit-Cost Analysis Input	Value
Retail Rate	\$4.87 / Mcf
Commodity Cost	\$3.81 / Mcf
Demand Cost	\$127.59 / Mcf / Year
Peak Reduction Factor	1%
Environmental Damage Factor	\$0.00 / Mcf (\$0.27 Mcf)
Variable O & M Cost	\$0.03 / Mcf
Total Sales	49,693,865 Mcf Growth Rate = 4.4%
Total Customers	173,958 Growth Rate = 2.7%
Utility Discount Rate	10%
Social Discount Rate	3%
Effective State and Federal Tax Rate	38%
Net Operating Income as % of Total Operating Income	5.8%
Program Costs (Direct Installation Costs)	\$246,382.75
Program Participants	161
Per-Participant Energy Savings	16.8 Mcf
Measure Lifetime	20 years

## Table V.13 - Benefit-Cost Inputs

Appendix F contains the detailed BENCOST Model Inputs and Outputs.

## E. Participant Bill Savings

Based on the estimates of energy savings, the annual bill savings for program participants can be estimated at \$81.84 (1998\$) based on an average of January - March, 1998 residential retail rates. This represents 16.7% of the typical participant's annual natural gas bill. The bill savings estimates were derived from a net energy savings of 16.8 Mcf with a retail rate of \$4.87 per Mcf and a monthly customer charge of \$5.10.

## VI. Overall Conclusions and Recommendations

#### A. General Conclusions

WKG CARES appears to be functioning effectively with minimal administrative support required, either from WKG itself, or from the Collaborative under whose auspices the program functions. The program is a good example of the way in which a DSM program can be designed that capitalizes on the strengths of a number of partners in the implementation process.

WKG CARES focuses on qualifying low-income residential homeowners that meet firm federal income guidelines. The program reaches these customers through promotional or marketing channels already established by the five CAP agencies who currently participate in the delivery of program services. WKG CARES promotional efforts, by both the CAPs and WKG are somewhat weak and could be improved upon, particularly if WKG were authorized to commit to some amount of paid-for advertising.

Since the CAP agencies already have staffs that install weatherization measures, there is no need for WKG to secure its own implementation contractor. As a result, the program manifests a high degree of inherent efficiency as its infrastructure is already in place via the agencies. The CAPs are positive in their approach to serving low-income customers and appreciative of the opportunity to partner with WKG in this effort. Although program parameters were established regarding allowable program measures, the agencies are afforded considerable latitude in the way in which they commit WKG Program budgets.

Overall program goals were well-defined initially, and status reports are provided monthly by WKG. However, there isn't a clear understanding among *all* of the CAPs as to the current goals or their expected level of participation. The CAPs do recognize that the original program limit of \$1,500 per customer was altered to allow expenditures of up to \$2,000 per customer, as long as the overall average remained at \$1,500. WKG CARES has not met its participation targets during the first year of operation, due at least partially to the fact that program funds cannot be used for program promotion. Besides the inability to promote the program, there are other barriers to participation that are the result of federal or state regulation that would be difficult if not impossible to alter. It is possible that

AEG Applied Energy Group
by undertaking a prudent amount of paid advertising, WKG CARES participation levels could be improved to the point of goal attainment.

There is a certain lack of standardization in the ways in which an educational "component" is delivered coincident with weatherization services. Similarly, positive public relations opportunities are lost for WKG as the CAPs do not uniformly explain to customers that a portion of the benefits they will receive are being provided by WKG.

More than 50% of customers who participated in the program indicated an improvement in their opinion of WKG once gaining an understanding of WKG's role in the program. Few however, understood WKG's contributory role in the program at the time of the survey.

Overall, average customer reaction to the various elements of the program, i.e., ease of scheduling, opinion of the workforce, quality, cost savings, etc., were very positive, with each major scoring topic averaging between 1.23 and 1.50 on a scale of 1 to 5 (1 = "extremely satisfied"; 5 = "extremely dissatisfied").

The program delivers real benefit for participants in terms of actual energy-use reduction, while increasing comfort levels. The impact evaluation demonstrated that, collectively, the 161 customers whose post-treatment gas consumption history was sufficient to allow calculation of savings attributable to the program, saved 2,705 Mcf of gas.

The net per-participant energy savings has been estimated as 16.8 Mcf for those WKG customers who participated in the program between November 1996 and September 1997. This represents approximately a 19% reduction in energy consumption. Program participants thereby experienced an average bill reduction of \$82 (a decrease of 16.7% in a typical participant's natural gas bill). Additionally, the fact that the program passes the Societal benefit-cost test, with a score of 1.12 (excluding environmental externality benefits), which was a key target in the original program design.

#### B. Recommendations

The following recommendations are suggested to improve the implementation of WKG CARES. This is already an effective DSM program and community service. These recommendations are offered as a means of further improving an already valuable program.

From the process perspective:

Secure permission from the Collaborative to prudently fund paid promotional efforts.

The effectiveness of a promotional strategy relying solely on free advertising time/space was limited in this program. If a *small* portion of the program funding surplus can be used to buy appropriated advertising coverage, the investment may not only attract further participants, but will also help WKG and the Collaborative in maximizing the benefits made available from the program over its three-year implementation period. It is recognized that the level of such spending must be tightly controlled to ensure continued positive benefit/cost analysis results.

 Hold more frequent status meetings with CAP agencies. Use these as an opportunity for cross-sharing of ideas as well as a vehicle for the transmission of goal achievement reports.

Throughout the course of the evaluation, it became apparent that there are certain program elements where a common understanding among the CAPs and WKG is not present. One means of improving the situation is to reinstate the periodic meetings between all parties that were present at the beginning of the program. The cross-sharing of ideas, as well as the exploration of common problems and concerns may help to heighten the effectiveness and efficiency of the program. Develop an educational supplement that can be uniformly distributed by all CAPs to all incoming participants that heightens the conservation message while clearly conveying the role played by WKG.

There is no standard "educational component" offered by all CAPS. While each agency reports that it provides information to all participants, it is likely that the lack of a formalized "curriculum" or set of educational materials results in somewhat incomplete transfer of energy-saving suggestions and ideas to participants. It is worthwhile for WKG to consider either preparing materials on its own or securing ready-made booklets, brochures, etc., that it can supply to the CAPs for subsequent distribution to participants.

 Contact the remaining CAP agencies that share a portion of WKG's territory and solicit their participation in the program.

Three additional CAP agencies serve WKG customers. They are not included in the current roster of participating agencies, nor are they understood as having a significant WKG customer population. Nonetheless, WKG should contact these agencies for the purposes of assessing its interest in joining with the other five CAPs. Since there is no incremental expense on WKG's part to add additional CAP agencies, any contribution of additional participants from the additional agencies will only stand to benefit the program.

• Design and implement written contracts with each of the participating CAP agencies.

Written contracts do not presently capture the relationship between the agencies and WKG. For the purposes of liability protection, as well as to exactly define the responsibilities of all parties, written contracts should be set in place.

# From the impact perspective:

- These results indicate that the WKG CARES Program should continue as is with no modifications in delivery or measure specification suggested by the level of the achieved program impacts.
  - The original program planning and evaluation of measures to be included have been confirmed as a cost effective mix of weatherization components. The installation of these measures should continue without alteration throughout the balance of the program cycle.
- The impact evaluation could not determine whether there is any degradation in the savings level through time. WKG should consider performing an evaluation of year-one participants following year-three to determine if such savings degradation has occurred.

Such an analysis would determine the level of long-term benefit made available to customers through program participation. The evaluation will also show if customers have traded off further increases in comfort level for decreased bill savings, or whether some treatable element in the home's building shell has further deteriorated, thus reducing the effectiveness of the installed measures. A study of this type would not be appropriate until after the end of the three-year program cycle.

# Appendix A - Bill Stuffer

¥.

# Appendix B - Monthly Summary Report

Y.

# WKG CARES MONTHLY REPORT (interim) - October 1997

								commisted Audit
				Avg/Home	<b>-</b>	COMES		
A MANAGE		FUNDS					Ter Yes	Prod Year
AGENCI			TARY METTON	Program Year	Begin	000	TADE 501	
	Begin	October	LINE WITHOUT	1001	Relance		1997	1997
			1997	ICCT				6
	DOTETCO	12 22 22 C	610 261 47	\$1,308.78	28	2	00	
1.4.100	\$37,752.91	90. ULC. 18		1 0 0 0 0 0 T		0	9	-
		40.00	S8.268.79	107 · 0/ 5 · 75				
TINA Grass	\$8,268.73			27 872 75	55	9	61	0
	ATC ACC ED	\$10,807.13	587,763.03			╀	ļ	~
Pennyr11e	OCC '0/ &		AC 757 10	\$1.255.14	44	9	0	1
	\$52 072 45	\$10,684.65	04-101 70¢			ľ	52	5
Southern KI		22 200 00	448.966.75	\$1,483.84	28	n		
	\$41,466.75			55 55 55		91	180	1
MORT AL		220.502.34	\$247,019.76	\$1,576.18	407			
sub total	41.17C'0125		23 189 R2	\$185.50				
	<u>\$11,389.82</u>				-1			
Collaborative		10 000	\$280.409.58	\$1,557.83				
CENT TOTAL	\$249,907.24	1 \$30°, 305 - 31			ב			

G. Smith, A. Chevront, S. Crocker, K. Bowmar, M. Marks H. Jones, B. Vincent, S. Mhite, M. McGill, S. Nest

copies:

Y,

Appendix C - Program Manager Protocol, CAP Agency Protocol & Participant Telephone Survey Instrument

Wester	n Kentucky Gas WKG CARES Participant Survey
Customer Information:	
Name:	
Address:	
Telephone Number:	
Survey Number:	
<u>Response Codes:</u>	
1. Busy signal	7. Left message
2. Call back	8. Moved
3. Wrong number	9. Disconnected
4. No answer	10. Completed
5. Refused	11. Has no phone
6. Left message	12. Other:

CONTACT RECORD												
Attempts	Interviewer	Date	Time	Response								
1												
2			······································									
3												
4												
5												

# [INTERVIEWER: ASK TO SPEAK TO THE PERSON NAMED ON THE SAMPLE]

Introduction: Good morning/afternoon/evening, my name is \_\_\_\_\_\_ and I'm calling long distance from Applied Energy Group, a market research company. We are conducting a brief survey on behalf of Western Kentucky Gas. The purpose of this survey is to discuss the energy saving measures that were recently installed in your home. This survey will only take a few minutes of your time. Your feedback is very important to Western Kentucky Gas , and please be assured that your answers will be keep strictly confidential.

- 1. I would like to talk to you about the energy saving improvements that were installed in your home at [READ ADDRESS]. Is that the address that I have reached you at?
  - yes [continue]
  - no [terminate]
  - don't know [terminate]
  - **refused** [terminate]
- 2. Are you a Western Kentucky Gas customer?
  - yes [continue]
    no [terminate]
    don't know [terminate]
  - **u** refused [terminate]
- 3. Did you have energy savings measures installed in your home by [INSERT APPROPRIATE CAP AGENCY NAME] over the past year or so, free of charge?
  - □ yes [continue]
  - no [terminate]
  - don't know [terminate]
  - **u** refused [terminate]
- 4. Are you aware that Western Kentucy Gas paid for some of the energy savings measures that were installed at your home?
  - yes
  - 🗋 no
  - don't know
  - c refused
- 5. How did you learn about this program?



# 6. Who installed the energy saving measures in your home? [READ LIST]

- U Western Kentucky Gas
- the [insert CAP agency name] agency
- the State of Kentucky
- don't know
- refused
- 7. Please tell me how satisfied you were with each of the following using a scale of 1 to 5 where "1" is "extremely satisfied" and "5" is "extremely dissatisfied". How satisfied were you ...?

N

	Extremely <u>Satisfied</u>		Satisfied		Extremely Dissatisfied	Don't <u>Know</u>
a. The amount of time it took to install the measures your home	1	2	3	4	5	6
<ul> <li>b. The information you received about the program</li> </ul>	1	2	3	4	5	6
c. The ease of scheduling the installation	1	2	3	4	5	6
d. The quality of the measures installed	1	2	3	4	5	6
e. The friendliness and helpfulness of the contractor who installed the measures	1	2	3	4	5	6
f. The energy cost savings as a result from the equipment installed	1	2	3	4	5	6

#### 8. Did the installer...

	Yes	No	<u>Don't Know</u>
a. Arrive at the scheduled time? [keep the appointment]	1	2	З
b. Cause any damage during the installation?	1	2	3
c. Clean up after the installation?	1	2	3

Now we'd like to know whether or not you feel that any of the following conditions have improved since your participation in the WKG Cares program. Do you feel that the following has improved "a great deal", "somewhat" or "not at all"?

		A Great Deal	Somewhat	Not At All	<u>Don't Know</u>
a	. The comfort of your home	1	2	3	4
ь	. Your overall health	1	2	3	4
c	. The safety of your home	1	2	3	4
d	. The overall condition of your home	1	2	3	4
e	. Your ability to pay your energy bills	1	2	3	4
f.	Your ability to pay bills other than energy bills	1	2	3	4
g	. Whether the value of your home has increased	1	2	3	4

10. Since the measures have been installed, have you increased your comfort level in your home by increasing the temperature settings on ...

		<u>Yes</u>	<u>No</u>	<u>Don't know</u>
a.	the heating system thermostat?	1	2	3
b.	the water heater?	1	2	3

11. Again using a scale of 1 to 5 where "1" is "extremely satisfied" and "5" is "extremely dissatisfied": How satisfied are you with Western Kentucky Gas as your energy service provider overall? *(circle one choice)* 

Extremely <u>Satisfied</u>		<u>Satisfied</u>		Extremely Dissatisfied	Don't <u>Know</u>
1	2	3	4	5	6

- 12. Has your overall opinion of Western Kentucky Gas improved, fallen, or remained the same as a result of your participation in the program? *(circle one choice)* 
  - improved
  - fallen
  - **remained the same**
  - don't know
  - refused

If not satisfied in 11 or 12, why are you not satisfied with Western Kentucky Gas ? [RECORD VERBATIM]

- 13. Have you participated in any other energy conservation programs in the last year?
  - yes [go to question 13a]
    no [go to question 15]
    don't know [go to question 15]
- 13a. If yes, who offered it?

[IF SPONSORED BY ANY ORGANIZATION OTHER THAN A STATE AGENCY, GO TO QUESTION 15; ELSE GO TO QUESTION 14.]

14. Have you received any other weatherization services from any other state agencies or programs?

yesno

#### My last few questions are for statistical purposes only.

- 15. First, do you own or rent your home?
  - 🛛 own
  - c rent
  - don't know
  - refused
- 16. How long have you been at this address? [ IF LESS THAN ONE YEAR, RECORD MONTHS]

RECORD YEARS \_\_\_\_\_ RECORD MONTHS \_\_\_\_\_

WKG CARES - Participant Survey

4/7/98 🛚 5

17. Approximately, in what year was your home built? [READ LIST IF NECESSARY]

- □ after 1995
- D between 1990 to 1994
- D between 1985 to 1989
- D between 1980 to 1984
- D between 1970 to 1979
- D between 1960 to 1969
- D between 1950 to 1959
- D between 1940 to 1949, or
- □ 1939 or before
- don't know

# 18. Do you use an additional or supplemental source of heat for your home?

- Yes [go to question 19]
- □ No [skip to question 20]

19. What is the additional or supplemental source of heat for your home? [DO NOT READ LIST. *Multiple answers accepted*.]

- **G** gas from underground pipes serving the neighborhood/natural gas
- □ gas bottled, tank, or LP/Propane
- □ electricity
- □ fuel oil
- kerosene
- 🛛 wood
- don't know

20. Have you made any major changes in your home since the energy saving measures were installed? [*new major appliances, more/less people in home, people home more/less than before, added rooms/shut off rooms*] [RECORD VERBATIM]

21. Including yourself, how many people live in your household at least six months of the year?

RECORD #: \_\_\_\_\_

- 22. How many of these people are.. [READ LIST]?

23. What is the age of the head of your household? [IF ONLY ONE IN HOUSEHOLD, ASK:] What is your approximate age please? [READ LIST IF NECESSARY] Under 18 18 - 19 20 - 24 25 - 34 35 - 44 45 - 54 55 - 64 65 - 74 75 or older Refused [DO NOT READ]

# 24. RECORD FROM OBSERVATION

Male

Female

THOSE ARE ALL OF MY QUESTIONS. THANK YOU VERY MUCH FOR DOING THIS SURVEY.

# WKG CARES 1997 Program Evaluation Process Evaluation Program Manager Protocol

- A. Introduction
- 1. Title/role
- 2. What are your general responsibilities?
- 3. How long have you been involved with the WKG CARES Program? What are your specific responsibilities with regard to WKG CARES?
- 4. Please give a complete description of the WKG CARES Program.

# B. Goals

- 1. What do you see as the goal/purpose of WKG CARES?
- 2. Would you say the program is currently achieving its goals and objectives? In what ways is it achieving them? [*PROBE: Do you think these goals/purposes are achievable under the current program design and with current resources? Why or why not?*]
- 3. Does WKG CARES have any specific customer relations goals? If yes, what are they?
- C. Program Promotion
- 1. Please describe how potential program participants are selected. How do you define an eligible customer? How were these eligibility standards determined?
- 2. Do you have input to the way in which the program is promoted by the CAP agencies? Are you satisfied or dissatisfied with this process?
- 3. Are any promotional activities undertaken by WKG independent of these agencies? How are any such promotions coordinated with the CAPs?
- 4. Do you think participants understand WKG's role in the improvements they receive? Would you like to change this in anyway?

- 1

# D. Program Delivery

- 1. Please describe the process through which a customer receives services through the CAP agencies and how WKG CARES is folded into the process. Take me step by step through the program delivery, telling me who does what and what happens.
- 2. Have any problems been discovered with services provided by the CAPS?
- 3. Have there been any changes to the overall process since the program began, apart from those already discussed? If so, what and why?
- 4. How satisfied do you think participants are with the program? Why?
- 5. What obstacles do the CAP agency work crews encounter? How do they deal with those obstacles? Are you satisfied with the supervision of the crews that is supplied by the CAP agencies?
- 6. How often are post-inspections done by the CAP agencies? Does each participant have a post-inspection? What specific quality control measures does WKG have in place? Has WKG performed additional post-inspections for quality control? How often is this done? What kinds of problems have been discovered? If discovered, are the agencies willing to remedy the issue?
- 7. Do you feel the program needs to offer any additional services? If so, which services?

### E. Program Management

- 1. Please describe the management structure for the WKG CARES program. Who are the key players in the program delivery from both WKG and the agency perspective? What are their roles? Please describe any changes since the program started.
- 2. Has the program management structure on either side changed since the program began?
- 3. Describe how WKG staff and the CAP agencies work together. How successful/satisfactory is that relationship? Why?
- 4. Is WKG satisfied with the CAP agencies' work on the program to date? Why/why not?
- 5. Did all of the five agencies presently involved with the program, begin with the initiation of the program last year? Have any been added or dropped? Why?

# F. Data Tracking/Evaluation

- 1. How does WKG track information on WKG CARES? Who collects and stores the information? What information is collected? How is it stored?
- 2. What reports do you receive on program progress from the agencies? Are they adequate? Have you given the agencies specific forms and report formats they must prepare and submit? Are they doing so?
- 3. Do you have any suggestions for improving the data tracking system and reporting?
- 4. What kinds of evaluation efforts do you think are important for this program? What information would you like to have about how the program is working?

## G. Summary Comments

- 1. What are the greatest strengths of the WKG CARES program? Likewise, what are the program's weaknesses?
- 2. What do you feel are the biggest challenges in the program's future?

.

3. Do you have any additional comments?

# WKG CARES 1997 Program Process Evaluation Community Action Agency Protocol

#### A. Introduction

I'd like to start with some background information.

- 1. What is your official title, and what are your responsibilities?
- 2. How large is your organization? [*PROBE for number of employees and approximate size of area served/number of people served.*]
- 3. What do you see as the purpose of the WKG CARES program? What is your understanding of the program goals and objectives?
- 4. What are your organization's goals for your participation in this program?

a. Do you establish goals cooperatively with WKG?

b. Are you reaching any goals that you established with WKG? *[PROBE for specific numbers*] How often do you talk about program status with WKG?

#### **B.** Program Implementation

- 1. Who in your organization, besides yourself, is involved in the program? Are you still involved?
- 2. What is your understanding of the reasons why your organization was involved in the delivery of WKG program services? [*PROBE: What were your expectations*?] How did your experiences differ from your expectations?
- 3. What specific services does your organization provide outside of the WKG CARES program? What additional services can you provide to your clients because of the WKG CARES program? How do you determine the level of additional funds to use for customers who are eligible for the WKG CARES program versus those that do not qualify for the program?
- 4. Have you or your staff received any instruction about the program from WKG? If so, what type and when? Is this sufficient?
- 5. Has your staff had many questions and concerns about the WKG CARES program? Does your staff have any questions or concerns that have developed since the program began? If so, what type of questions or concerns? How have they been answered or handled by WKG?

- 6. Please describe the process through which an eligible low-income household receives expanded services from your organization under the WKG CARES program.
- 7. How is eligibility established for this program? [*PROBE to see if they think this criteria is sufficient means for establishing eligibility.*]
- 8. Are there any measures not eligible for the WKG CARES funds that you think should be included? Why?

# C. Program Promotion

- 1. Please describe how your agency was recruited to become involved with the WKG CARES program. What led you to participate in the program? What do you believe are the benefits of participation?
- 2. Prior to your involvement with the program, did you have any concerns about it? If so, what were they, and how were they resolved?
- 3. How do you promote the services your organization offers? Do you do anything extra or different to promote the program to households you believe will be eligible for WKG benefits. How does promotion differ for ineligible versus eligible households?
- 4. Is there an educational component that you promote? Is it offered to all of your organization's retrofit clients or only to recipients of specific measures? Please explain.

### D. Customer Response

- 1. What energy and program issues do customers appear to be most concerned with?
- 2. Do customers appear to be satisfied with the energy efficiency services provided by your organization. What about with measures facilitated through the WKG CARES program?
- 3. How important do you think the non-quantifiable services (i.e. increased comfort, health, safety, improved ability to pay electricity bill, improved condition of household) are to program participants?

# E. Interaction with WKG

- 1. With whom do you work at WKG?
- 2. How would you describe the relationship between your organization and WKG? [*PROBE for the strengths, weaknesses, potential improvements*]

- 3. How is the level of communication between your organization and WKG? [*PROBE for the strengths, weaknesses, ways to improve*]
- 4. How frequent are communications with WKG?
- 5. Based on your experience in the program, are there any changes that you think should be considered to make your dealings with WKG more effective?

## F. Data Tracking

- 1. How do you track customers who receive program services?
- 2. What records do you keep on services delivered? Are the records of customers who receive benefits from the WKG CARES program different form those that do not? Please explain.
- 3. How would you describe the tracking methods used? (paper, computer, etc.)
- 4. Does WKG provide you with specific forms which you must submit for eligible customers? What is your opinion of these forms?
- 5. How often do you submit invoices to WKG? Are you paid in a timely fashion?
- 6. Does WKG ever accompany your staff during a post inspection? How often? How would you describe the level of control exercised by WKG? Does WKG perform any quality control?

## G. Conclusion

- 1. To conclude, what do you perceive to be the strengths of WKG CARES program over the past year?
- 2. In what ways could the program be improved?
- 3. Do you have any additional comments?

Appendix D - PRISM Modeling Output

X.

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
1	3	PART	0.9822	54.55	2.61	0.543	0.160	0.278	0.035	821.67	39.82
2	6	PART	0.9967	63.62	1.29	0.478	0.103	0.242	0.011	1107.51	20.83
3	16	PART	0.9846	46.00	1.50	0.094	0.059	0.202	0.021	269.82	16.03
4	19	PART	0.9927	64.15	1.68	-0.061	0.185	0.309	0.018	1203.34	37.43
5	23	PART	0.9951	61.80	1.41	-0.016	0.147	0.369	0.020	1283.37	34.92
6	25	PART	0.9837	64.00	2.46	0.827	0.164	0.168	0.014	964.30	32.59
7	30	PART	0.9976	67.19	0.95	0.016	0.100	0.239	0.007	1110.56	19.23
8	33	PART	0.9955	66.85	1.36	-0.005	0.072	0.126	0.005	572.23	13.76
9	35	PART	0.9824	59.03	2.94	0.234	0.074	0.102	0.012	388.56	17.53
10	36	PART	0.9935	61.00	1.77	-0.037	0.140	0.271	0.018	891.56	29.83
11	37	PART	0.9877	64.80	2.09	-0.034	0.101	0.138	0.010	553.96	22.10
12	47	PART	0.9946	58.71	1.71	0.347	0.075	0.191	0.014	685.41	17.28
13	48	PART	0.9919	71.54	2.12	-0.038	0.086	0.088	0.005	485.47	13.49
· 14	49	PART	0.9950	59.71	1.42	0.561	0.073	0.186	0.011	781.40	17.28
15	50	PART	0.9889	66.32	2.13	0.292	0.104	0.120	0.008	640.04	19.68
16	51	PART	0.9602	43.82	3.54	0.439	0.109	0.294	0.075	441.50	31.95
17	52	PART	0.7625	41.00	3.92	0.726	0.447	0.494	0.164	621.58	124.93
18	53	PART	0.9898	54.49	1.65	0.441	0.058	0.134	0.011	460.85	14.23
19	54	PART	0.9751	54.21	2.54	-0.106	0.166	0.251	0.032	513.27	41.24
20	59	PART	0.9912	59.57	2.21	1.063	0.086	0.165	0.015	896.88	19.62
21	62	PART	0.9926	69.35	1.96	0.157	0.103	0.122	0.007	683.71	17.84
22	63	PART	0.9938	64.67	1.15	0.144	0.038	0.125	0.010	563.28	18.52
23	64	PART	0.9909	66.70	1.80	0.283	0.118	0.150	0.008	778.21	23.18
24	65	PART	0.9965	64.00	1.06	0.087	0.094	0.247	0.009	1005.27	20.75
25	66	PART	0.9964	61.59	1.28	0.189	0.119	0.311	0.015	1143.77	25.58
26	67	PART	0.9369	70.53	5.18	0.257	0.383	0.154	0.023	926.95	71.25
27	72	PART	0.9979	59.80	1.06	0.469	0.031	0.122	0.005	550.67	7.12
28	83	PART	0.9778	64.82	2.99	0.088	0.361	0.322	0.033	1355.16	73.07
29	84	PART	0.8769	67.00	5.71	-0.087	0.491	0.169	0.031	743.08	93.39
30	85	PART	0.9931	63.46	1.81	0.058	0.063	0.103	0.006	415.48	12.53
31	128	PART	0.9940	66.00	1.50	0.153	0.184	0.283	0.013	1288.42	35.23
32	131	PART	0.9940	63.37	1.73	0.300	0.128	0.230	0.014	984.82	25.48
33	139	PART	0.9736	58.00	3.22	1.138	0.145	0.168	0.023	885.35	34.37
34	154	PART	0.8672	85.00	-9.00	-1.133	0.720	0.177	0.022	1321.22	129.99
35	155	PART	0.9729	55.75	3.25	0.477	0.118	0.155	0.023	550.41	29.35
36	156	PART	0.9931	64.12	1.72	0.736	0.154	0.250	0.015	1260.21	30.28
37	157	PART	0.9855	65.15	2.44	0.686	0.099	0.102	0.008	678.75	18.73
38	166	PART	0.9579	60.00	4.22	0.078	0.109	0.100	0.017	344.58	26.35



OBS HOUSEID LABEL R2 TAU\_H SE\_TAUH BASE SE\_BASE HSLOP SE\_HSLOP POSTNAC SE\_NAC

1	1	PART	0.9894 62.19	1.98	0.691	0.088	0.150	0.015	810.60	19.73
2	2	PART	0.9580 64.59	3.59	0.327	0.103	0.081	0.013	448.09	21.71
3	3	PART	0.9769 65.56	2.76	0.328	0.124	0.123	0.014	645.79	24.81
4	4	PART	0.7255 79.00	30.24	-0.037	1.517	0.066	0.021	501.51	73.91
5	6	PART	0.9902 60.25	1.31	0.709	0.102	0.254	0.018	1070.68	25.49
6	7	PART	0.9549 71.00	5.68	0.108	0.349	0.136	0.020	790.70	44.88
7	8	PART	0.9832 60.17	2.21	0.566	0.073	0.117	0.013	580.83	16.43
8	9	PART	0.9766 68.10	2.88	0.178	0.247	0.191	0.018	987.48	41.94
9	10	PART	0.9208 71.72	4.70	0.088	0.175	0.072	0.011	445.25	30.31
10	11.	PART	0.9969 63.25	0.99	0.309	0.075	0.240	0.011	1021.57	16.04
11	12	PART	0.9937 58.35	1.01	0.941	0.095	0.295	0.017	1187.47	23.12
12	13	PART	0.9857 62.98	2.38	0.086	0.148	0.222	0.024	858.15	31.88
13	14	PART	0.9830 59.71	1.87	0.367	0.059	0.101	0.010	448.50	13.95
14	15	PART	0.9080 62.73	5.75	0.199	0.225	0.133	0.035	562.02	47.31
15	16	PART	0.9423 49.59	3.67	0.081	0.060	0.123	0.043	222.28	24.35
16	17	PART	0.9665 50.64	3.26	0.450	0.046	0.113	0.033	357.52	18.11
17	18	PART	0.9881 59.38	1.43	0.321	0.041	0.086	0.007	380.15	9.71
18	20	PART	0.9890 61.08	1.78	0.661	0.058	0.109	0.010	608.07	12.78
19	21	PART	0.9928 62.87	1.61	-0.026	0.057	0.136	0.010	494.74	12.77
20	22	PART	0.9964 62.15	0.99	0.187	0.040	0.126	0.006	516.47	8.77
21	24	PART	0.9974 58.51	0.75	0.316	0.027	0.130	0.005	492.01	6.68
22	25	PART	0.9859 61.35	1.79	0.703	0.094	0.162	0.015	809.47	21.45
23	26	PART	0.9798 59.72	2.29	0.606	0.051	0.079	0.009	466.18	11.49
24	27	PART	0.9632 58.70	2.99	-0.011	0.072	0.089	0.015	256.98	16.92
25	29	PART	0.9798 57.84	2.06	1.008	0.084	0.152	0.018	790.37	21.05
26	30	PART	0.9956 65.43	1.23	0.002	0.086	0.198	0.010	838.81	17.16
27	31	PART	0.9816 64.34	2.22	0.297	0.110	0.134	0.013	643.64	22.88
28	32	PART	0.9829 70.52	2.34	0.421	0.095	0.087	0.006	621.97	16.59
29	33	PART	0.9897 65.11	1.84	0.014	0.063	0.096	0.007	403.50	12.43
30	34	PART	0.7607 48.00	5.96	0.770	0.268	0.291	0.191	683.82	110.88
31	35	PART	0.9819 25.33	-9.00	0.160	0.056	2.388	0.103	318.89	20.26
32	36	PART	0.9940 61.59	1.19	-0.021	0.075	0.198	0.012	677.36	17.19
33	38	PART	0.9853 62.51	2.34	0.871	0.154	0.235	0.026	1173.94	33.51
34	39	PART	0.9806 60.00	2.14	0.452	0.096	0.151	0.017	640.31	22.58
35	40	PART	0.9726 65.15	3.00	1.422	0.176	0.163	0.020	1198.11	34.66
36	41	PART	0.9868 67.43	2.48	0.118	0.154	0.159	0.014	787.00	25.94
37	43	PART	0.9844 69.06	2.86	-0.048	0.138	0.108	0.009	529.19	20.27
38	44	PART	0.9577 67.18	3.83	0.296	0.118	0.072	0.010	441.66	20.88
	2 3 4 5 6 7 8 9 0 1 1 2 3 1 4 5 6 7 8 9 0 1 1 2 3 1 4 5 6 7 8 9 0 1 1 2 3 4 5 6 7 8 9 0 1 1 2 3 4 5 6 7 8 9 0 1 2 2 3 4 2 5 6 7 8 9 0 1 2 3 3 3 3 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 5 6 7 8 9 0 1 2 3 3 4 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2       2       PART         3       3       PART         4       4       PART         5       6       PART         6       7       PART         7       8       PART         9       10       PART         9       10       PART         10       11       PART         11       12       PART         12       13       PART         13       14       PART         14       15       PART         15       16       PART         16       17       PART         17       18       PART         18       20       PART         20       22       PART         21       24       PART         22       25       PART         23       26       PART         24       27       PART         25       29       PART         26       30       PART         27       31       PART         33       38       PART         30       34       PART         31 </td <td>2         2         PART         0.9580         64.59           3         3         PART         0.9769         65.56           4         4         PART         0.7255         79.00           5         6         PART         0.9902         60.25           6         7         PART         0.9549         71.00           7         8         PART         0.9766         68.10           9         10         PART         0.9208         71.72           10         11         PART         0.9969         63.25           11         12         PART         0.9937         58.35           12         13         PART         0.9857         62.98           13         14         PART         0.9830         59.71           14         15         PART         0.9830         59.71           14         15         PART         0.9831         59.38           18         20         PART         0.99423         49.59           16         17         PART         0.9890         61.08           19         21         PART         0.9816         62.15</td> <td>2         PART         0.9580         64.59         3.59           3         3         PART         0.9769         65.56         2.76           4         4         PART         0.7255         79.00         30.24           5         6         PART         0.9902         60.25         1.31           6         7         PART         0.9549         71.00         5.68           7         8         PART         0.9832         60.17         2.21           8         9         PART         0.9208         71.72         4.70           10         11         PART         0.9208         71.72         4.70           10         11         PART         0.9937         58.35         1.01           12         13         PART         0.9857         62.98         2.38           13         14         PART         0.9830         59.71         1.87           14         15         PART         0.9830         59.71         1.87           15         16         PART         0.9843         49.59         3.67           16         17         PART         0.9826         50.64</td> <td>2         2         PART         0.9580         64.59         3.59         0.327           3         3         PART         0.9769         65.56         2.76         0.328           4         4         PART         0.7255         79.00         30.24         -0.037           5         6         PART         0.9902         60.25         1.31         0.709           6         7         PART         0.9928         60.17         2.21         0.566           8         9         PART         0.9208         71.72         4.70         0.088           10         PART         0.9969         63.25         0.99         0.309           11         PART         0.9969         63.25         0.99         0.309           11         PART         0.9965         62.98         2.38         0.086           13         PART         0.9805         59.71         1.87         0.367           14         15         PART         0.9980         61.08         1.78         0.661           17         B         PART         0.9928         62.87         1.61         -0.026           20         22         &lt;</td> <td>2         PART         0.9580         64.59         3.59         0.327         0.103           3         PART         0.9769         65.56         2.76         0.328         0.124           4         4         PART         0.7255         79.00         30.24         -0.037         1.517           5         6         PART         0.9902         60.25         1.31         0.709         0.102           6         7         PART         0.9928         60.17         2.21         0.566         0.073           8         9         PART         0.9969         63.25         0.99         0.309         0.075           10         PART         0.9969         63.25         0.99         0.309         0.075           11         12         PART         0.9965         62.98         2.38         0.086         0.148           13         14         PART         0.9655         50.64         3.26         0.450         0.041           14         PART         0.9964         62.15         0.99         0.321         0.041           15         PART         0.9964         62.15         0.99         0.187         0.040     <!--</td--><td>2         PART         0.9580         64.59         3.59         0.327         0.103         0.081           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123           4         4         PART         0.7255         79.00         30.24         -0.037         1.517         0.066           5         6         PART         0.9902         60.25         1.31         0.709         0.102         0.254           6         7         PART         0.9549         71.00         5.68         0.108         0.349         0.136           7         8         PART         0.9926         63.25         0.99         0.309         0.075         0.240           11         12         PART         0.9927         58.35         1.01         0.941         0.095         0.225           12         13         PART         0.9805         62.73         5.75         0.199         0.225         0.101           14         15         PART         0.9805         59.71         1.87         0.367         0.059         0.102           14         15         PART         0.9806</td><td>2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013           3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014           4         PART         0.9725         79.00         30.24         -0.037         1.517         0.066         0.021           5         6         PART         0.9926         0.255         1.31         0.709         0.102         0.254         0.018           6         7         PART         0.9926         60.275         1.31         0.709         0.102         0.254         0.013           8         PART         0.9926         63.25         0.99         0.309         0.075         0.240         0.011           10         11         PART         0.9836         52.75         0.199         0.225         0.133         0.035           11         12         PART         0.9830         59.71         1.87         0.367         0.059         0.101         0.010           14         5         PART         0.9830         59.71         1.87         0.361         0.060         <th< td=""><td>2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013         448.09           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014         645.79           4         4         PART         0.9549         71.00         5.68         0.102         0.254         0.018         1070.68           6         7         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9366         63.25         0.99         0.309         0.075         0.240         0.011         1021.57           11         12         PART         0.9969         63.25         0.399         0.305         0.225         0.017         148.52           13         14         PART         0.9830         59.71         1.87         0.367         0.255         0.101         0.010</td></th<></td></td>	2         2         PART         0.9580         64.59           3         3         PART         0.9769         65.56           4         4         PART         0.7255         79.00           5         6         PART         0.9902         60.25           6         7         PART         0.9549         71.00           7         8         PART         0.9766         68.10           9         10         PART         0.9208         71.72           10         11         PART         0.9969         63.25           11         12         PART         0.9937         58.35           12         13         PART         0.9857         62.98           13         14         PART         0.9830         59.71           14         15         PART         0.9830         59.71           14         15         PART         0.9831         59.38           18         20         PART         0.99423         49.59           16         17         PART         0.9890         61.08           19         21         PART         0.9816         62.15	2         PART         0.9580         64.59         3.59           3         3         PART         0.9769         65.56         2.76           4         4         PART         0.7255         79.00         30.24           5         6         PART         0.9902         60.25         1.31           6         7         PART         0.9549         71.00         5.68           7         8         PART         0.9832         60.17         2.21           8         9         PART         0.9208         71.72         4.70           10         11         PART         0.9208         71.72         4.70           10         11         PART         0.9937         58.35         1.01           12         13         PART         0.9857         62.98         2.38           13         14         PART         0.9830         59.71         1.87           14         15         PART         0.9830         59.71         1.87           15         16         PART         0.9843         49.59         3.67           16         17         PART         0.9826         50.64	2         2         PART         0.9580         64.59         3.59         0.327           3         3         PART         0.9769         65.56         2.76         0.328           4         4         PART         0.7255         79.00         30.24         -0.037           5         6         PART         0.9902         60.25         1.31         0.709           6         7         PART         0.9928         60.17         2.21         0.566           8         9         PART         0.9208         71.72         4.70         0.088           10         PART         0.9969         63.25         0.99         0.309           11         PART         0.9969         63.25         0.99         0.309           11         PART         0.9965         62.98         2.38         0.086           13         PART         0.9805         59.71         1.87         0.367           14         15         PART         0.9980         61.08         1.78         0.661           17         B         PART         0.9928         62.87         1.61         -0.026           20         22         <	2         PART         0.9580         64.59         3.59         0.327         0.103           3         PART         0.9769         65.56         2.76         0.328         0.124           4         4         PART         0.7255         79.00         30.24         -0.037         1.517           5         6         PART         0.9902         60.25         1.31         0.709         0.102           6         7         PART         0.9928         60.17         2.21         0.566         0.073           8         9         PART         0.9969         63.25         0.99         0.309         0.075           10         PART         0.9969         63.25         0.99         0.309         0.075           11         12         PART         0.9965         62.98         2.38         0.086         0.148           13         14         PART         0.9655         50.64         3.26         0.450         0.041           14         PART         0.9964         62.15         0.99         0.321         0.041           15         PART         0.9964         62.15         0.99         0.187         0.040 </td <td>2         PART         0.9580         64.59         3.59         0.327         0.103         0.081           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123           4         4         PART         0.7255         79.00         30.24         -0.037         1.517         0.066           5         6         PART         0.9902         60.25         1.31         0.709         0.102         0.254           6         7         PART         0.9549         71.00         5.68         0.108         0.349         0.136           7         8         PART         0.9926         63.25         0.99         0.309         0.075         0.240           11         12         PART         0.9927         58.35         1.01         0.941         0.095         0.225           12         13         PART         0.9805         62.73         5.75         0.199         0.225         0.101           14         15         PART         0.9805         59.71         1.87         0.367         0.059         0.102           14         15         PART         0.9806</td> <td>2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013           3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014           4         PART         0.9725         79.00         30.24         -0.037         1.517         0.066         0.021           5         6         PART         0.9926         0.255         1.31         0.709         0.102         0.254         0.018           6         7         PART         0.9926         60.275         1.31         0.709         0.102         0.254         0.013           8         PART         0.9926         63.25         0.99         0.309         0.075         0.240         0.011           10         11         PART         0.9836         52.75         0.199         0.225         0.133         0.035           11         12         PART         0.9830         59.71         1.87         0.367         0.059         0.101         0.010           14         5         PART         0.9830         59.71         1.87         0.361         0.060         <th< td=""><td>2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013         448.09           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014         645.79           4         4         PART         0.9549         71.00         5.68         0.102         0.254         0.018         1070.68           6         7         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9366         63.25         0.99         0.309         0.075         0.240         0.011         1021.57           11         12         PART         0.9969         63.25         0.399         0.305         0.225         0.017         148.52           13         14         PART         0.9830         59.71         1.87         0.367         0.255         0.101         0.010</td></th<></td>	2         PART         0.9580         64.59         3.59         0.327         0.103         0.081           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123           4         4         PART         0.7255         79.00         30.24         -0.037         1.517         0.066           5         6         PART         0.9902         60.25         1.31         0.709         0.102         0.254           6         7         PART         0.9549         71.00         5.68         0.108         0.349         0.136           7         8         PART         0.9926         63.25         0.99         0.309         0.075         0.240           11         12         PART         0.9927         58.35         1.01         0.941         0.095         0.225           12         13         PART         0.9805         62.73         5.75         0.199         0.225         0.101           14         15         PART         0.9805         59.71         1.87         0.367         0.059         0.102           14         15         PART         0.9806	2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013           3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014           4         PART         0.9725         79.00         30.24         -0.037         1.517         0.066         0.021           5         6         PART         0.9926         0.255         1.31         0.709         0.102         0.254         0.018           6         7         PART         0.9926         60.275         1.31         0.709         0.102         0.254         0.013           8         PART         0.9926         63.25         0.99         0.309         0.075         0.240         0.011           10         11         PART         0.9836         52.75         0.199         0.225         0.133         0.035           11         12         PART         0.9830         59.71         1.87         0.367         0.059         0.101         0.010           14         5         PART         0.9830         59.71         1.87         0.361         0.060 <th< td=""><td>2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013         448.09           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014         645.79           4         4         PART         0.9549         71.00         5.68         0.102         0.254         0.018         1070.68           6         7         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9366         63.25         0.99         0.309         0.075         0.240         0.011         1021.57           11         12         PART         0.9969         63.25         0.399         0.305         0.225         0.017         148.52           13         14         PART         0.9830         59.71         1.87         0.367         0.255         0.101         0.010</td></th<>	2         2         PART         0.9580         64.59         3.59         0.327         0.103         0.081         0.013         448.09           3         3         PART         0.9769         65.56         2.76         0.328         0.124         0.123         0.014         645.79           4         4         PART         0.9549         71.00         5.68         0.102         0.254         0.018         1070.68           6         7         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9549         71.00         5.68         0.178         0.247         0.136         0.020         790.70           7         8         PART         0.9366         63.25         0.99         0.309         0.075         0.240         0.011         1021.57           11         12         PART         0.9969         63.25         0.399         0.305         0.225         0.017         148.52           13         14         PART         0.9830         59.71         1.87         0.367         0.255         0.101         0.010



OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	45	PART	0.9341	66.40	4.86	0.784	0.323	0.170	0.031	1043.15	60.24
40	46	PART	0.9745	58.36	2.46	0.101	0.078	0.117	0.016	371.15	18.73
41	47	PART	0.9510	48.86	2.16	0.483	0.079	0.196	0.043	466.78	29.14
42	48	PART	0.9874	62.34	2.21	0.338	0.055	0.093	0.010	457.00	11.79
43	49	PART	0.9552	61.68	3.28	0.405	0.128	0.123	0.020	575.00	29.60
44	50	PART	0.9768	59.98	2.10	0.884	0.110	0.164	0.018	840.65	25.53
45	52	PART	0.9739	71.96	3.82	-0.227	0.305	0.171	0.018	904.00	42.79
46	53	PART	0.9120	56.39	3.92	0.424	0.108	0.100	0.024	407.40	27.69
47	54	PART	0.9022	49.46	4.63	0.153	0.226	0.344	0.129	590.90	72.57
48	55	PART	0.9848	70.92	2.61	0.182	0.220	0.178	0.014	1046.73	33.33
49	56	PART	0.9486	70.00	4.71	0.075	0.143	0.062	0.009	355.19	21.37
50	57	PART	0.9893	65.70	1.90	0.138	0.059	0.085	0.006	416.78	11.55
51	58	PART	0.9079	63.42	5.76	0.493	0.268	0.150	0.039	752.17	57.04
52	59	PART	0.9691	65.47	3.44	0.520	0.151	0.127	0.018	728.85	29.61
53	60	PART	0.9730	71.00	4.20	0.131	0.201	0.107	0.012	637.53	26.99
54	61	PART	0.9874	57.21	1.71	0.047	0.068	0.159	0.016	440.33	16.54
55	64	PART	0.9732	66.40	2.98	0.541	0.165	0.143	0.016	833.66	31.53
56	65	PART	0.9571	70.08	4.25	-0.089	0.232	0.112	0.014	562.47	36.33
57	66	PART	0.9558	68.94	4.88	0.082	0.269	0.135	0.021	707.15	42.83
58	67	PART	0.9878	68.70	2.09	-0.074	0.156	0.170	0.012	817.84	26.94
59	68	PART	0.9758	60.49	2.22	0.364	0.113	0.152	0.017	627.71	26.19
60	70	PART	0.9760	56.10	2.35	0.392	0.056	0.103	0.015	399.37	13.85
61	71	PART	0.8737	68.84	8.08	-0.039	0.437	0.130	0.033	634.52	69.88
62	72	PART	0.9853	60.48	2.01	0.133	0.056	0.097	0.010	364.55	13.09
63	73	PART	0.9802	67.00	2.63	0.288	0.167	0.149	0.014	787.32	29.07
64	74	PART	0.9520	66.61	4.09	0.467	0.141	0.089	0.014	570.86	26.67
65	75	PART	0.9503	68.09	4.56	-0.010	0.187	0.093	0.014	445.73	30.18
66	76	PART	0.9860	60.23	1.64	0.238	0.071	0.128	0.011	495.47	16.42
67	77	PART	0.9939	50.23	2.01	-0.017	0.060	0.179	0.032	290.16	18.96
68	82	PART	0.9777	61.79	2.48	0.552	0.083	0.110	0.013	584.06	18.50
69	83	PART	0.9853	67.00	2.33	0.472	0.260	0.275	0.023	1431.65	46.87
70	84	PART	0.9229	60.91	4.26	0.467	0.203	0.152	0.034	674.91	48.73
71	85	PART	0.9910	70.00	2.12	-0.051	0.067	0.069	0.004	344.81	9.90
72	86	PART	0.7505	73.00	13.16	0.783	0.342	0.048	0.017	576.49	46.19
73	87	PART	0.9380	72.00	5.32	0.322	0.296	0.108	0.016	743.64	43.02
74	88	PART	0.9524	65.40	4.57	0.415	0.322	0.213	0.039	1053.58	61.78
75	89	PART	0.9795	71.00	2.99	1.182	0.112	0.078	0.007	863.34	16.91
76	90	PART	0.9541	67.85	4.44	-0.067	0.220	0.123	0.019	563.48	39.16

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	92	PART	0.9181	65.06	5.21	0.798	0.165	0.085	0.018	646.41	32.89
78	93	PART	0.9752	57.00	2.38	0.590	0.084	0.141	0.020	587.57	20.78
79	94	PART	0.9793	72.00	3.11	0.374	0.104	0.067	0.006	522.97	14.94
80	95	PART	0.9085	77.49	9.50	0.177	0.440	0.072	0.012	587.56	38.55
81	97	PART	0.9871	65.84	2.33	1.157	0.145	0.185	0.017 y	1222.20	27.43
82	98	PART	0.8922	70.17	5.37	0.014	0.622	0.199	0.032	1062.46	99.08
83	99	PART	0.9577	73.00	4.71	0.329	0.349	0.132	0.015	914.57	45.40
84	100	PART	0.9225	60.46	4.81	0.886	0.122	0.090	0.022	613.87	27.66
85	101	PART	0.9652	68.29	3.89	0.362	0.177	0.106	0.013	648.73	29.07
86	102	PART	0.9683	70.00	3.56	0.372	0.293	0.169	0.018	1029.00	45.17
87	104	PART	0.9692	71.71	4.58	0.393	0.277	0.136	0.017	916.64	37.11
88	105	PART	0.7526	73.00	14.83	0.520	0.296	0.037	0.013	415.49	34.32
89	106	PART	0.9516	71.84	5.97	0.533	0.211	0.080	0.013	654.03	27.87
· 90	107	PART	0.9204	71.00	6.33	0.283	0.205	0.068	0.012	476.66	30.45
91	108	PART	0.8989	64.00	4.81	0.215	0.102	0.055	0.011	293.24	21.54
92	109	PART	0.8938	53.82	5.09	0.789	0.129	0.126	0.044	556.80	36.17
93	110	PART	0.9760	72.00	3.20	0.549	0.241	0.147	0.013	1048.14	35.61
94	111	PART	0.9840	61.00	1.92	0.869	0.074	0.118	0.011	712.47	16.75
95	112	PART	0.9686	73.77	4.19	0.799	0.408	0.177	0.018	1398.24	52.51
96	113	PART	0.9817	72.12	2.79	0.404	0.358	0.249	0.019	1591.94	52.08
97	114	PART	0.9754	69.24	3.21	0.127	0.205	0.137	0.014	743.65	32.43
98	115	PART	0.9863	73.56	2.87	0.221	0.157	0.102	0.007	714.69	19.45
99	116	PART	0.9766	72.31	3.18	-0.118	0.216	0.131	0.011	722.36	31.54
100	117	PART	0.9703	60.48	1.94	0.427	0.123	0.164	0.017	687.51	29.46
101	118	PART	0.8088	81.00	37.23	-0.184	3.702	0.115	0.026	903.37	112.68
102	119	PART	0.8794	88.00	-9.00	-0.429	0.367	0.092	0.011	841.93	55.95
103	120	PART	0.9371	59.98	3.90	0.627	0.181	0.157	0.032	725.48	43.23
104	121	PART	0.9690	61.31	2.69	0.081	0.061	0.069	0.009	263.22	13.88
105	123	PART	0.8529	76.27	12.05	-0.019	0.451	0.059	0.013	400.78	40.82
106	124	PART	0.9737	64.42	2.64	-0.036	0.129	0.132	0.015	516.01	27.01
107	125	PART	0.7006	80.00	35.51	0.234	1.306	0.045	0.014	452.80	54.41
108	126	PART	0.9626	65.73	4.08	0.323	0.114	0.085	0.014	485.57	21.44
109	127	PART	0.8638	77.00	12.77	0.041	0.457	0.055	0.011	405.10	36.53
110	128	PART	0.9616	69.79	4.37	0.021	0.436	0.216	0.028	1139.80	65.58
111	129	PART	0.9579	72.00	4.74	0.836	0.156	0.067	0.008	691.99	21.21
112	130	PART	0.9890	67.00	1.87	0.622	0.068	0.084	0.006	613.25	12.31
113	131	PART	0.9516	71.20	5.24	0.303	0.337	0.139	0.020	885.08	47.66
114	132	PART	0.9471	66.74	4.43	0.529	0.195	0.114	0.019	708.85	36.31

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	133	PART	0.9626	51.64	2.66	0.123	0.086	0.195	0.042	402.73	28,65
116	134	PART	0.9807	71.00	2.99	0.498	0.179	0.126	0.011	876.30	26.94
117	135	PART	0.9924	66.58	1.69	0.042	0.067	0.120	0.008	552.84	13.42
118	136	PART	0.9798	60.84	2.28	-0.018	0.102	0.169	0.020	551.67	25.16
119	137	PART	0.7536	84.00	107.26	-0.435	7.816	0.075	0.016	550.07	95.30
120	138	PART	0.9556	72.21	4.49	0.774	0.286	0.124	0.015	1004.06	42.18
121	139	PART	0.9665	69.52	4.19	0.450	0.276	0.146	0.018	917.16	40.76
122	140	PART	0.8464	72.00	8.60	0.518	0.229	0.052	0.013	489.09	33.75
123	141	PART	0.9496	54.98	4.04	0.251	0.122	0.160	0.042	462.29	32.80
124	142	PART	0.9781	75.43	3.99	1.419	0.236	0.100	0.008	1186.33	24.96
125	143	PART	0.9775	68.33	3.34	0.790	0.203	0.147	0.016	1007.04	32.29
126	144	PART	0.9554	71.00	5.58	0.005	0.148	0.059	0.009	325.95	19.17
127	145	PART	0.9944	56.00	1.27	1.023	0.054	0.201	0.016	870.52	13.57
128	146	PART	0.9950	66.40	1.45	0.196	0.063	0.118	0.006	598.63	11.31
129	147	PART	0.9901	60.00	1.59	0.240	0.116	0.253	0.021	885.39	26.21
130	148	PART	0.9095	64.95	5.57	1.260	0.220	0.113	0.026	925.92	45.81
131	149	PART	0.8102	84.00	90.49	0.101	14.734	0.168	0.030	1621.28	179.58
132	150	PART	0.9867	63.21	1.88	0.663	0.098	0.153	0.013	819.46	21.12
133	151	PART	0.9465	69.62	4.72	0.353	0.164	0.074	0.011	515.09	26.38
134	152	PART	0.9865	62.32	1.90	0.718	0.212	0.346	0.031	1506.82	46.79
135	153	PART	0.9638	70.85	4.41	1.102	0.163	0.081	0.010	846.95	24.12
136	154	PART	0.9918	62.35	1.80	0.447	0.196	0.401	0.034	1609.45	41.78
137	155	PART	0.9894	66.79	1.90	0.146	0.107	0.147	0.011	717.52	20.44
138	156	PART	0.9455	53.57	3.10	0.615	0.185	0.298	0.067	850.87	54.82
139	157	PART	0.9945	69.05	1.67	0.598	0.065	0.087	0.004	658.41	9.75
140	158	PART	0.8700	75.77	10.03	0.303	0.267	0.045	0.010	417.85	28.72
141	159	PART	0.7285	70.88	9.98	0.330	0.341	0.061	0.018	453.23	54.34
142	160	PART	0.9696	66.17	3.19	0.716	0.162	0.128	0.015	822.09	30.09
143	161	PART	0.9850	61.46	1.92	0.641	0.077	0.125	0.012	664.26	17.53
144	162	PART	0.8991	77.68	9.86	0.482	0.597	0.094	0.016	866.73	52.68
145	163	PART	0.9741	72.28	3.97	0.268	0.221	0.116	0.012	777.83	28.64
146	165	PART	0.9762	69.50	3.27	0.306	0.146	0.097	0.010	611.22	22.87
147	166	PART	0.9602	45.00	3.24	0.224	0.043	0.163	0.064	255.25	24.48
148	1080	PART	0.9749	66.30	2.84	0.105	0.107	0.097	0.011	466.98	20.32

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
1	1	PART	0.9423	62.00	4.28	0.190	0.426	0.294	0.048	1130.96	86.38
2	4	PART	0.8970	57.88	7.47	0.322	0.542	0.339	0.118	1077.18	128.25
3	7	PART	0.9264	56.78	6.30	1.063	0.334	0.268	0.081	1093.50	78.45
4	8	PART	0.9790	62.16	2.99	0.612	0.137	0.161	0.019	808.23	29.78
5	9	PART	0.9859	66.60	2.30	0.326	0.236	0.261	0.020	1319.77	46.02
6	10	PART	0.9834	64.00	2.69	0.093	0.151	0.175	0.018	739.27	30.87
7	11	PART	0.8315	58.83	9.02	0.549	0.768	0.344	0.137	1233.14	177.73
8	12	PART	0.9603	59.95	4.76	0.929	0.355	0.318	0.063	1358.39	78.49
9	13	PART	0.9093	41.85	4.17	0.203	0.463	0.937	0.308	803.49	121.37
10	14	PART	0.9936	61.26	1.88	0.357	0.060	0.128	0.010	573.16	12.91
11	15	PART	0.9900	57.28	2.15	0.020	0.075	0.152	0.015	421.04	18.19
12	17	PART	0.9891	54.13	2.03	0.452	0.045	0.115	0.013	418.53	10.62
13	18	PART	0.9538	61.00	4.64	0.241	0.171	0.135	0.025	549.28	37.67
• 14	20	PART	0.9922	57.78	1.81	1.099	0.076	0.183	0.015	914.61	17.67
15	21	PART	0.9885	58.59	2.45	-0.073	0.143	0.271	0.030	774.23	33.97
16	22	PART	0.9886	60.00	2.33	0.354	0.126	0.218	0.022	832.31	28.21
17	24	PART	0.9944	56.09	1.43	0.268	0.068	0.215	0.015	639.10	16.29
18	26	PART	0.9842	61.52	2.94	0.612	0.133	0.179	0.021	852.08	28.79
19	29	PART	0.9955	57.89	1.56	0.942	0.080	0.251	0.018	1055.45	18.88
20	31	PART	0.9949	65.31	1.31	0.340	0.129	0.250	0.012	1200.68	25.54
21	32	PART	0.9807	63.57	2.61	0.342	0.141	0.163	0.016	764.45	28.40
22	40	PART	0.9803	65.90	2.76	1.508	0.248	0.238	0.023	1606.17	49.46
23	41	PART	0.9827	50.00	2.56	0.166	0.062	0.182	0.030	356.40	16.64
24	42	PART	0.8842	54.00	5.21	-0.085	0.227	0.185	0.053	373.00	57.51
25	44	PART	0.9836	63.88	2.71	0.437	0.146	0.173	0.017	850.73	30.99
26	45	PART	0.9988	64.17	0.71	0.910	0.097	0.405	0.011	1975.76	19.82
27	46	PART	0.9945	57.64	1.66	0.121	0.087	0.241	0.018	714.54	21.90
28	55	PART	0.9912	65.59	1.78	0.527	0.166	0.245	0.015	1262.29	33.31
29	56	PART	0.9944	64.00	1.53	0.075	0.061	0.120	0.007	509.74	12.43
30	57	PART	0.9533	61.30	4.24	0.005	0.150	0.123	0.021	428.11	31.51
31	58	PART	0.9594	55.00	3.68	0.925	0.283	0.352	0.066	1161.39	63.56
32	68	PART	0.9747	60.85	3.95	0.020	0.160	0.174	0.028	596.48	35.03
33	70	PART	0.9933	57.26	1.48	0.401	0.039	0.103	0.007	426.90	9.10
34	71	PART	0.9906	63.00	2.10	-0.004	0.138	0.259	0.022	984.75	33.05
35	73	PART	0.8709	70.49	8.02	0.363	0.379	0.103	0.024	702.14	65.18
36	74	PART	0.9980	59.52	0.91	0.523	0.062	0.271	0.011	1037.90	14.26
37	76	PART	0.9796	55.00	2.73	0.374	0.101	0.172	0.024	539.33	23.70
38	82	PART	0.9790	62.16	2.99	0.612	0.137	0.161	0.019	808.23	29.78

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
39	86	PART	0.9888	53.69	1.66	1.236	0.104	0.263	0.024	1013.85	25.09
40	88	PART	0.9744	60.67	4.03	0.569	0.290	0.317	0.052	1267.58	64.38
41	90	PART	0.9034	60.00	6.80	0.029	0.244	0.158	0.047	520.02	58.77
42	91	PART	0.9733	63.00	3.29	0.277	0.197	0.187	0.023	815.10	41.62
43	92	PART	0.9585	53.36	4.61	0.449	0.316	0.390	0.095	978.49	80.23
44	93	PART	0.9964	62.00	1.15	-0.007	0.056	0.187	0.009	672.60	13.27
45	94	PART	0.9846	60.62	2.78	0.580	0.078	0.112	0.013	584.00	17.11
46	95	PART	0.9943	69.47	1.67	0.456	0.094	0.137	0.007	889.05	16.54
47	97	PART	0.9977	61.23	1.13	0.978	0.097	0.347	0.016	1557.72	21.08
48	98	PART	0.9745	65.54	2.93	0.265	0.307	0.261	0.027	1234.15	60.87
49	99	PART	0.9958	61.83	1.54	0.648	0.102	0.265	0.016	1183.07	22.29
50	100	PART	0.9957	58.31	1.49	0.617	0.051	0.164	0.011	701.72	12.17
51	101	PART	0.9263	58.11	5.32	0.847	0.388	0.289	0.070	1139.03	90.13
- 52	102	PART	0.9893	64.38	2.07	0.073	0.197	0.313	0.024	1311.65	44.11
53	103	PART	0.9818	51.66	2.70	0.868	0.131	0.311	0.051	891.25	33.45
54	106	PART	0.9799	55.47	3.01	1.136	0.162	0.273	0.042	1074.84	37.67
55	107	PART	0.9895	62.63	2.16	0.286	0.116	0.185	0.015	797.25	25.14
56	108	PART	0.9748	62.98	3.64	0.158	0.109	0.109	0.015	474.59	24.01
57	109	PART	0.9502	52.00	4.38	0.731	0.231	0.308	0.081	848.15	57.45
58	110	PART	0.9946	64.41	1.49	0.690	0.145	0.292	0.016	1453.02	29.49
59	111	PART	0.9893	58.66	2.12	0.629	0.079	0.156	0.015	694.57	18.59
60	112	PART	0.9744	65.55	3.30	1.751	0.260	0.215	0.024	1577.24	49.32
61	113	PART	0.9850	66.80	2.40	0.430	0.349	0.364	0.029	1847.32	67.26
62	114	PART	0.9926	62.93	1.85	0.074	0.131	0.247	0.017	964.11	28.14
63	115	PART	0.9935	68.28	1.62	0.252	0.089	0.129	0.007	738.03	16.32
64	116	PART	0.9898	66.16	2.16	0.002	0.147	0.185	0.013	833.74	26.94
65	118	PART	0.9870	67.52	2.56	0.455	0.239	0.251	0.021	1375.03	43.09
66	119	PART	0.9862	67.92	2.54	0.962	0.230	0.235	0.019	1506.42	42.09
67	121	PART	0.9977	57.24	1.01	0.184	0.032	0.147	0.007	466.36	7.54
68	122	PART	0.8575	64.86	8.85	0.850	0.316	0.115	0.038	796.33	64.51
69	123	PART	0.9829	58.00	2.75	-0.059	0.117	0.189	0.024	517.03	27.86
70	124	PART	0.9957	58.81	1.28	-0.018	0.078	0.240	0.014	712.66	18.09
71	125	PART	0.9784	56.00	2.98	0.620	0.108	0.171	0.025	653.76	25.36
72	126	PART	0.9757	58.21	2.94	0.377	0.117	0.153	0.020	579.94	27.70
73	127	PART	0.9893	63.50	2.18	0.029	0.135	0.205	0.017	814.91	28.66
74	129	PART	0.9699	57.29	3.69	0.985	0.146	0.181	0.032	851.76	34.58
75	130	PART	0.9589	60.37	4.76	0.703	0.183	0.156	0.031	769.79	39.88
76	132	PART	0.9543	65.93	4.12	0.550	0.281	0.176	0.025	985.49	56.11

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
77	134	PART	0.9858	59.39	2.53	0.939	0.154	0.249	0.028	1117.16	35.09
78	135	PART	0.9342	53.57	4.36	0.051	0.319	0.339	0.082	737.12	72.88
79	136	PART	0.8354	54.00	7.40	-0.072	0.328	0.191	0.082	391.61	76.41
80	137	PART	0.9566	67.55	3.89	0.190	0.344	0.210	0.028	1081.74	67.20
81	140	PART	0.9719	60.71	3.71	0.724	0.134	0.138	0.021	729.26	29.37
82	142	PART	0.9727	57.06	3.44	1.922	0.171	0.219	0.035	1290.73	39.83
83	145	PART	0.9897	60.00	2.11	1.230	0.095	0.174	0.015	1008.12	21.09
84	146	PART	0.9683	73.85	5.58	0.149	0.311	0.096	0.010	672.36	30.61
85	147	PART	0.9968	57.40	1.13	0.291	0.053	0.206	0.011	671.18	12.58
86	148	PART	0.9709	59.92	4.04	1.138	0.163	0.172	0.029	966.32	36.12
87	150	PART	0.9743	57.54	3.47	0.646	0.189	0.246	0.039	917.43	44.20
88	151	PART	0.9943	60.69	1.85	0.668	0.052	0.119	0.009	641.07	12.38
89	152	PART	0.9815	62.99	3.01	0.507	0.273	0.313	0.035	1378.17	57.78
· 90	160	PART	0.9748	67.37	3.07	0.675	0.243	0.190	0.019	1155.36	46.36
91	161	PART	0.8597	65.00	6.86	0.699	0.591	0.216	0.055	1170.50	119.52
92	162	PART	0.9895	72.43	2.47	0.469	0.194	0.157	0.010	1124.47	28.00
93	163	PART	0.9848	63.35	2.50	0.415	0.149	0.190	0.018	890.94	32.14



OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
1	1	PART	0.9890	60.24	1.66	0.702	0.086	0.179	0.016	840.47	20.79
2	2	PART	0.9773	60.44	2.22	0.351	0.070	0.102	0.012	466.50	16.49
3	3	PART	0.9865	62.32	1.86	0.360	0.084	0.147	0.014	671.73	19.05
4	4	PART	0.7385	76.08	16.61	0.038	0.798	0.071	0.021	520.76	67.46
5	6	PART	0.9913	58.77	1.17	0.695	0.096	0.286	0.020	1110.37	25.04
6	7	PART	0.9556	67.62	4.95	0.220	0.293	0.153	0.025	820.58	44.90
7	8	PART	0.9891	58.00	1.57	0.583	0.057	0.136	0.013	601.43	13.78
8	9	PART	0.9835	65.18	2.06	0.245	0.178	0.219	0.019	1027.72	35.21
9	10	PART	0.9261	69.11	4.13	0.118	0.168	0.080	0.012	459.02	29.64
10	11	PART	0.9957	61.07	0.98	0.340	0.085	0.275	0.014	1065.20	19.26
11	12	PART	0.9914	56.78	1.07	0.938	0.111	0.340	0.023	1238.62	28.73
12	13	PART	0.9910	60.44	1.49	0.121	0.113	0.259	0.021	901.02	26.20
13	14	PART	0.9819	58.37	1.78	0.367	0.061	0.113	0.012	463.19	15.01
· 14	15	PART	0.9208	58.82	4.20	0.236	0.196	0.167	0.040	587.02	46.57
15	16	PART	0.9458	48.00	2.78	0.077	0.057	0.154	0.047	240.76	22.84
16	17	PART	0.9615	49.00	2.56	0.452	0.047	0.141	0.039	376.23	20.26
17	18	PART	0.9939	57.76	0.96	0.318	0.029	0.099	0.006	394.09	7.33
18	19	PART	0.7568	18.00	-9.00	0.188	0.188	8.118	1.455	368.27	74.03
19	20	PART	0.9910	59.40	1.45	0.668	0.050	0.123	0.010	627.08	11.97
20	21	PART	0.9768	62.89	2.77	-0.032	0,101	0.139	0.018	513.45	23.10
21	22	PART	0.9900	60.63	1.44	0.195	0.064	0.139	0.011	536.18	15.07
22	24	PART	0.9942	57.16	0.97	0.320	0.041	0.147	0.009	513.49	10.28
23	25	PART	0.9900	58.93	1.36	0.719	0.077	0.190	0.015	837.39	19.23
24	26	PART	0.9769	57.61	2.12	0.621	0.052	0.091	0.012	480.86	12.80
25	27	PART	0.9704	58.67	2.51	-0.018	0.065	0.093	0.013	268.96	15.55
26	29	PART	0.9712	57.13	2.04	0.993	0.102	0.166	0.022	810.79	26.05
27	30	PART	0.9980	62.68	0.74	0.053	0.050	0.228	0.009	874.93	11.53
28	31	PART	0.9763	62.00	2.45	0.309	0.115	0.154	0.019	668.44	25.99
29	32	PART	0.9919	69.05	1.55	0.436	0.062	0.092	0.005	638.73	11.33
30	33	PART	0.9885	63.00	1.84	0.028	0.062	0.107	0.009	418.91	13.20
31	34	PART	0.7699	48.00	6.15	0.735	0.262	0.314	0.211	700.96	104.71
32	35	PART	0.9819	24.05	-9.00	0.160	0.056	6.430	0.276	661.66	27.99
33	36	PART	0.9838	60.06	1.76	-0.019	0.123	0.221	0.022	707.88	29.27
34	38	PART	0.9944	60.36	1.16	0.891	0.092	0.271	0.017	1216.18	21.40
35	39	PART	0.9883	57.61	1.63	0.472	0.074	0.179	0.018	669.44	18.60
36	40	PART	0.9774	63.31	2.63	1.434	0.145	0.180	0.023	1223.24	31.57
37	41	PART	0.9938	65.60	1.49	0.139	0.094	0.175	0.011	818.11	17.73
38	43	PART	0.9873	65.88	2.09	0.023	0.092	0.123	0.010	551.98	17.72

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	44	PART	0.9683	63.87	3.17	0.325	0.084	0.085	0.013	458.24	18.17
40	45	PART	0.9508	62.31	3.68	0.872	0.232	0.207	0.039	1079.66	52.08
41	46	PART	0.9700	56.07	2.18	0.125	0.085	0.138	0.020	393.40	21.63
42	47	PART	0.9404	46.00	2.11	0.492	0.088	0.284	0.075	509.31	39.61
43	48	PART	0.9806	60.07	2.21	0.358	0.066	0.106	0.013	474.22	15.21
44	49	PART	0.9673	59.15	2.45	0.418	0.108	0.145	0.021	596.68	26.58
45	50	PART	0.9469	59.72	3.38	0.897	0.168	0.170	0.032	866.20	39.65
46	52	PART	0.9782	69.19	3.07	-0.126	0.258	0.188	0.018	936.86	38.96
47	53	PART	0.9006	56.49	3.76	0.414	0.119	0.102	0.025	414.52	30.80
48	54	PART	0.8477	50.00	5.25	0.103	0.282	0.366	0.168	631.33	93.84
49	55	PART	0.9915	68.58	1.79	0.244	0.151	0.195	0.012	1079.05	24.86
50	56	PART	0.9495	67.00	4.17	0.107	0.125	0.070	0.011	367.57	21.23
51	57	PART	0.9955	62.37	1.11	0.165	0.034	0.101	0.006	433.78	7.58
• 52	58	PART	0.8887	60.46	5.48	0.548	0.281	0.176	0.052	781.70	65.45
53	59	PART	0.9722	62.26	2.80	0.564	0.125	0.150	0.021	754.92	28.10
54	. 60	PART	0.9758	68.63	3.76	0.181	0.176	0.117	0.014	659.24	25.98
55	61	PART	0.9915	53.57	1.59	0.093	0.053	0.210	0.025	478.47	15.31
56	64	PART	0.9696	64.03	2.77	0.573	0.166	0.161	0.020	860.09	34.13
57	65	PART	0.9716	67.02	2.91	-0.038	0.164	0.128	0.014	586.24	29.44
58	66	PART	0.9645	64.66	3.40	0.198	0.191	0.160	0.023	738.57	38.37
59	67	PART	0.9900	65.74	1.74	0.001	0.125	0.194	0.014	852.83	24.56
60	68	PART	0.9698	58.62	2.33	0.368	0.125	0.175	0.024	653.10	30.54
61	70	PART	0.9819	56.22	1.70	0.384	0.050	0.108	0.012	414.02	12.44
62	71	PART	0.8351	69.48	9.00	-0.072	0.514	0.129	0.037	654.71	78.89
63	72	PART	0.9745	58.63	2.65	0.146	0.074	0.110	0.017	380.58	18.19
64	73	PART	0.9886	64.08	1.72	0.344	0.110	0.171	0.013	815.62	22.12
65	74	PART	0.9590	63.00	3.31	0.512	0.113	0.106	0.017	589.54	24.76
66	75	PART	0.9572	65.52	3.89	0.023	0.149	0.104	0.016	461.99	28.03
67	76	PART	0.9969	58.34	0.72	0.232	0.033	0.149	0.006	517.28	8.12
68	77	PART	0.9886	50.72	2.59	-0.012	0.081	0.181	0.042	305.70	25.02
69	82	PART	0.9845	59.49	1.71	0.569	0.066	0.127	0.012	602.93	16.03
70	83	PART	0.9865	64.44	1.89	0.548	0.215	0.312	0.026	1485.90	45.14
71	84	PART	0.9325	59.33	3.44	0.458	0.188	0.172	0.035	700.45	46.29
72	85	PART	0.9955	67.38	1.36	-0.018	0.042	0.076	0.004	358.56	6.96
73	86	PART	0.7824	72.21	11.76	0.762	0.283	0.051	0.018	583.27	40.98
74	87	PART	0.9553	69.00	4.13	0.388	0.229	0.121	0.016	766.98	36.59
75	88	PART	0.9696	62.07	3.14	0.479	0.223	0.255	0.040	1099.72	49.74
76	89	PART	0.9880	68.00	1.95	1.221	0.077	0.088	0.006	881.05	13.13

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	90	PART	0.9716	64.39	2.74	-0.008	0.149	0.144	0.017	589.78	30.81
78	92	PART	0.9375	60.65	3.88	0.835	0.128	0.108	0.022	666.12	29.73
79	93	PART	0.9840	54.30	1.92	0.615	0.065	0.176	0.024	617.03	17.48
80	94	PART	0.9889	69.66	2.15	0.400	0.069	0.073	0.005	537.74	10.96
81	95	PART	0.9193	74.91	6.82	0.238	0.302	0.077	0.012	606.86	35.95
82	97	PART	0.9910	63.18	1.69	1.226	0.110	0.210	0.016	1257.81	22.95
83	98	PART	0.8750	67.00	4.97	0.183	0.591	0.221	0.040	1105.32	106.53
84	99	PART	0.9627	71.19	4.13	0.361	0.283	0.141	0.017	940.95	41.34
85	100	PART	0.9097	59.46	4.75	0.893	0.129	0.097	0.025	627.08	30.40
86	101	PART	0.9741	65.55	2.90	0.410	0.131	0.119	0.014	670.24	25.06
87	102	PART	0.9792	66.67	2.41	0.463	0.198	0.194	0.018	1066.23	37.16
88	104	PART	0.9771	68.98	3.67	0.464	0.218	0.150	0.018	945.18	32.59
89	105	PART	0.7603	73.00	14.61	0.504	0.297	0.038	0.013	422.56	32.65
90	106	PART	0.9630	69.00	4.32	0.583	0.165	0.089	0.012	671.20	24.25
91	107	PART	0.9312	68.90	5.46	0.305	0.172	0.074	0.013	490.20	28.64
92	108	PART	0.9038	61.12	4.00	0.225	0.089	0.064	0.014	302.47	21.12
93	109	PART	0.9236	51.89	4.77	0.793	0.104	0.154	0.059	579.19	33.46
94	110	PART	0.9832	69.68	2.53	0.616	0.183	0.160	0.013	1078.06	29.80
95	111	PART	0.9894	59.69	1.42	0.867	0.059	0.131	0.010	730.64	14.01
96	112	PART	0.9811	72.03	3.02	0.844	0.279	0.189	0.016	1435.50	39.20
97	113	PART	0.9806	69.78	2.69	0.500	0.332	0.270	0.024	1634.60	53.62
98	114	PART	0.9792	66.94	2.67	0.180	0.164	0.151	0.015	770.04	29.88
99	115	PART	0.9932	71.87	1.97	0.244	0.098	0.109	0.006	733.95	13.24
100	116	PART	0.9837	70.13	2.57	-0.071	0.176	0.142	0.011	751.35	26.41
101	117	PART	0.9823	58.67	1.30	0.431	0.095	0.188	0.015	716.57	23.54
102	118	PART	0.8229	81.00	39.33	-0.296	4.154	0.120	0.025	934.94	95.51
103	119	PART	0.8993	88.00	-9.00	-0.570	0.346	0.096	0.010	870.07	50.97
104	120	PART	0.9436	58.58	3.38	0.629	0.172	0.176	0.035	750.38	42.84
105	121	PART	0.9824	59.57	1.81	0.077	0.045	0.078	0.008	274.36	10.74
106	123	PART	0.8654	74.00	9.92	0.016	0.361	0.063	0.014	415.20	38.61
107	124	PART	0.9798	62.51	2.30	-0.036	0.105	0.149	0.017	539.50	23.78
108	125	PART	0.7327	74.00	15.19	0.362	0.451	0.051	0.017	466.93	48.32
109	126	PART	0.9767	63.00	2.75	0.347	0.081	0.099	0.013	504.18	17.06
110	127	PART	0.8692	74.00	9.75	0.101	0.331	0.059	0.013	419.15	35.46
111	128	PART	0.9620	67.26	3.78	0.129	0.381	0.238	0.032	1176.30	63.55
112	129	PART	0.9450	70.63	5.64	0.853	0.179	0.071	0.011	706.01	24.36
113	130	PART	0.9944	64.26	1.18	0.645	0.043	0.097	0.005	630.42	8.88
114	131	PART	0.9430	69.15	5.01	0.364	0.334	0.150	0.024	912.00	51.59

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	132	PART	0.9712	63.82	3.12	0.560	0.126	0.132	0.019	730.67	26.89
116	133	PART	0.9552	51.00	2.33	0.128	0.093	0.216	0.044	425.11	30.64
117	134	PART	0.9867	68.70	2.26	0.548	0.134	0.138	0.011	901.39	22.62
118	135	PART	0.9904	63.42	1.67	0.092	0.066	0.140	0.011	577.73	15.24
119	136	PART	0.9811	58.85	2.12	-0.003	0.099	0.194	0.024	582.08	25.61
120	137	PART	0.7895	83.00	146.79	-0.487	11.321	0.080	0.017	570.81	95.80
121	138	PART	0.9693	69.81	3.50	0.818	0.214	0.136	0.015	1028.88	35.11
122	139	PART	0.9573	67.82	4.33	0.492	0.272	0.155	0.023	939.62	44.89
123	140	PART	0.8701	69.00	6.99	0.538	0.190	0.059	0.014	500.87	31.16
124	141	PART	0.9657	57.00	2.59	0.210	0.106	0.149	0.024	473.76	26.57
125	142	PART	0.9817	72.67	2.99	1.495	0.156	0.108	0.009	1210.65	22.14
126	143	PART	0.9839	65.34	2.37	0.861	0.144	0.168	0.016	1038.31	27.27
127	144	PART	0.9686	68.47	4.21	0.033	0.111	0.065	0.009	337.85	16.29
128	145	PART	0.9795	55.31	2.06	1.043	0.104	0.217	0.030	899.64	27.33
129	146	PART	0.9916	64.90	1.66	0.217	0.075	0.128	0.009	619.64	14.83
130	147	PART	0.9911	59.63	1.44	0.235	0.110	0.265	0.020	920.30	25.16
131	148	PART	0.9231	61.33	4.14	1.291	0.185	0.138	0.030	950.62	42.48
132	149	PART	0.8409	78.25	17.65	0.705	2.351	0.182	0.040	1674.99	131.65
133	150	PART	0.9912	61.01	1.31	0.669	0.076	0.177	0.012	846.73	17.49
134	151	PART	0.9491	67.28	4.18	0.386	0.145	0.082	0.013	531.61	25.41
135	152	PART	0.9938	59.90	1.14	0.742	0.139	0.407	0.026	1572.47	33.25
136	153	PART	0.9720	67.41	3.40	1.160	0.127	0.092	0.011	865.69	21.27
137	154	PART	0.9875	60.06	1.81	0.536	0.233	0.460	0.045	1681.51	53.65
138	155	PART	0.9717	65.46	2.81	0.159	0.170	0.158	0.019	742.15	33.60
139	156	PART	0.9206	50.76	3.70	0.664	0.211	0.387	0.132	908.65	78.36
140	157	PART	0.9939	66.79	1.54	0.629	0.057	0.096	0.005	675.39	10.21
141	158	PART	0.8982	73.00	7.56	0.331	0.203	0.049	0.010	427.81	24.94
142	159	PART	0.7392	63.00	7.03	0.471	0.231	0.081	0.028	479.14	52.44
143	160	PART	0.9850	63.41	2.12	0.739	0.099	0.148	0.015	845.83	21.15
144	161	PART	0.9938	59.88	1.12	0.634	0.048	0.142	0.009	684.74	11.58
145	162	PART	0.9222	74.52	6.20	0.571	0.371	0.102	0.016	889.67	45.86
146	163	PART	0.9883	70.52	2.73	0.299	0.149	0.124	0.010	800.88	19.36
147	165	PART	0.9813	67.48	2.68	0.335	0.116	0.106	0.010	629.34	19.98
148	166	PART	0.9441	47.86	3.33	0.201	0.051	0.128	0.043	247.50	22.27
149	1080	PART	0.9850	64.00	2.07	0.130	0.080	0.109	0.010	486.13	16.23

.

## PRISM Output Participant Pre-Treatment Louisville Station

)	OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
							•					
	1	38	PART	0.9970	67.00	1.29	0.356	0.111	0.222	0.008	1186.95	19.65
	2	39	PART	0.9979	60.72	1.13	0.473	0.048	0.168	0.007	751.68	10.34
	3	96	PART	0.9272	47.00	3.56	0.798	0.436	0.634	0.157	1138.92	111.67
	4	117	PART	0.9465	65.76	5.47	0.843	0.180	0.089	0.014	707.61	33.32
	5	143	PART	0.9777	65.94	3.12	1.290	0.249	0.220	0.021	1463.93	44.77
	6	158	PART	0.9988	62.90	0.79	0.590	0.029	0.119	0.003	675.96	6.24
OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC	
------	---------	-------	--------	-------	---------	--------	---------	-------	----------	---------	--------	
- 1	1	PART	0.9859	62.40	2.20	0.696	0.100	0.157	0.017	844.74	23.17	
2	2	PART	0.9719	62.28	2.62	0.351	0.078	0.092	0.012	471.70	18,00	
3	3	PART	0.9865	64.18	1.78	0.346	0.092	0.133	0.011	678.57	19.59	
4	4	PART	0.7239	78.96	23.57	-0.056	1.157	0.068	0.021	521.96	70.86	
5	6	PART	0.9919	59.71	1.31	0.706	0.092	0.267	0.020	1125.68	23.81	
6	7	PART	0.9606	68,77	4.71	0.200	0.280	0.146	0.021	830.08	43.01	
7	8	PART	0.9870	59.40	1.97	0.578	0.063	0.124	0.013	607.88	14.88	
8	9	PART	0.9804	66.60	2.30	0.222	0.195	0.205	0.019	1039.15	38.93	
9	10	PART	0.9176	70.54	4.52	0.103	0.172	0.076	0.012	462.98	31.10	
10	11	PART	0.9970	63.12	0.89	0.329	0.078	0.245	0.010	1080.47	16.61	
11	12	PART	0.9959	57.36	0.90	0.949	0.076	0.321	0.017	1252.03	19.29	
12	13	PART	0.9922	62.96	1.65	0.094	0.108	0.226	0.017	912.12	24.10	
13	14	PART	0.9820	60.03	2.04	0.361	0.061	0.101	0.011	467.01	14.41	
· 14	15	PART	0.9229	61.00	4.46	0.228	0.195	0.147	0.033	595.12	43.74	
15	16	PART	0.9655	47.98	2.19	0.076	0.045	0.152	0.035	247.46	19.59	
16	17	PART	0.9770	47.00	1.67	0.463	0.036	0.169	0.034	395.32	17.90	
17	18	PART	0.9953	58.36	0.98	0.322	0.026	0.094	0.005	399.32	6.43	
18	19	PART	0.7625	37.00	4.34	0.062	0.221	0.639	0.305	335.52	74.68	
19	20	PART	0.9915	61.36	1.58	0.660	0.049	0.110	0.009	631.59	11.29	
20	21	PART	0.9844	63.35	2.22	-0.029	0.104	0.135	0.012	523.71	20.42	
21	22	PART	0.9949	62.00	1.10	0.196	0.047	0.128	0.007	544.68	10.60	
22	24	PART	0.9973	58.66	0.84	0.314	0.029	0.132	0.006	518.34	6.94	
23	25	PART	0.9919	60.05	1.38	0.719	0.069	0.177	0.013	846.95	16.79	
24	26	PART	0.9770	59.00	2.27	0.614	0.053	0.083	0.011	483.73	12.66	
25	27	PART	0.9692	59.28	2.67	-0.016	0.067	0.088	0.013	273.67	15.79	
26	29	PART	0.9779	59.00	2.16	0.977	0.091	0.147	0.018	814.44	22.35	
27	30	PART	0.9979	64.33	0.73	0.038	0.057	0.210	0.007	889.05	12.36	
28	31	PART	0.9806	63.55	2.26	0.305	0.115	0.141	0.014	677.44	24.74	
29	32	PART	0.9893	70.00	1.77	0.431	0.073	0.089	0.005	644.15	13.16	
30	33	PART	0.9906	64.26	1.51	0.021	0.057	0.101	0.007	426.55	12.20	
31	34	PART	0.8205	47.00	4.57	0.725	0.231	0.350	0.190	732.70	104.01	
32	35	PART	0.9819	24.00	-9.00	0.160	0.056	4.507	0.194	496.31	23.44	
33	36	PART	0.9909	61.00	1.37	-0.011	0.090	0.208	0.015	723.08	21.40	
34	38	PART	0.9940	62.53	1.39	0.880	0.097	0.239	0.016	1229.14	21.85	
35	39	PART	0.9902	59.23	1.62	0.470	0.069	0.160	0.014	675.13	16.54	
36	40	PART	0.9755	64.88	2.82	1.410	0.165	0.168	0.020	1233.46	34.38	
37	41	PART	0.9927	67.39	1.71	0.107	0.110	0.162	0.010	825.34	19.31	
38	43	PART	0.9840	67.09	2.48	0.003	0.114	0.116	0.011	557.42	20.22	

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	44	PART	0.9633	65.26	3.08	0.318	0.098	0.079	0.010	462.00	19.91
40	45	PART	0.9483	64.04	3.78	0.851	0.268	0.190	0.031	1092.02	55,60
41	46	PART	0.9697	59.00	2.94	0.097	0.088	0.114	0.019	391.35	20.75
42	47	PART	0.9496	47.00	1.86	0.490	0.080	0.245	0.054	506.36	33.57
43	48	PART	0.9867	61.55	2.16	0.347	0.053	0.098	0.011	480.06	12.30
44	49	PART	0.9716	59.15	2.45	0.435	0.100	0.144	0.020	609.82	25.00
45	50	PART	0.9589	60.13	2.95	0.895	0.146	0.165	0.026	874.88	34.19
46	52	PART	0.9809	71.07	3.29	-0.200	0.278	0.178	0.016	948.26	36.98
47	53	PART	0.9013	55.79	4.68	0.428	0.116	0.105	0.031	423.69	31.45
48	54	PART	0.8724	52.00	4.80	0.054	0.262	0.310	0.115	630.73	81.50
49	55	PART	0.9913	70.00	1.89	0.211	0.163	0.185	0.011	1092.15	25.12
50	56	PART	0.9419	68.71	4.74	0.094	0.135	0.065	0.010	371.12	23.11
51	57	PART	0.9926	64.28	1.43	0.153	0.048	0.092	0.006	438.60	10.05
52	58	PART	0.9045	63.00	5.28	0.517	0.288	0.154	0.037	788.65	61.26
53	5 <del>9</del>	PART	0.9737	64.01	2.72	0.559	0.136	0.136	0.016	763.91	28.27
54	60	PART	0.9745	69.95	3.82	0.159	0.174	0.111	0.013	666.58	26.71
55	61	PART	0.9931	53.65	1.44	0.094	0.047	0.204	0.021	486.70	14.32
56	64	PART	0.9785	64.70	2.40	0.571	0.137	0.156	0.016	871.97	29.32
57	65	PART	0.9629	68.56	3.64	-0.052	0.193	0.120	0.015	593.76	34.18
58	66	PART	0.9650	67.19	3.76	0.139	0.216	0.144	0.019	743.70	38.18
59	67	PART	0.9906	66.80	1.69	-0.012	0.122	0.184	0.012	864.47	24.24
60	68	PART	0.9750	60.00	2.35	0.361	0.114	0.160	0.020	660.48	26.68
61	70	PART	0.9861	55.50	1.89	0.394	0.043	0.110	0.013	420.12	11.00
62	71	PART	0.8478	70.06	8.91	-0.076	0.512	0.126	0.033	663.77	76.16
63	72	PART	0.9805	60.35	2.53	0.137	0.065	0.099	0.013	384.59	15.08
64	73	PART	0.9887	65.61	1.78	0.314	0.112	0.160	0.012	825.40	22.32
65	74	PART	0.9605	65.56	3.34	0.473	0.124	0.095	0.013	592.85	25.01
66	75	PART	0.9557	66.98	3.97	0.004	0.153	0.098	0.015	467.25	29.04
67	76	PART	0.9948	59.05	1.03	0.244	0.043	0.140	0.008	525.52	10.23
68	77	PART	0.9932	52.55	2.48	-0.035	0.071	0.153	0.028	300.51	19.07
69	82	PART	0.9855	60.20	1.86	0.570	0.064	0.121	0.012	610.75	15.25
70	83	PART	0.9847	65.97	2.09	0.511	0.232	0.291	0.025	1502.00	48.58
71	84	PART	0.9349	60.40	3.98	0.462	0.186	0.160	0.034	709.86	44.95
72	85	PART	0.9943	69.06	1.54	-0.033	0.048	0.071	0.004	362.65	7.87
73	86	PART	0.7790	74.00	12.08	0.742	0.338	0.048	0.015	586.43	42.97
74	87	PART	0.9463	71.00	5.31	0.356	0.285	0.112	0.017	773.52	40.44
75	88	PART	0.9676	64.91	3.45	0.423	0.257	0.222	0.032	1106.36	52.77
76	89	PART	0.9862	69.00	2.13	1.221	0.081	0.084	0.006	886.44	14.14

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	90	PART	0.9632	66.56	3.34	-0.041	0.178	0.131	0.017	595.10	35.38
78	92	PART	0.9320	62.33	4.33	0.836	0.135	0.098	0.020	672.37	30.57
79	93	PART	0.9888	54.69	1.74	0.622	0.053	0.166	0.020	622.63	15.03
80	94	PART	0.9853	71.85	3.02	0.374	0.093	0.068	0.006	541.05	12.74
81	95	PART	0.9110	76.70	8.37	0.199	0.392	0.073	0.012	609.08	37.99
82	97	PART	0.9918	65.09	1.56	1.177	0.118	0.193	0.012	1267.22	22.44
83	98	PART	0.8667	67.29	5.77	0.214	0.598	0.216	0.045	1121.02	111.15
84	99	PART	0.9580	73.00	5.02	0.322	0.369	0.133	0.016	948.55	45.36
85	100	PART	0.9017	61.45	5.90	0.885	0.136	0.086	0.025	630.90	30.87
86	101	PART	0.9719	67.61	3.24	0.377	0.145	0.110	0.013	675.24	26.23
87	102	PART	0.9734	68.00	2.85	0.449	0.235	0.183	0.019	1076.77	41.98
88	104	PART	0.9768	70.23	3.72	0.454	0.231	0.143	0.015	954.61	32.47
89	105	PART	0.7466	73.00	15.76	0.517	0.318	0.038	0.014	425.47	34.74
90	106	PART	0.9561	70.86	5.46	0.552	0.193	0.083	0.013	676.06	26.76
91	107	PART	0.9213	70.38	6.14	0.298	0.196	0.069	0.013	494.69	30.58
92	108	PART	0.9103	62.00	4.47	0.231	0.092	0.061	0.013	308.46	21.09
93	109	PART	0.9255	52.00	4.29	0.792	0.106	0.150	0.051	584.69	33.26
94	110	PART	0.9793	71.65	3.29	0.566	0.232	0.149	0.014	1086.80	33.37
95	111	PART	0.9877	61.28	1.67	0.865	0.063	0.119	0.010	736.75	14.71
96	112	PART	0.9768	73.00	3.39	0.845	0.347	0.182	0.015	1448.01	44.86
97	113	PART	0.9851	71.87	2.89	0.391	0.337	0.254	0.021	1651.28	47.57
98	114	PART	0.9780	68.39	2.85	0.160	0.178	0.142	0.014	778.95	30.66
99	115	PART	0.9946	73.90	2.06	0.180	0.112	0.104	0.005	739.66	12.43
100	. 116	PART	0.9812	71.67	3.11	-0.100	0.200	0.135	0.012	758.64	28.56
101	117	PART	0.9794	59.64	1.54	0.448	0.101	0.174	0.015	725.40	25.13
102	118	PART	0.8110	82.00	64.34	-0.314	6.891	0.117	0.026	939.32	108.24
103	119	PART	0.8911	87.00	-9.00	-0.380	0.342	0.094	0.010	874.78	52.99
104	120	PART	0.9448	58.70	3.78	0.655	0.171	0.171	0.036	763.61	42.17
105	121	PART	0.9767	61.03	2.18	0.080	0.051	0.071	0.008	278.31	12.05
106	123	PART	0.8667	74.76	10.24	0.014	0.364	0.061	0.013	418.33	39.44
107	124	PART	0.9787	63.78	2.37	-0.035	0.118	0.138	0.015	547.51	25.62
108	125	PART	0.7201	80.00	35.04	0.209	1.293	0.047	0.015	466.17	49.50
109	126	PART	0.9726	65.00	2.89	0.327	0.098	0.090	0.010	507.57	18.81
110	127	PART	0.8665	76.00	11.39	0.063	0.413	0.056	0.011	420.82	36.16
111	128	PART	0.9610	68.32	3.89	0.105	0.390	0.228	0.030	1194.57	65.47
112	129	PART	0.9509	72.05	5.82	0.832	0.195	0.068	0.010	710.11	23.16
113	130	PART	0.9931	65.63	1.32	0.637	0.048	0.090	0.005	635.36	9.89
114	131	PART	0.9501	70.83	5.35	0.319	0.332	0.142	0.022	920.62	48.92

OBS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	132	PART	0.9699	64.99	3.07	0.551	0.138	0.125	0.016	739.60	28.57
116	133	PART	0.9605	49.36	3.05	0.152	0.087	0.241	0.069	446.39	37.26
117	134	PART	0.9847	70.08	2.55	0.533	0.153	0.130	0.010	909.64	24.20
118	135	PART	0.9934	65.34	1.31	0.068	0.059	0.128	0.007	584.49	12.78
119	136	PART	0.9776	60.56	2.53	-0.001	0.107	0.173	0.023	589.98	26.71
120	137	PART	0.7718	85.00	306.16	-0.550	23.254	0.077	0.016	573.94	100.81
121	138	PART	0.9629	71.49	4.42	0.793	0.265	0.128	0.016	1036.70	38.86
122	139	PART	0.9656	68.57	3.87	0.481	0.247	0.151	0.019	952.04	41.04
123	140	PART	0.8630	70.00	7.74	0.541	0.200	0.056	0.014	504.51	32.37
124	141	PART	0.9503	58.23	3.75	0.210	0.131	0.134	0.028	476.48	32.06
125	142	PART	0.9822	74.99	3.41	1.422	0.200	0.102	0.008	1213.80	22.65
126	143	PART	0.9843	67.60	2.56	0.811	0.153	0.153	0.014	1045.08	27.10
127	144	PART	0.9593	70.80	5.37	0.004	0.137	0.060	0.009	341.31	18.52
128	145	PART	0.9837	57.00	2.08	1.020	0.094	0.191	0.024	900.24	23.16
129	146	PART	0.9913	66.06	1.69	0.199	0.082	0.122	0.008	628.45	15.36
130	147	PART	0.9934	60.38	1.36	0.236	0.094	0.252	0.018	935.46	21.42
131	148	PART	0.9235	62.43	4.53	1.301	0.187	0.128	0.029	960.06	42.82
132	149	PART	0.8300	80.00	17.57	0.590	2.451	0.175	0.035	1679.87	135.81
133	150	PART	0.9923	62.54	1.40	0.669	0.072	0.161	0.011	856.94	16.56
134	151	PART	0.9610	68.00	3.63	0.382	0.129	0.079	0.010	536.09	22.59
135	152	PART	0.9944	61.05	1.08	0.755	0.129	0.377	0.021	1595.21	30.48
136	153	PART	0.9715	69.18	3.41	1.142	0.130	0.086	0.010	870.89	21.41
137	154	PART	0.9902	61.67	1.92	0.502	0.202	0.420	0.040	1705.31	46.27
138	155	PART	0.9778	66.21	2.44	0.156	0.152	0.152	0.015	752.80	30.27
139	156	PART	0.9231	51.00	3.64	0.669	0.210	0.370	0.121	922.85	77.32
140	157	PART	0.9948	68.11	1.47	0.620	0.056	0.090	0.004	680.17	9.36
141	158	PART	0.8902	74.00	8.04	0.327	0.224	0.048	0.009	430.58	26.49
142	159	PART	0.7338	67.39	8.69	0.398	0.288	0.068	0.022	474.42	54.15
143	160	PART	0.9798	64.99	2.51	0.725	0.124	0.137	0.014	853.73	25.56
144	161	PART	0.9915	61.27	1.37	0.636	0.056	0.130	0.009	692.55	13.26
145	162	PART	0.9189	76.00	7.81	0.528	0.495	0.099	0.015	894.14	47.32
146	163	PART	0.9846	73.00	3.50	0.218	0.216	0.116	0.009	806.83	22.82
147	165	PART	0.9771	68.97	3.09	0.320	0.131	0.099	0.010	635.36	22.43
148	166	PART	0.9540	45.00	2.63	0.222	0.045	0.168	0.054	268.97	23.52
149	1080	PART	0.9836	65.01	2.06	0.118	0.084	0.104	0.009	493.21	17.11

S	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
				~ ~ ~			0 057	0 407	0 004	707 00	
1	580002013800	NPART	0.9976	68.69	1.08	0.373	0.057	0.127	0.004	767.28	10.37
2	585018022900	NPART	0.9969	66.59	1.18	0.704	0.046	0.101	0.004	709.24	8.74
3	590011001500	NPART	0.9963	66.00	1.12	0.283	0.058	0.118	0.004	619.13	11.42
4	590011001605	NPART	0.9979	68.27	0.79	-0.007	0.038	0.112	0.003	542.04	8.23
5	590011007000	NPART	0.9943	67.30	1.42	0.456	0.172	0.272	0.012	1430.75	33.70
6	590017004000	NPART	0.9957	63.73	1.45	0.320	0.067	0.143	0.007	671.02	13.62

I

s	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
1	500003118400	NPART	0.9934	72.00	1.87	0.132	0.120	0.132	0.007	807.86	16.22
2	500003120800	NPART	0.9775	63.58	2.45	0.695	0.116	0.141	0.016	795.81	25.32
3	500003143201	NPART	0.9777	65.04	2.55	0.409	0.160	0.168	0.018	848.20	32.04
4	500003220641	NPART	0.9760	63.88	2.60	0.247	0.077	0.089	0.010	437.25	16.70
5	500005115602	NPART	0.9798	68.00	2.83	0.430	0.134	0.107	0.010	671.68	21.75
6	500007377900	NPART	0.7309	61.02	8.74	1.110	0.593	0.204	0.091	1086.64	134.45
7	500008129800	NPART	0.9436	75.00	6.20	1.082	0.540	0.146	0.019	1354.53	58.21
8	500009318100	NPART	0.9785	65.94	2.71	0.919	0.077	0.078	0.009	676.65	15.31
9	500010523502	NPART	0.9787	66.87	2.72	0.453	0.217	0.206	0.021	1104.35	41.61
10	500011181901	NPART	0.9951	64.91	1.22	1.111	0.086	0.200	0.010	1228.95	18.12
11	500016212601	NPART	0.9724	62.71	2.84	0.452	0.127	0.145	0.020	696.34	28.42
12	500017250901	NPART	0.9801	67.55	2.80	0.227	0.111	0.098	0.010	541.83	19.56
13	500018446600	NPART	0.9904	57.00	1.26	1.213	0.101	0.281	0.022	1182.74	25.60
14	500018477800	NPART	0.9815	69.29	2.80	0.810	0.214	0.166	0.015	1144.55	33.77
15	500018506200	NPART	0.9902	60.70	1.44	0.541	0.113	0.248	0.019	1012.86	27.04
16	500018562000	NPART	0.8329	60.63	6.45	0.553	0.433	0.212	0.072	896.62	103.78
17	500020296100	NPART	0.9651	63.33	3.06	0.462	0.059	0.058	0.008	388.94	12.92
18	501006019200	NPART	0.9942	70.22	1.56	0.440	0.131	0.175	0.008	1096.11	19.83
19	501006050900	NPART	0.9856	64.60	2.00	0.486	0.150	0.208	0.018	1022.78	31.49
20	501006052101	NPART	0.9808	66.51	2.54	0.192	0.076	0.078	0.008	416.72	14.45
1	506002056800	NPART	0.9911	64.10	1.52	0.143	0.068	0.121	0.008	531.26	14.13
22	506002077900	NPART	0.9853	63.95	2.15	0.112	0.072	0.103	0.010	446.23	15.31
23	506002084300	NPART	0.9921	65.91	1.68	0.338	0.061	0.101	0.007	563.20	11.85
24	506002125400	NPART	0.9853	63.15	2.02	0.961	0.114	0.170	0.016	991.45	24.43
25	506003071305	NPART	0.7478	73.20	13.46	0.392	0.050	0.007	0.002	185.19	6.39
26	506003077200	NPART	0.9884	66.95	2.03	0.362	0.060	0.077	0.006	485.19	11.33
27	515005032902	NPART	0.9904	64.00	1.51	0.624	0.171	0.305	0.021	1429.23	36.36
28	515007009000	NPART	0.8843	59.00	4.50	0.973	0.160	0.106	0.026	670.77	38.33
29	515007014307	NPART	0.9959	57.68	0.84	0.163	0.027	0.110	0.005	363.06	6.76
30	515008021001	NPART	0.9554	67.63	4.10	0.549	0.283	0.167	0.024	988.40	50.27
31	515008022303	NPART	0.9876	62.23	1.81	0.787	0.095	0.164	0.014	875.61	21.06
32	515014036400	NPART	0.9746	56.94	2.00	1.219	0.142	0.248	0.030	1095.98	36.84
33	515020016700	NPART	0.9567	56.36	2.81	0.457	0.120	0.165	0.029	583.93	31.52
34	515102005200	NPART	0.8138	88.00	-9.00	-0.316	0.173	0.034	0.005	258.42	26.09
35	518019019002	NPART	0.9458	69.67	4.71	0.571	0.199	0.089	0.013	671.50	31.85
36	530002014600	NPART	0.9778	68.03	3.20	0.518	0.156	0.114	0.012	738.39	24.58
37	530004023301	NPART	0.9495	55.12	2.83	0.269	0.058	0.080	0.015	285.20	15.58
38	530011004702	NPART	0.9734	67.88	3.04	0.440	0.131	0.102	0.011	645.74	23.58

s	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	531015009905	NPART	0.9868	68.00	1.95	-0.088	0.163	0.202	0.014	938.20	31.17
40	531015013403	NPART	0.9834	59.33	1.60	0.282	0.104	0.193	0.017	689.53	25.57
41	531015022301	NPART	0.9915	64.33	1.51	0.207	0.066	0.120	0.008	555.86	14.03
42	532003016401	NPART	0.9679	68.77	4.04	0.101	0.245	0.145	0.019	760.93	38.36
43	532003019202	NPART	0.9789	59.89	1.74	-0.032	0.106	0.177	0.017	544.64	26.16
44	534007003801	NPART	0.9924	66.27	1.56	0.802	0.168	0.273	0.016	1498.31	31.62
45	534013001602	NPART	0.9469	61.16	3.41	-0.018	0.116	0.103	0.018	340.83	26.64
46	534016005500	NPART	0.9722	75.06	3.78	0.197	0.274	0.116	0.010	839.88	32.06
47	534016008700	NPART	0.8931	62.36	5.52	.0.290	0.137	0.078	0.021	387.52	31.17
48	534018009900	NPART	0.9829	69.60	2.62	0.584	0.190	0.156	0.013	1020.21	30.54
49	537001000201	NPART	0.9738	73.04	4.05	0.764	0.307	0.144	0.014	1149.58	36.66
50	537001017100	NPART	0.9679	67.24	3.70	0.767	0.339	0.225	0.029	1324.39	57.10
51	537002006702	NPART	0.9712	60.62	2.46	0.294	0.090	0.112	0.014	472.22	21.03
52	537002030200	NPART	0.9763	70.00	3.44	0.146	0.310	0.188	0.018	1044.67	43.82
53	537003016500	NPART	0.9854	62.78	2.03	0.210	0.084	0.130	0.012	554.87	18.49
54	537003030701	NPART	0.9868	61.48	1.76	0.574	0.059	0.104	0.009	566.39	13.37
55	537004000302	NPART	0.9840	68.35	2.48	0.089	0.161	0.148	0.012	755.11	26.79
56	537007022100	NPART	0.9918	73.00	2.12	0.141	0.131	0.113	0.006	731.19	16.41
57	537017035801	NPART	0.9827	61.68	2.00	0.328	0.064	0.099	0.010	463.15	14.62
58	537017047501	NPART	0.9971	65.75	0.94	0.019	0.027	0.078	0.003	343.69	5.45
9	537019009200	NPART	0.9812	67.30	2.98	-0.069	0.109	0.093	0.009	405.77	18.05
60	540115007300	NPART	0.9368	60.91	3.85	0.581	0.147	0.122	0.024	617.22	35.05
61	541006031600	NPART	0.9938	59.72	1.08	0.501	0.070	0.204	0.012	815.42	16.66
62	541007002301	NPART	0.9747	64.17	2.65	0.700	0.146	0.153	0.018	862.76	30.51
63	541009010700	NPART	0.9901	66.31	1.76	0.450	0.146	0.210	0.014	1093.92	27.93
64	541009019200	NPART	0.9881	66.36	1.94	0.586	0.085	0.110	0.008	704.11	16.15
65	541012039900	NPART	0.9877	70.83	2.31	0.463	0.224	0.207	0.014	1305.18	35.19
66	542014008952	NPART	0.9783	62.60	2.46	0.719	0.058	0.075	0.009	535.71	12.89
67	550001038700	NPART	0.9837	68.83	2.86	-0.101	0.374	0.315	0.029	1536.22	59.24
68	550003052601	NPART	0.9939	59.04	0.99	0.279	0.074	0.225	0.012	771.63	17.66
69	550006006501	NPART	0.9868	65.22	2.02	0.788	0.145	0.192	0.016	1093.35	28.51
70	550009076206	NPART	0.9913	64.19	1.48	0.880	0.099	0.179	0.012	1034.33	20.82
71	550014018900	NPART	0.9903	63.68	1.52	0.840	0.136	0.253	0.018	1285.42	30.43
72	550014025300	NPART	0.9284	54.98	3.97	0.727	0.261	0.306	0.079	972.42	71.81
73	550016016100	NPART	0.9247	70.00	5.56	1.060	0.302	0.112	0.019	977.19	47.57
74	550020050000	NPART	0.9662	52.64	2.62	0.687	0.144	0.300	0.059	842.12	45.53
75	550021023700	NPART	0.9815	64.47	2.25	0.523	0.078	0.098	0.010	584.67	16.64
76	550104003400	NPART	0.8474	54.00	5.45	0.887	0.322	0.257	0.096	879.12	88.08

	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	553007005001	NPART	0.9772	64.33	2.48	0.418	0,142	0.159	0.017	788.12	30.04
78	555004016803	NPART	0.8755	54.38	5.42	1.359	0.189	0.164	0.060	861.19	51.92
79	555005016800	NPART	0.9856	72.11	2.52	0.482	0.118	0.093	0.007	718.10	17.12
80	555008022602	NPART	0.9941	59.00	0.99	0.315	0.060	0.187	0.010	672.19	14.33
81	555012028400	NPART	0.9809	62.47	2.18	0.386	0.099	0.139	0.015	645.50	22.50
82	555013028700	NPART	0.9936	58.00	0.97	0.068	0.062	0.202	0.012	591.73	15.60
83	555015009007	NPART	0.9678	49.20	1.99	0.417	0.044	0.104	0.017	310.15	12.67
84	555019023192	NPART	0.9930	61.64	1.32	0.002	0.064	0.160	0.011	554.34	14.86
85	555021029100	NPART	0.9759	67.78	3.47	-0.117	0.227	0.167	0.020	753.12	37.50
86	560013031400	NPART	0.9935	60.87	1.11	0.703	0.045	0.123	0.007	663.91	11.00
87	560015018800	NPART	0.9526	49.93	3.11	0.650	0.172	0.383	0.106	855.80	66.50
88	560016037700	NPART	0.9531	57.61	2.83	0.435	0.066	0.082	0.014	382.97	17.28
89	560016100900	NPART	0.9194	61.65	4.51	0.591	0.153	0,107	0.024	587.77	35.72
) Ó	560017023404	NPART	0.9851	57.58	1.65	0.239	0.096	0.213	0.021	668.58	24.89
91	562006032700	NPART	0.9820	69.59	2.87	0.173	0.162	0.123	0.011	698.20	24.74
92	562007009500	NPART	0.9787	74.50	3.70	0.116	0.228	0.109	0.009	742.39	26.29
93	562008002303	NPART	0.9796	57.29	1.82	0.536	0.119	0.223	0.024	795.09	29.99
94	562008020500	NPART	0.9789	60.19	2.04	0.595	0.066	0.099	0.011	532.94	15.51
95	562017012100	NPART	0.9903	67.39	1.88	1.096	0.101	0.131	0.009	1010.58	18.25
<b>9</b> 6	562021013900	NPART	0.9807	65.11	2.44	-0.039	0.140	0.159	0.016	647.64	28.29
7	564010006302	NPART	0.9926	65.54	1.53	0.166	0.050	0.090	0.006	442.44	10.02
98	564011023900	NPART	0.9793	70.75	3.37	0.078	0.213	0.132	0.012	748.39	29.73
99	564011024000	NPART	0.8817	75.82	10.59	-0.037	0.135	0.021	0.004	127.96	12.99
100	565001014600	NPART	0.9743	67.09	3.39	0.680	0.155	0.116	0.014	781.65	26.25
101	570004021700	NPART	0.9766	57.64	1.99	0.846	0.168	0.291	0.034	1105.92	42.44
102	570007017101	NPART	0.7849	70.85	11.13	0.541	0.912	0.170	0.054	1132.04	135.26
103	570008008300	NPART	0.9409	75.00	6.47	0.301	0.252	0.065	0.009	538.71	26.78
104	570010011500	NPART	0.9319	64.71	4.68	0.052	0.216	0.130	0.026	550.41	45.23
105	570010042400	NPART	0.8855	58.23	4.49	1.027	0.125	0.087	0.022	621.27	30.62
106	570011001800	NPART	0.9559	62.87	3.79	0.696	0.210	0.180	0.032	920.22	46.00
107	570011003001	NPART	0.9741	63.33	2.76	0.301	0.081	0.089	0.011	447.81	17.47
108	570120002600	NPART	0.9906	64.86	1.67	0.261	0.074	0.126	0.009	613.74	15.39
109	570305000900	NPART	0.9390	67.57	5.04	0.166	0.188	0.090	0.016	485.92	32.29
110	572017002703	NPART	0.9004	57.79	3.04	0.150	0.519	0.428	0.079	1239.64	130.12
111	573012001102	NPART	0.9965	59.10	0.81	0.491	0.043	0.166	0.007	675.25	10.04
112	574018001602	NPART	0.9703	64.00	2.74	0.382	0.147	0,146	0.018	/14.00	31.14
113	574018025100	NPART	0.9732	75.85	4,41	0.494	0.335	0.130	0.012	1068.52	35.54
114	580002013800	NPARE	0.9838	70.77	3.10	0.301	0.164	0.118	0.010	154.82	22.89

	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	585018022900	NPART	0.9777	67.67	3.04	0.703	0.123	0.102	0.011	740.68	21.76
116	590011001500	NPART	0.9936	67.10	1.46	0.325	0.078	0.125	0.007	693.67	14.07
117	590011001605	NPART	0.9878	70.80	2.29	-0.070	0.102	0.094	0.006	486.31	16.08
118	590011007000	NPART	0.9862	72.54	2.53	0.292	0.245	0.191	0.013	1237.32	35.83
119	590017004000	NPART	0.9887	66.47	2.00	0.162	0.105	0.139	0.011	679.99	19.97

	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
1	500003118400	NPART	0.9879	66.00	2.24	0.230	0.126	0.151	0.012	757.92	24.01
2	500003120800	NPART	0.9946	62.04	1.55	0.403	0.078	0.177	0.011	788.52	16.78
3	500003143201	NPART	0.9924	55.75	1.80	0.557	·0.129	0.354	0.032	1074.37	30.13
4	500003220641	NPART	0.9860	62.22	2.52	0.181	0.074	0.104	0.010	446.18	15.94
5	500005115602	NPART	0.9654	57.39	4.00	0.759	0.151	0.173	0.033	750.44	35.69
6	500007377900	NPART	0.9873	61.00	2.38	1.184	0.168	0.256	0.025	1304.94	36.66
7	500008129800	NPART	0.9983	70.65	0.84	0.855	0.073	0.185	0.004	1347.26	12.69
8	500009318100	NPART	0.9432	62.51	5.08	0.877	0.171	0.112	0.022	734.92	36.33
9	500010523502	NPART	0.9535	61.31	4.84	0.636	0.286	0.218	0.042	990.73	61.66
10	500016212601	NPART	0.9672	60.78	3.95	0.377	0.195	0.191	0.031	782.48	43.58
11	500017250901	NPART	0.9813	63.30	2.71	0.276	0.126	0.143	0.015	653.04	26.44
12	500018446600	NPART	0.9949	55.90	1.37	1.033	0.111	0.361	0.024	1275.13	25.68
13	500018477800	NPART	0.9792	65.60	2.98	1.081	0.205	0.184	0.019	1199.84	38.18
14	500018506200	NPART	0.9480	58.11	4.42	0.671	0.299	0.266	0.053	1007.17	69.29
15	500018562000	NPART	0.9710	62.88	3.55	0.563	0.218	0.205	0.028	981.96	46.82
16	501006019200	NPART	0.9849	66.85	2.58	0.658	0.192	0.200	0.017	1170.58	37.24
17	501006050900	NPART	0.9899	59.00	1.99	0.393	0.154	0.301	0.027	1057.04	35.60
18	501006052101	NPART	0.9660	59.29	3.79	0.216	0.083	0.086	0.014	344.88	19.16
19	506002056800	NPART	0.9568	60.45	4.87	0.152	0.170	0.145	0.030	535.36	38.18
20	506002077900	NPART	0.9526	55.45	4.17	0.133	0.170	0.186	0.040	498.07	40.50
	506002084300	NPART	0.9551	57.03	4.43	0.398	0.192	0.195	0.042	667.24	45.37
22	506002125400	NPART	0.9745	61.75	3.67	1.180	0.173	0.183	0.027	1081.72	38.61
23	506003077200	NPART	0.9853	65.85	2.44	0.229	0.081	0.091	0.008	485.92	16.21
24	515005032902	NPART	0.9947	56.86	1.53	0.794	0.159	0.487	0.036	1580.71	37.66
25	515007009000	NPART	0.9955	59.75	1.38	0.892	0.037	0.107	0.006	665.07	8.71
26	515007014307	NPART	0.9827	51.17	2.02	0.385	0.062	0.150	0.018	407.02	15.86
27	515008021001	NPART	0.9701	61.64	3.81	0.938	0.288	0.276	0.041	1317.86	62.85
28	515008022303	NPART	0.9399	60.00	5.12	1.056	0.216	0.155	0.033	883.13	47.34
29	515014036400	NPART	0.9906	59.69	2.21	1.440	0.094	0.185	0.018	1111.33	21.37
30	515020016700	NPART	0.9945	55.38	1.51	0.433	0.059	0.195	0.015	626.76	13.50
31	515102005200	NPART	0.9937	65.00	1.56	0.181	0.026	0.046	0.003	262.88	5.13
32	518019019002	NPART	0.9690	55.36	3.38	0.848	0.131	0.178	0.031	737.40	30.35
33	532003016401	NPART	0.9859	66.31	2.56	0.437	0.158	0.171	0.015	933.54	30.46
34	532003019202	NPART	0.9773	59.00	3.39	0.215	0.128	0.161	0.024	567.17	29.19
35	537001000201	NPART	0.9901	71.88	2.51	0.423	0.171	0.149	0.010	1036.89	25.22
36	537001017100	NPART	0.9962	63.21	1.34	0.639	0.116	0.294	0.015	1366.75	24.18
37	537002006702	NPART	0.9777	59.00	3.31	0.433	0.092	0.120	0.018	520.43	21.15
38	537002030200	NPART	0.9915	62.46	2.10	0.237	0.166	0.292	0.024	1167.74	36.11

s	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
39	537003016500	NPART	0.9970	58.57	1.18	0.229	0.055	0.211	0.011	706.33	13.22
40	537003030701	NPART	0.9936	54.39	1.45	0.595	0.047	0.156	0.012	566.52	11.40
41	537004000302	NPART	0.9774	64.68	3.25	0.320	0.160	0.156	0.019	767.03	33.29
42	537007022100	NPART	0.9928	65.00	1.56	0.437	0.088	0.146	0.008	777.99	17.61
43	537012007100	NPART	0.9915	60.11	2.14	0.073	0.100	0.199	0.018	670.04	22.22
44	537017035801	NPART	0.9688	57.36	3.48	0.285	0.082	0.098	0.016	373.07	19.00
45	537017047501	NPART	0.9946	62.00	1.41	0.057	0.039	0.088	0.005	338.49	8.27
46	537019009200	NPART	0.9877	58.49	2.26	-0.007	0.095	0.176	0.018	514.45	21.98
47	540004004700	NPART	0.9901	54.82	1.62	0.973	0.059	0.152	0.013	705.75	14.15
48	540115007300	NPART	0.9866	55.59	2.13	0.610	0.092	0.187	0.020	678.16	21.35
49	541006031600	NPART	0.9941	60.35	1.61	0.240	0.135	0.318	0.021	1132.72	29.91
50	541007002301	NPART	0.9896	66.23	1.90	0.549	0.123	0.157	0.010	907.75	23.71
51	541009010700	NPART	0.9827	66.07	2.42	0.468	0.266	0.267	0.022	1368.29	51.93
52	541009019200	NPART	0.9906	62.99	2.16	0.611	0.083	0.134	0.011	734.16	17.57
53	541012039900	NPART	0.9738	63.00	3.05	0.802	0.288	0.282	0.033	1369.96	60.83
54	542014008952	NPART	0.9502	67.26	4.52	0.593	0.130	0.070	0.010	549.10	24.00
55	560013031400	NPART	0.9936	57.02	1.55	0.646	0.059	0.167	0.013	682.44	13.95
56	560015018800	NPART	0.9906	52.77	1.81	0.497	0.107	0.326	0.034	833.61	26.90
57	560016037700	NPART	0.9662	55.36	3.55	0.416	0.101	0.132	0.024	469.18	23.84
58	560016100900	NPART	0.9807	61.16	3.13	0.473	0.100	0.123	0.015	595.07	21.59
9	560017023404	NPART	0.9926	56.62	1.77	0.244	0.087	0.229	0.019	685.65	20.36
60	562006032700	NPART	0.9965	64.41	1.19	0.273	0.068	0.171	0.007	801.79	14.18
61	562007009500	NPART	0.9805	70.25	3.00	0.247	0.169	0.123	0.011	767.03	28.92
62	562008002303	NPART	0.9726	58.24	3.14	-0.032	0.214	0.280	0.039	798.98	49.98
63	562008020500	NPART	0.9936	57.36	1.48	0.561	0.049	0.132	0.009	566.97	11.63
64	562009011501	NPART	0.9843	70.04	2.54	0.133	0.274	0.232	0.017	1306.58	47.78
65	562017012100	NPART	0.9797	60.23	3.13	1.058	0.117	0.149	0.019	871.84	26.09
66	562021013900	NPART	0.9993	58.56	0.58	0.027	0.030	0.240	0.006	717.86	7.05
67	564010006302	NPART	0.9846	60.47	1.86	0.087	0.071	0.116	0.009	416.22	16.32
68	564011023900	NPART	0.9952	64.35	1.42	0.249	0.078	0.166	0.009	770.77	15.94
69	564011024000	NPART	0.9820	57.89	3.02	0.026	0.033	0.052	0.007	155.36	7.76
70	565001014600	NPART	0.9929	62.46	1.74	0.512	0.077	0.154	0.011	757.41	16.62
71	570004021700	NPART	0.9907	52.99	2.04	0.922	0.142	0.431	0.051	1212.94	35.27
72	570007017101	NPART	0.9947	54.00	1.08	0.790	0.118	0.426	0.025	1218.61	28.41
73	570008008300	NPART	0.9828	62.50	2.69	0.645	0.102	0.124	0.013	697.07	21.75
74	570010011500	NPART	0.9799	56.13	2.78	0.235	0.139	0.220	0.030	640.06	32.21
75	570010042400	NPART	0.9801	57.17	2.86	1.037	0.078	0.118	0.016	696.88	18.13
76	570011001800	NPART	0.9907	61.85	2.22	0.621	0.137	0.239	0.021	1083.89	30.08

SS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
77	570011003001	NPART	0.9885	61.45	2.39	0.375	0.094	0.151	0.015	664.91	20.72
78	570120002600	NPART	0.9903	63.55	2.04	0.238	0.096	0.154	0.012	693.79	20.41
79	570305000900	NPART	0.9960	55.57	1.19	0.095	0.057	0.220	0.014	569.36	13.55
80	572017002703	NPART	0.9785	68.68	3.09	0.793	0.361	0.282	0.027	1726.26	64.88
81	573012001102	NPART	0.9989	54.57	0.64	0.497	0.023	0.183	0.006	598.05	5.60
82	574018001602	NPART	0.9704	65.81	3.66	0.371	0.175	0.132	0.016	719.12	32.97
83	574018025100	NPART	0.9882	65.61	2.26	0.767	0.177	0.215	0.017	1221.47	33.35

BS	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
1	500003118400	NPART	0.9962	69.74	1.34	0.192	0.082	0.143	0.006	834.58	12.40
2	500003120800	NPART	0.9790	61.44	2.13	0.715	0.108	0.160	0.018	819.97	25,13
3	500003143201	NPART	0.9836	62.76	2.15	0.442	0.121	0.191	0.020	878.82	27,66
4	500003220641	NPART	0.9824	61.68	1.99	0.259	0.063	0.101	0.010	452.68	14.70
5	500005115602	NPART	0.9808	65.98	2.62	0.455	0.113	0.117	0.012	689.28	21,20
6	500007377900	NPART	0.7093	62.10	10.12	1.040	0.614	0.198	0.102	1098.05	138.29
7	500008129800	NPART	0.9556	70.59	4.61	1.265	0.352	0.167	0.023	1392.92	51.48
8	500009318100	NPART	0.9841	63.00	2.01	0.939	0.061	0.092	0.009	693.65	13.27
9	500010523502	NPART	0.9835	64.48	2.06	0.492	0.181	0.234	0.021	1144.41	37.17
10	500011181901	NPART	0.9946	62.41	1.15	1.140	0.083	0.230	0.013	1266.60	18.95
11	500016212601	NPART	0.9743	60.31	2.24	0.486	0.118	0.167	0.021	725.76	28.49
12	500017250901	NPART	0.9819	65.27	2.28	0.259	0.095	0.109	0.010	562.90	18.59
13	500018446600	NPART	0.9852	54.68	1.96	1.262	0.123	0.341	0.047	1241.97	35.58
14	500018477800	NPART	0.9895	67.00	1.91	0.861	0.146	0.185	0.013	1181.81	25.23
15	500018506200	NPART	0.9935	58.63	1.14	0.569	0.092	0.289	0.020	1063.12	23.20
16	500018562000	NPART	0.8508	59.86	5.97	0.513	0.409	0.233	0.078	929.99	100.64
17	500020296100	NPART	0.9766	60.14	2.11	0.475	0.046	0.070	0.008	401.02	11.18
18	501006019200	NPART	0.9977	67.83	0.91	0.503	0.074	0.193	0.006	1127.89	12.65
19	501006050900	NPART	0.9914	61.66	1.39	0.536	0.108	0.244	0.018	1060.67	24.80
20	501006052101	NPART	0.9831	64.14	2.09	0.211	0.068	0.087	0.008	430.13	13.78
1	506002056800	NPART	0.9968	61.83	0.89	0.157	0.038	0.138	0.006	550.33	8.60
22	506002077900	NPART	0.9928	61.39	1.29	0.128	0.048	0.119	0.008	463.37	10.96
23	506002084300	NPART	0.9959	63.44	1.11	0.364	0.040	0.115	0.006	580.58	8.56
24	506002125400	NPART	0.9936	60.87	1.24	0.978	0.073	0.195	0.013	1018.66	17.02
25	506003071305	NPART	0.7548	78.00	21.75	0.371	0.104	0.007	0.002	185.99	6.04
26	506003077200	NPART	0.9949	64.00	·1.28	0.391	0.035	0.089	0.005	499.90	7.59
27	515005032902	NPART	0.9903	61.62	1.48	0.676	0.161	0.349	0.027	1479.86	37.44
28	515007009000	NPART	0.8811	57.94	4.82	0.965	0.161	0.117	0.034	684.95	41.12
29	515007014307	NPART	0.9792	56.99	1.76	0.155	0.062	0.120	0.013	376.32	16.17
30	515008021001	NPART	0.9633	64.05	3.10	0.635	0.224	0.196	0.027	1024.00	45.91
31	515008022303	NPART	0.9924	59.56	1.25	0.807	0.072	0.195	0.014	906.49	17.38
32	515014036400	NPART	0.9657	55.00	1.94	1.261	0.165	0.291	0.039	1141.49	44.60
33	515020016700	NPART	0.9604	53.49	3.30	0.487	0.114	0.210	0.052	621.12	36.47
34	515102005200	NPART	0.8275	88.00	-9.00	-0.364	0.171	0.036	0.005	268.11	25.10
35	518019019002	NPART	0.9556	67.01	3.84	0.607	0.165	0.100	0.014	691.00	28.47
36	530002014600	NPART	0.9718	65.50	3.12	0.570	0.145	0.127	0.016	761.70	27.53
37	530004023301	NPART	0.9411	54.22	3.18	0.269	0.062	0.089	0.021	295.78	17.72
38	530011004702	NPART	0.9739	65.45	2.57	0.471	0.116	0.114	0.012	667.38	23.18

	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	531015009905	NPART	0.9842	66.00	1.87	-0.049	0.167	0.223	0.018	978.39	34.74
40	531015013403	NPART	0.9915	57.57	1.08	0.285	0.073	0.224	0.015	724.09	18.90
41	531015022301	NPART	0.9977	62.37	0.73	0.213	0.031	0.136	0.005	577.63	7.25
42	532003016401	NPART	0.9800	65.92	2.65	0.172	0.158	0.164	0.017	791.10	30.00
43	532003019202	NPART	0.9802	58.22	1.64	-0.033	0.102	0.204	0.022	578.01	27.09
44	534007003801	NPART	0.9905	63.82	1.77	0.871	0.165	0.308	0.025	1544.88	35.58
45	534013001602	NPART	0.9584	59.35	2.71	-0.018	0.100	0.118	0.018	358.29	24.35
46	534016005500	NPART	0.9756	73.00	3.36	0.261	0.236	0.124	0.011	868.33	29.79
47	534016008700	NPART	0.8976	59.50	4.33	0.307	0.126	0.093	0.024	404.38	31.86
48	534018009900	NPART	0.9780	67.43	2.76	0.652	0.195	0.170	0.017	1052.08	34.11
49	537001000201	NPART	0.9857	70.73	2.86	0.832	0.186	0.156	0.013	1181.51	26.29
50	537001017100	NPART	0.9767	64.13	2.53	0.876	0.243	0.260	0.028	1371.69	48.54
51	537002006702	NPART	0.9461	59.72	3.23	0.300	0.122	0.119	0.021	484.73	29.20
52	537002030200	NPART	0.9754	68.00	3.22	0.211	0.284	0.203	0.022	1075.24	44.26
53	537003016500	NPART	0.9738	61.32	2.43	0.218	0.111	0.141	0.018	570.48	24.97
54	537003030701	NPART	0.9987	59.60	0.50	0.580	0.018	0.117	0.003	579.81	4.32
55	537004000302	NPART	0.9881	65.43	1.95	0.155	0.123	0.168	0.013	783.58	23.25
56	537007022100	NPART	0.9946	70.50	1.62	0.191	0.091	0.123	0.006	751.24	13.08
57	537017035801	NPART	0.9729	59.69	2.30	0.348	0.079	0.112	0.015	480.78	19.08
58	537017047501	NPART	0.9913	64.31	1.52	0.033	0.048	0.084	0.006	357.35	9.64
	537019009200	NPART	0.9815	64.57	2.40	-0.024	0.089	0.104	0.011	424.06	17.69
60	540115007300	NPART	0.9429	57.08	3.15	0.638	0.138	0.155	0.031	649.60	34.91
61	541006031600	NPART	0.9935	58.45	1.04	0.497	0.072	0.226	0.014	842.64	17.92
62	541007002301	NPART	0.9899	61.37	1.50	0.715	0.085	0.180	0.014	889.02	19.46
63	541009010700	NPART	0.9923	63.59	1.49	0.501	0.114	0.242	0.017	1134.59	24.77
64	541009019200	NPART	0.9957	63.78	1.14	0.607	0.045	0.127	0.007	725.21	9.80
65	541012039900	NPART	0.9905	68.18	1.80	0.571	0.180	0.230	0.015	1351.98	30.96
66	542014008952	NPART	0.9730	60.12	2.25	0.738	0.062	0.087	0.011	551.07	14.95
67	550001038700	NPART	0.9906	65.67	1.80	0.088	0.236	0.359	0.026	1606.31	44.56
68	550003052601	NPART	0.9920	57.17	1.05	0.303	0.083	0.258	0.017	808.41	21.12
69	550006006501	NPART	0.9897	63.74	1.74	0.803	0.117	0.208	0.017	1117.66	25.06
70	550009076206	NPART	0.9920	61.57	1.32	0.897	0.088	0.210	0.014	1066.06	20.11
71	550014018900	NPART	0.9883	61.55	1.41	0.869	0.142	0.289	0.022	1332.75	33.89
72	550014025300	NPART	0.9167	57.32	3.43	0.583	0.297	0.281	0.061	979.91	76.61
73	550014026103	NPART	0.7229	44.00	4.99	0.689	0.326	0.570	0.404	800.55	188.05
74	550016016100	NPART	0.9246	68.09	5.10	1.097	0.276	0.121	0.022	1001.56	47.61
75	550020050000	NPART	0.9782	49.00	1.68	0.716	0.107	U.446	0.079	928.28	46.05
76	550021023700	NPART	0.9828	61.71	1.95	0.544	0.069	0.114	0.012	602.36	16.23

S	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	550104003400	NPART	0.8191	50.00	5.33	0.973	0.332	0.365	0.188	947.55	125.75
78	553007005001	NPART	0.9857	61.91	1.84	0.429	0.105	0.183	0.017	814.66	24.18
79	555004016803	NPART	0.8672	53.85	5.95	1.362	0.196	0.176	0.075	877.29	57.20
80	555005016800	NPART	0.9845	70.44	2.66	0.500	0.122	0.100	0.008	734.45	17.84
81	555008022602	NPART	0.9880	57.65	1.35	0.307	0.085	0.211	0.018	700.55	21.71
82	555012028400	NPART	0.9653	60.75	2.57	0.399	0.128	0.155	0.022	667.99	31.35
83	555013028700	NPART	0.9877	56.00	1.22	0.095	0.086	0.236	0.020	626.12	22.62
84	555015009007	NPART	0.9628	47.95	2.73	0.408	0.046	0.132	0.035	330.67	18.15
85	555019023192	NPART	0.9859	59.56	1.65	0.024	0.089	0.183	0.018	582.10	22.17
86	555021029100	NPART	0.9744	64.27	2.73	-0.005	0.191	0.194	0.023	791.18	38.23
87	555104016702	NPART	0.8657	39.98	2.25	0.241	0.490	2.769	1.022	1848.90	315.98
88	560013031400	NPART	0.9928	59.72	1.07	0.696	0.047	0.135	0.008	682.38	11.68
89	560015018800	NPART	0.9739	48.94	2.11	0.637	0.124	0.454	0.091	909.15	49.92
90	560016037700	NPART	0.9733	52.16	2.00	0.476	0.045	0.124	0.021	411.33	14.53
91	560016100900	NPART	0.9217	60.23	3.94	0.597	0.148	0.119	0.026	606.00	36.33
92	560017023404	NPART	0.9857	55.63	1.67	0.273	0.095	0.249	0.028	708.71	26.50
93	562006032700	NPART	0.9962	66.47	1.11	0.239	0.063	0.139	0.006	722.77	11.29
94	562007009500	NPART	0.9866	72.30	2.62	0.164	0.145	0.117	0.009	765.08	19.92
95	562008002303	NPART	0.9950	56.05	0.76	0.520	0.060	0.255	0.013	830.12	15.84
96	562008020500	NPART	0.9958	57.89	0.84	0.600	0.028	0.117	0.006	551.16	7.24
	562017012100	NPART	0.9938	65.48	1.31	1.128	0.072	0.144	0.008	1037.76	14.56
-98	562021013900	NPART	0.9887	62.07	1.67	0.005	0.094	0.187	0.016	679.47	21.68
99	564010006302	NPART	0.9991	62.65	0.51	0.182	0.016	0.105	0.003	460.83	3.55
100	564011023900	NPART	0.9902	67.14	1.90	0.172	0.124	0.152	0.010	778.63	20.33
101	564011024000	NPART	0.9065	71.34	7.26	-0.009	0.085	0.024	0.005	133.52	11.08
102	565001014600	NPART	0.9802	63.00	2.36	0.753	0.102	0.140	0.016	810.00	22.58
103	570004021700	NPART	0.9866	56.00	1.23	0.847	0.130	0.338	0.029	1154.38	34.56
104	570007017101	NPART	0.8290	65.37	7.98	0.716	0.647	0.212	0.069	1179.48	121.68
105	570008008300	NPART	0.9609	72.67	4.35	0.322	0.151	0.071	0.009	552.98	21.12
106	570010011500	NPART	0.9529	61.38	3.23	0.070	0.164	0.158	0.027	576.67	37.75
107	570010042400	NPART	0.9104	56.97	3.67	1.010	0.111	0.100	0.024	634.84	28.77
108	570011001800	NPART	0.9552	61.22	3.11	0.694	0.204	0.201	0.033	948.20	46.55
109	570011003001	NPART	0.9850	60.71	1.87	0.310	0.059	0.105	0.010	464.01	13.86
110	570120002600	NPART	0.9925	61.93	1.36	0.304	0.060	0.147	0.010	638.85	13.98
111	570305000900	NPART	0.9627	64.32	3.32	0.198	0.128	0.105	0.015	503.48	25.37
112	572017002703	NPART	0.9299	57.19	2.20	0.160	0.447	0.470	0.067	1331.72	111.76
113	573012001102	NPART	0.9946	57.49	0.95	0.492	0.052	0.189	0.011	701.60	13.03
114	574018001602	NPART	0.9642	61.47	2.68	0.416	0.148	0.169	0.024	743.17	34.48

.

<b>P</b> F	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	574018025100	NPART	0.9778	73.80	3.44	0.566	0.257	0.139	0.012	1101.82	31.79
116	580002013800	NPART	0.9915	67.55	1.99	0.387	0.103	0.133	0.009	781.45	16.66
117	585018022900	NPART	0.9858	66.05	2.08	0.714	0.095	0.112	0.009	759.95	17.46
118	590011001500	NPART	0.9955	65.16	1.08	0.341	0.059	0.138	0.006	716.10	11.74
119	590011001605	NPART	0.9943	67.99	1.34	-0.025	0.060	0.105	0.005	505.99	10.90
120	590011007000	NPART	0.9931	70.00	1.67	0.378	0.165	0.210	0.011	1277.09	25.28
121	590017004000	NPART	0.9897	65.25	1.71	0.170	0.098	0.150	0.011	705.17	19.24

Ĵ	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	PRENAC	SE_NAC
1	530002014600	NPART	0.9883	61.67	2.75	0.639	0.094	0.129	0.013	701.36	20.41
2	530004023301	NPART	0.9921	59.72	1.88	0.292	0.054	0.103	0.008	440.62	12.50
3	530011004702	NPART	0.9882	63.76	2.49	0.416	0.084	0.104	0.009	572.77	17.15
4	531015009905	NPART	0.9726	64.04	3.71	-0.050	0.259	0.201	0.025	807.28	51.72
5	531015013403	NPART	0.9899	66.48	2.08	0.043	0.105	0.141	0.009	672.12	22.28
6	531015022301	NPART	0.9911	67.07	1.79	-0.033	0.090	0.120	0.007	559.27	17.59
7	534007003801	NPART	0.9909	66.27	2.10	0.735	0.212	0.256	0.016	1443.83	39.94
8	534013001602	NPART	0.7908	49.00	7.48	0.166	0.202	0.133	0.064	269.67	53.57
9	534016005500	NPART	0.9950	69.85	1.36	0.333	0.082	0.123	0.005	791.99	15.31
10	534016008700	NPART	0.9867	55.68	2.19	0.339	0.061	0.113	0.012	409.86	14.52
11	534018009900	NPART	0.9946	70.32	1.53	0.597	0.103	0.136	0.006	975.14	17.53
12	550001038700	NPART	0.9896	65.36	2.45	0.080	0.300	0.358	0.028	1601.36	57.04
13	550003052601	NPART	0.9402	67.38	5.95	0.192	0.260	0.114	0.020	621.03	47.28
1'4	550006006501	NPART	0.9649	65.11	4.08	0.649	0.370	0.241	0.031	1280.91	73.47
15	550009076206	NPART	0.9896	65.00	2.16	0.812	0.155	0.189	0.013	1110.96	30.59
16	550014018900	NPART	0.9810	65.15	2.77	0.842	0.200	0.188	0.017	1124.53	40.60
17	550014025300	NPART	0.8216	64.40	9.83	0.610	0.398	0.115	0.038	706.08	81.99
18	550014026103	NPART	0.8543	50.94	6.38	0.357	0.326	0.242	0.088	572.59	80.55
19	550016016100	NPART	0.9944	64.14	1.62	1.105	0.093	0.163	0.009	1078.71	18.55
20	550020050000	NPART	0.9208	46.78	5.25	0.943	0.566	0.769	0.266	1353.24	152.47
	550021023700	NPART	0.9894	64.64	2.55	0.512	0.075	0.093	0.008	581.87	14.64
22	550104003400	NPART	0.9948	65.86	1.73	0.652	0.083	0.141	0.008	871.81	16.30
23	553007005001	NPART	0.9892	62.97	2.31	0.511	0.115	0.160	0.013	806.56	24.96
24	555004016803	NPART	0.9781	63.65	3.28	1.058	0.086	0.080	0.009	707.60	18.19
25	555005016800	NPART	0.9889	68.72	2.22	0.374	0.104	0.106	0.007	681.83	19.33
26	555008022602	NPART	0.9965	61.72	1.37	0.301	0.062	0.158	0.008	683.65	13.50
27	555012028400	NPART	0.9745	62.00	3.59	0.380	0.151	0.136	0.018	641.02	31.39
28	555013028700	NPART	0.9929	60.18	1.80	-0.012	0.083	0.160	0.011	528.06	18.44
29	555015009007	NPART	0.9281	56.00	4.85	0.432	0.088	0.067	0.015	331.43	20.15
30	555019023192	NPART	0.8794	54.73	7.58	-0.035	0.257	0.153	0.056	352.43	61.03
31	555021029100	NPART	0.9807	61.11	3.32	0.434	0.195	0.218	0.028	925.91	43.07
32	555104016702	NPART	0.9833	64.75	2.90	0.117	0.235	0.237	0.023	1052.50	48.54

31

s	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
1	500003118400	NPART	0.9978	70.87	1.09	0.168	0.065	0.137	0.004	842.16	9.48
2	500003120800	NPART	0.9820	62.56	2.16	0.712	0.102	0.150	0.016	830.30	23.72
3	500003143201	NPART	0.9862	63.90	1.99	0.428	0.125	0.180	0.016	890.17	26.93
4	500003220641	NPART	0.9764	63.89	2.62	0.240	0.082	0.090	0.010	455.73	17.73
5	500005115602	NPART	0.9829	67.32	2.55	0.432	0.116	0.111	0.010	695.95	20.18
6	500007377900	NPART	0.7200	61.00	9.13	1.109	0.588	0.206	0.099	1124.92	138.80
7	500008129800	NPART	0.9515	72.00	5.41	1.248	0.420	0.158	0.023	1401.61	54.41
8	500009318100	NPART	0.9857	64.38	1.83	0.933	0.060	0.086	0.007	698.40	12.94
9	500010523502	NPART	0.9840	65.75	2.04	0.474	0.180	0.220	0.018	1156.97	37.02
10	500011181901	NPART	0.9962	64.00	0.92	1.130	0.077	0.211	0.009	1279.47	16.48
11	500016212601	NPART	0.9854	61.70	1.87	0.474	0.087	0.154	0.015	732.99	21.17
12	500017250901	NPART	0.9901	66.32	1.67	0.245	0.071	0.104	0.007	567.72	14.05
13	500018446600	NPART	0.9916	55.21	1.26	1.260	0.093	0.320	0.027	1248.41	26.09
14	500018477800	NPART	0.9889	68.30	2.02	0.842	0.152	0.174	0.012	1188.13	26.32
15	500018506200	NPART	0.9946	59.68	1.19	0.574	0.084	0.266	0.017	1071.95	20.87
16	500018562000	NPART	0.8261	61.07	6.84	0.545	0.439	0.211	0.077	937.45	105.82
17	500020296100	NPART	0.9800	61.53	2.26	0.474	0.042	0.064	0.008	403.59	10.18
18	501006019200	NPART	0.9992	69.13	0.53	0.472	0.043	0.184	0.003	1140.18	7.21
19	501006050900	NPART	0.9913	63.27	1.52	0.521	0.122	0.224	0.015	1075.58	25.99
20	501006052101	NPART	0.9880	65.12	1.76	0.205	0.059	0.083	0.006	436.20	11.79
1	506002056800	NPART	0.9987	63.06	0.61	0.157	0.028	0.129	0.004	560.33	5.78
22	506002077900	NPART	0.9957	62.40	1.13	0.131	0.039	0.113	0.006	472.56	8.67
23	506002084300	NPART	0.9968	64.91	1.04	0.347	0.039	0.107	0.005	587.03	7.95
24	506002125400	NPART	0.9958	62.16	1.09	0.975	0.061	0.182	0.010	1031.88	13.64
25	506003071305	NPART	0.7655	77.00	16.37	0.376	0.073	0.007	0.002	186.40	5.98
26	506003077200	NPART	0.9946	65.90	1.23	0.367	0.039	0.082	0.004	503.72	7.93
27	515005032902	NPART	0.9929	63.00	1.35	0.661	0.154	0.324	0.020	1502.77	33.32
28	515007009000	NPART	0.8891	58.00	4.97	0.976	0.160	0.115	0.033	694.41	39.67
29	515007014307	NPART	0.9830	57.40	1.86	0.160	0.055	0.115	0.013	383.52	14.15
30	515008021001	NPART	0.9591	65.84	3.27	0.601	0.234	0.181	0.025	1032.63	48.65
31	515008022303	NPART	0.9953	60.58	1.10	0.811	0.056	0.182	0.011	917.63	13.47
32	515014036400	NPART	0.9737	56.14	1.91	1.226	0.144	0.265	0.032	1140.79	38.25
33	515020016700	NPART	0.9677	52.86	2.66	0.505	0.097	0.214	0.045	634.53	34.13
34	515102005200	NPART	0.8248	85.00	280.12	-0.233	9.746	0.035	0.006	269.33	40.83
35	518019019002	NPART	0.9614	68.00	3.61	0.602	0.155	0.095	0.012	695.29	26.94
36	530002014600	NPART	0.9770	66.77	2.84	0.536	0.133	0.122	0.013	767.78	25.55
37	530004023301	NPART	0.9544	53.61	2.87	0.273	0.053	0.092	0.020	302.56	16.87
38	530011004702	NPART	0.9809	66.37	2.19	0.463	0.101	0.109	0.010	671.69	20.23

,

S	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
39	531015009905	NPART	0.9883	67.00	1.72	-0.063	0.143	0.212	0.015	986.36	29.87
40	531015013403	NPART	0.9937	58.15	1.04	0.294	0.064	0.211	0.013	733.15	16.36
41	531015022301	NPART	0.9943	63.52	1.13	0.219	0.053	0.126	0.007	584.50	11.88
42	532003016401	NPART	0.9817	67.17	2.64	0.136	0.165	0.156	0.015	798.97	29.06
43	532003019202	NPART	0.9897	58.42	1.27	-0.017	0.073	0.198	0.015	590.84	19.59
44	534007003801	NPART	0.9933	65.46	1.36	0.817	0.152	0.286	0.016	1559.66	30.67
45	534013001602	NPART	0.9581	59.70	3.12	-0.006	0.100	0.113	0.019	365.19	24.58
46	534016005500	NPART	0.9815	74.00	3.03	0.237	0.221	0.120	0.009	872.46	26.53
47	534016008700	NPART	0.9116	60.35	4.65	0.305	0.118	0.088	0.023	409.20	29.23
48	534018009900	NPART	0.9849	68.38	2.30	0.638	0.164	0.163	0.013	1060.03	28.72
49	537001000201	NPART	0.9834	71.66	3.31	0.817	0.221	0.151	0.013	1188.13	28.84
50	537001017100	NPART	0.9781	65.85	2.47	0.811	0.236	0.241	0.024	1379.14	47.42
51	537002006702	NPART	0.9564	60.79	3.36	0.295	0.110	0.111	0.019	491.83	25.95
52	537002030200	NPART	0.9771	69.00	3.11	0.195	0.271	0.195	0.019	1088.63	42.81
53	537003016500	NPART	0.9817	62.80	2.34	0.210	0.095	0.131	0.014	580.58	21.35
54	537003030701	NPART	0.9983	60.47	0.65	0.584	0.021	0.111	0.004	588.18	4.96
55	537004000302	NPART	0.9870	66.84	2.11	0.128	0.131	0.158	0.013	793.71	24.81
56	537007022100	NPART	0.9947	71.71	1.73	0.182	0.093	0.117	0.006	760.25	13.04
57	537017035801	NPART	0.9729	61.00	2.40	0.338	0.079	0.104	0.013	487.17	18.69
58	537017047501	NPART	0.9921	65.22	1.44	0.022	0.047	0.081	0.005	362.80	9.37
P	537019009200	NPART	0.9903	65.75	1.75	-0.048	0.066	0.100	0.007	429.09	13.02
-60	540115007300	NPART	0.9561	59.56	3.41	0.595	0.124	0.134	0.025	647.95	30.40
61	541006031600	NPART	0.9951	59.21	0.98	0.500	0.062	0.214	0.012	857.01	15.28
62	541007002301	NPART	0.9887	62.58	1.76	0.721	0.091	0.168	0.014	902.86	20.91
63	541009010700	NPART	0.9940	64.84	1.25	0.492	0.105	0.227	0.012	1150.02	22.58
64	541009019200	NPART	0.9938	65.00	1.28	0.605	0.057	0.119	0.007	733.28	12.12
65	541012039900	NPART	0.9938	69.37	1.47	0.546	0.144	0.218	0.011	1361.17	25.16
66	542014008952	NPART	0.9845	61.44	1.88	0.732	0.046	0.080	0.008	554.90	11.16
67	550001038700	NPART	0.9914	67.24	1.86	-0.023	0.248	0.338	0.022	1618.16	43.41
68	550003052601	NPART	0.9895	57.88	1.54	0.304	0.095	0.243	0.022	818.08	24.32
<b>69</b>	550006006501	NPART	0.9904	65.00	1.55	0.781	0.125	0.196	0.013	1132.56	25.00
70	550009076206	NPART	0.9949	62.68	1.14	0.910	0.071	0.196	0.011	1081.15	16.25
71	550014018900	NPART	0.9945	63.00	1.01	0.858	0.106	0.265	0.013	1344.58	24.00
72	550014025300	NPART	0.9035	53.63	4.19	0.765	0.300	0.338	0.101	1027.45	89.35
73	550014026103	NPART	0.7681	44.87	5.91	0.660	0.293	0.502	0.355	798.14	164.21
74	550016016100	NPART	0.9405	69.00	4.48	1.078	0.243	0.117	0.018	1007.22	42.50
75	550020050000	NPART	0.9831	49.49	1.80	0.720	0.093	0.411	0.074	935.35	43.56
76	550021023700	NPART	0.9898	63.82	1.74	0.525	0.060	0.102	0.008	605.58	13.10

s	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
77	550104003400	NPART	0.8262	49.00	5.53	1.005	0.316	0.391	0.220	981.31	137.05
78	553007005001	NPART	0.9887	62.89	1.88	0.440	0.097	0.173	0.016	829.93	22.09
79	555004016803	NPART	0.8730	53.46	5.22	1.366	0.189	0.178	0.068	889.20	57.15
80	555005016800	NPART	0.9840	70.91	2.72	0.516	0.117	0.097	0.008	743.02	18.37
81	555008022602	NPART	0.9908	58.00	1.40	0.316	0.076	0.204	0.017	712.74	19.00
82	555012028400	NPART	0.9706	62.09	2.57	0.398	0.121	0.143	0.019	675.42	28.72
83	555013028700	NPART	0.9904	57.61	1.47	0.068	0.075	0.210	0.019	626.92	19.47
84	555015009007	NPART	0.9642	47.00	2.08	0.418	0.045	0.142	0.031	342.32	18.14
85	555019023192	NPART	0.9841	60.79	2.09	0.028	0.095	0.168	0.019	589.05	23.07
86	555021029100	NPART	0.9708	66.00	3.03	-0.054	0.220	0.180	0.021	798.47	41.65
87	555104016702	NPART	0.9108	37.60	1.60	0.294	0.403	4.142	1.349	2281.48	385.02
88	560013031400	NPART	0.9928	60.83	1.27	0.698	0.047	0.125	0.008	687.70	11.54
89	560015018800	NPART	0.9724	49.40	2.43	0.639	0.130	0.422	0.096	917.87	55.75
90	560016037700	NPART	0.9730	53.50	2.19	0.467	0.047	0.110	0.018	410.75	15.12
91	560016100900	NPART	0.9224	61.27	4.21	0.598	0.144	0.111	0.025	612.02	35.62
92	560017023404	NPART	0.9831	55.72	1.89	0.283	0.102	0.242	0.030	718.86	28.86
93	562006032700	NPART	0.9954	68.00	1.28	0.209	0.073	0.131	0.006	730.80	12.43
94	562007009500	NPART	0.9826	73.94	3.47	0.121	0.206	0.112	0.009	770.60	24.00
95	562008002303	NPART	0.9948	56.39	1.02	0.526	0.061	0.244	0.016	839.97	16.01
96	562008020500	NPART	0.9906	58.81	1.52	0.602	0.044	0.109	0.009	556.26	10.85
7	562017012100	NPART	0.9952	66.59	1.15	1.110	0.064	0.137	0.006	1043.39	13.01
-98	562021013900	NPART	0.9915	64.28	1.52	-0.028	0.092	0.167	0.011	682.56	19.62
99	564010006302	NPART	0.9976	64.21	0.74	0.178	0.028	0.097	0.003	466.47	5.94
100	564011023900	NPART	0.9854	68.94	2.44	0.136	0.153	0.141	0.011	785.39	25.20
101	564011024000	NPART	0.9030	74.00	8.57	-0.026	0.113	0.022	0.004	134.21	11.75
102	565001014600	NPART	0.9871	64.79	1.91	0.729	0.088	0.128	0.011	812.99	18.86
103	570004021700	NPART	0.9854	55.92	1.77	0.864	0.133	0.333	0.038	1173.43	37.27
104	570007017101	NPART	0.8175	67.92	9.18	0.639	0.714	0.191	0.062	1185.20	126.48
105	570008008300	NPART	0.9544	74.00	5.08	0.315	0.201	0.068	0.008	556.68	23.58
106	570010011500	NPART	0.9479	64.54	3.67	0.029	0.189	0.135	0.022	579.84	40.50
107	570010042400	NPART	0.9074	54.18	3.81	1.059	0.107	0.114	0.031	649.53	29.60
108	570011001800	NPARI	0.9595	61.74	3.25	0.725	0.190	0.193	0.032	965.54	44.46
109	570011003001	NPARI	0.9833	62.35	2.09	0.311	0.063	0.095	0.010	4/0.16	14.39
110	570120002600	NPART	0.9928	62.86	1.49	0.304	0.061	0.139	0.010	547.41	14.11
111	570305000900	NPARI	0.9596	66.30	3.68	0.169	0.143	0.097	0.014	507.95	20.80
112	572017002703	NPARI	0.9284	5/.66	2.44	0.191	0.439	0.444	0.007	1340./3	0.00
113	573012001102	NPARI	0.99/1	59.10	0.80	0.484	0.039	0.170	0.008	707.95	9.28
114	5/4018001602	NPAHI	0.9713	63.05	2.61	0.412	0.14/	0.154	0.019	750.30	32.13

S	HOUSEID	LABEL	R2	TAU_H	SE_TAUH	BASE	SE_BASE	HSLOP	SE_HSLOP	POSTNAC	SE_NAC
115	574018025100	NPART	0.9791	75.00	3.37	0.531	0.284	0.134	0.010	1105.54	31.32
116	580002013800	NPART	0.9932	68.77	1.81	0.360	0.093	0.127	0.008	788.27	15.20
117	585018022900	NPART	0.9857	67.65	2.30	0.689	0.095	0.104	0.009	763.58	17.50
118	590011001500	NPART	0.9954	66.33	1.11	0.333	0.061	0.131	0.006	724.30	12.09
119	590011001605	NPART	0.9925	69.32	1.60	-0.035	0.072	0.099	0.005	512.21	12.59
120	590011007000	NPART	0.9910	71.74	2.09	0.324	0.196	0.198	0.012	1288.66	29.19
121	590017004000	NPART	0.9941	66.30	1.29	0.155	0.075	0.142	0.007	711.40	14.87

# Appendix E - SAS Statistical Output

#### Univariate Procedure

riable=DIFFNAC

## Moments

# Quantiles(Def=5)

1

N	. 126	Sum Wgts	126	100% Ma	x 896.1	99%	636.35
Mean	163.0139	Sum	20539.75	75% Q3	282.19	95%	468.79
Std Dev	201.0589	Variance	40424.68	50% Me	d 163.625	90%	418.85
Skewness	0.09665	Kurtosis	1.277538	25% Q1	52.72	10%	-76.49
USS	8401350	CSS	5053085	0% Mi	n -461.71	5%	-190.47
CV	123.3385	Std Mean	17.91175			18	-288.23
T:Mean=O	9.100947	Pr> T	0.0001	Range	1357.81		
Num ^= 0	126	Num > 0	105	Q3-Q1	229.47		
M(Sign)	42	Pr>= M	0.0001	Mode	-461.71		
Sgn Rank	3055.5	Pr>= S	0.0001				

Lowest	Obs	Highest	Obs	
-461.71(	55)	510.93(	108)	
-288.23(	28)	532.24(	52)	
-282.42(	13)	556.42(	34)	
-249.12(	113)	636.35(	95)	
-200.61(	12)	896.1(	57)	

## Univariate Statistics for Changes in Mean Normalized Annual Consumption Participant Truncated Sample

## Univariate Procedure

riable=PRENAC

#### Moments

## Quantiles(Def=5)

N	114	Sum Wgts	114	100%	Max	1847.32	99%	1606.17
Mean	877.7175	Sum	100059.8	75%	Q3	1110.56	95%	1453.02
Std Dev	313.3579	Variance	98193.15	50%	Med	833.025	90%	1290.73
Skewness	0.497992	Kurtosis	-0.08676	25%	Q1	672.36	10%	485.47
USS	98920068	CSS	11095826	0%	Min	269.82	5%	421.04
CV	35.70145	Std Mean	29.34865				1%	344.58
T:Mean=0	29.90657	Pr> T	0.0001	Range	)	1577.5		
Num ^= 0	114	Num > O	114	Q3-Q1		438.2		
M(Sign)	57	Pr>= M	0.0001	Mode		808.23		
Sgn Rank	3277.5	Pr>= S	0.0001					

Lowest	Obs	Highest	Obs
269.82(	3)	1463.93(	113)
344.58(	29)	1557.72(	70)
388.56(	7)	1577.24(	82)
415.48(	22)	1606.17(	49)
418.53(	40)	1847.32(	83)

#### Univariate Procedure

riable=POSTNAC

## Moments

# Quantiles(Def=5)

N	114	Sum Wgts	114	100%	Max	1634.6	99%	1435.5
Mean	708.6424	Sum	80785.23	75%	Q3	846.73	95%	1229.14
Std Dev	273.569	Variance	74840.01	50%	Med	662.265 🕔	90%	1079.05
Skewness	0.823937	Kurtosis	0.634974	25%	Q1	513.45	10%	403.5
USS	65704758	CSS	8456921	0%	Min	222.28	5%	344.81
CV	38.60467	Std Mean	25.62209				1%	255.25
T:Mean=0	27.65748	Pr> T	0.0001	Range	)	1412.32		
Num ^= 0	114	Num > 0	114	Q3 - Q1	!	333.28		
M(Sign)	57	Pr>= M	0.0001	Mode		627.08		
Sgn Rank	3277.5	Pr>= S	0.0001					

Lowest	Obs	Highest	Obs
222.28(	3)	1238.62(	36)
255.25(	29)	1257.81(	70)
274.36(	88)	1431.65(	20)
302.47(	78)	1435.5(	82)
318.89(	7)	1634.6(	83)

#### Univariate Procedure

riable=DIFFNAC

#### Moments

## Quantiles(Def=5)

N	114	Sum Wgts	114	100%	Max	485.76	99%	468.79
Mean	169.0752	Sum	19274.57	75%	Q3	272.92	95%	436.62
Std Dev	149.5799	Variance	22374.16	50%	Med	167.9	90%	392.49
Skewness	0.078662	Kurtosis	-0.50428	25%	Q1	69.67	10%	-31.81
USS	5787131	CSS	2528280	0%	Min	-167.11	5%	-76.49
CV	88.46949	Std Mean	14.00944				1%	-165.98
T:Mean=0	12.06866	Pr> T	0.0001	Range	)	652.87		
Num ^= 0	114	Num > O	100	Q3-Q1		203.25		
M(Sign)	43	Pr>= M	0.0001	Mode		-167.11		
Sgn Rank	2932.5	Pr>= S	0.0001					

Lowest	Obs	Highest	Obs
-167.11(	26)	440.09(	87)
-165.98(	39)	442.8(	16)
-113.48(	59)	448.36(	60)
-97.53(	37)	468.79(	74)
-77.63(	13)	485.76(	107)

## Univariate Statistics for Changes in Mean Normalized Annual Consumption Non-Participant Full Sample

### Univariate Procedure

iable=DIFFNAC

#### Moments

## Quantiles(Def=5)

1

N	118	Sum Wgts	118	100%	Max	417.89	99%	394.54
Mean	-0.30195	Sum	-35.63	75%	Q3	58.56	95%	206.89
Std Dev	161.9079	Variance	26214.18	50%	Med	-2.8	90%	142.44
Skewness	-3.49955	Kurtosis	28.16627	25%	Q1	-45.66	10%	-115.83
USS	3067070	CSS	3067059	0%	Min	-1228.98	5%	-197.05
CV	-53620.9	Std Mean	14.90483				1%	-321.37
T:Mean=O	-0.02026	Pr> T	0.9839	Range	)	1646.87		
Num ^= 0	118	Num > 0	58	Q3-Q1		104.22		
M(Sign)	- 1	Pr>= M	0.9267	Mode		-1228.98		
Sgn Rank	181.5	Pr>= S	0.6280					

Lowest	Obs	Highest	Obs
1228.98(	118)	233.7(	55)
-321.37(	103)	290.08(	53)
-236.62(	116)	293.86(	33)
-225.55(	104)	394.54(	83)
-220.05(	102)	417.89(	106)

#### Univariate Procedure

riable=PRENAC

#### Moments

## Quantiles(Def=5)

N	112	Sum Wgts	112	100%	Max	1601.36	99%	1580.71
Mean	787.9088	Sum	88245.79	75%	Q3	986.345	95%	1368.29
Std Dev	304.7559	Variance	92876.14	50%	Med	718.49	90%	1221.47
Skewness	0.612018	Kurtosis	-0.07108	25%	Q1	572.68	10%	440.62
USS	79838889	CSS	10309251	0%	Min	155.36	5%	344.88
CV	38.67908	Std Mean	28.79672				18	262.88
T:Mean=O	27.36106	Pr> T	0.0001	Range	9	1446		
Num ^= 0	112	Num > 0	112	Q3-Q1	l	413.665		
M(Sign)	56	Pr>= M	0.0001	Mode		767.03		
Sgn Rank	3164	Pr>= S	0.0001					

Lowest	Obs	Highest	Obs
155.36(	70)	1369.96(	55)
262.88(	36)	1430.75(	5)
269.67(	91)	1443.83(	90)
331.43(	110)	1580.71(	30)
338.49(	49)	1601.36(	95)



## Univariate Statistics for Changes in Mean Normalized Annual Consumption Non-Participant Truncated Sample

#### Univariate Procedure

iable=POSTNAC

#### Moments

## Quantiles(Def=5)

N	112	Sum Wgts	112	100%	Max	1618.16	99%	1559.66
Mean	786.8521	Sum	88127.44	75%	QЗ	996.79	95%	1351.98
Std Dev	294.9265	Variance	86981.65	50%	Med	736.915	90%	1179.48
Skewness	0.549836	Kurtosis	0.014334	25%	Q1	578.91	10%	452.68
USS	78998228	CSS	9654963	0%	Min	133.52	5%	365.19
CV	37.48182	Std Mean	27.86794				1%	268.11
T:Mean=O	28.23503	Pr> T	0.0001	Range	)	1484.64		
Num ^= 0	112	Num > 0	112	Q3-Q1		417.88		
M(Sign)	56	Pr>= M	0.0001	Mode		133.52		
Sgn Rank	3164	Pr>= S	0.0001					

Lowest	Obs	Highest	Obs
133.52(	70)	1371.69(	41)
268.11(	36)	1392.92(	13)
302.56(	85)	1479.86(	30)
342.32(	110)	1559.66(	90)
357.35(	49)	1618.16(	95)

## Univariate Statistics for Changes in Mean Normalized Annual Consumption Non-Participant Truncated Sample

#### Univariate Procedure

hiable=DIFFNAC

## Moments

## Quantiles(Def=5)

N	112	Sum Wgts	112	100%	Max	233.7	99%	206.89
Mean	1.056696	Sum	118.35	75%	Q3	56.225	95%	148.35
Std Dev	92.63318	Variance	8580.906	50%	Med	-4.285	90%	127.44
Skewness	-0.10268	Kurtosis	0.454219	25%	Q1	-45.135	10%	-107.71
USS	952605.6	CSS	952480.6	0%	Min	-236.62	5%	-179.08
cv	8766.3	Std Mean	8.753013				1%	-225.55
T:Mean=O	0.120724	Pr> T	0.9041	Range	;	470.32		
Num ^= 0	112	Num > O	54	Q3-Q1		101.36		
M(Sign)	-2	Pr>= M	0.7770	Mode		-236.62		
Son Rank	68	Pr>= S	0.8445					

Lowest	Obs	Highest	Obs
-236.62(	111)	193.43(	5)
- 225 . 55 (	100)	195.55(	9)
-220.05(	99)	200.9(	78)
-197.05(	96)	206.89(	12)
-181.6(	105)	233.7(	53)

Appendix F - BENCOST Cost-Effectiveness Output

Conservation Improvement Program (CIP) BENCOST FOR GAS CIPS-- Cost-Effectiveness Analysis

Company:	Western Kentucky Gas
Project	WKG Cares

#### Input Data

1) Retail Rate (\$/MCF) =	\$4.87
Escalation Rate =	3.00%
2) Commodity Cost (\$/MCF) =	\$3.81
Escalation Rate =	3.00%
3) Demand Cost (\$/Unit/Yr) =	\$127.59
Escalation Rate =	3.00%
4) Peak Reduction Factor =	1.00%
5) Variable O&M (\$/MCF) =	\$0.03
Escalation Rate =	3.00%
6) Environmental Damage Factor =	\$0.2700
Escalation Rate =	3.00%
7) Total Sales ≕	49,693,865
Growth Rate ≖	4.40%
8) Total Customers ≖	173,958
Growth Rate =	2.70%
9) Utility Discount Rate =	10.00%
10) Social Discount Rate =	3.00%
11) General Input Data Year =	1998
12) Project Analysis Year =	1998
13) Effective Fed & State Income Tax Rate ≈	38.00%
14) Net Operating Income Before Taxes as % Total Operating Income	5.80%

15) Utility Project Costs	
Administrative Costs =	\$0
Direct Operating Costs =	\$246,383
Incentive Costs =	\$0
Total Utility Project Costs =	\$246,383

16) Direct Participant Costs (\$/Part.) =	\$0
17) Other Participant Costs (Annual \$/Part.) = Escalation Rate =	\$0.00 3.00%
18) Project Life (Years) ≖	20
19) Avg. Energy Reduction (Project) =	
20) Avg. Consumption (MCF/Part.) =	
21) Avg. MCF/Part. Saved =	16.8 √0.0
22) Number of Participants =	161 0
23) Incentive/Participant =	\$0 \$0





## **Conservation Improvement Program (CIP)** BENCOST FOR GAS CIPS-- Cost-Effectiveness Analysis

## Summary Information

Company:	Western Kentucky Gas
Project:	WKG Cares

### **Cost Summary**

Utility Cost per Participant (First Year) =	\$1,530.33
Total Energy Reduction (MCF) Societal Cost per MCF	43,277 \$5.69
Cost per Participant per MCF (First Year) =	\$91.09

# Test Results

	NPV	B/C
Cost Comparison Test	(\$80,662)	0.66
Revenue Requirements Test	(\$87,314)	0.65
Societal Benefit Test	\$41,711	1.17
Participant Test	\$151,232	681.35

Table 1 Cost Comparison Test

#### Company: Western Kentucky Gas WKG Cares Project:

This test compares the cost of energy saved to the total cost of saving that same amount of energy.

	Cost of Ener	gy Saved				·····	Project Cost			Cost of
Year	Total Energy Reduction (A)	Commodity Cost (B)	Variable O & M Cost Savings (C)	Peak Demand Reduction (D)	Demand Cost (E)	Annual Cost of Energy Saved (F)	Utility Project Costs (G)	Lost Margin (H)	Annual Project Costs (I)	Energy Saved Less Project Cost (J)
1998	2,705	\$3.81	\$81	27.05	\$127.59	\$13,837	\$246,383	(650)	\$245,732	(\$231,895)
1999	2,705	3.92	84	27.05	131.42	14,253	0	(670)	(670)	14,923
2000	2,705	4.04	86	27.05	135.36	14,680	0	(690)	(690)	15,370
2001	2,705	4.16	89	27.05	139.42	15,121	0	(711)	(711)	15,831
2002	2,705	4.29	91	27.05	143.60	15,574	0	(732)	(732)	16,306
2003	2,705	4.42	94	27.05	147.91	16,041	0	(754)	(754)	16,795
2004	2,705	4.55	97	27.05	152.35	16,523	0	(777)	(777)	17,299
2005	2,705	4.69	100	27.05	156.92	17,018	0	(800)	(800)	17,818
2006	2,705	4.83	103	27.05	161.63	17,529	0	(824)	(824)	18,353
2007	2,705	4.97	106	27.05	166.48	18,055	0	(849)	(849)	18,903
2008	2,705	5.12	109	27.05	171.47	18,596	0	(874)	(874)	19,471
2009	2,705	5.27	112	27.05	176.61	19,154	0	(900)	(900)	20,055
2010	2,705	5.43	116	27.05	181.91	19,729	0	(927)	(927)	20,656
2011	2,705	5.60	119	27.05	187.37	20,321	0	(955)	(955)	21,276
2012	2,705	5.76	123	27.05	192.99	20,930	0	(984)	(984)	21,914
2013	2,705	5.94	126	27.05	198.78	21,558	0	(1,013)	(1,013)	22,572
2014	2,705	6.11	130	27.05	204.74	22,205	0	(1,044)	(1,044)	23,249
2015	2,705	6.30	134	27.05	210.89	22,871	0	(1,075)	(1,075)	23,946
2016	2,705	6,49	138	27.05	217.21	23,557	0	(1,107)	(1,107)	24,665
2017	2,705	6.68	142	27.05	223.73	24,264	0	(1,141)	(1,141)	25,405
Γotal ≠ NPV =	43,277					\$371,818 159,069	\$246,383 246,383	(\$17,478) (6,652)	\$233,272 239,731	\$45,649 (80,662)

Total NPV ■	(\$80,662)
8enefit/Cost Ratio =	0.66

(A) = Energy Reduction/Part. (21) x Participants (22) (B) = Commodity Cost (2) (C) = (A) x Variable O&M (5) (D) = (A) x Peak Reduction Factor (4) (E) = Demand Cost (3)

(F) = (A)x(B) + (C) + (D)x(E)
(G) = Total Utility Project Costs (15)
(H) = [1 - Effective Tax Rate (13) x % Net Income Before Taxes (14)] x [(A) x Retail Rate (1) - (F)]
(I) = (G) + (H)
(J) = (F) - (I)

#### Table 2 Revenue Requirements Test

Company: Western Kentucky Gas Project: WKG Cares

	Decreases				Increases		
Year	Total Energy Savings (A)	Variable O & M Cost Savings (B)	Peak Demand Savings (C)	Annual Total Decrease (D)	Utility Program Costs (E)	Annual Total Increase (F).	Net Change (G)
1000		• •	 	A ( 0, 0.077			
1998	\$10,305	\$81	\$3,451	\$13,837	\$246,383	\$246,383	(\$232,545)
1999	10,614	84	3,555	14,253	-	-	14,253
2000	10,933	86	3,661	14,680	0	0	14,680
2001	11,261	89	3,771	15,121	0	0	15,121
2002	11,599	91	3,884	15,574	0	0	15,574
2003	11,947	94	4,001	16,041	0	0	16,041
2004	12,305	97	4,121	16,523	0	0	16,523
2005	12,674	100	4,244	17,018	0	0	17,018
2006	13,054	103	4,372	17,529	0	0	17,529
2007	13,446	106	4,503	18,055	0	0	18,055
2008	13,849	109	4,638	18,596	0	0	18,596
2009	14,265	112	4,777	19,154	0	0	19,154
2010	14,693	116	4,920	19,729	0	0	19,729
2011	15,134	119	5,068	20,321	0	0	20,321
2012	15,588	123	5,220	20,930	0	0	20,930
2013	16,055	126	5,377	21,558	0	0	21,558
2014	16,537	130	5,538	22,205	0	0	22,205
2015	17,033	134	5,704	22,871	0	0	22,871
2016	17,544	138	5,875	23,557	0	0	23,557
2017	18,070	142	6,051	24,264	0	0	24,264
Total =	\$276,907	\$2,180	\$92,731	\$371,818	\$246,383	\$246,383	\$125,436
NPV =	118,465	933	39,672	159,069	246,383	246,383	(87,314)

 Total NPV =
 (\$87,314)

 Benefit/Cost Ratio =
 0.65

(A) = Energy Reduction/Part. (21) x Participants (22) x Commodity Cost (2)

(B) = Energy Reduction/Part. (21) x Participants (22) x Variable O&M (5)

(C) = Energy Reduction/Part. (21) x Participants (22) x Peak Reduction Factor (4)

x Demand Cost (3)

(D) = (A) + (B) + (C)

(E) = Total Utility Project Costs (15) (F) = (E)

(G) = (D) - (F)

Company:	Western Kentucky Gas
Project:	WKG Cares

	Decreases					Increases				
Year	Total Energy Savings (A)	Variable O & M Cost Savings (B)	Total Demand Savings (C)	Avoided Environmental Damage Costs (D)	Annual Total Decrease (E)	Utility Program Costs (F)	Total Participants' Costs (G)	Incentives Paid to Participants (H)	Annual Total Increase (I)	Net Change (J)
1998	\$10,305	\$81	\$3,451	\$730	\$14,568	\$246,383	\$0	\$0	\$246 383	(\$231.815)
1999	10,614	84	3,555	752	15,005		0	0	0	15 005
2000	10,933	86	3,661	775	15,455	0	0	Ō	Ō	15 455
2001	11,261	89	3,771	798	15,919	Ó	Ō	0	0	15,919
2002	11,599	91	3,884	822	16,396	0	0	0	Ō	16.396
2003	11,947	94	4,001	847	16,888	0	0	0	0	16,888
2004	12,305	97	4,121	872	17,395	0	0	0	0	17,395
2005	12,674	100	4,244	898	17,917	0	0	0	0	17,917
2006	13,054	103	4,372	925	18,454	0	0	0	0	18,454
2007	13,446	106	4,503	953	19,008	0	0	0	0	19,008
2008	13,849	109	4,638	981	19,578	0	0	0	0	19,578
2009	14,265	112	4,777	1,011	20,165	0	0	0	0	20,165
2010	14,693	116	4,920	1,041	20,770	0	0	0	0	20,770
2011	15,134	119	5,068	1,072	21,393	0	0	0	0	21,393
2012	15,588	123	5,220	1,105	22,035	0	0	0	0	22,035
2013	16,055	126	5,377	1,138	22,696	0	0	0	0	22,696
2014	16,537	130	5,538	1,172	23,377	0	0	0	0	23,377
2015	17,033	134	5,704	1,207	24,078	0	0	0	0	24,078
2016	17,544	138	5,875	1,243	24,801	0	0	0	0	24,801
2017	18,070	142	6,051	1,281	25,545	0	0	0	0	25,545
Total =	\$276,907	\$2,180	\$92,731	\$14,720	\$391,442	\$246,383	\$0	\$0	\$246,383	\$47,258
NPV =	206,106	1,623	69,021	11,344	288,094	246,383	0	0	246,383	41,711
Total NP	∨=	\$41,711								

Benefit/Cost Ratio = 1.17 -

(A) = Energy Reduction/Part. (21) x Participants (22) x Commodity Cost (2)

(B) = Energy Reduction/Part. (21) x Participants (22) x Variable O&M (5)
 (C) = Energy Reduction/Part. (21) x Participants (22) x Peak Reduction Factor (4) x Demand Cost (3)

(D) = Energy Reduction/Part. (21) x Participants (22) x Environmental Damage Factor (6)

(E) = (A) + (B) + (C) + (D)

(F) = Total Utility Project Costs (15) (G) = Direct (16) x Other (17) Participant Costs x Participants (22) (H) = Incentive Costs (15) (I) = (F) + (G) - (H)(J) = (E) - (I)
Project:

WKG Cares

		Benefits						Costs					
Year	Ratio of Part. to Total Customers (A)	Incentives Received (B)	Total Energy Reduction (C)	Retail Rate (D)	Peak Demand Reduction (E)	Demand Cost (F)	Total Annual Benefits (G)	Direct Part. Costs (H)	Other Part. Costs (I)	Utility Project Costs (J)	Lost Margin (K)	Total Annual Costs (L)	Annual Benefits Less Costs (M)
1998	0.0009	\$0	2,705	\$4.87	27.05	\$127.59	\$13,176	\$0	\$0	\$228	(\$1)	\$227	\$12,948
1999	0.0009	0	2,705	5.02	27.05	131.42	13,571	0	0	0	-1	(1)	13,571
2000	0.0009	0	2,705	5.17	27.05	135.36	13,978	0	0	0	-1	(1)	13,978
2001	0.0009	0	2,705	5.32	27.05	139.42	14,397	0	0	0	-1	(1)	14,398
2002	0.0008	0	2,705	5.48	27.05	143.60	14,829	0	0	0	-1	(1)	14,829
2003	0.0008	0	2,705	5.65	27.05	147.91	15,274	0	0	0	-1	(1)	15,274
2004	0.0008	0	2,705	5.82	27.05	152.35	15,732	0	0	0	-1	(1)	15,732
2005	0.0008	0	2,705	5.99	27.05	156.92	16,204	0	0	0	-1	(1)	16,204
2006	0.0007	0	2,705	6.17	27.05	161.63	16,690	0	0	0	-1	(1)	16,690
2007	0.0007	0	2,705	6.35	27.05	166.48	17,190	0	0	0	-1	(1)	17,191
2008	0.0007	0	2,705	6.54	27.05	171.47	17,706	0	0	0	-1	(1)	17,706
2009	0.0007	0	2,705	6.74	27.05	176.61	18,237	0	0	0	-1	(1)	18,238
2010	0.0007	0	2,705	6.94	27.05	181.91	18,784	0	0	0	-1	(1)	18,785
2011	0.0007	0	2,705	7.15	27.05	187.37	19,347	0	0	0	-1	(1)	19,348
2012	0.0006	0	2,705	7.37	27.05	192.99	19,928	0	0	0	-1	(1)	19,928
2013	0.0006	0	2,705	7.59	27.05	198.78	20,525	0	0	0	-1	(1)	20,526
2014	0.0006	0	2,705	7.81	27.05	204.74	21,141	0	0	0	-1	(1)	21,142
2015	0.0006	0	2,705	8.05	27.05	210.89	21,775	0	0	0	-1	(1)	21,776
2016	0.0006	0	2,705	8.29	27.05	217.21	22,428	0	0	0	-1	(1)	22,429
2017	0.0006	0	2,705	8.54	27.05	223.73	23,101	0	0	0	-1	(1)	23,102
Total = NPV =			54,096				\$354,012 \$151,453.80	\$0 0	\$0 0	\$228 228		\$216 222	\$353,797 151,232

Total NPV = \$151,232 Benefit/Cost Ratio = 681.35

(A) = Participants (22)/Total Customers (8)

(B) = Incentive Costs (15) (C) = Energy Reduction/Part. (21) x Participants (22)

(D) = Retail Rate (1)

(E) = (A) x Peak Reduction Factor (4) (F) = Demand Cost (3) (G) = (B) + (C)x(D) + (E)x(F)x(A)

(H) = Direct Participants Costs (16) x Participants (22)

(I) = Other Participants Costs (17) x Participants (22)
 (J) = Other Participant Costs (17) x Participants (22)
 (J) = (A) x Total Utility Project Costs (15)
 (K) = (A) x [1 - Effective Tax Rate (13) x % Net Income Before Taxes (14)] x ((C) x [(D) - Commodity Cost (2)] - (A) x (E) x (F)}

(L) = (H) + (I) + (J) + (K)(M) = (G) - (L)

# Concervation Improvement Program (CIP) BENCOST FOR GAS CIPS-- Cost-Effectiveness Analysis



Western Ke<mark>ntucky G</mark>as MKO Cares Company:

Project

i mut Data	
i) Rc: il Rate ({//wCP) = Et relation Rato =	04.87 3.00%
2) Commodity Cost (\$/MCF) = Escalation Rate =	\$3.81 3.00%
3) Domand Cost (CAUniWr) = Esculation Rate =	\$127.59 3.00%
4) Peak Reduction Factor =	1.00%
5) Variable O&M (\$/MCF) = Escalation Raio ==	\$0.03 3.00%
6) Environment <sup>-</sup> I Damage Factor = Escalation Rate =	€0.0000 3.00%
7) Total Sales = Growin Rato =	49,693,855 4.40%
<ul> <li>7) Total Sales ≠ Growth Rato ∞</li> <li>8) Tot 1 Customers ∞ Growth Rate ≠</li> </ul>	49,693,865 4,40% 173,958 2,70%
<ul> <li>7) Total Sales = Growth Rato =</li> <li>8) Tot i Customors = Growth Rate =</li> <li>9) Utility Discount Rate =</li> </ul>	49,693,855 4,40% 173,958 2.70% 10.00%
<ul> <li>7) Total Salos ≠ Growth Rato ≠</li> <li>8) Tot 1 Customers ≠ Growth Rate ≠</li> <li>9) Utility Discount Rate ≠</li> <li>10) Social Discount Rate ≠</li> </ul>	49,693,865 4.40% 173,958 2.70% 10.00% 3.00%
<ul> <li>7) Total Sales ≠ Growth Rato ≠</li> <li>8) Total Customors ≠ Growth Rato ≠</li> <li>9) Utility Discount Rato ≠</li> <li>10) Social Discount Rato ≠</li> <li>11) General Input Data Your =</li> </ul>	49,693,865 4.40% 173,958 2.70% 10.00% 3.00% 1\$98
<ul> <li>7) Total Salos = Growth Rato =</li> <li>8) Total Customers = Growth Rato =</li> <li>9) Utility Discount Rato =</li> <li>10) Social Discount Rato =</li> <li>11) General Input Data Your =</li> <li>12) Project Analysis Your =</li> </ul>	49,693,865 4.40% 173,958 2.70% 10.00% 3.00% 1998 1998
<ul> <li>7) Total Salos ≠ Growth Rato ≠</li> <li>8) Tot 1 Customors = Growth Rato ≠</li> <li>9) Utility Discount Rato ≠</li> <li>10) Social Discount Rato ≠</li> <li>11) General Input Data Your =</li> <li>12) Project Analysis Year ≠</li> <li>13) Effective Fed &amp; State Income Tax Rate =</li> </ul>	49,693,865 4.40% 173,958 2.70% 10.00% 3.00% 1998 1998 38.00%

15) Utility Project Costs	
Administrativo Costs =	S0
Direct Operating Costs =	3246,383
Incentive Costs ==	50
Total Utility Project Costs =	246 583

ar an gang ay an an ay a Ra a watay gana ana a sa a sana dan kanakan kasakan na sanakan an a sa ay an aya ka sa

16) Direct Panicipant Costs (\$/Part.) =	30
17) Other Participant Costs (Annual \$/Part.) = Escalation Rate =	\$0.00 3.00%
18) Project Life (Years) =	20
19) Avg. Energy Reduction ⊜roject) =	
20) Avg. Consumption (MCF/Part.) =	
21) Avg. MCF/Part. Saved =	16.8 0.0
22) Number of Participants =	161 0
23) Incentive/Participant =	୍ବିପ ବିପ



## **Conservation Improvement Program (CIP)** BENCOST FOR GAS CIPS-- Cost-Effectiveness Analysis

## **Summary Information**

Company:	Western Kentucky Gas
Project:	WKG Cares

## Cost Summary

Utility Cost per Participant (First Year) =	\$1,530.33
Total Energy Reduction (MCF) Societal Cost per MCF	43,277 \$5.69
Cost per Participant per MCF (First Year) =	\$91.09

#### **Test Results**

	NPV	B/C
Cost Comparison Test	(\$80,662)	0.66
Revenue Requirements Test	(\$87,314)	0.65
Societal Benefit Test	\$30,367	1.12
Participant Test	\$151,232	681.35

Table 1 **Cost Comparison Test** 

Company:	Western Kentucky Gas
Project:	WKG Cares

	Cost of Ener	gy Saved					Project Cost			Cost of
Year	Total Energy Reduction (A)	Commodity Cost (B)	Variable O & M Cost Savings <sup>-</sup> (C)	Peak Demand Reduction (D)	Demand Cost (E)	Annual Cost of Energy Saved (F)	Utility Project Costs (G)	Lost Margin (H)	Annual Project Costs (I)	Energy Saved Less Project Cost (J)
1998	2,705	\$3.81	\$81	27.05	\$127.59	\$13,837	\$246,383	(650)	\$245.732	(\$231 895
1999	2,705	3.92	84	27.05	131.42	14,253	0	. (670)	(670)	14,923
2000	2,705	4.04	86	27.05	135.36	14,680	0	(690).	(690)	15.370
2001	2,705	4.16	89	27.05	139.42	15,121	0	(711)	(711)	15,831
2002	2,705	4.29	91	27.05	143.60	15,574	0	(732)	(732)	16,306
2003	2,705	4.42	94	27.05	147.91	16,041	0	(754)	(754)	16,795
2004	2,705	4.55	97	27.05	152.35	16,523	0	(777)	(777)	17,299
2005	2,705	4.69	100	27.05	156.92	17,018	0	(800)	(800)	17,818
2006	2,705	4.83	103	27.05	161.63	17,529	0	(824)	(824)	18,353
2007	2,705	4.97	106	27.05	166.48	18,055	0	(849)	(849)	18,903
2008	2,705	5.12	109	27.05	171.47	18,596	0	(874)	(874)	19,471
2009	2,705	5.27	112	27.05	176.61	19,154	0	(900)	(900)	20,055
2010	2,705	5.43	116	27.05	181.91	19,729	0	(927)	(927)	20,656
2011	2,705	5.60	119	27.05	187.37	20,321	0	(955)	(955)	21,276
2012	2,705	5.76	123	27.05	192.99	20,930	0	(984)	(984)	21,914
2013	2,705	5,94	126	27.05	198.78	21,558	0	(1,013)	(1,013)	22,572
2014	2,705	6.11	130	27.05	204.74	22,205	0	(1,044)	(1,044)	23,249
2015	2,705	6.30	134	27.05	210.89	22,871	0	(1,075)	(1,075)	23,946
2016	2,705	6.49	138	27.05	217.21	23,557	0	(1,107)	(1,107)	24,665
2017	2,705	6.68	142	27.05	223.73	24,264	0	(1,141)	(1,141)	25,405
fotal = NPV =	43,277					\$371,818 159,069	\$246,383 246,383	(\$17,478) (6,652)	\$233,272 239,731	\$45,6 <b>49</b> (80,662

Total NPV = Benefit/Cost Ratio =

#### (\$80,662) 0.66

(A) = Energy Reduction/Part. (21) x Participants (22)
(B) = Commodity Cost (2)
(C) = (A) x Variable O&M (5)
(D) = (A) x Peak Reduction Factor (4)
(E) = Demand Cost (3)

(F) = (A)x(B) + (C) + (D)x(E) (G) = Total Utility Project Costs (15) (H) = [1 - Effective Tax Rate (13) x % Net Income Before Taxes (14)] x [(A) x Retail Rate (1) - (F)] T = (0) + 20(I) = (G) + (H)(J) = (F) - (I)

#### Table 2 Revenue Requirements Test

Company: Project: Western Kentucky Gas WKG Cares

	Decreases				Increases			
Year	Total Energy Savings (A)	Variable O & M Cost Savings (B)	Peak Demand Savings (C)	Annual Total Decrease (D)	Utility Program Costs (E)	Annual Total Increase (F)	Net Change (G)	
1008	\$10 305	\$81	\$3.451	\$13,837	\$246 383	\$246 383	(\$232 545)	
1990	10,505	901 94	3 555	14 253	φ <b>2</b> 40,505	φ <b>240,</b> 303	(4232,343)	
2000	10,014	86	3,555	14,200	0	٥	14,233	
2000	11 261	80	3 771	14,000	0	0	14,000	
2001	11,201	03 Q1	3 884	15,721	0	0	15,721	
2002	11 947	94	4 001	16,0/4	Ő	0	16 041	
2003	12 305	97	4,001	16 523	0	· 0	16 523	
2004	12,505	100	4 244	17 018	0	0	17.018	
2005	13 054	103	4 372	17,579	0	0	17 529	
2000	13 446	105	4,572	18 055	0	0	18 055	
2007	13 840	100	4,505	18 596	0	0	18 506	
2000	14 265	112	4,000	19 154	0	0	10,000	
2000	14 693	116	4 920	19 729	ů	Õ	19,729	
2010	15 134	119	5,068	20,321	0	Ő	20,321	
2012	15 588	123	5 220	20,930	0	õ	20,930	
2012	16 055	126	5 377	21 558	0	Ő	21,558	
2014	16,537	130	5,538	22,205	0	0	22 205	
2015	17 033	134	5 704	22,871	0	0	22 871	
2016	17.544	138	5,875	23,557	0	0	23.557	
2017	18,070	142	6,051	24,264	0	0	24,264	
Total =	\$276,907	\$2,180	\$92,731	\$371,818	\$246,383	\$246,383	\$125,436	
NPV =	118,465	933	39,672	159,069	246,383	246,383	(87,314)	

Total NPV =	(\$87,314)
Benefit/Cost Ratio =	0.65

(A) = Energy Reduction/Part. (21) x Participants (22) x Commodity Cost (2)

(B) = Energy Reduction/Part. (21) x Participants (22) x Variable O&M (5)

(C) = Energy Reduction/Part. (21) x Participants (22) x Peak Reduction Factor (4)

x Demand Cost (3)

(D) = (A) + (B) + (C)

(E) = Total Utility Project Costs (15) (F) = (E) (G) = (D) - (F) Table 3 Societal Benefit Test

	Decreases					Increases				
Year	Total Energy Savings (A)	Variable O & M Cost Savings (B)	Total Derrand Savings (C)	Avoided Environmental Damage Costs (D)	Annual Total Decrease (E)	Utility Program Costs (F)	Total Participants' Costs (G)	Incentives Paid to Participants (H)	Annual Total Increase (I)	Net Change (J)
1998	\$10,305	\$81	\$3,451	<b>\$</b> 0	\$13,837	\$246,383	\$0	\$0	\$246,383	(\$232,545)
1999	10,614	84	3,555	0	14,253	0	0	· 0	0	14,253
2000	10,933	86	3,661	0	14,680	· 0	0	0	0	14,680
2001	11,261	89	3,771	0	15,121	0	0	0	0	15,121
2002	11,599	91	3,884	0	15,574	0	0	0	0	15,574
2003	11,947	94	4,001	0	16,041	0	0	0	0	16,041
2004	12,305	97	4,121	0	16,523	0	0	0	0	16,523
2005	12,674	100	4,244	0	17,018	0	0	0	0	17,018
2006	13,054	103	4,372	0	17,529	0	0	0	0	17,529
2007	13,446	106	4,503	0	18,055	0	0	0	0	18,055
2008	13,849	109	4,638	0	18,596	0	0	0	0	18,596
2009	14,265	112	4,777	0	19,154	0	0	0	0	19,154
2010	14,693	116	4,920	0	19,729	0	0	0	0	19,729
2011	15,134	119	5,068	0	20,321	0	0	0	0	20,321
2012	15,588	123	5,220	0	20,930	0	0	0	0	20,930
2013	16,055	126	5,377	0	21,558	0	0	0	0	21,558
2014	16,537	130	5,538	0	22,205	0	0	0	0	22,205
2015	17,033	134	5,704	0	22,871	0	0	0	0	22,871
2016	17,544	138	5,875	0	23,557	0	0	0	0	23,557
2017	18,070	142	6,051	0	24,264	0	0	0	0	24,264
Total =	\$276,907	\$2,180	\$92,731	\$0	\$371,818	\$246,383	\$0	\$0	\$246,383	\$32,538
NPV =	206,106	1,623	69,021	0	276,750	246,383	0	0	246,383 -	30,367
Total NP	V=	\$30,367								

Benefit/Cost Ratio =

(A) = Energy Reduction/Part. (21) x Participants (22) x Commodity Cost (2)

1.12

Western Kentucky Gas

WKG Cares

(B) = Energy Reduction/Part. (21) x Participants (22) x Variable O&M (5)

(C) = Energy Reduction/Part. (21) x Participants (22) x Peak Reduction Factor (4) x Demand Cost (3) (D) = Energy Reduction/Part. (21) x Participants (22) x Environmental Damage Factor (6)

(E) = (A) + (B) + (C) + (D)

(F) = Total Utility Project Costs (15)
(G) = Direct (16) x Other (17) Participant Costs x Participants (22)
(H) = Incentive Costs (15)
(I) = (F) + (G) - (H)
(J) = (E) - (I) Table 4 Participant Test

Company:	Western Kentucky Gas
Project:	WKG Cares

	_	Benefits						Costs					
Year	Ratio of Part, to Total Customers (A)	Incentives Received (B)	Total Energy Reduction (C)	Retail Rate (D)	Peak Demand Reduction (E)	Demand Cost (F)	Total Annual Benefits (G)	Direct Part. Costs (H)	Other Part. Costs (I)	Utility Project Costs (J)	Lost Margin (K)	Total Annual Costs (L)	Annual Benefits Less Costs (M)
1998	0.0009	\$0	2,705	\$4.87	27.05	\$127.59	\$13,176	\$0	\$0	\$228	(\$1)	\$227	\$12,948
1999	0.0009	0	2,705	5.02	27.05	131.42	13,571	0	0	0	-1	(1)	13,571
2000	0.0009	0	2,705	5.17	27.05	135.36	13,978	0	0	0	-1	ά	13,978
2001	0.0009	0	2,705	5.32	27.05	139.42	14,397	0	0	0	-1	à	14,398
2002	0.0008	0	2,705	5.48	27.05	143.60	14,829	0	0	0	-1	(1)	14,829
2003	0.0008	. 0	2,705	5.65	27.05	147.91	15,274	0	0	0	-1	(1)	15,274
2004	0.0008	0	2,705	5.82	27.05	152.35	15,732	0	0	0	-1	(i)	15,732
2005	0.0008	0	2,705	5.99	27.05	156.92	16,204	0	0	0	-1	(1)	16,204
2006	0.0007	0	2,705	6.17	27.05	161.63	16,690	0	0	0	-1	(i)	16,690
2007	0.0007	0	2,705	6.35	27.05	166.48	17,190	0	0	0	-1	(1)	17,191
2008	0.0007	0	2,705	6.54	27.05	171.47	17,706	0	0	0	-1	(1)	17,706
2009	0.0007	0	2,705	6.74	27.05	176.61	18,237	0	0	0	-1	(1)	18,238
2010	0.0007	0	2,705	6.94	27.05	181.91	18,784	0	0	0	-1	(1)	18,785
2011	0.0007	0	2,705	7.15	27.05	187.37	19,347	0	0	0	-1	(1)	19,348
2012	0.0006	0	2,705	7.37	27.05	192.99	19,928	0	0	0	-1	(1)	19,928
2013	0.0006	0	2,705	7.59	27.05	198.78	20,525	0	0	0	-1	(1)	20,526
2014	0.0006	0	2,705	7.81	27.05	204.74	21,141	· 0	0	0	-1	(1)	21,142
2015	0.0006	0	2,705	8.05	27.05	210.89	21,775	0	0	0	-1	(1)	21,776
2016	0.0006	0	2,705	8.29	27.05	217.21	22,428	0	0	0	-1	(1)	22,429
2017	0.0006	0	2,705	8.54	27.05	223.73	23,101	0	0	0	-1	(1)	23,102
Total =			54,096				\$354,012	\$0	\$0	\$228		\$216	\$353,797
NPV =							\$151,453.80	0	0	228		222	151,232

Total NPV =			\$151,232
Benefit/Cost	Ratio	3	681.35

(A) = Participants (22)/Total Customers (8)

(B) = Incentive Costs (15)

(C) = Energy Reduction/Part. (21) x Participants (22) (D) = Retail Rate (1) (E) = (A) x Peak Reduction Factor (4)

(F) = Demand Cost (3)

(G) = (B) + (C)x(D) + (E)x(F)x(A)

(H) = Direct Participants Costs (16) x Participants (22) (I) = Other Participant Costs (17) x Participants (22)

 (J) = (A) x Total Utility Project Costs (15)
 (K) = (A) x [1 - Effective Tax Rate (13) x % Net Income Before Taxes (14)] x ((C) x [(D) - Commodity Cost (2)] - (A) x (E) x (F)} (L) = (H) + (I) + (J) + (K)(M) = (G) - (L)

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20. b. Witness: Smith and Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

b. Explain why the Applied Energy Group, Inc. ("AEG") was retained to perform these evaluations, given Mr. Marks association with AEG.

#### Response:

The WKG CARES Collaborative requested proposals to provide process and impact evaluation from several companies and selected AEG as the best qualified. AEG has performed evaluations for over 400 DSM programs for utilities throughout the United States. AEG employed evaluation techniques, which relied upon pre- and post-treatment billing data. This evaluation technique, unlike engineering analysis, is based upon actual billing data before and after the customer participates in the program. This is considered to be the technique of choice in the industry because it is virtually free from subjective input. AEG's knowledge of the program through its previous work for the Collaborative was viewed as a positive since it provided the background that was required to conduct a comprehensive a evaluation in a timely and cost efficient manner.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 c Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

Ŋ,

c. Explain in detail why actual customer savings were not developed as part of the impact evaluation.

## Response:

Actual customer savings were developed as stated on Page 12, Lines 2-6 of my testimony. Further detail is provided in the attachment entitled "Process and Impact Evaluation of Western Kentucky Gas WKG CARES Program" pages 49-57.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 d Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

Y

d. Explain how the impacts of the weather experienced during the evaluation period were taken into consideration when pre- and post-treatment consumption values were determined.

#### Response:

The PRISM software incorporates the impacts of weather in its methodology. A detailed technical discussion of how this is accomplished is contained on pages 43 - 46 of the attachment entitled "Process and Impact Evaluation of Western Kentucky Gas WKG CARES Program".

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 e Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

Ń

e. Provide the workpapers showing the calculation of the TRC, UCT, RIM, and PT values for the impact evaluation.

#### Response:

See Appendix F in the attachment entitled "Process and Impact Evaluation of Western Kentucky Gas WKG CARES Program". Note that the software we employed used slightly different names for the tests. In Appendix F, Table 1, the Cost Comparison test is the same as the UCT; Table 2 has the Revenue Requirement Test which is the same as the RIM; Table 3 has the Societal Benefits Test which is the same as the TRC; and Table 4 has the Participant test which is the same as the PT. When externalities are added to the Societal Benefits Test, it is equal to the Societal Test.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 f Witness: Marks

## Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

f. Explain why the Societal Test is referenced on page 13 of the Marks Testimony instead of the TRC.

## Response:

The name "Societal Test (without externalities)" has been substituted for TRC. There is no difference in how the benefit/cost ratios are calculated.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 g Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

g. Provide the calculations and assumptions used to determine the environmental externality adder reflected in the Societal Test. Include an explanation as to why it was believed an environmental externality adder should have been included in the analysis.

#### Response:

By definition, the "Societal Test" includes externalities. Depending upon where the analysis is conducted, from a geographic and regulatory perspective, the value for the environmental externality can range from zero to a positive value. In order to provide the full range of potential benefits attributable to WKG CARES, we included a Societal Test with and without externalities (TRC). We were unable to find an estimate for externalities that had been used recently in Kentucky, so we utilized a value from a recent project we had done in Minnesota. We felt that Minnesota was a reasonable proxy for Kentucky.

The question of whether an environmental externality adder should have been included in the analysis can be answered as follows. AEG wanted to provide the Collaborative with a comprehensive evaluation. Including benefit/cost results with and without externalities provides a more comprehensive analysis than one which omitted externalities. By including externalities, AEG was not attempting to make a policy statement regarding their appropriateness for use in Kentucky.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 20 h Witness: Marks

#### Data Request:

20. Concerning the process and impact evaluations of WKG CARES:

h. Did the calculation of the TRC during the impact evaluation include a determination of Western's lost revenues associated with WKG CARES? If no, explain why the exclusion of lost revenues would not skew the results of the TRC.

#### Response:

The TRC Test does not include lost revenues. Lost revenues are a transfer between the utility and the participant. A copy of the Test as defined by the California Commission is attached. Lost revenues are included in the RIM Test.

ANU ATTACHMENT D.R. ZOh. STANDARD PRACTICE MANUAL Economic Analysis 1 of Demand-Side Management Programs 3 0 0 میں 3 2 22 ß 5 Ð **California Energy Commission** Energy Efficiency and Local Assistance Division • Don Schwartz, Project Leader 916 654 45-,7 California Public Utilities Commission **Division of Ratepayer Advocates** • Don Schultz, Project Leader

## Chapter 4 TOTAL RESOURCE COST TEST<sup>1</sup>

**Definition:** The Total Resource Cost Test measures the net costs of a demand-side management program as a resource option based on the total costs of the program, including both the participants' and the utility's costs.

The test is applicable to conservation, load management, and fuel substitution programs. For fuel substitution programs, the test measures the net effect of the impacts from the fuel not chosen versus the impacts from the fuel that is chosen as a result of the program. TRC test results for fuel substitution programs should be viewed as a measure of the economic efficiency implications of the total energy supply system (gas and electric).

A variant on the TRC test is the Societal Test. The Societal Test differs from the TRC test in that it includes the effects of externalities (e.g. environmental, national security), excludes tax credit benefits, and uses a different (societal) discount rate.

Benefits and Costs: This test represents the combination of the effects of a program on both the customers participating and those not participating in a program. In a sense, it is the summation of the benefit and cost terms in the Participant and the Ratepayer Impact Measure tests, where the revenue (bill) change and the incentive terms intuitively cancel (except for the differences in net and gross savings).

The <u>benefits</u> calculated in the Total Resource Cost Test are the avoided supply costs--the reduction in transmission, distribution, generation, and capacity costs valued at marginal cost--for the periods when there is a load reduction. The avoided supply costs should be calculated using net program savings, savings net of changes in energy use that would have happened in the absence of the program. For fuel substitution programs, benefits include the avoided device costs and

<sup>1</sup>This test was previously called the All Ratepayers Test.

avoided supply costs for the energy-using equipment not chosen by the program participant.

The <u>costs</u> in this test are the program costs paid by both the utility and the participants plus the increase in supply costs for the periods in which load is increased. Thus <u>all</u> equipment costs, installation, operation and maintenance, cost of removal (less salvage value), and administration costs, no matter who pays for them, are included in this test. Any tax credits are considered a reduction to costs in this test. For fuel substitution programs, the costs also include the increase in supply costs for the utility providing the fuel that is chosen as a result of the program.

How the Results Can be Expressed: The results of the Total Resource Cost Test can be expressed in several forms: as a net present value, a benefit-cost ratio, or as a levelized cost. The net present value is the primary unit of measurement for this test. Secondary means of expressing TRC test results are a benefit-cost ratio and levelized costs. The Societal Test--expressed in terms of net present value, a benefit-cost ratio, or levelized costs--is also considered a secondary means of expressing results. Levelized costs as a unit of measurement are inapplicable for fuel substitution programs, since these programs represent the net change of alternative fuels which are measured in different physical units (e.g. kWh or therms). Levelized costs are also not applicable for load building programs.

Net present value  $(NPV_{TRC})$  is the discounted value of the net benefits to this test over a specified period of time.  $NPV_{TRC}$  is a 'measure of the change in the total resource costs due to the program. A net present value above zero indicates that the program is a less expensive resource than the supply option upon which the marginal costs are based.

The benefit-cost ratio (BCR<sub>TRC</sub>) is the ratio of the discounted total benefits of the program to the discounted total costs over some specified time period. It gives an indication of the rate of return of this program to the utility and its ratepayers. A benefit-cost ratio above one indicates that the program is beneficial to the utility and its ratepayers on a total resource cost basis.

26

The levelized cost is a measure of the total costs of the program in a form that is sometimes used to estimate costs of utilityowned supply additions. It presents the total costs of the program to the utility and its ratepayers on a per kilowatt, per kilowatt-hour, or per therm basis levelized over the life of the program.

The Societal Test is structurally similar to the Total Resource Cost Test. It goes beyond the TRC test in that it attempts to quantify the change in the total resource costs to society as a whole rather than to only the service territory (the stility and its ratepayers). In taking society's perspective, the Societal Test utilizes essentially the same input variables as the TRC Test, but they are defined with a broader societal point of view. More specifically, the Societal Test differs from the TRC Test in at least one of five ways. First, the Societal Test may use higher marginal costs than the TRC test if a utility faces marginal costs that are lower than other utilities in the state or than its out-of-state suppliers. Marginal costs used in the Societal Test would reflect the cost to society of the more expensive alternative resources. Second, these marginal costs might also contain externality costs of power generation not captured by the market system. Third, tax credits are treated as a transfer payment in the Societal Test, and thus are left out. Fourth, in the case of capital expenditures, interest payments are considered a transfer payment since society actually expends the resources in the first year. Therefore, capital costs enter the calculations in the year in which they occur. And finally, a societal discount rate should be used.<sup>2</sup>

Strengths of the Total Resource Cost Test: The primary strength of the Total Resource Cost (TRC) test is its scope. The test includes total costs (participant plus utility) and also has the potential for capturing total benefits (avoided supply costs plus, in the case of the societal test variation, externalities). To the extent supply-side project evaluations also include total costs of generation

<sup>&</sup>lt;sup>2</sup>Many economists have pointed out that use of a market discount rate in social cost-benefit analysis undervalues the interests of future generations. Yet if a market discount rate is not used, comparisons with alternative investments are difficult to make.

and/or transmission, the TRC test provides a useful basis for comparing demand- and supply-side options.

Since this test treats utility incentives paid to participants and revenue shifts as transfer payments (from all ratepayers to participants through increased revenue requirements), the test results are unaffected by the uncertainties of projected average rates, thus reducing the uncertainty of the test results. Average rates and assumptions associated with how other options are financed (analogous to the issue of incentives for DSM programs) are also excluded from most supply-side cost determinations, again making the TRC test useful for comparing demand-side and supply-side options.

Weakness of the Total Resource Cost Test: The treatment of revenue shifts and incentive payments as transfer payments--identified previously as a strength--can also be considered a weakness of the TRC test. While it is true that most supply-side cost analyses do not include such financial issues, it can be argued that DSM programs <u>should</u> include these effects since, in contrast to most supply options, DSM programs do result in lost revenues.

In addition, the costs of the DSM "resource" in the TRC test are based on the total costs of the program, including costs incurred by the participant. Supply-side resource options are typically based only on the costs incurred by the utility.

Finally, the TRC test cannot be applied meaningfully to load building programs, thereby limiting the ability to use this test to compare the full range of demand-side management options.

Formulas: The formulas for the net present value (NPV<sub>TRC</sub>), the benefit-cost ratio (BCR<sub>TRC</sub>) and levelized costs are presented below:

 $\begin{array}{rcl} \mathsf{NPV}_{\mathsf{TRC}} &=& \mathsf{B}_{\mathsf{TRC}} &=& \mathsf{C}_{\mathsf{TRC}} \\ & & \mathsf{BCR}_{\mathsf{TRC}} &=& \mathsf{B}_{\mathsf{TRC}} & / & \mathsf{C}_{\mathsf{TRC}} \\ & & \mathsf{LC}_{\mathsf{TRC}} &=& \mathsf{LCRC} & / & \mathsf{IMP} \end{array}$ 

28

where:

NPV	Net present value of total costs of the resource
BCRTRC	= Benefit-cost ratio of total costs of the resource
LCTRC	<ul> <li>Levelized cost per unit of the total cost of the resource (cents per kWh for conservation programs; dollars per kW for load management programs)</li> </ul>
BTRC	= Benefits of the program
CTRC	= Costs of the program
LCRC	= Total resource costs used for levelizing
IMP	= Total discounted load impacts of the program

The B<sub>TRC</sub>, C<sub>TRC</sub>, LCRC, and IMP terms are further defined as follows:

$$B_{TRC} = \sum_{t=1}^{N} \frac{UAC_{t} + TC_{t}}{(i+d)^{t-1}} + \sum_{t=1}^{N} \frac{UAC_{at} + PAC_{at}}{(1+d)^{t-1}}$$

$$C_{TRC} = \sum_{t=1}^{N} \frac{UC_{t} + PC_{t} + UIC_{t}}{(1+d)^{t-1}}$$

$$LCRC = \sum_{t=1}^{N} \frac{UC_{t} + PC_{t} - TC_{t}}{(1+d)^{t-1}}$$

$$IMP = \frac{\sum_{t=1}^{N} \left[ \left( \sum_{i=1}^{I} \Delta EN_{it} \right) \text{ or } \left( \Delta DN_{it} \text{ where } i = \text{peak period} \right) \right]}{(1+d)^{t-1}}$$

[All terms have been defined in previous chapters.]

The first summation in the  $B_{TRC}$  equation should be used for conservation and load management programs. For fuel substitution programs, both the first and second summations should be used.

Examples of the Use of the Total Resource Cost Test: Table 4 demonstrates the use and calculation of the Total Resource Cost Test

for each of five program types. Detailed input and output data and calculations are presented for each example program type in Appendix B.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 21 Witness: Smith & Marks

#### Data Request:

On page 13 of the Marks Testimony is the statement that Western is not attempting to recover the revenue erosion caused by WKG CARES. However, throughout its application, Western has cited one of the reasons for the proposed revenue increase has been the impact of residential energy conservation efforts. Is it correct that Western is attempting to recover these lost revenues through its general rate increase, rather than through the DSM tariff? Explain the response.

#### Response:

Western made no attempt to either include or exclude revenue erosion from WKG CARES in the general rate increase proposed by Western. Any revenues lost during the pilot period are permanently lost.

Western would not dispute the assertion that the annual revenue lost during calendar 2000, due to WKG CARES, would be captured through Western's proposed general increase in rates. However, Western would not capture any additional revenue lost from those WKG CARES customers added after 2000.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 22 Witness: Marks

#### Data Request:

22. On page 14 of the Marks Testimony is a discussion of the benefit to Western's uncollectible accounts as a result of WKG CARES. If the actual impact on uncollectible accounts was not determined as part of the impact evaluation, explain in detail the basis for the assumption that uncollectible accounts were impacted favorably by WKG CARES.

#### Response:

AEG has conducted evaluations of other utility low income programs. In one such evaluation, we were asked by the Collaborative that was responsible for Detroit Edison's low income program to evaluate the impact on uncollectible accounts. Detroit Edison's program was similar in nature to WKG CARES. We found that Detroit Edison's program did have the effect of reducing the amount of uncollectible accounts. We were also told by Dr. Gil Peach, who was the Technical Advisor for the Detroit Edison Collaborative and a nationally recognized expert in low income program evaluation, that such a finding was consistent with a number of other utility programs which he had evaluated. While we did not conduct this analysis for WKG CARES, it is reasonable to hypothesize that such a benefit "might" also exist for this program as well.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 23. Witness: Marks

## Data Request:

23. Provide copies of all program materials associated with WKG CARES, including the program parameters established regarding allowable program measures.

## Response:

See attachment entitled, "Pre-Implementation Benefit/Cost Screening Results".

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 24 a Witness: Marks

#### Data Request:

24. Provide the following information concerning WKG CARES:

a. Describe the specific changes in customers' consumption patterns that Western is attempting to influence through WKG CARES.

#### Response:

Since no specific page or line in my testimony has been cited, it is assumed that this question is being asked from a generic perspective. The WKG CARES Program provides improvements in overall home thermal integrity in general accordance with DOE guidelines. While customer consumption patterns may change as a result of the program, it is not the specific intent of the program to cause this change in behavior. Savings can and will occur if participating customers continue to use energy in the same fashion after they participate in the program as they did before they participated in the program.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 24. b. Witness: Marks

## Data Request:

24. Provide the following information concerning WKG CARES:

b. Indicate whether WKG CARES is consistent with Western's most recent long-range integrated resource plan ("IRP"). Include copies of Western's most recent IRP.

N,

#### Response:

Western does not have an IRP.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 24 c Witness: Marks

#### Data Request:

24. Provide the following information concerning WKG CARES:

c. Does WKG CARES result in any unreasonable prejudice or disadvantage to any class of customers? Explain the response.

#### Response:

The WKG CARES Program is targeted to residential families that qualify for DOE assistance through local CAP Agencies. The guidelines for acceptance for the DOE Program are mandated federally and generally supported by local government and states. WKG CARES is offering enhanced services to customers who are eligible for Federal assistance because the available Federal dollars are often not enough to do the job properly and completely. Furthermore, as federal dollar allocations shrink, there are less funds available for eligible families. The program was designed for the benefit of lowincome customers. Within that segment of society, it provides equal and fair treatment to everyone. The only other qualifying factor is that the home must be served by WKG.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 24. d. Witness: Marks

#### Data Request:

24. Provide the following information concerning WKG CARES:

d. Describe the extent to which WKG CARES programs are available, affordable, and useful to all customers.

#### Response:

The WKG CARES programs are available for use only by low-income WKG customers. As noted in Exhibit MM-2, the proposed rate will cost about \$.25 per month for each residential customer. The prospective annual program expense of \$200,000 equates to spending which is less than 0.3% of the normalized annual bill for all residential customers. This amount seems affordable. That the programs are useful to all customers is demonstrated by the positive TRC cost/benefit ratio.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 25 Witness: Smith & Marks

#### Data Request:

Given the provisions of KRS 278.285, explain in detail why Western waited until the filing of this rate proceeding to recover the costs for WKG CARES, which began in 1996.

#### Response:

KRS 278.285 provides no specific guidance regarding the time period over which DSM program costs may be collected. It merely provides that the issue may be addressed as part of a proceeding for approval of new rate schedules or in a separate proceeding which would be limited to a review of demand-side management and associated rate recovery issues.

This was Western's first DSM program. Western believed it was necessary for the program to produce data upon which to base a filing for rate recovery. Western indicated this intention in its letter responding to Mr. Mills in November 1996. The WKG CARES process and impact evaluation report was not available until the timeframe during which the decision to file this rate proceeding was made. The WKG CARES program began in late November 1996. The first full winter heating season to provide information for the process and impact evaluation report did not end until spring of 1998. The report was completed in July 1998 and presented to the Collaborative in September 1998. Since the decision to file this rate proceeding was made shortly thereafter, it was decided that this filing was the best way to address the issue of WKG CARES cost recovery.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 26 Witness: Smith & Marks

#### Data Request:

On page 18 of the Marks Testimony is the statement that Western is seeking to recover the costs associated with the 3-year pilot program which was approved by the Commission in the October 20, 1995 Order in Case No. 95-010. The October 20, 1995 Order approving the October 9, 1995 Settlement in Case No. 95-010 specifically stated that:

In all other respects this proposal mirrors the July 18, 1995 Settlement. Those provisions not addressed herein which were previously addressed and accepted in the Commission's Order of August 10, 1995 are approved without discussion.<sup>1</sup>

The pilot DSM program is not discussed in the October 20, 1995 Order, and the Commission stated in the August 10, 1995 Order that it made no decision or findings of fact related to the DSM provisions included in the Settlement. Explain in detail how Western has concluded that the Commission approved the 3-year, pilot program.

#### Response:

Correctly stated, the Commission approved the settlement, and in so doing commenced a process for developing DSM programs that qualified for rate recovery. The August 10, 1995 Order recognized that DSM programs would be developed and that Western had committed funding for this purpose. Clearly, the Commission understood that the purpose of the DSM programs would be to reduce the energy bills of low income customers via Demand Side Management, and that Western would eventually seek rate recovery.

<sup>1</sup> Case No. 95-010, October 20, 1995 Order, at 3.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 27. a. Witness: Marks

#### Data Request:

27. Provide the following information concerning the costs associated with WKG CARES:

a. A schedule of the costs incurred during the 3-year pilot. These amounts should be identified using Western's account numbers, with account titles. The amounts should be presented in total and fiscal year amounts.

Response:

Fiscal YEAR	Account Number	Account Name	Amount
1997	18607069909	DSM Project – KY	\$249,907.24
1998	18607069909	DSM Project - KY	\$334,832.29
1999 (through May)	18607069909	DSM Project - KY	\$133,721.90
Total	18607069909	DSM Project - KY	\$718,461.43

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 27. b. Witness: Adams

## Data Request:

27. Provide the following information concerning the costs associated with WKG CARES:

b. Were the pilot costs expensed or deferred by Western? If deferred, provide the Western detailed account numbers utilized and the basis for Western assuming deferral was appropriate.

## Response:

The pilot costs were expensed.

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 27. c. Witness: Marks

## Data Request:

27. Provide the following information concerning the costs associated with WKG CARES:

c. A schedule of the costs proposed for the 1999 - 2002 period. These amounts should be identified using Western's account numbers, with account titles. The amounts should be presented in total and fiscal year amounts.

Response:

Fiscal YEAR	Account Number	Account Name	Amount
2000	18607069909	DSM Project - KY	\$200,000.00
2001	18607069909	DSM Project - KY	\$200,000.00
2002	18607069909	DSM Project – KY	\$200,000.00
Total	18607069909	DSM Project - KY	\$600,000.00

,

Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 28 Witness: David H. Doggette

#### Data Request:

Provide an analysis that examines the completion percentage for capital budget projects. The analysis should be by individual capital project, by year, for fiscal years 1994 through 1998. The analysis should show the completion percentage (actual expenditure to budget amount) for each project in each fiscal year.

#### **Response:**

See attached schedules entitled "WKG CAPITAL BUDGET PROJECTS" for each of the fiscal years from 1994 through 1998, pages 1 to 28.

Table 1 on page 8 in the testimony of David H. Doggette contains data that differs from the schedules attached. The difference is attributable to rounding of data to the nearest dollar amount. Also, the data listed in Table 1 for FY 1998 does not include overhead dollars. Therefore, the attached schedule reflects higher dollar amounts for FY 1998. The "Actual Dollars" that should be listed in Table 1 for FY 1998 should be \$11,459,605 and the "Budgeted Dollars" should be listed as \$10,194,434 which produces a difference of \$1,265,171.
## WKG CAPITAL BUDGET PROJECTS FY 1994 Completion Percentage

Budget	No Description B Operating Area	udget A	Amount	Expendit	ture Amount	Completion %
	WKG Company Office				١	
39009	Improvement-Leased Premises	\$	-	\$	9,597	<del>#####</del>
39160	Office Machines	\$	28,510	\$	23,926	83.9
39170	Other Office Equipment	\$	-	\$	276	<del>#####</del>
39211	Vehicles	\$	30,868	\$	28,621	92.7
39711	Communication Equip	\$	60,473	\$	62,587	103.5
39985	Mainframe Hardware	\$	, -	\$	413	<del>#####</del>
39987	PC Software	\$	451	\$	442	98.0
99800	Retirements & Salvage	\$	5	\$	(12,883)	<del>#####</del>
	Total	\$	120,307	\$	112,979	93.9
	Owensboro Operations					
33400	Field M&R Station	\$	-	\$	2,122	<del>#####</del>
36740	Mains Cathodic Protection	\$	11,000	\$	4,684	42.6
37610	Blanket Mains System Improven	nen \$	255,980	\$	268,902	105.0
37620	Blanket Mains Public Improveme	ent \$	65,608	\$	28,152	42.9
37630	Blanket Mains Leakage	\$	39,255	\$	48,925	124.6
37820	US 60 E Relocation	\$	31,736	\$	24,640	77.6
38010	Services System Improvement	\$	317,000	\$	564,647	178.1
38030	Services Leakage	\$	43,000	\$	207,339	482.2
38510	Hancock Co Asphalt	\$	40,000	\$	11,306	28.3
38700	Other Equipment	\$	1,901	\$	-	0.0 Deferred
39009	Improvements- Leased Premises	s \$	28,111	\$	17,077	60.7
39150	Office Furniture	\$	1,038	\$	-	0.0 Deferred
39211	Vehicles	\$	74,030	\$	103,031	139.2
39221	CNG Project Phase II	\$	15,120	\$	17,062	112.8
39400	Tools, Shop Equipment	\$	15,923	\$	10,752	67.5
39600	Backhoe Trailer Ramp Lift	\$	840	\$	1,110	132.1
39985	Mainframe Hardware	\$	5,260	\$	3,204	60.9
39986	PC Hardware	\$	-	\$	1,259	<del>#####</del>
98100	Main Ext Forfeitures	\$	(41,574)	\$	(54,640)	131.4
99200	AIC Center St Relocation	\$	-	\$	(23,491)	<del>#####</del>
99800	Retirements	\$	15	\$	22,865	<del>#####</del>
	Total	\$	904,243	\$	1,258,946	139.2



WKG	Technical	Services
-----	-----------	----------

35500	Measuring & RegulatingStation	\$ 126,317	\$ 32,520	25.7
39150	Office Furniture	\$ 552	\$ 55,703	<del>#####</del>
39211	Vehicles	\$ 38,076	\$ 27,042	71.0
39711	Communication Equipment	\$ 4,290	\$ 4,308	100.4
39986	PC Hardware	\$ 14,709	\$ 21,186	144.0
39987	PC Software	\$ 12,561	\$ 13,832	<b>λ</b> 110.1
99800	Retirements & Salvage	\$ 1	\$ (2,400)	<del>#####</del>
	Total	\$ 196,506	\$ 152,191	77.4
M	adisonville Operations			
36710	Bremner Cookie 4" Pipeline	\$ -	\$ 3,917	<del>#####</del>
36720	Marion Trans Line Relocate	\$ -	\$ 32	<del>######</del>
36740	Transm Rectifier & Groundbed	\$ 30,958	\$ -	0.0 Deferred
37610	Mains- System Improvements	\$ 362,500	\$ 227,951	62.9
37620	Mains- Public (Hwy) Relocations	\$ 145,779	\$ 58,334	40.0
37630	Mains- Leakage	\$ 91,492	\$ 365,287	399.3
37640	Mains- Cathodic Protection	\$ 45,774	\$ 39,725	86.8
37810	M&R Sta Equipment	\$ 10,152	\$ 9,673	95.3
37820	M&R Sta Equipment- Public Impr	\$ -	\$ 3,629	<del>#####</del>
37920	City Gate M&R Sta Pub Impr	\$ 3,000	\$ -	0.0 Deferred
38010	Services- New	\$ 205,900	\$ 607,526	295.1
38020	Services-Pub-Impr-SkyLine Dr	\$ 18,415	\$ -	0.0 Deferred
38030	Services- Leakage	\$ 114,500	\$ 202,346	176.7
38700	Other Equipment	\$ 16,331	\$ 14,848	90.9
39009	Improvements- Leased Mad'vI WH	\$ 10,150	\$ 61,974	610.6
39150	Office Furniture	\$ 136,623	\$ 2,491	1.8
39160	Office Machines	\$ 2,150	\$ 2,479	115.3
39211	Vehicles	\$ 172,190	\$ 169,695	98.6
39400	Tools, Shop Equip	\$ 22,794	\$ 17,416	76.4
39600	Power Operated Equip	\$ 28,614	\$ 26,324	92.0
39711	Communication Equip	\$ 7,755	\$ 240	3.1
39986	PC Software	\$ -	\$ 100	<del>#####</del>
98100	Main Ext Forfeiture	\$ (2,872)	\$ (3,376)	117.5
98300	Public Works Reimbursement	\$ -	\$ (8,765)	<del>        </del>
99100	Public Works Reimbursement- Gla	\$ (11,850)	\$ (34,419)	290.5
99200	Hwy Relocation AIC	\$ -	\$ (12,013)	<del>#####</del>
99800	Retirements	\$ 30	\$ 75,382	<del>#####</del>
	Total	\$ 1,410,385	\$ 1,830,796	129.8

### Paducah Operations

36720	Transm Mains Public Improv	\$ 200,000	\$ -	0.0	Deferred
36740	Transm Mains Cathodic Protection	\$ 2,000	\$ -	0.0	Deferred
36920	M&R Sta Public Improvement	\$ 1,500	\$ 767	51.1	
37610	Mains- System Improve	\$ 563,513	\$ 281,325	49.9	
37620	Mains- Public Improve	\$ 85,770	\$ 10,471	12.2	
37630	Mains- Leakage	\$ 67,514	\$ 20,460	× <b>30.3</b>	
37640	Mains- Cathodic Protection	\$ 10,000	\$ 25,616	256.2	
38010	Services- New	\$ 257,000	\$ 427,722	166.4	
38030	Services- Leakage	\$ 193,000	\$ 32,330	16.8	
38700	Other Equipment	\$ 1,038	\$ 786	75.7	
39150	Office Equipment	\$ 154,835	\$ -	0.0	Deferred
39160	Office Machines	\$ 995	\$ 985	99.0	
39170	Other Office Equipment	\$ 14,776	\$ -	0.0	Deferred
39211	Vehicles	\$ 38,041	\$ 47,627	125.2	
39400	Tools, Shop Equip	\$ 5,530	\$ 4,677	84.6	
39985	Mainframe Hardware	\$ 270	\$ 117	43.3	
39986	PC Hardware	\$ -	\$ 114	<del>#####</del>	
98100	Main Extension Forfeiture	\$ (13,286)	\$ (15,274)	115.0	
99100	Pub Works Reimbursement Gla	\$ (5,000)	\$ 9,271	<del>#####</del>	
99800	Retirements & Salvage+C94	\$ 10	\$ (14,484)	<del>#####</del>	
	Ũ				
	Total	\$ 1,577,506	\$ 832,510	52.8	
Bo	wling Green Operations				
36710	Cemetery Rd System Improv	\$ 370,000	\$ 297,981	80.5	
36740	Transm Mains- Cathodic Protection	\$ 2,000	\$ -	0.0	
36910	M&R Sta Sys Improv	\$ 63,406	\$ 275	0.4	
37610	Mains System Improv	\$ 580,297	\$ 546,533	94.2	
37620	Mains- Public Improv	\$ 85,825	\$ 15,553	18.1	
37630	Mains- Leakage	\$ 98,406	\$ 150,268	152.7	
37640	Mains- Cathodic Protection	\$ 20,000	\$ 16,905	84.5	
37800	M&R Sta Sys Improv	\$ 43,400	\$ 90,775	209.2	
37910	City Gate M&R Sta Equip Sys Imp	\$ 61,293	\$ 31,200	50.9	
38010	Services- New	\$ 265,000	\$ 481,077	181.5	
38020	Services- Public Improve	\$ 44,909	\$ 1,580	3.5	
38030	Services- Leakage	\$ 150,000	\$ 270,342	180.2	
38510	Ind M&R Equip Sys Improv	\$ 40,000	\$ 5,394	13.5	
38700	Other Equipment	\$ 3,735	\$ 2,947	78.9	
39009	Improvements- Leased Glasgow W	\$ 30,443	\$ 38,519	126.5	
39150	Office Furniture	\$ 91,694	\$ 89,757	97.9	
39160	Office Machines	\$ 985	\$ 733	74.4	
39170	Other Office Equipment	\$ 10,571	\$ 7,012	66.3	
39211	Vehicles	\$ 97,190	\$ 88,974	91.5	
39400	Tools, Shop Equip	\$ 8,271	\$ 5,866	70. <del>9</del>	
39694	Backhoe	\$ 47,500	\$ 53,404	112.4	

39711	Communication Equipment	\$	1,155	\$	3,646	315.7	
39985	Mainframe Hardware	\$	275	\$	117	42.5	
39986	PC Hardware	\$	-	\$	126	######	
98100	Main Extension Forfeiture	\$	(7,189)	\$	(3,852)	53.6	
99100	Public Works Reimbursement	\$	(5,000)	\$	-	0.0	Deferred
99800	Retirements	\$	16	\$	37,875	<del>######</del>	
	Total	\$	2,104,182	\$	2,233,007	106.1	
D	anville Operations						
36730	Line 133 12" Replacement	\$	175,000	\$	172,438	98.5	
36740	Trans Mains-Cathodic Protection	\$	4,761	\$	452	9.5	
36910	M&R Sta- Aiken Rd PS Odorizer	\$	29,117	\$	-	0.0	Deferred
39620	M&R Sta- Aiken Rd PS Rebuild	\$	75,000	\$	-	0.0	Deferred
37610	Mains- Sys Improve	\$	369,158	\$	374,299	101.4	
37620	Mains- Public (Hwy) Relocat	\$	270,403	\$	106,390	39.3	
37630	Mains- Leakage	\$	169,441	\$	159,417	94.1	
37640	Mains- Cathodic Protection	\$	25,070	\$	33,190	132.4	
37810	M&R Sta Equip Sys Improv	\$	-	\$	7,725	<del>          </del>	
37910	Shelbyville Ind M&R Sta	\$	11,231	\$	47,558	423.5	
38010	Services- New	\$	305,400	\$	861,470	282.1	
38030	Services- Leakage	\$	24,000	\$	95,027	395.9	
38510	Ind M&R ATGTBT Meter	\$	-	\$	17,681	<del>#####</del>	
387000	Other Equipment	\$	12,532	\$	10,561	84.3	
39004	Heating & A/C Equip	\$	1,430	\$	1,228	85.9	
39009	Improv Shelbyvi WH	\$	28,329	\$	16,489	58.2	
39150	Office Furniture- Danville	\$	48,943	\$	42,575	87.0	
39160	Office Machines- Danville	\$	6,256	\$	3,236	51.7	
39211	Vehicles	\$	56,488	\$	55,182	97.7	
39400	Tools, Shop Equip	\$	20,646	\$	20,462	99.1	
39600	Power Op Equip- Forklift	\$	23,900	\$	24,524	102.6	
39693	Ditchers & Trenchers	\$	111,400	\$	93,355	83.8	
39711	Communication Equip	\$	4,200	\$	16,010	381.2	
39985	Mainframe Hardware	\$	1,051	\$	581	55.3	
98100	Main Ext Forfeiture	\$	(25,162)	\$	(35,479)	141.0	
99100	Public Works Reimbursement	\$	(128,798)	\$	(4,621)	3.6	
99800	Retirements & Salvage	\$	31	\$	(107,851)	<del>#####</del>	
	Total	\$	1,619,827	\$	2,011,899	124.2	
W	/KG Overheads	\$	196,461	\$	176,032	89.6	
	Grand Total	\$ <sup>-</sup>	11,453,427	\$1	0,872,491	94.9	



I.

## WKG CAPITAL BUDGET PROJECTS FY 1995

Completion Percentage

Budget No. Description **Budget Amount Expenditure Amount Completion % Operating Area** WKG Company Office 39009 5,809 Improvement-Leased Premises \$ 5,714 \$ 101.7 Office Furniture & Equipment 39100 \$ 2,726 \$ 3,060 112.3 4,788 39103 **Office Machines** \$ \$ 4,863 98.5 Vehicles & CNG Conversion \$ \$ 39200 87,941 77,729 88.4 \$ \$ 39700 Communication Equip 54,638 46.972 86.0 \$ \$ 39905 Mainframe Hardware 6,108 5,603 91.7 \$ \$ 39906 PC Hardware 13,506 6,973 51.6 \$ 39907 PC Software \$ 1,233 66.0 1,869 99800 **Retirements & Salvage** \$ \$ (1,500)24.9 (6,019) Total \$ \$ 171,346 150,667 87.9 **Owensboro** Operations 36700 **Trans Mains Cathodic Protection** \$ 1.000 \$ 0.0 36701 **TP Mains Steel- Public Reloc** \$ 350,020 \$ 0.0 Deferred \$ 36900 10 M & R Sta Old KY 54 \$ 12,240 11,190 91.4 \$ \$ 35,498 36900 20 **Commonwealth Purchase** 8,542 415.6 \$ \$ 24,915 37600 40 Mains Cathodic Protection 10,000 249.2 \$ \$ 207,965 37601 10 Blanket Mains- Sys Improv 332,938 62.5 37601 20 Blanket Mains Public Improvements \$ 282.671 \$ 174,069 61.6 \$ 144,674 175,651 121.4 37601 30 **Blanket Mains Leakage** \$ \$ \$ 6.020 84.8 37800 20 US 60 E Relocation 7,096 38000 10 Services System Improvement \$ 400,000 \$ 570,828 142.7 \$ 100,000 \$ 187,178 187.2 38000 30 Services Leakage \$ \$ 187,382 ###### 38200 10 Meter Installations New 1 \$ \$ Meter Installations Public Improv 1 165,271 ###### 38200 20 \$ \$ 38700 Other Equipment 2,588 3,824 147.8 \$ \$ 6,327 **Improvements-** Leased Premises 5,529 114.4 39009 \$ **Office Furniture & Equipment** \$ 876 733 83.7 39100 \$ \$ 152.624 Vehicles 143.574 106.3 39200 \$ \$ 8,881 128.3 39400 **Tools, Shop Equipment** 6,920 \$ \$ 30,950 102.0 39603 Ditchers 30.354 \$ 382 \$ 148 38.7 39700 **Communication Equip** \$ 15,265 \$ 14,055 92.1 Mainframe Hardware 39905 \$ (11,400)\$ (11, 403)100.0 Main Ext Forfeitures 98100 \$ ###### \$ (19, 532)98300 **Public Works Reimbursements** 0 \$ \$ 106,859 ###### 1,072 Retirements 99800

Total

l

\$ 1,844,343

\$ 2,039,433

110.6

## WKG CAPITAL BUDGET PROJECTS FY 1995 Completion Percentage

Budget No. Oj	Description E perating Area	Budget .	Amount	Expenditu	ire Amount	Completion %
O	wensboro Storage & Transmissio	n			<u>``</u>	
35100 02	Structures & Improvements	\$	58,480	\$	54,382	93.0
35200 20	Well Workovers	\$	466,785	\$	254,946	54.6
35400 20	Compressor Station Equip	\$	5,000	\$	9,134	182.7
36701 20	Spencer Co 4" Relocation	\$	172,547	\$	173,221	100.4
36900 20	M&R Sta- Public Improvement	\$	13,511	\$	7,708	57.0
37601 20	US 231 Hartford-Beaver Dam 6	"Rel\$	199,895	\$	253,082	126.6
38700	Other Equipment	\$	3,622	\$	3,743	103.3
39004	A/C Equipment	\$	1,320	\$	1,454	110.2
39200	Vehicles	\$	23,140	\$	28,048	121.2
39400	Tools Shop Equip	\$	4,058	\$	4,956	122.1
99100	Public Works Reimbursement-	\$	-	\$	(61,756)	<del>#####</del>
99800	Retirements & Salvage	\$	(1,492)	\$	(6,686)	448.1
	Total Resp Ctr	• \$	946,866	\$	722,232	76.3
W	KG Measurement Center					
35500 20	M&R Equipment	\$	3,600	\$	4,438	123.3
37000	Communication Equip	\$	10,800	\$	10,055	93.1
38100 10	Meters, New	\$	1,192,613	\$	1,174,693	98.5
38100 20	Meters, Replacement	\$	796,624	\$	689,849	86.6
38300 10	House Regs,new	\$	38,354	\$	50,769	132.4
38300 20	House Regs- Replacement	\$	375,034	\$	326,147	87.0
38500 20	Ind M&R Equip Public Improve	\$	420,380	\$	304,677	72.5
39009	Improvements- Leased Premise	es \$	78,719	\$	79,943	101.6
39200	Vehicles	\$	21,760	\$	20,092	92.3
39400	Tools, Shop Equip	\$	11,321	\$	11,700	103.3
39700	Communication Equip	\$	583	\$	-	0.0
39705	Communication Equip- Telemeter	erin \$	1,800	\$	2,526	140.3
39905	Mainframe Hardware	\$	6,108	\$	5,603	91.7
39906	PC Hardware	\$	300	\$	524	174.7
39907	PC Software	\$	3,000	\$	1,629	54.3
99800	Retirements & Salvage	\$	(2,842)	\$	(157,854)	<del>#####</del>
	Total	\$	2,958,154	\$	2,524,791	85.4

39400

39600

39603

39700

39800

Tools, Shop Equip

Ditchers

Misc Equip

**Power Operated Equip** 

**Communication Equip** 

# WKG CAPITAL BUDGET PROJECTS FY 1995

Completion Percentage

**Budget No.** Description **Budget Amount Expenditure Amount** Completion % **Operating Area** WKG Technical Services 36900 Measuring & Reg Station Equip 0 \$ ###### \$ (26)38700 \$ Other Equipment \$ 0 40,812 ####### **Office Furniture & Equipment** \$ \$ 39100 385 260 67.5 \$ \$ 39211 Vehicles 129,220 99.0 130,564 39600 **Power Operated Equip** \$ 17,971 \$ 19,574 108.9 39700 **Communication Equipment** \$ \$ 350 0.0 39905 Mainframe Hardware \$ \$ 5,650 6,107 92.5 39906 PC Hardware \$ \$ 35,500 191.6 18,530 39907 PC Software \$ \$ 23,308 ` 4.005 17.2 99800 Retirements & Salvage \$ \$ (5,529)63.4 (8,721) Total \$ 188.494 \$ 229,466 121.7 Madisonville Operations **Transm Rectifier & Groundbed** \$ 46,429 \$ 36,094 77.7 Deferred 36700 40 M & R Station Equip \$ \$ 9,955 ####### 36900 10 \$ \$ 29,958 Mains - Cathodic Protection 23,407 128.0 37640 \$ \$ Mains- Stl System Improve 381,558 164,429 43.1 37601 10 Mains- Public (Hwy) Relocations \$ \$ 38,594 37601 20 360,417 10.7 \$ Mains- Leakage \$ 199,800 458,842 229.7 37601 30 \$ ####### 37602 20 Mains- PE Public (Hwy) Relocation \$ 3,180 \$ \$ 37900 20 City Gate M&R Sta Pub Impr 2.999 2,679 89.3 \$ 38000 10 Services-New \$ 326,353 490,539 150.3 \$ \$ 10.584 38000 20 Services-Pub-Impr-SkyLine Dr 18,415 57.5 38000 30 Services-Leakage \$ 109,000 \$ 280,294 257.2 Meter Installations New \$ \$ 64,107 ###### 38200 10 1 Meter Set Replacement \$ 1 \$ 49,858 ####### 38200 20 \$ \$ Indust M&R Equip KY 1789 Reloc 0 19.346 ###### 38500 10 \$ \$ 38700 Other Equipment 14,806 14,253 96.3 \$ \$ 39009 Improvements- Leased Mad'vI WH 296,652 302,922 102.1 \$ \$ Office Furniture & Equipment 254,036 122,276 48.1 39100 \$ **Office Machines** \$ 2,866 11,189 390.4 39103 \$ \$ 50,333 103.8 39200 Vehicles 48,493 \$ \$ 0.0 39300 Stores Equipment 13,327

\$

\$

\$

\$

\$

39.383

18,796

9,700

14,197

2,622

\$

\$

\$

\$

\$

36,559

16.861

9,535

10,734

668

92.8

89.7

98.3

75.6

25.5

## WKG CAPITAL BUDGET PROJECTS FY 1995 Completion Percentage

Budget No.	Description	Budget	Amount	Expenditu	ire Amount	Completion %
Ο	perating Area					
39905	Mainframe Hardware	\$	26,362	\$	20,610	78.2
98100	Main Ext Forfeiture	\$	(77,471)	\$	(76,378)	98.6
98300	Public Works Reimbursement	\$	-	\$	(41,977)	<del>#####</del>
99100	Public Works Reimbursement-	Gla \$	-	\$	(106,573)	<del>#####</del>
99800	Retirements	\$	14,928	\$	111,091	744.2
	Total	\$	2,147,077	\$	2,140,562	99.7
Pa	aducah Operations					
36510	Land & Land Rights	\$	-	\$	4,862	<del>#####</del>
36700 40	Transm Mains Cathodic Protect	tion \$	2,000	\$	-	0.0 Deferred
36900 20	M&R Sta Public Improvement	\$	26,800	\$	3,429	12.8
37600 40	Mains- Cathodic Protection	\$	20,000	\$	18,179	90.9
37601 10	Mains- System Improve	\$	544,594	\$	364,500	66.9
37601 20	Mains- Public Improve	\$	190,224	\$	196,978	103.6
37601 30	Mains- Leakage	\$	205,229	\$	222,152	108.2
37900 10	City Gate M&R Sta Chk Sta	\$	5,509	\$	7,570	137.4
38000 10	Services- New	\$	545,000	\$	447,544	82.1
38000 30	Services- Leakage	\$	45,000	\$	75,890	168.6
38200 10	Meter Installations	\$	1	\$	87,231	<del>///////</del>
38200 20	Meter Replacements	\$	1	\$	28,953	<del>#####</del>
38700	Other Equipment	\$	8,884	\$	8,034	90.4
39009	Improvements- Leased Premise	es \$	8,325	\$	1,680	20.2
39100	Office Furniture & Equipment	\$	169,610	\$	101,505	59.8
39200	Vehicles	\$	152,479	\$	154,730	101.5
39400	Tools, Shop Equip	\$	29,558	\$	32,253	109.1
39600	Power Operated Equip	\$	27,914	\$	27,007	96.8
39603	Ditchers	\$	82,851	\$	68,261	82.4
39604	Backhoes	\$	55,692	\$	53,103	95.4
39700	Communication Equip	\$	382	\$	-	0.0
39905	Mainframe Hardware	\$	9,312	\$	8,528	91.6
98100	Main Extension Forfeiture	\$	(18,671)	\$	(14,863)	79.6
99100	Pub Works Reimbursement Gla	\$	(15,100)	\$	(69,462)	460.0
99800	Retirements & Salvage	\$	(18,354)	\$	(32,526)	177.2
	Total	\$	2,077,240	\$	1,795,538	86.4

WKG CAPITAL BUDGET PROJECTS FY 1995

**Completion Percentage** 

**Budget No.** Description **Budget Amount** Expenditure Amount Completion % **Operating Area Bowling Green Operations** \_` 36700 40 Transm Mains- Cathodic Protection \$ 2.000 \$ 0.0 \$ 36700 10 Cemetery Rd System Improv \$ 183.871 159.473 86.7 6" TP Line to Elrod Rd \$ 379,096 \$ 377,094 99.5 36701 20 \$ 8" TP line to US 31W \$ 390,624 102.6 36701 20 380,859 \$ \$ M&R Sta US 31W, Elrod Rd 103,945 138,830 133.6 36900 20 \$ 37600 40 Mains- Cathodic Protection \$ 21,000 7,816 37.2 \$ \$ Mains System Improv 927,182 711.308 76.7 37601 10 37601 20 Mains- Public Improv \$ 776.867 \$ 435.632 56.1 \$ \$ 37601 30 Mains-Leakage 171.830 285.506 166.2 \$ \$ 950 18.3 37800 M&R Sta Sys Improv 5.200 \$ \$ 0.0 Deferred 37900 20 City Gate M&R Sta Equip Pub Imp 15.000 \$ \$ 549.543 104.7 38000 10 Services-New 525,000 \$ \$ 38000 20 Services- Public Improve (3,933)##### \$ \$ 38000 30 Services-Leakage 85.000 240,413 282.8 \$ \$ Meter Installations 181.684 ##### 38200 10 1 \$ \$ ###### Meter Replacements 1 119,144 38200 20 \$ \$ Ind M&R Equip Sys Improv 40.000 0.0 38500 10 \$ \$ 38500 20 Ind M&R Equip Pub Improv 22.000 15.854 72.1 \$ \$ 107.1 Other Equipment 7.809 8.361 38700 \$ \$ 39009 Improvements- Leased Premises 8.200 4,263 52.0 \$ \$ 2,867 105.9 39100 **Office Furniture & Equipment** 2,707 \$ \$ 422 87.9 39103 Office Machines 480 \$ \$ 259,630 100.8 39200 Vehicles 257,495 \$ \$ 33,867 32,859 97.0 39400 Tools, Shop Equip \$ \$ 76.4 39600 **Power Operated Equip** 18,320 13,996 \$ \$ 7,693 80.1 Ditchers 9,600 39603 \$ \$ 1,797 173.6 39700 **Communication Equipment** 1,035 \$ \$ 90.3 Communication Equip- Mob Radio 2,600 2,347 39701 \$ 39905 Mainframe Hardware \$ 9,462 8,584 90.7 \$ \$ Main Extension Forfeiture (37, 430)(148, 397)396.5 98100 \$ 98300 **Public Works Reimbursement** \$ (7, 120)##### Deferred \$ \$ (120,000)0.0 Deferred 99100 **Public Works Reimbursement** \$ 90,161 ####### \$ 99800 **Retirements & Salvage** (14, 263)\$ 3,887,401 101.8

Total

\$ 3,818,734

## WKG CAPITAL BUDGET PROJECTS FY 1995 Completion Percentage

Budget No.	Description B	udget	Amount	Expenditu	ire Amount	Completion %
O	perating Area					
Da	anville Operations					
36600 20	Structures & Improvements	\$	1 702	\$	1.505	88.4
36700 40	Trans Mains-Cathodic Protection	n \$	3 000	\$ \$	-	0.0
36701 20	Shelbyville Ind Tie-back	 \$	273 400	\$	260 005	95.1
36701 30	Line 133 12" Replacement	\$	280 748	\$	241 210	85.9
36901 20	M&R Sta- Aiken Rd PS Rebuild	\$	161,197	\$	183.064	113 6 Deferred
37600 40	Mains- Cathodic Protection	\$	40.000	\$	63,966	159.9
37601 10	Mains- Sys Improve	\$	447,850	\$	386.771	86.4
37601 20	Mains- Public (Hwy) Relocat	\$	275,802	\$	333.935	121.1
37601 30	Mains-Leakage	\$	204,179	\$	57,140	28.0
37800 10	M&R Sta Equip Sys Improv	\$	14.234	\$	9.852	69.2
37800 20	M&R Sta Equip Sys Improv	\$	7.100	\$	6.684	
37900 20	Shelbyville Ind M&R Sta	\$	27.000	\$	30.624	94.1
38000 10	Services- New	\$	423,997	\$	646,960	152.6
38000 30	Services- Leakage	\$	70.000	\$	67.425	96.3
38200 10	Meter Installations	\$	1	\$	112.928	######
38200 20	Meter Replacements	\$	1	\$	76.672	######
38500 10	Ind M&R ATGTBT Meter	\$	61.878	\$	15.533	25.1
387000	Other Equipment	\$	3.016	\$	4.528	150.1
39003	Improvements	\$	5,600	\$	455	8.1
39004	Heating & A/C Equip	\$	-	\$	2,310	<del>######</del>
39009	Improv Leased Premises	\$	41,123	\$	48,394	117.7
39100	Office Furniture & Equip- Danville	e \$	23,059	\$	371	1.6
39103	Office Machines- Danville	\$	3,053	\$	1,596	52.3
39200	Vehicles	\$	37,870	\$	47,285	124.9
39300	Stores Equipment	\$	639	\$	948	148.4
39400	Tools, Shop Equip	\$	17,305	\$	17,339	100.2
39600	Power Op Equip- Forklift	\$	11,334	\$	8,088	71.4
39604	Backhoes	\$	59,755	\$	60,102	100.6
39700	Communication Equip	\$	1,157	\$	888	76.8
39702	Communication Equip - Fixed Ra	adio \$	-	\$	2,255	
39905	Mainframe Hardware	\$	15,418	\$	14,186	92.0
98100	Main Ext Forfeiture	\$	(68,198)	\$	(67,272)	98.6
99100	Public Works Reimbursement	\$	-	\$	(104,926)	<del>#####</del>
99800	Retirements & Salvage	\$	(3,304)	\$	35,406	<del>#####</del>
	Total	\$	2,439,916	\$	2,566,227	105.2
\\//	KG Overheads					
39900		\$	0	\$	(598,260)	<del>#####</del>
	Grand Total	\$	16,592,171	\$	15,458,057	93.2

## WKG CAPITAL BUDGET PROJECTS FY 1996 **Completion Percentage**

Budget No. Description Operating Area		Budget /	Amount	Ac	tual Amount	Completion %
v	VKG Company Office					
39009	Improvement-Leased Premise	es \$	7,755	\$	8,354	107.7
39100	Office Furniture & Equipment	\$	895	\$	15,237	<del>#####</del>
39200	Vehicles & CNG Conversion	\$	64,831	\$	50,954	78.6
39200 98	Transportation Equip	\$	(2,500)	\$	(3,800)	152.0
39700	Communication Equip	\$	1,375	\$	900	65.4
39906	PC Hardware	\$	20,581	\$	25,461	123.7
39907	PC Software	\$	1,790	\$	20,556	<del>#####</del>
99800	Retirements & Salvage	\$	-	\$	(2,400)	#####
	Total	\$	94,727	\$	115,262	121.7
0	wensboro Operations					
36900 20	Mains	\$	8,542	\$	-	0.0
37600 40	Mains Cathodic Protection	\$	21,000	\$	41,563	197.9
37601 10	Mains - Steel Sys Improv	\$	-	\$	10,158	<del>#####</del>
37601 20	Mains - Steel Relocate	\$	11,287	\$	64,289	569.6
37601 30	Mains - Steel Leakage	\$	101,221	\$	80,925	79.9
37601 81	Main Ext. Forfeiture	\$	(23,002)	\$	(15,934)	69.3
37601 98	Mains - Steel Retire	\$	9,839	\$	7,777	79.0
37602 10	Mains - Plastic Sys Improve	\$	280,000	\$	255,944	91.4
37602 20	Mains - Plastic	\$	183,110	\$	-	0.0
37602 30	Mains - Plastic Leakage	\$	-	\$	12,621	<del>#####</del>
37602 98	Mains - Plastic Retire	\$	-	\$	2,742	<del>#####</del>
37602 91	Mains	\$	(2)	\$	-	0.0
38000 10	Services New	\$	546,400	\$	500,940	91.7
38000 30	Services Leakage	\$	170,000	\$	234,779	138.1
38000 98	Services Retire	\$	-	\$	87,638	<del>#####</del>
38200 10	Meter Installations	\$	92,100	\$	156,002	169.4
38200 20	Meter Installations Public Impr	rov \$	30,000	\$	125,760	419.2
38200 98	Meter Installations Retire	\$	22,500	\$	9,081	40.4
38700	Other Equipment	\$	6,826	\$	5,734	84.0
39009	Improvements- Leased Premis	ses \$	13,400	\$	10,222	76.3
39103	Office Machines	\$	-	\$	540	#####
39200	Vehicles	\$	272,000	\$	243,917	89.7
39200 98	Transportation Equipment	\$	(22,500)	\$	(23,331)	103.7
39400	Tools, Shop Equipment	\$	4,172	\$	4,977	119.3
39600	Power Operated Equipment	\$	10,200	\$	8,734	85.6
39700	Communication Equip	\$	382	\$	545	142.7

DR 28		Ρ	age 13 of 28			
39702	Communication Equip - FI	\$	-	\$	833	<del>#####</del>
99800	Retirements	\$	-	\$	(4,811)	<del>#####</del>
	Total	\$	1,737,475	\$	1,821,646	104.8
C	wensboro Storage & Transmission					
33400	Field Meas & Reg	\$	1,308	\$	1,976	151.1
35200 20	Well Workovers	\$	324,614	\$	118,100	36.4
35400 98	Compressor Station Equip	\$	7,750	\$	-	0.0
35600	Purification Equip	\$	22,800	\$	20,244	88.8
36600	Structures & Improv	\$	15,000	\$	10,883	72.6
36701 20	TP Mains - Stl Public Improvemen	\$	1.849.640	\$	600.681	32.5
36701 30	TP Mains - Steel Leakage	\$	49.500	\$	6.477	13.1
36701 91	Mains - Steel	Ŝ	(24,198)	\$	-	
36701 98	Mains - Stl Retire	\$	13,800	\$	2.077	15.1
36900 20	M&R Sta- Public Improvement	\$	232,500	\$	143.696	61.8
37601 20	Mains - Steel Public Improv	ŝ	6 088	\$	-	0.0
37601 98	Mains - Steel Retire	\$	4 879	\$	11 292	231.4
38700	Other Equipment	\$	965	\$	625	64.8
39200	Vehicles	ŝ	43 889	\$	35 468	80.8
39400	Tools Shon Equin	ŝ	123 781	\$	126 021	101.8
39600	Power Operated Equipment	ŝ	37 966	\$		0.0
99100	Public Works Reimbursement-	ŝ	-	ŝ	(21.691)	<del>#####</del>
99800	Retirements & Salvage	\$	-	\$	2,756	<del>#####</del>
	Total Resp Ctr	\$	2,710,282	\$	1,058,605	39.1
W	KG Measurement Center					
37000	Commication Equip	\$	80.000	\$	72.125	90.2
38100 10	Meters, New	ŝ	2.099.393	\$	1.932.554	92.1
38100 20	Meters, Replacement	\$	77.224	\$	134.039	173.6
38100 98	Meters Retire	\$	(300)	\$	(339.362)	<del>         </del>
38300 10	House Reas.new	\$	472.064	\$	406.094	86.0
38300 98	House Reas Retire	\$	(300)	\$	-	0.0
38500 20	Ind M&R Equip Public Improve	\$	308,825	\$	217,654	70.5
38500 91	Industrial Measuring and	\$	(150,000)	\$	-	0.0
38500 98	Industrial M&R Retire	\$	(2,574)	\$	3,138	<del>#####</del>
38700	Other Equipment	\$	780	\$	746	95.6
39009	Improvements- Leased Premises	\$	11,999	\$	44,443	370.4
39103	Office Machines	\$	17,913	\$	13,372	74.6
39200	Vehicles	\$	56,492	\$	52,398	92.8
39200 98	Vehicles Retire /Salvage	\$	(5,000)	\$	(6,640)	132.8
39400	Tools, Shop Equip	\$	1,380	\$	190	13.8
39705	Communication Equip- Telemeteri	\$	6,135	\$	6,790	110.7
39906	PC Hardware	\$	6,204	\$	29,644	477.8
39907	PC Software	\$	5,005	\$	613	12.2

	Total	\$	2,985,240	\$	2,567,798	86.0
v	VKG Technical Services					
35200	Wells	\$	26 000	\$	_	0.0
39103	Office Machines	\$	20,000	\$	20 295	74 7
39200	Vehicles	ŝ	58 814	₩ \$	52 929	90 0
39200 98	Vehicles Retire / Salvage	ŝ	(2,500)	ŝ	(3 448)	137.9
39905	Mainframe Hardware	ŝ	3.001	\$	7,407	246.8
39906	PC Hardware	ŝ	97,430	\$	98.601	101.2
39907	PC Software	ŝ	17 048	\$	14 184	83.2
39908	Application Software	ŝ	930	\$	1 032	111.0
99800	Retirements & Salvage	\$	-	\$	(1.920)	<del>#####</del>
	<b>T</b>	•	007.004	•	(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
	I otal	\$	227,894	\$	189,080	83.0
Μ	ladisonville Operations					
36700 40	Transm Rectifier & Groundbed	\$	1,900	\$	-	0.0 Deferred
36701	Mains - Steel	\$	18,001	\$	16,846	93.6
36900 10	M & R Station Equip	\$	8,530	\$	41,998	492.4
36900 20	M & R Station	\$	25,842	\$	12,531	48.5
37500	Structures and Improve	\$	-	\$	2,724	<del>#####</del>
37600	Mains - Cathodic Protection	\$	22,100	\$	19,380	87.7
37601 10	Mains- Stl System Improve	\$	14,507	\$	8,531	58.8
37601 10	Hudson Foods	\$	-	\$	102,646	<del>#####</del>
37601 20	Mains- Public (Hwy) Relocations	\$	190,986	\$	38,642	20.2
37601 30	Mains- Leakage	\$	336,296	\$	213,195	63.4
37601 81	Main Ext Forfeiture	\$	(70,280)	\$	(19,025)	27.1
37601 83	Mains - Pub Works Reimburse	\$	-	\$	(5,864)	<del>#####</del>
37601 98	Mains - Steel Retire	\$	(29,403)	\$	(38,249)	130.1
37602 10	Mains- PE Sys Improve	\$	239,406	\$	198,157	82.8
37602 20	Mains- PE Public Relocations	\$	-	\$	1,461	<del>#####</del>
37602 30	Mains - Plastic Leakage	\$	25,080	\$	103,863	414.1
37602 98	Mains - Plastic Retire	\$	220	\$	2,512	<del>#####</del>
37800	Meas and Reg. Sta Equip	\$	50,824	\$	35,849	70.5
38000 10	Services- New	\$	550,000	\$	513,082	93.3
38000 20	Services Replace	\$	29,980	\$	-	0.0
38000 30	Services- Leakage	\$	192,771	\$	135,944	70.5
38000 98	Services Retirement	\$	252	\$	62,350	
38200 10	Meter Installations New	\$	54,629	\$	123,632	226.3
38200 20	Meter Set Replacement	\$	19,576	\$	63,362	323.7
38200 98	Meter Retire	\$	18,976	\$	16,021	84.4
38700	Other Equipment	\$	3,030	\$	5,130	169.3
39009	Improvements-	\$	1,200	\$	1,421	118.4
39100	Office Furniture & Equipment	\$	1,225	\$	2,541	207.4
39103	Office Machines	\$	1,201	\$	7,961	662.9
39200	Vehicles	\$	46,083	\$	43,475	94.3

DR 28		P	age 15 of 28			
39200 98	Vehicles Retire / Salvage	\$	(5,000)	\$	(6,800)	136.0
39400	Tools, Shop Equip	\$	22,581	\$	23,081	102.2
39400 98	Tools, Shop Equip	\$	8	\$	(800)	<del>#####</del>
39700	Communication Equip	\$	1,215	\$	419	34.5
39701	Communication Equip - MO	\$	2,800	\$	2,190	78.2
39905	Mainframe Hardware	\$	3,791	\$	2,659	70.1
99800	Retirements	\$	-	\$	(4,868)	<del>#####</del>
	Total	\$	1,778,327	\$	1,725,997	97.1
P	aducah Operations					
36700 40	Transm Mains Cathodic Protectio	\$	500	\$	-	0.0 Deferred
36900 20	M&R Sta Public Improvement	\$	167,803	\$	54,124	32.3
37600 40	Mains- Cathodic Protection	\$	30,000	\$	10,069	33.6
37601 10	Mains- System Improve	\$	38,384	\$	6,480	16.9
37601 20	Mains- Public Improve	\$	420,200	\$	72,808	17.3
37601 20	Husbands Rd Relocation	\$	212,800	\$	164,884	77.5
37601 20	West Paducah 6" HP Feeder	\$	1,400,000	\$	793,822	56.7
37601 30	Mains- Leakage	\$	149,650	\$	95,035	63.5
37601 81	Main Ext Forfeiture	\$	(36,100)	\$	(26,633)	73.8
37601 91	Mains - Stl Reimburse	\$	(158,172)	\$	(22,699)	14.4
37601 98	Mains - Steel Retirement	\$	3,956	\$	17,047	430.9
37602 10	Mains - Plastic Sys Improv	\$	452,769	\$	146,032	32.3
37602 10	Grand Lakes Subdivision	\$	97,231	\$	85,326	87.8
37602 20	Mains - Plastic	\$	141,668	\$	117,389	82.9
37602 30	Mains - Plastic Leakage	\$	42,555	\$	5,333	12.5
37602 91	Mains - Plastic	\$	(88,747)	\$	(25,307)	28.5
37602 98	Mains - Plastic Retirements	\$	2	\$	961	<del>#####</del>
37800 10	Meas. And Reg. Sta Equip	\$	-	\$	9,321	#####
37800 20	Meas. And Reg. Sta Equip	\$	-	\$	5,493	#####
37900 20	City Gate M&R Sta Chk Sta	\$	-	\$	209	#####
38000 10	Services- New	\$	517,000	\$	521,482	100.9
38000 30	Services-Leakage	\$	50,000	\$	37,001	74.0
38000 40	Services	\$	7,000	\$	-	0.0
38000 98	Services	\$	3,000	\$	35,102	#####
38200 10	Meter Installations	\$	33,000	\$	102,051	309.2
38200 20	Meter Replacements	\$	29,000	\$	29,679	102.3
38200 98	Meter Installations	\$	8,000	\$	11,259	140.7
38700	Other Equipment	\$	8,954	\$	6,529	72.9
39009	Improvements- Leased Premises	\$	9,801	\$	13,782	140.6
39100	Office Furniture & Equipment	\$	-	\$	2,486	<del>#####</del>
39103		\$	1,001	\$	808 155 050	80.7 04 F
39200	Vehicles	\$	184,568	<b>Þ</b>	105,953	84.5
39400	Loois, Shop Equip	\$	8,211	\$	0,002	/ Y.Ŏ
39600	Power Operated Equip	\$	2,601	\$ \$	1,007	JO.1
39605	vveiders	\$	4,501	\$	3,000	00.9 07 0
39701	Communication Equip - MO	\$	0,502	\$	0,309	97.0 ######
99800	Retirements & Salvage	\$	-	Þ	(7,795)	<del>######</del>

Page 16 of 28

	Total	\$	3,747,638	\$ 2,435,824	65.0
Bo	wling Green Operations				
36700 40	Transm Mains- Cathodic Protec	tio \$	2,000	\$ -	0.0
36701 20	TP Mains - Stl Public Improv	\$	-	\$ 20,986	<del>####</del>
36900	Meas and Reg. Station	\$	30,000	\$ 34,743	115.8
37600 40	Mains- Cathodic Protection	\$	20,000	\$ 10,978	54.9
37601 10	Mains System Improv	\$	1,032,000	\$ 639,863	62.0
37601 20	US 68-80 Relocation	\$	77,166	\$ 82,409	106.8
37601 30	Mains- Leakage	\$	175,000	\$ 288,954	165.1
37601 81	Mains - Steel	\$	(68,998)	\$ (65,733)	95.3
37601 91	Mains - Steel	\$	(197,571)	\$ _	0.0
37601 98	Mains - Steel Retirements	\$	8,440	\$ 26,012	308.2
37602	Mains - Plastic	\$	-	\$ 52,704	<del>#####</del>
37800	M&R Sta Sys Improv	\$	-	\$ 3,166	<del>#####</del>
37900 20	City Gate M&R Sta Equip Pub 1	m \$	14,997	\$ -	0.0 Deferred
38000 10	Services- New	\$	700,000	\$ 413,288	59.0
38000 30	Services- Leakage	\$	150,000	\$ 135,031	90.0
38000 98	Services Retirements	\$	5,000	\$ 62,588	<del>#####</del>
38200 10	Meter Installations	\$	100,000	\$ 147,740	147.7
38200 20	Meter Replacements	\$	100,000	\$ 42,743	42.7
38200 98	Meter Retirements	\$	-	\$ 12,333	<del>#####</del>
38500 10	Ind M&R Equip Sys Improv	\$	40,000	\$ -	0.0
38700	Other Equipment	\$	8,805	\$ 11,124	126.3
39000	Structures & Improvement	\$	-	\$ 2,666	<del>######</del>
39009	Improvements - Leased Pre	\$	3,749	\$ -	0.0
39100	Office Furniture & Equipment	\$	44,808	\$ -	0.0
39103	Office Machines	\$	10,241	\$ 4,342	42.4
39200	Vehicles	\$	226,273	\$ 198,749	87.8
39200 98	Vehicles Retire / Salvage	\$	(15,000)	\$ (23,119)	154.1
39400	Tools, Shop Equip	\$	44,537	\$ 19,275	43.3
39604	Backhoes	\$	59,360	\$ 58,574	98.7
39700	Communication Equipment	\$	2,575	\$ 680	26.4
39800	Misc. Equipment	\$	3,339	\$ -	0.0
39905	Mainframe Hardware	\$	6,882	\$ 5,318	77.3
99800	Retirements & Salvage	\$	-	\$ (7,960)	<del>#####</del>
	Total	\$	2,583,603	\$ 2,177,453	84.3

## Danville Operations

36700 40	Trans Mains-Cathodic Protection	\$ 10,875	\$ 7,761	71.4
37601 20	TP Mains - Stl Relocate	\$ -	\$ 10,064	<del>#####</del>
36701 30	12" L-133 Replacement	\$ 299,260	\$ 249,071	83.2
36701 98	TP Mains - Steel Retirement	\$ 400	\$ 132	33.0
36900	Meas. And Reg. Sta	\$ 1,766	\$ 3,098	175.4 Deferred
37600 40	Mains- Cathodic Protection	\$ 25,000	\$ 16,782	67.1
37601 10	Mains- Sys Improve	\$ 4,500	\$ 6,620	147.1
37601 20	Mains- Public (Hwy) Relocat	\$ 4,700	\$ 42,455	903.3
37601 30	Mains- Leakage	\$ 75,000	\$ 117,886	157.2
37601 81	Main Ext Forfeiture	\$ (42,804)	\$ (14,577)	34.1
37601 98	Mains - Steel Retirement	\$ -	\$ 3,049	<del>#####</del>
37602 10	Mains - Plastic Sys Improv	\$ 380,000	\$ 279,225	73.5
37602 30	Mains - Plastic Leakage	\$ -	\$ 226,986	######
37602 91	Mains - Plastic	\$ -	\$ (1,028)	<del>#####</del>
37800 20	Meas & Reg. Sta	\$ 25,001	\$ -	0.0
38000 10	Services- New	\$ 553,850	\$ 620,985	112.1
38000 30	Services- Leakage	\$ 100,080	\$ 76,775	76.7
38000 98	Services Retirement	\$ -	\$ 25,707	<del>#####</del>
38200 10	Meter Installations	\$ 34,980	\$ 141,697	405.1
38200 20	Meter Replacements	\$ 51,930	\$ 86,091	165.8
38200 98	Meter Retirement	\$ 1	\$ 12,409	<del>#####</del>
38500 10	Ind M&R Meter	\$ 40,000	\$ -	0.0
387000	Other Equipment	\$ 5,323	\$ 8,289	155.7
39003	Improvements	\$ 2,700	\$ 2,511	93.0
39009	Improv Leased Premises	\$ 11,720	\$ 6,379	54.4
39100	Office Furniture & Equip- Danville	\$ 36,165	\$ 1,473	4.1
39103	Office Machines- Danville	\$ 12,896	\$ 11,012	85.4
39200	Vehicles	\$ 261,452	\$ 239,445	91.6
39200 98	Vehicles Retire /Salvage	\$ (25,000)	\$ (26,352)	105.4
39400	Tools, Shop Equip	\$ 31,509	\$ 27,490	87.2
39600	Power Op Equip- Forklift	\$ (999)	\$ (400)	40.0
39604	Backhoes	\$ (3,499)	\$ (5,900)	168.6
39700	Communication Equip	\$ 900	\$ 812	90.2
39701	Communication Equip - MO	\$ 3,990	\$ 3,410	85.5
39905	Mainframe Hardware	\$ 3,491	\$ 2,659	76.2
99100	Public Works Reimbursement	\$ -	\$ (18,269)	<del>#####</del>
99800	Retirements & Salvage	\$ -	\$ (1,200)	<del>#####</del>
	Total	\$ 1,905,187	\$ 2,162,547	113.5
v	VKG Overheads			<del>#####</del>
39900		\$ 0	\$ -	
	Grand Total	\$ 17,770,373	\$ 14,254,212	80.2

ł

## WKG CAPITAL BUDGET PROJECTS FY 1997 Completion Percentage

Budget No. Description Operating Area		Budget Amount		Expendi	ture Amount	Completion %	
v	VKG Company Office						
39100	Office Furniture & Equipment	\$	917	\$	875	95.4	
39103	Office Machines	\$	8.879	\$	2,754	31.0	
39200	Vehicles	\$	24.577	\$	43.833	178.3	
39700	Communication Equip	\$	1.571	\$	1.134	72.2	
39906	PC Hardware	Ś	17.736	\$	25,183	70.4	
39907	PC Software	\$	8,974	\$	14,283	159.2	
	Total	\$	62,654	\$	88,063	140.6	
С	wensboro Operations						
36700 40	Trans Mains Cathodic Protecti	on \$	2,594	\$	-	0.0 Deferred	
36701 83	US 60 E Relocation Reimburse	e \$	-	\$	(62,790)	<del>#####</del>	
36900 20	Fairview Dr Reg Sta	\$	-	\$	1,417	<del>######</del>	
37400 20	Land Mosleyville PS	\$	-	\$	3,276	<del>#####</del>	
37600 40	Mains Cathodic Protection	\$	45,635	\$	17,612	38.6	
37600 40	Mains CP Retirement	\$	176	\$	6	3.4	
37601 10	Blanket Mains- Stl Sys Improv	\$	48,113	\$	24,107	50.1	
37601 20	Newbolt Rd. 4" HP	\$	180,004	\$	254,483	141.4	
37601 30	Blanket Stl Mains Leakage	\$	76,635	\$	58,413	76.2	
37601 98	Mains Stl Retirement	\$	11,327	\$	10,381	91.6	
37602 10	Mains PE Sys Improv	\$	315,321	\$	320,783	101.7	
37602 20	Mains PE Whitesville KY 54 Re	eloc \$	119,109	\$	220,654	185.3	
37602 30	Mains PE Leakage	\$	26,528	\$	23,308	87.9	
37602 81	Main Ext Forfeitures	\$	(20,515)	\$	(25,461)	124.1	
37602 98	Main PE Retirement	\$	5,094	\$	5,179	101.7	
37800 20	M&R Sta Yager Mat'l	\$	-	\$	6,745	<del>#####</del>	
37900 20	Owensboro TB # 5 Reg Sta	\$	131,000	\$	90,618	69.2	
37900 98	Retire Dublin Reg Sta	\$	2,000	\$	1,950	97.5	
38000 10	Services New	\$	532,122	\$	557,316	104.7	
38000 30	Services Leakage	\$	196,500	\$	242,968	123.6	
38000 98	Services Retirement	\$	89,375	\$	98,948	110.7	
38200 10	Meter Installations New	\$	44,016	\$	245,396	557.5	
38200 20	Meter Installations Replace	\$	130,214	\$	138,913	106.7	
38200 98	Meter Retirements	\$	16,425	\$	30,998	188.7	
38700	Other Equipment	\$	10,472	\$	11,098	106.0	
39000	Structures & Improv Bon Harbo	or \$	10,136	\$	9,639	95.1	
39200	Vehicles	\$	331,370	\$	271,336	81.9	
39200 98	Vehicles Retire /Salvage	\$	(27,500)	\$	(6,200)	22.5	

DR 28			Page 19 of 28							
39400	Tools, Shop Equipment	\$	4,044	\$	3,306	81.8				
39700	Communication Equip	\$	639	\$	527	82.4				
	Total	\$ 2	,280,834	\$ 2	,554,925	112.0				

Budget N	lo. Description E Operating Area	Budget ,	Amount	Expend	iture Amount \	Completion %	
	Owensboro Storage & Transmiss	ion					
35100 10	Structures & Improvements	\$	7,370	\$	6,550	88.9	
35100 20	Structures & Improvements	\$	937	\$	950	101.4	
35100 98	Structures Retirements	\$	800	\$	589	73.6	
35200 20	Well Workovers	\$	678,551	\$	472,419	69.6	
35200 40	Well Workovers	\$	12,073	\$	9,844	81.5	
3520098	Well Work Retirement	\$	23,993	\$	1,565	6.5	
35400 20	Compressor Station Equip	\$	32,750	\$	26,294	267.1	
35400 98	Compressor Station Equip Re	tire \$	7,682	\$	941	3.6	
35600 20	Purification Equip	\$	6,578	\$	15,384	233.9	
35600 98	Purification Equip Retirement	\$	-	\$	99	<del>#####</del>	
36600 20	Structures ANR Intertie	\$	28,958	\$	19,721	68.1	
36700 40	Cathodic Protection	\$	-	\$	4,106	<del>######</del>	
36701 20	Madisonville 8" Loop	\$	949,147	\$	68,706	7.2	
36701 20	ANR Bon Harb/ Kirkwood Inte	rtie \$	73,796	\$	91,159	123.5	
36701 30	Madisonville 6" Replacement	\$	37,860	\$	41,417	109.4	
36701 98	Madisonville 6" Retire	\$	1,080	\$	525	48.6	
36900 20	M&R Sta-ANR Intertie	\$	578,005	\$	211,020	36.5	
37100	Other Equipment	\$	9,982	\$	9,706	97.2	
37601 20	Replace Cadiz TB	\$	5,879	\$	1,740	29.6	
37601 98	Replace Cadiz TB Retirement	\$	4,834	\$	860	17.8	
38700	Other Equipment	\$	4,176	\$	4,374	104.7	
39200	Vehicles	\$	79,564	\$	74,719	93.9	
39200 98	Vehicles -Retire-Salvage	\$	(7,500)	\$	-	0.0	
39400	Tools Shop Equip	\$	18,697	\$	17,542	93.8	
39400 98	Tools Shop Equip Retire/Salva	age \$	(999)	\$	-		
39600	Power Operated Equipment	\$	37,966	\$	34,426	90.7	
	Total Resp C	tr \$	2,592,179	\$	1,114,656	43.0	

WKG	Measurement	Center
-----	-------------	--------

Combined with WKG Tech Services

35500 20	M&R Equipment	#####
37000	Commication Equip	######
38100 10	Meters, New	######
38100 20	Meters, Replacement	######
38300 10	House Regs, new	######

38300 20	House Regs- Replacement				<del>#####</del> #	:
38500 20	Ind M&R Equip Public Improve				<del>######</del>	1
39009	Improvements- Leased Premises				<del>######</del>	1
39200	Vehicles				<del>#####</del> #	
39400	Tools, Shop Equip				<del>######</del>	
39700	Communication Equip				<del>######</del>	
39705	Communication Equip- Telemeteri	ng			<del>######</del>	
39905	Mainframe Hardware				<del>#####</del> #	
39906	PC Hardware				× <del>#####</del>	
39907	PC Software				<del>              </del>	
99800	Retirements & Salvage				######	
	Total	\$	-	\$ -		
v	VKG Technical Services					
35200 10	Wells	\$	24,314	\$ 23,676	97.4	
39100	Office Furniture & Equipment	\$	9,259	\$ 9,042	97.7	
39103	Office Machines	\$	17,679	\$ 12,097	68.4	
39200	Transportation Equipment	\$	85,329	\$ 46,270	54.2	
39200	Vehicle Retire / Salvage	\$	(7,500)	\$ (6,719)	89.6	
39700	Communication Equipment	\$	459	\$ 457	99.5	
39705	Commun EquipTelemeter	\$	1,048	\$ -	0.0	
39800	Miscellaneous Equipment	\$	-	\$ 2,312	<del>#####</del>	
39906	PC Hardware	\$	113,809	\$ 93,953	82.6	
39907	PC Software	\$	10,207	\$ 13,909	136.3	
39908	AM/FM Software	\$	262,731	\$ -	0.0	Deferred
	Total	\$	517,335	\$ 194,996	37.7	
N	adisonville Operations					
36700 40	Transm Cathodic Protection	\$	49,009	\$ 19,288	39.4	
36701 10	TP Mains - WK Training Ctr 4"	\$	-	\$ 343,114	<del>######</del>	
36701 30	TP Mains Leakage	\$	-	\$ 1,170	<del>#####</del>	
36900 10	M & R Station Equip Rev	\$	-	\$ 88,436	<del>#####</del>	
36900 20	M & R Station Equip Non-Revenu	\$	13,310	\$ 235	1.8	
37600 40	Cathodic Protection	\$	24,235	\$ 19,493	80.4	
37601 20	Mains- Public (Hwy) Relocations	\$	252,135	\$ -	0.0	Deferred
37601 30	Mains- Leakage	\$	183,400	\$ 171,634	93.6	
37601 83	Main Relocat Reimbursemenrt	\$	(1)	\$ (7,000)	#######	
37601 98	Main Retirements	\$	11,997	\$ 6,190	51.6	
37602 10	Mains PE Revenue	\$	181,390	\$ 139,904	77.1	
37602 20	Mains- PE Public (Hwy) Relocatio	\$	-	\$ 1,594	#####	
37602 30	Mains Install PE for Leakage	\$	34,060	\$ 71,196	209.0	
37602 81	Main Ext Forfeiture	\$	(30,360)	\$ (24,964)	82.2	
37602 83	Main Reimbursement	\$		\$ (2,202)	#######	
37602 98	Main PE Retirements	\$	10,000	\$ 3,113	31.1	
37800 20	M&R Sta Equipment	\$	22.869	\$ 28.979	126.7	

DR 28

Page 20 of 28

DR 28		Pa	ige 21 of 28			
37800 98	M&R Sta Retirement	\$	400	\$ _	0.0	
37900 10	M&R Sta Equip- City Gate	\$	-	\$ 19.249	<del>#####</del>	
37900 20	City Gate M&R Sta Pub Impr	\$	1,703	\$ 1.863	109.4	
38000 10	Services- New	Ś	556,292	\$ 515,921	92.7	
38000 20	Services Relocate	\$	39,274	\$ -	0.0 Deferre	ed
38000 30	Services- Leakage	\$	170,837	\$ 94,950	55.6	
38000 98	Services- Retirements	\$	38,481	\$ 45,011	117.0	
38200 10	Meter Installations New	\$	33,353	\$ 123,789	371.1	
38200 20	Meter Set Replacement	\$	113,001	\$ 94,158	83.3	
38200 98	Meter Set Retirement	\$	14,760	\$ 26,373	178.7	
38700	Other Equipment	\$	5,359	\$ 7,774	145.1	
39009	Improvements- Leased Mad'vI WH	\$	3,859	\$ 3,937	102.0	
39100	Office Furniture & Equipment	\$	2,758	\$ 1,553	56.3	
39103	Office Machines	\$	4,368	\$ 3,108	71.1	
39200	Vehicles	\$	262,787	\$ 238,980	90.9	
39200 98	Vehicles Retire / Salvage	\$	(25,000)	\$ (18,168)	72.7	
39400	Tools, Shop Equip	\$	24,034	\$ 24,521	102.0	
39400 98	Tools, Shop Equip	\$	4	\$ 99	<del>#####</del>	
99800	Retirements	\$	-	\$ (250)	<del>#####</del>	
	Total	\$	1,998,314	\$ 2,043,046	102.2	
F	Paducah Operations					
36701 20	TP Mains- Relocation	\$	-	\$ 22,326	<del>#####</del>	
36701 98	TP Mains- Retirement	\$	-	\$ 1,115	<del>#####</del>	
36900 20	M&R Sta	\$	125,734	\$ 138,655	110.3	
37600 40	Cathodic Protection	\$	59,493	\$ 19,571	32.9	
37600 83	Mains Reimbursement	\$	-	\$ (6,350)	######	
37600 40	C Prot Retirement	\$	30	\$ (890)	<del>#####</del>	
37601 20	Mains Relocation (Hwy)	\$	304,743	\$ 410,074	134.6	
37601 30	Mains	\$	83,407	\$ 69,890	83.8	
37601 81	Main Extension Forfeiture	\$	(6,866)	\$ (3,750)	54.6	



3760 3760 3760 (6,866) 37601 81 Main Extension Forfeiture \$ 37601 83 Main Public Works Reimburseme \$ \$ 4,202 Main Retirement 37601 98 37602 10 Mains PE Revenue \$ 410,082 Mains- PE Public (Hwy) Relocatio \$ 157,746 37602 20 Mains Install PE for Leakage (CI) 20,100 37602 30 \$ Main Ext Forfeiture \$ 37602 81 (59,914)\$ Main Reimbursement 37602 83 \$ 1,500 37602 98 Main PE Retirements \$ 37800 10 M&R Sta Equip \$ 1,384 37900 20 M&R Sta City Gate \$ 38000 10 Services-New 442,518 38000 30 Services-Leakage \$ 42,051 \$ Services- Retirements 30,600 38000 98

38200 20 38200 98

38200 10

Meter, Reg Installations

Meter, Reg Retirements

Meter, Reg Replacements

\$

\$

\$

39,169

84,364

13,050

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

\$

(45,738)

10,292

348,645

151,967

23,196

(52, 386)

(47, 148)

1,992

6,616

603,179

98,961

55,317

96,081

25,552

10,125

56

######

244.9

85.0

96.3

115.4

87.4

######

132.8

######

136.3

235.3

180.8

245.3

30.3

77.6

4.0

38500 20	Industrial Measuring	\$	3,930	\$	5,087	129.4
38600 20	Other Prop. On Customer's	\$	-	\$	11,799	<del>######</del>
38600 83	Other Prop. On Customer's	\$	-	\$	(6,106)	<del>#####</del>
38700	Other Equipment	\$	8,407	\$	8,049	95.7
39009	Improvements- Leased Premises	\$	12,791	\$	11,282	88.2
39200	Vehicles	\$	158,879	\$	122,694	77.2
39200 98	Vehicles Retire /Salvage	\$	(17,500)	\$	(8,180)	46.7
39400	Tools, Shop Equip	\$	9,757	\$	14,878	152.5
39600	Power Operated Equip	\$	21,592	\$	22,524	× 104.3
39700	Comm Equip Tel	\$	347	\$	-	0.0
	Total	\$	1,951,596	\$	2,119,373	108.6
	Bowling Green Operations					
36700 40	Transm Mains- Cathodic Protectio	\$	2 000	\$	-	0.0 Deferred
36900 20	M&R Sta	ŝ	39 333	\$	20,209	51.4
36900 98	M&R Sta Retire	ŝ	500	\$	534	106.8
37600 40	Mains- Cathodic Protection	ŝ	15 065	\$	19.962	132.5
37601 10	Mains System Improv	\$	102 653	\$	77.541	75.5
37601 30	Mains-Leakage	\$	172 744	\$	330,472	191.3
37601.81	Main Extension Forfeiture	ŝ	(11, 464)	\$	(10,167)	88.7
37601 98	Main Retirements	\$	9.525	ŝ	27 765	291.5
37602 10		ŝ	675,000	÷ S	385 791	57.2
37602 20	Mains- PE Public (Hwy) Relocatio	ŝ	5 961	\$	4 734	79.4
37602.30	Mains Install PE for Leakage	\$	77 830	\$	181 766	233.5
37602 81	Main PE Ext Forfeiture	ŝ	(45,156)	\$	(35,527)	78.7
37602.98	Main Retirements	\$	2,563	\$	1.252	48.8
37800 10	Meas & Reg Sta Fouip	ŝ	2,000	\$	6,697	325.6
38000 10	Services-New	\$	548,450	\$	541,209	98.7
38000 30	Services-Leakage	ŝ	187,330	\$	306.483	163.6
38000 30	Services-Retire	\$	50,000	\$	114,458	228.9
38200 10	Meter Installations	\$	62,225	Ś	238.910	383.9
38200 20	Meter Replacements	\$	97 268	\$	117,750	121.1
38200.98	Meter Retirements	ŝ	13 500	\$	17.613	130.5
38500 10	Ind M&R Fauin Sys Improv	\$	-	\$	12.542	######
38700	Other Equipment	ŝ	2 557	\$	3,637	142.2
39000	Structures and Improvements	ŝ	15 720	\$	16.395	104.3
39009	Improvements - lease pre	ŝ	5 351	\$	5,605	104.8
39100	Office Eurniture & Equipment	\$	54,886	\$	-,	0.0 Deferred
39103	Office Machines	\$	4,109	\$	1,770	43.1
39200	Vehicles	\$	156,065	\$	162,935	104.4
39200 98	Vehicles Rtire / Salvage	\$	(17,500)	\$	(1.900)	10.9
39400	Tools, Shop Equip	\$	5.761	\$	6,529	113.3
39700	Communication Equipment	\$	563	\$	,	0.0
39702	Communication Equip- Radios	\$	•	\$	5,988	<del>#####</del>
39702 98	CommEquip- Radios Retire	\$	-	\$	205	<del>#####</del>

DR 28

Page 22 of 28



Tools, Shop Equip	\$	
Communication Equipment	\$	
Communication Equip- Radios	\$	
CommEquip- Radios Retire	\$	
Tatal	¢	2
rotal	Φ	- <b>4</b> ,4

\$ 2,234,896

\$ 2,561,158

114.6



36600 10	Structures & Improvements	\$ 1,850	\$	1,809	97.8	
36700 40	Trans Mains-Cathodic Protection	\$ 3,900	\$	-	0.0	Deferred
36701 30	Replace 12" L-133	\$ 311,807	\$	7,379	2.4	Deferred
36701 98	Retire 12" L-133	\$ 512	\$	-	0.0	Deferred
39900 20	Meas. & Reg	\$ 71,157	\$	-	0.0	Deferred
39900 98	Meas. & Reg Retire	\$ 3,240	\$	- \	0.0	Deferred
37600 40	Mains- Cathodic Protection	\$ 32,500	\$	37,383	115.0	
37600 98	Mains- Cathodic Protection	\$ -	\$	87	<del>#####</del>	
37601 10	Mains- Stl Sys Improve	\$ 2,000	\$	147,639	<del>          </del>	
37601 20	Mains- Public (Hwy) Relocat	\$ 12,556	\$	13,075	104.1	
37601 30	Mains- Leakage	\$ 87,508	\$	172,237	196.8	
37601 83	Mains- Pub Works Reimbursemen	\$ (7,840)	\$	(25,149)	320.8	
37601 98	Mains- Retirement	\$ 2,488	\$	11,113	446.7	
37602 10	Mains PE Revenue	\$ 405,501	\$	262,933	64.8	
37602 20	Mains- PE Public (Hwy) Relocatio	\$ 75,878	\$	16,848	22.2	
37602 30	Mains Install PE for Leakage	\$ 15,000	\$	(65,441)	<del>######</del>	
37602 81	Main PE Ext Forfeiture	\$ (25,255)	\$	(16,645)		
37602 83	Main PE Pub Works Reimburse	\$ (50,175)	\$	-		
37602 98	Main Retirements	\$ 680	\$	197		
37800 10	Meas. & Reg	\$ -	\$	316	<del>#####</del>	
38000 10	Services- New	\$ 649,695	\$	651,909	100.3	
38000 30	Services- Leakage	\$ 93,565	\$	137,369	146.8	
38000 98	Services- Retirements	\$ 33,425	\$	45,574	136.3	
38200 10	Meter Installations	\$ 63,273	\$	147,514	233.1	
38200 20	Meter Replacements	\$ 115,542	\$	137,182	118.7	
38200 98	Meter Retirements	\$ 13,050	\$	19,035	145.9	
38700	Other Equipment	\$ 16,517	\$	14,720	89.1	
39009	Improv Leased Premises	\$ 4,441	\$	3,856	86.8	
39100	Office Furniture & Equip- Danville	\$ 37,967	\$	3,115	8.2	
39103	Office Machines- Danville	\$ 1,987	\$	2,092	105.3	
39200	Vehicles	\$ 251,375	\$	239,439	95.3	
39200 98	Vehicles Retire /Salvage	\$ (27,500)	\$	(12,888)	46.9	
39400	Tools, Shop Equip	\$ 8,498	\$	9,098	107.1	
39600	Power Op Equip- Forklift	\$ 18,152	\$	15,992	88.1	
39603	Ditchers & Trenchers	\$ 61,681	\$	50,842	82.4	
39603 98	Ditchers & Trenchers Retire	\$ (1,100)	\$	(4,000)	363.6	
	Total	\$ 2,283,875	\$ 2	2,024,629	88.6	
N	/KG Overheads					
39900		\$ 2,673,677	\$ 2	2,384,377	89.2	
	Grand Total	\$ 16,595,360	\$1	5,085,222	90.9	



## WKG CAPITAL BUDGET PROJECTS FY 1998 Completion Percentage

Budget No C	. Description I Operating Area	Budget A	mount	Expendit	ure Amount	Completion %
v	VKG Company Office				Ň	
39200 98	Vehicles Retire /Salvage	\$	-	\$	(10,392)	<del>#####</del>
39700	Communication Equip	\$	9,700	\$	9,392	96.8
39906	PC Hardware	\$	-	\$	25,978	<del>#####</del>
39907	PC Software	\$	900	\$	-	0.0
39908	Application Software	\$	• –	\$	40,846	#####
	Total	\$	10,600	\$	65,824	621.0
С	wensboro Operations					
36700	Trans Mains Cathodic Protec	tion \$	1,000	\$	-	0.0
36701	TP Mains Steel- Public Reloc	; \$	7,650	\$	-	0.0 Deferred
37600 40	Mains Cathodic Protection	\$	33,864	\$	26,661	78.7
37600 98	Cathodic Protection Retireme	ent \$	50	\$	19	38.0
37601 10	Mains - Stl Sys Improv	\$	112,829	\$	92,943	82.4
37601 20	Mains - Stl- Pub Improv	\$	100,500	\$	64,267	63.9
37601 30	Mains - Stl - Repl-Leakage	\$	48,600	\$	72,868	149.9
37601 83	Mains - Stl- Pub Works Reim	burs \$	-	\$	(18,877)	<del>#####</del>
37601 98	Mains - Steel- Retirements	\$	20,836	\$	14,860	71.3
37602 10	Mains - Plastic Sys Improv	\$	215,000	\$	195,379	90.9
37602 20	Mains - Plastic Pub Improv	\$	152,959	\$	82,516	53.9
37602 30	Mains - Plastic Leakage	\$	19,200	\$	15,360	80.0
37602 81	Mains - Plastic Reimburseme	ents \$	(14,898)	\$	(13,548)	90.9
37602 83	Mains - Plastic Reimburseme	ents \$	(3)	\$	(7,195)	<del>#####</del>
37602 98	Mains - Plastic Retirements	\$	900	\$	3,061	340.1
37800 20	Meas and Reg. Sta Equipme	nt \$	4,600	\$	5,369	116.7
37800 98	M&R Sta Retirement	\$	50	\$	39	78.0
38000 10	Services System Improveme	nt \$	400,649	\$	409,377	102.2
38000 30	Services Leakage	\$	87,000	\$	202,722	233.0
38000 98	Services Retirement	\$	45,000	\$	103,344	229.7
38200 10	Meter Installations New	\$	58,179	\$	172,671	296.8
38200 20	Meter Installations Public Imp	prov \$	75,640	\$	82,883	109.6
38200 98	Meter Retirements	\$	19,395	\$	45,932	236.8
38700	Other Equipment	\$	1,442	\$	1,398	96.9
39103	Office Machines	\$	582	\$	522	89.7
39200	Vehicles	\$	-	\$	2,393	<del>#####</del>
39200 98	Transportation Equipment Re	etire \$	-	\$	(26,645)	#####
39400	Tools, Shop Equipment	\$	3,763	\$	3,697	98.2



```
DR 28
```

Page 25 of 28

	Total	\$	1,394,787	\$ 1,532,016	109.8
--	-------	----	-----------	--------------	-------

Budget No. C	. Description B Operating Area	Budget	Amount	Expendit	ure Amount	Expenditure %
С	wensboro Storage & Transmis	sion				N.
36701 20 39200	Spencer Co 4" Relocation Vehicles	\$ \$	-	\$ \$	4,877 (6,689)	<del>#####</del> ######
	Total Resp	Ctr \$	-	\$	(1,812)	<del>#####</del>
v	VKG Measurement Center					
38500 20	Ind M&R Equip Public Improv	ve \$	-	\$	5,529	<del>#####</del>
38500 83	Ind M&R Equip Reimbursem	ent \$	-	\$	(19,234)	#####
	Total	\$	-	\$	(15,517)	<del>#####</del>
W	/KG Technical Services					
35100	Structures & Improv	\$	3,501	\$	3,072	87.7
36900 20	Measuring & Reg Station Equ	\$ qiu	80,210	\$	93,680	116.8
36900 98	Measuring & Reg Station Equ	uip \$	1	\$	1,309	<del>#####</del>
37601	Mains - Steel	\$	0	\$	871	<del>#####</del>
38100 10	Meters	\$	847,275	\$	642,338	75.8
38100 98	Meters	\$	106,500	\$	38,535	36.2
38300 10	House Regulators	\$	331,961	\$	134,766	40.6
38500 20	Industrial Measuring and	\$	280,776	\$	179,290	63.9
38500 83	Industrial Measuring and	\$	(150,000)	\$	(23,412)	15.6
38500 98	Industrial Measuring and	\$	6,000	\$	2,741	45.7
38700	Other Equipment (Distrib)	\$	83,790	\$	-	0.0
39200	Transportation Equipment	\$	-	\$	(2,212)	<del>#####</del>
39400	Tools Shop and Garage Equi	р\$	1,450	\$	1,354	93.4
39705	Communication Equipment	\$	1,050	\$	-	0.0
39908	Application Software- AM/FM	\$	-	\$	59,643	<del>#####</del>
	Total	\$	1,592,514	\$	1,131,975	71.1
М	adisonville Operations					
36700 40	Transm Rectifier & Groundbe	ed \$	1,092	\$	-	0.0 Deferred
36701 10	TP Mains - Sys Improv	\$	-	\$	249,137	<del>#####</del>
36701 20	TP Mains - Pub Improv	\$	-	\$	3,479	<del>#####</del>
36701 30	TP Mains - Steel Leakage	\$	15,742	\$	3,313	21.0
36701 83	Mains - Stl Reimbursements	\$	-	\$	(12,321)	<del>#####</del>

Page 26 of 28

36900 30	M & R Station Equip	\$	-	\$	1,072	<del>#####</del>
37600 40	Mains - Cathodic Protection	\$	11,012	\$	14,819	134.6
37601 10	Mains - Stl Sys Improv	\$	10,189	\$	16,078	157.8
37601 20	Mains - Stl- Pub Improv	\$	199,899	\$	79,398	39.7
37601 30	Mains - Stl - Repl-Leakage	\$	62,100	\$	39,260	63.2
37601 98	Mains - Steel- Retirements	\$	6,644	\$	6,331	95.3
37602 10	Mains - Plastic Sys Improv	\$	85,678	\$	68,091	79.5
37602 20	Mains - Plastic Pub Improv	\$	5.265	\$	6,458	122.7
37602 30	Mains - Plastic Leakage	\$	49.474	\$	14,269	× 28.8
37602 81	Mains - Plastic Reimbursements	\$	(300,415)	\$	(264,592)	88.1
37602 98	Mains - Plastic Retirements	\$	613	\$	442	72.1
37800 20	M&R Sta Equipment	\$	4.059	\$	4,078	100.5
38000 10	Services New	\$	263,421	\$	259.224	98.4
38000 30	Services Leakage	\$	75.345	\$	31,389	41.7
38000 98	Services Retirements	\$	33,425	\$	31.272	93.6
38200 10	Meter Installations New	\$	46.858	\$	35.959	76.7
38200 20	Meter Set Replacement	\$	48.628	\$	63.372	130.3
38200 98	Meter Set Retirements	\$	12,960	\$	8.930	68.9
38500 10	Indust M&R Equip	\$	-	\$	20.349	######
38700	Other Equipment	Ŝ	876	\$	847	96.7
39200	Vehicles	\$		\$	(15,964)	######
39400	Tools, Shop Equip	Ŝ	6.552	\$	4.718	72.0
	i i i i i i i i i i i i i i i i i i i	*	_,	•		
	Total	\$	639,417	\$	669,408	104.7
i	Paducah Operations					
36600	Structures & Improvements	\$	0	\$	15 868	<del>!! !! !! !! !!</del>
36700 40	TP Mains - Cathodic Protection	ŝ	500	÷ \$	0	0.0 Deferred
36701 30	TP Mains - Steel Leakage	ŝ	2 475	\$	-	0.0
36900 20	M&R Sta Public Improvement	\$	_,	\$	2.893	<del>######</del>
37600 40	Mains- Cathodic Protection	Ŝ	15.000	\$	8.211	54.7
37601 10	Mains- System Improve	Ŝ	62.010	\$	53.734	86.7
37601 20	Mains- Public Improve	\$	47,390	\$	36,510	77.0
37601 30	Mains- Leakage	\$	28,350	\$	1,061	3.7
37601 83	Mains - Steel Reimbursements	\$	, _	\$	(70,076)	<del>#####</del>
37601 98	Mains - Steel- Retirements	\$	152	\$	3,092	<del>#####</del>
37602 10	Mains - Plastic Sys Improv	\$	212,500	\$	312,473	147.0
37602 20	Mains - Plastic Pub Improv	\$	5,458	\$	11,289	206.8
37602 30	Mains - Plastic Leakage	\$	18,000	\$	4,715	26.2
37602 81	Mains - Plastic Reimbursements	\$	(35,192)	\$	(24,143)	68.6
37602 83	Mains - Plastic Reimbursements	\$	(24,261)	\$	(12,956)	53.4
37602 98	Mains - Plastic Retirements	\$	151	\$	3,113	<del>#####</del>
37800 10	Meas and Reg. Sta. Equip	\$	-	\$	12,184	<del>#####</del>
38000 10	Services- New	\$	339,500	\$	375,756	110.7
38000 30	Services- Leakage	\$	33,000	\$	19,910	60.3
38000 98	Sanvisos Potiromants	\$	18 000	¢	20 514	164.0
	Services Remements	Ψ	10,000	Ψ	23,014	104.0
38200 10	Meter Installations	\$	90,125	\$	72,725	80.7

DR 28		F	Page 27 of 28		
38200 98	Meter Installations	\$	15,705	\$ 9,507	60.5
38500 10	Industrial Measuring and	\$	-	\$ 7,958	<del>#####</del>
38700	Other Equipment	\$	1,402	\$ 1,213	86.5
39200 98	Vehicle Retirement	\$	-	\$ (13,132)	#####
39400	Tools, Shop Equip	\$	450	\$ -	0.0
	Total	\$	889,224	\$ 941,048	105.8
Ε	Bowling Green Operations			Ň	
36700 40	Transm Mains- Cathodic Protecti	\$	708	\$ -	0.0
36701 30	TP Mains - SteelLeakage	\$	9,908	\$ 4,505	45.5
37600 40	Mains- Cathodic Protection	\$	23,647	\$ 16,102	68.1
37601 10	Mains System Improv	\$	39,768	\$ 621,996	<del>#####</del>
37601 20	Mains- Public Improv	\$	-	\$ 36,957	<del>#####</del>
37601 30	Mains- Leakage	\$	123,525	\$ 210,403	170.3
37601 81	Mains - Steel Reimbursement	\$	(45,612)	\$ (44,330)	97:2
37601 98	Mains - Steel Retirement	\$	7,756	\$ 16,311	210.3
37602 10	Mains - Plastic Sys Improv	\$	634,039	\$ 507,213	80.0
37602 30	Mains - Plastic Leakage	\$	80,125	\$ 127,828	159.5
37602 81	Mains - Plastic Reimbursements	\$	(152,280)	\$ (151,507)	99.5
37602 98	Mains - Plastic Retirements	\$	887	\$ 664	74.9
37800 10	M&R Sta Sys Improv	\$	-	\$ 12,803	<del>#####</del>
37900 20	City Gate M&R Sta Equip Pub 1	\$	45,554	\$ 5,492	12.1 Deferred
38000 10	Services- New	\$	450,877	\$ 482,356	107.0
38000 30	Services-Leakage	\$	148,718	\$ 161,718	108.7
38000 98	Services	\$	75,625	\$ 58,330	77.1
38200 10	Meter Installations	\$	106,558	\$ 244,720	229.7
38200 20	Meter Replacements	\$	84,280	\$ 50,904	60.4
38200 98	Meter Installations	\$	21,807	\$ 14,422	66.1
38500 10	Ind M&R Equip Sys Improv	\$	23,892	\$ 2,123	8.9
39103 98	Office Mach Retirements	\$	-	\$ (619)	<del>######</del>
38700	Other Equipment	\$	1,923	\$ 2,023	105.2
39200	Vehicles	\$	-	\$ (29,034)	#####
39400	Tools, Shop Equip	\$	7,479	\$ 3,634	48.6
39600	Power Operated Equip	\$	511	\$ (520)	<del>#####</del>
39701	Commun Equip- Mobile Radio	\$	5,460	\$ -	0.0
39702	Commun Equip - Fixed Base Rad	\$	4,353	\$ 4,846	111.3
	Total	\$	1,699,508	\$ 2,359,340	138.8
D	anville Operations				
36700 40	Trans Mains-Cathodic Protection	\$	2,250	\$ -	0.0 Deferred
36701 20	Mains - Steel Pub Improv	\$	23,500	\$ -	0.0 Deferred
36900	Measuring & Reg. Sta	\$	8,000	\$ -	0.0 Deferred
37600	Mains- Cathodic Protection	\$	24,250	\$ 11,925	49.2
37601 10	Mains- Sys Improve	\$	25,900	\$ 9,437	36.4
37601 20	Mains- Public Improve	\$	-	\$ 8,574	<del>#####</del>

Page 28 of 28

37601 30	Mains- Leakage	\$ 43,200	\$	114,765	265.7
37601 81	Mains - Steel Reimbursements	\$ (58,764)	\$	(5,049)	8.6
37601 98	Mains - Steel- Retirements	\$ 300	\$	4,254	######
37602 10	Mains - Plastic Sys Improv	\$ 280,262	\$	133,530	47.6
37602 20	Mains - Plastic Pub Improv	\$ 5,168	\$	17,629	341.1
37602 30	Mains - Plastic Leakage	\$ 4,800	\$	26,525	552.6
37602 81	Mains - Plastic Reimbursements	\$ (47,043)	\$	(40,459)	86.0
37602 83	Mains - Plastic Reimbursements	\$ -	\$	(10,408)	<del>#####</del>
37602 98	Mains - Plastic Retirements	\$ 450	\$	1,042	231.6
37900 20	Harrodsburg M&R Sta Repl	\$ -	\$	28,612	` ######
38000 10	Services- New	\$ 492,901	\$	319,214	64.8
38000 30	Services- Leakage	\$ 44,400	\$	62,777	141.4
38000 98	Services Retirements	\$ 22,500	\$	41,611	184.9
38200 10	Meter Installations	\$ 73,947	\$	78,978	106.8
38200 20	Meter Replacements	\$ 57,146	\$	90,037	157.6
38200 98	Meter Inst Retirements	\$ 15,480	\$	30,886	199.5
38500 10	Ind M&R Sta Caterpiller	\$ 35,100	\$	15,680	44.7
38700	Other Equipment	\$ 8,555	\$	2,140	25.0
39100	Office Furniture & Equip- Danville	\$ -	\$	(1,342)	<del>#####</del>
39200 98	Vehicle Retirements	\$ -	\$	(23,561)	<del>#####</del>
39400	Tools, Shop Equip	\$ 8,365	\$	526	6.3
	Total	\$ 1,070,667	\$	917,323	85.7
	WKG Overheads				
39900		\$ 2,897,717	\$	3,860,000	133.2
	Grand Total	\$ 10,194,434	\$1	1,459,605	112.4

,

. .

. .

-

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 29 a through c Witness: David H. Doggette

#### Data Request:

Refer to the Application, Volume 3 of 10, Tab 1, titled "FR 10(9)(b)," page 3 of 6, the forecasted test year capital budget.

a. Does Western assume that all the capital projects included on this schedule will be completed and included in rate base by the end of the forecasted test year?

#### **Response:**

Yes.

b. If yes to part (a), explain in detail the basis for this assumption and why it is reasonable.

#### **Response:**

Given the difficulty in predicting weather, WKG has found it prudent to base its capital budget spending on normal weather conditions, normal growth and prioritized as stated in Volume 2 of 10, Tab 5, Testimony of David H. Doggette, page 4. Therefore, deviation from capital spending normally would be related to prudent management as a result of abnormal weather conditions. To a certain degree deviations could also result from external factors such as delays in state highway projects.

c. If no to part (a), provide a schedule showing the amounts from the capital budget that Western included in the rate base by the end of the forecasted test year.



### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 30 Witness: Betty L. Adams

### Data Request:

Refer to the Application, Volume 3 of 10, Tab 7, titled "FR 10(9)(h)2." Provide all the assumptions used by Western to determine the amounts shown for the base year, and the forecasted year, and the 2000 through 2003 fiscal years. Include an explanation as to why each assumption is reasonable.

#### Response:

See the assumptions discussed in the testimony and data requests responses, DR 13 a and b, DR 14 and DR 15 of Pat Reddy.



15

.

.

.

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 31 Witness: Rebecca M. Buchanan

### Data Request:

31. Refer to the Application, Volume 3 of 10, Tab 7, titled "FR 10(9)(h)4." Provide a revision of this schedule that fixes the forecasted required rate of return at 9.97 percent for each year shown.

### Response:

31.) See attached schedule "FR 10(9)(h)4" - Revised.

response to DR 31

Western Kentucky Gas Company Case No. 99-070 Forecasted Revenue Requirements

For the Test Year Twelve Months Ended December 31, 2000 & Four Forecasted Fiscal Years

FR 10(9)(h)4

Data.	: Base Period X Forecasted Period							
Type	of Filing: Original Updated X R	evised						Page 1 of 1
Work	(paper Reference No(s).						Witnes	s: Buchanan
		Test Period	Forecast FY	Test Period	FY 2000	FY 2001	FY 2002	FY 2003
		Supporting	Supporting	Forecasted	Forecasted	Forecasted	Forecasted	Forecasted
Line		Schedule	Schedule	Revenue	Revenue	Revenue	Revenue	Revenue
No.	Description	Reference	Reference	Requirement	Requirement	Requirement	Requirement	Requirement
				000\$	\$000	\$000	\$000	\$000
-	Rate Base	B-1	FR10(9)(h)12	130,464	132,853	131,203	129,522	127,044
C4	Operating Income at Current Rates	<u>-</u> -	FR10(9)(h)1	4,631	5,201	4,746	4,884	4,540
ę	Rete of Return at Current Rates (2 / 1)	J-1.1		3.55%	3.91%	3.62%	3.77%	3.57%
4	Forecasted Required Rate of Return	J-1	FR10(9)(h)11	9.97%	9.97%	9.97%	9.97%	9.97%
5	Required Operating Income (1 x 4)	ი -		13,009	13,245	13,081	12,913	12,666
9	Operating Income Deficiency (5 - 2)	С-1		8,378	8,044	8,335	8,029	8,126
7	Gross Revenue Conversion Factor	T	r	1.686137	1.686137	1.686137	1.686137	1.686137
ω	Revenue Deficiency (6 x 7)	5		14,128	13,564	14,055	13,539	13,703
თ	Operating Revenues at Current Rates	С- 1	FR10(9)(h)1	120,503	120,558	120,999	121,768	122,454
10	Revenue Required to Support Forecasted ROR (8 +	9 C-1	·	134,631	134,122	135,054	135,307	136,157

Note: Per Staff's direction, Capital Structure for all forecasted years assumes overall rate of return fixed at 9.97%.

### Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 32 Witness: Gruber

#### Data Request:

Concerning Western Kentucky Gas Resources, Inc. ("WKG Resources"):

a. Describe the nature of WKG Resources' business operations.

b. Indicate when WKG Resources was established.

c. Indicate whether any of Western's assets, liabilities, capital, or personnel were transferred to WKG Resources.

#### Response:

a. WKG Resources name was changed to Atmos Exploration and Production, Inc. effective April 29, 1999. Atmos Exploration and Production, Inc. has one natural gas supply agreement with a natural gas vehicle fueling station in Owensboro, Kentucky and storage field operation agreements with two industrial customers served by Western Kentucky Gas.

b. Western Kentucky Gas Resources (WKG Resources) was incorporated as Western Kentucky Gas Company in 1934. It provided local distribution service as Western Kentucky Gas Company pursuant to the Commission's jurisdiction until 1980. By Order of November 19, 1980, in Case No. 7975 the Commission approved the acquisition of Western Kentucky Gas Company by Texas American Energy (TAE). In a 1980 post-merger transaction, the name of Western Kentucky Gas Company was changed to Western Kentucky Gas Resources, Inc. at which time it became a subsidiary of TAE. WKG Resources has not provided any regulated utility services since that time. TAE provided distribution service in Kentucky through a division named Western Kentucky Gas Company which was not a separate corporate entity.

In Case No. 9992, TAE applied to the Commission for approval of an intra-corporate reorganization by which all of the assets and obligations of the Western Kentucky Gas Company division would be transferred to a corporate entity named Western Kentucky Utility. By Order of August 21, 1987, the Commission approved the reorganization. WKG Resources became a wholly owned subsidiary of Western Kentucky Utility pursuant to this reorganization. On December 18, 1987, in Case No. 10063, the Commission approved the
acquisition of Western Kentucky Utility by Atmos. At that time, WKG Resources became a wholly owned subsidiary of Atmos. In April 1999, the name of WKG Resources was changed to Atmos Exploration and Production, Inc.

.

۰.

c. None of Western's assets, liabilities or capital have been transferred or personnel currently dedicated to WKG Resources.

Y,



# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 33 Witness: Betty L. Adams

# Data Request:

Refer to the Application, Volume 3 of 10. Explain why the operating income amounts shown for the test year (January 2000 – December 2000) on the following schedules are not in agreement:

- A. Tab 3, titled "FR 10(9)(d)."
- B. Tab 7, titled "FR 10(9)(h)1 and FR 10(10)(i)1."
- C. Tab 7, titled "FR 10(9)(h)4."

#### Response:

The difference in A and B above is that the provision for income taxes was removed from the "Total operating expenses" section in "FR10(9)(h)1" and "FR10(10)(I)1" and entered as a line item after "Income Before Taxes". This was changed to correctly format taxes after all income and expenses were derived as part of the taxes were not Operating Income related.

The difference between A and C includes rate making adjustments as shown on Schedule C-1 in FR 10(10)(c), Volume 10, Tab 3.



# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 34 a Witness: Betty Adams

# Data Request:

- 34. Refer to the Application, Volume 9 of 10, Tab 2, titled "FR 10(9)(u)," Schedules 1 and 2. Provide the following information for Schedule 1 and Schedule 2:
  - a. An explanation of how the amounts shown under the column titled "WKG \$" were determined. Include any supporting workpapers, assumptions, and other documentation.

# Response:

- 1. Considering the Base Year consists of six months of actual expenses and six months of projected expenses, the "WKG \$" column was calculated by adding:
  - a. October 1998 through March 1999 actual expenses by Shared Service Unit and,
  - b. Projected expenses by Shared Service Unit for the months of April 1999 through September 1999.
  - c. The actual expenses consist of items which, when possible, have been directly coded to WKG plus items which, when direct coding is not possible, have been allocated according to each Shared Service Unit's allocation method (See DR Items 34 a and b).
- 2. Considering the Forecasted Year consists of twelve months of projected expenses, the "WKG \$" column was calculated by:
  - a. Applying the approved WKG allocation percentage by Shared Service Unit to each Shared Service Unit's total projected expenses for the Forecasted Year.

See attached Response to DR 34 – Revision to FR10(9)(u) Schedules 1 and 2

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 34 b Witness: Betty Adams

# Data Request:

- 34. Refer to the Application, Volume 9 of 10, Tab 2, titled "FR 10(9)(u)," Schedules 1 and 2. Provide the following information for Schedule 1 and Schedule 2:
  - b. The determination of the percentages shown under the column titled "WKG %." Include any supporting workpapers, assumption, and other documentation.

I.

# Response:

- 1. Considering the Base Year consists of six months of actual expenses and six months of projected expenses, the "WKG %" column was calculated by:
  - Dividing the WKG actual expenses for the months of October
     1998 through March 1999 plus the projected expenses of April
     1999 through September 1999 by the total Atmos expenses for the same periods.
- 2. Considering the Forecasted Year consists of twelve months of projected expenses, the "WKG %" column was calculated by:
  - a. Utilizing the approved WKG allocation percentage by Shared Service Unit as set forth in each SSU's Contract with the Business Units.

See attached Response to DR 34 – Revision to FR10(9)(u) Schedules 1 and 2

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 34 c Witness: Betty Adams

# Data Request:

. . ..

34. Refer to the Application, Volume 9 of 10, Tab 2, titled "FR 10(9)(u)," Schedules 1 and 2. Provide the following information for Schedule 1 and Schedule 2:

١,

c. For each Shared Service Unit listed on the schedule, an explanation of why the particular allocation methodology is reasonable for the allocation of the particular unit costs.

# Response:

When determining allocation factors in a multi-jurisdictional utility like Atmos, we try to strike a balance between relative simplicity and understandability with having a reasonable amount of precision. The factors used for this reporting period were deemed by management to be reasonable and supportable without undue complexity.

- 1. Accounting expenses are allocated based on the following factors:
  - a. Number of states the Tax department prepares individual tax returns by state and the Financial Reporting department prepares other state filings.
  - b. Number of employees the Payroll department processes paychecks with each business unit having a different number of employees.
  - c. Number of invoices processed the Accounts Payable department processes a different number of invoices for each business unit.
  - d. Number of hours worked the Gas Accounting department based the total workload per business unit on recent.
  - e. Number of customers the Financial Systems department services systems with capacities which are determined by how many customers are serviced by each business unit.
- 2. Planning & Budgeting
  - a. Number of hours worked The number of hours worked per business unit is the most reasonable allocation.
- 3. Call Center
  - a. Number of customers the number of calls handled for each business unit is directly related to the number of customers in each business unit.
- 4. Management Committee

- a. Average of all SSU allocations since the Management Committee oversees all other SSU's and functions, the most reasonable allocation is an average of all the other SSU allocations.
- 5. Gas Control
  - a. Number of SCADA points Gas delivery monitor points along the distribution system; reflects the diversity of the gas control function as it relates to each business unit.
  - b. Number of pipeline contracts each business unit has a different number of pipeline contracts to manage.
  - c. Number of storage contracts each business unit has a different number of storage contracts to manage.
- 6. Gas Supply
  - a. Number of hours worked the Forecasting and Administration functions of Gas Supply can best be measured by how many hours are spent on each business unit.
  - b. Number of storage contracts each business unit has a different number of storage contracts to manage.
  - c. Bcf gas purchased the Procurement function can be measured by how many Bcf is purchased for each business unit.
  - d. Number of supply contracts the Procurement function can also be measured by how many supply contracts it must manage for each business unit.
  - e. Number of Electronic Bulletin Boards monitored The Nominations department can be measured by how many Electronic Bulletin Boards it monitors for each business unit.
  - f. Number of nominations The Nonimations department can also be measured by how many nominations are requested for each business unit.
- 7. Human Resources
  - a. Functions such as Employment Services, Employee Relations, Employee Development and Employee Communications service a different number of employees within each business unit.
  - b. Total salary dollars the Compensation Services function manages the incentive plan which is based on salary dollars.
  - c. Number of management positions the Professional Development function provides management training for select employees of the business units.
  - d. Square footage of buildings the Facilities function is allocated based on the square footage of buildings within each business unit.
- 8. Information Technology
  - a. Number of employees the Help Desk Support and Telecommunications functions support the differing number of employees within each business unit.
  - b. Number of customers the Enterprise Servers Support and Information Systems Support functions service systems within each business unit which support a differing number of customers.

- c. Number of PC's the PC Support function is allocated according to how many PC's each business unit maintains.
- 9. Internal Audit
  - a. Number of hours worked The number of hours worked per business unit is the most reasonable allocation.
- 10. Investor Relations
  - a. Gross plant investment This was determined to be the most reasonable methodology for this department.
- 11. Legal and Corporate Secretary
  - a. Number of hours worked The number of hours worked per business unit is the most reasonable allocation.
  - b. Number of storage boxes archived the Storage function archives different numbers of boxes for each business unit.
- 12. Rates
  - a. Number of hours worked The number of hours worked per business unit is the most reasonable allocation.
- 13. Billing & Remittance Processing
  - a. Number of customers the Remittance Processing, Credit, and Collection functions all are directly related to how many customers each business unit has.
  - b. Number of payments processed the Pay Center, Telepay, and Bank Draft functions all process customer payments and can be allocated in the same manner.
  - c. Number of statements processed the Mass Mail function mails a different number of statements for each business unit.
- 14. Treasury, Risk Management & Purchasing
  - a. Number of employees the Mail & Supply function delivers supplies to each business unit as ordered which is directly related to the number of employees per business unit.
  - b. Number of customers the Treasury function manages short-term debt and equity based on the number of customers within each business unit.
  - c. Gross plant investment the Treasury function manages long-term debt based on the gross plan investment of each business unit.
  - d. Dollars of revenue insurance policies are based on revenue.
  - e. Number of purchase orders the Purchasing function issues purchase orders for each business unit according to their material and supply needs.
  - f. Number of vehicles the Fleet Administration function manages company vehicles for each business unit.

#### Response to DR 34 - Revision to FR10(9)(u) Base Year: October 1998 - September 1999

Shared Service Unit	WKG \$	WKG % Methodology	WKG Oct-Mar	Total Oct-Mar	WKG Apr-Sep	Total Apr-Sep	WKG	Total
Accounting	825,460	14.8% A, B, C, D, E	407,535	2,746,114	417,925	2,844,756	825,460	5,590,870
Planning & Budgeting	134,116	13.6% D	77,936	568,878	56,180	417,230	134,116	986,108
Call Center	1,212,222	17.5% E	571,316	3,264,664	640,906	3,662,320	1,212,222	6,926,984
Management Committee	885,431	16.6% Y	522,719	3,166,006	362,712	2,166,360	885,431	5,332,366
Gas Control	124,150	23.4% F, G, H	55,098	250,993	69,052	279,210	124,150	530,203
Gas Supply	367,276	24.7% D, H, I, J, K, L, M	173,583	703,725	193,693	783,200	367,276	1,486,925
Human Resources	2.024.872	18.4% B, N, O, P	986,190	4,800,587	1,038,682	6,220,976	2,024,872	11,021,563
Information Technology	936.215	17.7% B, E, Q,	467,271	2,642,380	468,944	2,645,999	936,215	5,288,379
Internal Audit	111,455	17.3% D	50,104	288,534	61,351	354,025	111,455	642,559
Investor Relations	288,368	15.4% R	117,219	761,248	171,149	1,115,125	288,368	1,876,373
Legal and Corporate Secretary	844,737	10.2% D. E. X	517,343	6,162,131	327,394	2,096,579	844,737	8,258,710
Rates	143,132	10.8% D	71,471	661,534	71,661	667,462	143,132	1,328,996
Billing & Remittance Processing	1.028.181	17.8% E. S. T	275,655	1,520,295	752,526	4,263,696	1,028,181	5,783,991
Technical Services	15.087	15.6% See Part 1	15,087	97,021	-	· -	15,087	97,021
Treasury, Risk Momt & Purchasing	913.924	18.1% B, E, R, U, V, W	543,247	2,710,287	370,677	2,349,176	913,924	5,059,463
Sub-Total	9.854.626	16.4%	4,851,774	30,344,397	5,002,852	29,866,114	9,854,626	60,210,511
Overhead Capitalized	(1,781,898)	16.6% Y, Z	(892,620)	(5,400,000)	(889,278)	(5,309,502)	(1,781,898)	(10,709,502)
Maintenance	59,393	16.5% See Part 1	59,393	359,301	•	-	59,393	359,301
Depreciation and Amortization	1.658.377	7.9% E. Y. AA	1,177,375	15,587,467	481,002	5,535,498	1,658,377	21,122,965
Taxes-Other	211.688	17.1% Y	134,690	806,108	76,998	435,000	211,688	1,241,108
Total	10.002,186	13.8%	5,330,612	41,697,273	4,671,574	30,527,110	10,002,186	72,224,383
	(0.716.607)							

(9,716,607) Total costs are a combination of direct charges and allocated charges.

- Allocated charges are based on:
- Α Number of states
- в Number of employees ¢ Number of invoices processed
- D Hours worked
- E Number of customers F Number of SCADA po Number of SCADA points
- G Number of pipeline contracts
- Number of storage contracts H
- Bcf gas purchased 1
- Number of supply contracts J
- Number of Electronic Bulletin Boards monitored κ
- L Number of Nominations
- М Bcf gas stored
- Total salary dollars
- N O P Number of management positions Square footage of buildings Number of PC's
- Q R
- Gross plant investment
- s Number of payments processed
- т Number of statements processed
- Ù Dollars of revenue
- ۷ Number of purchase orders
- w Number of vehicles
- X Y Number of storage boxes archived
- Average of all Shared Service Unit allocation percentages
- z Capital budget breakdown
- AA Severance costs

## Response to DR 34 - Revision to FR10(9)(u) Forecasted Year: January 2000 - December 2000



Shared Service Unit	WKG \$	WKG % Methodology	WKG	Total
Accounting	833,427	14.8% A, B, C, D, E	833,427	5,641,990
Planning & Budgeting	111,600	13.5% D	111,600	829,000
Call Center	1,272,133	17.5% E	1,272,133	7,269,330
Management Committee	712,855	16.7% Y	712,855	4,257,649
Gas Control	136,951	24.7% F, G, H	136,951	553,760
Gas Supply	384,849	24.7% D, H, I, J, K, L, M	384,849	1,556,140
Human Resources	2,090,632	16.7% B, N, O, P	2,090,632	12,525,196
Information Technology	928,202	17.7% B, E, Q,	928,202	5,237,344
Internal Audit	121,806	17.3% D	\121,806	702,880
Investor Relations	341,249	15.3% R	341,249	2,223,419
Legal and Corporate Secretary	674,378	15.7% D, E, X	674,378	4,304,202
Rates	158,870	10.7% D	158,870	1,479,023
Billing & Remittance Processing	1,505,052	17.6% E, S, T	1,505,052	8,527,392
Treasury, Risk Mgmt & Purchasing	760,962	15.8% B, E, R, U, V, W	760,962	4,807,726
Sub-Total	10,032,966	16.7%	10,032,966	59,915,051
Overhead Capitalized	(1,605,750)	17.1% Y, Z	(1,605,750)	(9,375,000)
Maintenance	-			
Depreciation and Amortization	1,473,750	9.5% E, Y, AA	1,473,750	15,519,750
Taxes-Other	152,000	16.8% Y	152,000	906,000
Total	10,052,966	15.0%	10,052,966	66,965,801

Total costs are a combination of direct charges and allocated charges.

Allocated charges are based on:

- A Number of states
- B Number of employees
- C Number of invoices processed
- D Hours worked
- E Number of customers
- F Number of SCADA points
- G Number of pipeline contracts
- H Number of storage contracts
- I Bcf gas purchased
- J Number of supply contracts
- K Number of Electronic Bulletin Boards monitored
- L Number of Nominations
- M Bcf gas stored
- N Total salary dollars
- O Number of management positions
- P Square footage of buildings
- Q Number of PC's
- R Gross plant investment
- S Number of payments processed
- T Number of statements processed
- U Dollars of revenue
- V Number of purchase orders
- W Number of vehicles
- X Number of storage boxes archived
- Y Average of all Shared Service Unit allocation percentages
- Z Capital budget breakdown
- AA Severance costs

Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 35 a. and b. Witness: Rebecca M. Buchanan

# Data Request:

35. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-2.2.
a. For both the base and forecasted periods, provide a schedule that links the plant additions shown on Schedule B-2.2 with the capital budget projects shown in Volume 3 of 10, Tab 1, Exhibit DHD-1.

b. For the base period, provide the workpapers, analysis, assumptions, and other documentation that support the amounts shown for "Retirements" and "Transfers/Reclassifications." Include a description for each retirement or transfer/reclassification shown on the schedule.

#### Response:

35a.) For the **base period**, workpaper WP B-2 B 09 is the schedule that links the plant additions on B-2.2 with the capital budget projects shown in Volume 3 of 10, Tab 1, Exhibit DHD-1, page 1 of 6.

Please find attached WP B-2 B 09, page 4 of 4, where the totals are shown on line 106 (note that in the original filing, the final column of this workpaper titled "service program additional" was unintentionally excluded from the print range). The columns have been labeled with letters A through F to facilitate this response.

WP B-2 B 09, page 4 of 4, line 106,	col. A	\$ 5,461,802
"	col. D	1,840,198
"	col. E	1,106,000
Exhibit DHD-1, page 1 of 6, line 91	FY 99	\$ 8,408,000
WP B-2 B 09, page 4 of 4, line 106,	col. B	19,235,840
"	col. C	873,888
"	col. F	2,632,460
		\$31,150,189

The total of columns A through F is \$31,150,189. This amount is carried from WP B-2 B 09 to Schedule B-2.3, sheet 3 of 8, line 37, additions less retirements, that is \$31,722,633 less \$572,444 equals \$31,150,189 net. To this amount, add 16.657% of the Division 02 General Office net additions and transfers, which are found on Schedule B-2.3, sheet 4 of 8, line 32. Division 02 additions of \$20,622,549 plus transfers of <\$515> gives net General Office additions of \$20,622,034, multiplied by the 16.657% residual factor equals \$3,435,012 net.

Sched. B-2.3, p. 3 of 8, net additions & retirements	\$31,150,189	Div. 09
Sched. B-2.3, p. 4 of 8, net additions & transfers * 16.6579	% 3,435,012	Div. 02
Equals the net additions, retirements & transfers B-2.2, p.3	<u>\$34,585,201</u>	Total

For the **forecasted test period**, Schedule B-2.3, is the schedule that links the forecasted test period plant additions on B-2.2 with the capital budget projects shown in Volume 3 of 10, Tab 1, Exhibit DHD-1, page 3 of 6. Refer to Schedule B-2.3, Sheet 7 of 8, line 37 additions, for the Division 09 amount \$9,718,882. This agrees with the amount on DHD-1, page 3 of 6, line 91. To this amount, add 16.657% of the Division 02 General Office net additions, which are found on Schedule B-2.3, sheet 8 of 8, line 33. Net General Office Additions of \$10,992,001, multiplied by the 16.657% residual factor equals \$1,830,938 net.

Sched. B-2.3, p. 7 of 8, net additions	\$ 9,718,882	Div. 09
Sched. B-2.3, p. 8 of 8, net additions * 16.657%	1,830,938	Div. 02
Equals the net additions, retirements & transfers B-2.2, p.6	<u>\$11,549,819</u>	Total

35b.) The source documents for the base period Retirements/Reclassifications are attached, and numbered 35 b pages 1 through 18. A description of each retirement and transfer is attached and numbered 35 b pages 19 through 21.



# ·

response to DR 35a

Western Kentucky Gas Company Computation of 13 Month Average Ptart Balances - Western Div, 09 Antropers WP 52 B 03 Base Period 10/163 - 92039

Base Peri	od 10/1/98 -	66/05/6		×	8	v	٩	ш	u.
			WKG Div 09	WKG Direct	service prog.				service
<sup>2</sup>	Aot N	Account Title	Plant Balance Sen-99	Additions FY 99	WKG adds Andi 1999	WKG OH	WKG OH	02 OH	program
			Projected	Budgeted	2001 1144	16.000%	33 692%	20.250%	
8		Distribution Plant							
49	374.10	Land Town Border	61,710	_		•	0	•	
ន	374.30	Land Other	2,784			•	0	0	
51	374.20	Right of Way	44,872			•	0	0	
23	375.10	Structures & Improvements T.B.	106,376			•	•	•	
ន	375.02	Structures & Improvements Other	0	_		•	•	•	
3	375.03	Improvements	7,518	_		•	0	•	
ន	375.20	Land Rights	46,591			•	•	•	
8	376.00	Mains	71,878,855	1,946,304		311,409	656,751	394,121	
6	378.10	Meas. & Reg. Sta. Equipment General	2,137,306	150,490		24,078	50,703	30,474	
<b>8</b>	379.30	Meas & Reg. Sta. Equipment T.b.	1,803,836	90,002		14,400	30,324	18,225	
8	380.00	Services	43,729,545	1,914,168		306,267	644,924	387,614	
83	381.00	melers	18,839,620	488,201		78,112	164,485	939'86'	
<del>ء</del> د	201.00 00.000	V & P Guages	575'R01			- -	0 0	0 101 02	
38	302.00		14,004,050	4/4,050		18.4	426,931	96,125 24,735	
3 3	00.500	Keguators Service	107'01'9'5	tro'ont		190'/I	128,45	21,583	
8 8		Neglatura Neutel Unite Den Instalhähme	2001 (201) 2002 (201)	1.000		0.05	750	- <u>-</u>	
3 8	365.10	Ind. Meas. & Red. Sta. Echimment	3058.017	74 400		11 904	75 067	15,065	
20	2			2001 C		200111	100'07	200/21	
8		Total Distribution Plant	160,086,162	5,247,145		EC9,543	1,767,875	1,062,533	
69									
۶ ;		General Plant				•	•		
< f		Land	44'/2	~ -		•	-	•	
2 8	20,055		210,015	_	134,435		<b>.</b>	2 0	
2 7		Air Conditioning Contract	ŝ	-		•			
1	30.05	Total Friends				• -			
76	390.09	Improvement to leased Premises	1.394.26	10.001		1.600	3.370	2.025	
11	391.00	Office Furniture & Equipment	1,861,880	1,500	283,245	240	505	Ъ́С Я	
78	391.83	Office Machines	200,475	-		o	0	0	
62	392.10	Transportation Equipment	6,044,07			Ð	•	•	
83	392.20	Trailers	165,97(	_		0	0	0	
5 8		Tada 0 Work Extrement				- ș	0 0 0	9	
8 8		1005 & FOA CQUINERII	o'nn'an Beateil	non't		3	9 <del>0</del> 0'l	80	
3 25	396.94	Backhoes	706.02				• •		
ß	396.95	Welders	92,41				0	0	
88	397.00	Communication Equipment - Phones	1,100,36	40,000	282,328	6,400	13,477	8,100	
87	397.20	Communication Equip Fixed Radios	21,69	~		•	0	0	
88	397.21	Communication Equipment - Mobile Radios	68,22	6,000		996 1	2,022	1,215	
88	77.765	Communication Equip Felemetering	114,69	• •		•	• •	•	
R 8	300.00	misuana auto Equipi an u Other Tencible Denorty	10'10						
5 8	309.84	Other Tennityle Proventy - CPU							
8	339.662	Other Tangible Procerty - MF Hardware	397.27			• =		• •	
\$	309.862	Other Tangible Property - PC Hardware	2,828,88	35,980	2,263,318	5.757	12.122	7.286	
8	399.87	Other Tang. Property - P.C. Software	306,17	3 10,000	103,668	1,600	3,369	2,025	
8	399.88	Other Tang. Property - Application Software	12,054,92		9,215,475	0	•	0	2,632,460
26	<b>399.89</b>	Other Tang. Property - System Software	-			o	•	0	
8	39r.xt	Server Hardware	632'37	-	695,971				
8	391,002	Server Software	228,31	-	228,311				
83	391.013	Network Cost	322,232	4.	332,234				
5	391.004	Start Up Cost	5,696,83 2 20 2 20 2		5,696,831	•	•	•	
žĘ	00.888	CUSTED LOAS	1,034,83	,		o	0	Þ	
<u>3</u>		Total General Plant	40,402,40	0 107,481	19,235,840	17,197	36,213	21,765	2,632,460
<u>1</u> 05									
ŝ		Total Gas Plant in Service	08'ZZE'6ZZ	0 5,461,802	19,235,840	673,888	1,840,198	1,106,000	2,632,460

tot of col. A+D+E \$8,408,000 from DHD-1, p.1

ł

vorkpaper 35a WP B-2 B 09 p4

•

#### R:\0113600 Plant Accounting\@Plant Accounting\TOPPER\Monthly Reports\Pg4 -Plant\[PG4 - 1999.xls]Oct'98-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF OCTOBER 31, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 10/31/98
	WESTER	N KENTUCKY DISTRIBUTION DIVISION					
00	3010000	Organization	8 170 77	0.00	0.00	0.00	8 179 77
09	3020000	Franchises & Consents	119,852.69	0.00	0.00	0.00	119,852.69
		Total Intangible Plant	128,182.41	0.00	0.00	0.00	128,182.41
		_				``	
	L-P PROI	DUCTION PLANT	0.00	0.00	0.00	0.00	0.00
09	3040010		0.00	0.00	0.00	0.00	0.00
09	3030000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110003	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges of Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Distant	0.00	0.00	0.00	0.00	0.00
09	3110030	Fiping Storoge Favinment	0.00	0.00	0.00	0.00	0.00
09	3110022	Venezizing Fouriement	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0.00	0.00	0.00	0.00	0.00
		Total I_P Production Plant	0.00	0.00	0.00	0.00	0.00
			0,00		0.00		
	NATURAI	L GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Field Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3,492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	4/,102.0/
09	3320020	Tributary Lines	528,218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sta. Equip.	0.00	0.00	0.00	0.00	100 469 81
09	3340000	Field Meas. & Reg. Sta. Equip. Purification Fourinment	198,408.81 44 369 30	0.00	0.00	0.00	44,369,30
09	550000	i u meation Equipment	44,505.50	0.00			
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261,126.69	0.00	0.00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09	3510020	Compressor Station Equipment	121,264.59	0.00	0.00	0.00	121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144,554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2,172,799.96	0.00	0.00	0.00	2,172,799.96
09	3520002	Well Equipment	535,976.49	0.00	0.00	0.00	535,976.49
09	3520010	Leaseholds	178,530.09	0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	34,014.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	I ributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470,683.43	0.00	0.00	0.00	4/0,083.43
09 09	3560000	vieas. & Keg. Equipment Purification Equipment	288,850.55 239,929.66	0.00	0.00	0.00	239,929.66
••							
		Total Storage Plant	4,884,110.51	.00	0.00	0.00	4,884,110.51



#### R:\0113600 Plant Accounting\@Plant Accounting\TOPPER\Monthly Reports\Pg4 -Plant\[PG4 - 1999.xls]Oct'98-A

1

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF OCTOBER 31, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 10/31/98
	TRANSM	ISSION PLANT					
09	3650010	Land	26,951.25	0.00	0.00	0.00	26,951.25
09	3650020	Rights of Ways	403,419.34	0.00	0.00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	0.00	0.00	0.00	32,921.36
09	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(5,427.87)	0.00	0.00 `	19,295,627.98
09	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas. & Reg. Equipment	2,976,155.16	0.00	0.00	0.00	2,976,155.16
		Total Transmission Plant	22,809,675.37	(5,427.87)	0.00	0.00	22,804,247.50
	DISTRIBU	TION PLANT					
09	3740010	Land Town Border	61,709.61	180.00	0.00	0.00	61,889.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783.89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,871.73
09	3750010	Structures & Improvements T.B.	106,375.76	0.00	0.00	0.00	106,375.76
09	3750002	Structures & Improvements Other	0.00	0.00	0.00	0.00	0.00
09	3750003	Improvements	7,517.58	0.00	0.00	0.00	7,517.58
09	3750020	Land Rights	46,591.01	0.00	0.00	0.00	40,591.01
09	3760000	Mains Mass & Den Ste Femineent Conserve	08,571,269.61	255,381.74	0.00	0.00	08,820,031.33
09	3780010	Meas. & Reg. Sta. Equipment General	1,881,300.42	23,739.04	0.00	0.00	1,505,520.00
09	3800000	Meas. & Reg. 5ta, Equipment 1.D.	1,030,884.23	118 070 61	0.00	0.00	40 795 547 36
09	3810000	Metera	40,470,371,73	0.00	0.00	0.00	18 009 962 39
09	3810000	V & P Cauges	109,502.35	0.00	0.00	0.00	109 523 93
09	3820000	Vot r Gauges Meter Installations	13 107 358 85	96 477 61	0.00	0.00	13 293 831.48
09	3830000	Regulators Service	3 478 991 68	0,472.00	0.00	0.00	3.428.991.68
09	3830020	Regulators Bellef	481 544 54	0.00	0.00	0.00	481.544.54
09	3840000	House Reg. Installations	154.276.36	0.00	0.00	0.00	154.276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2,931,579,79	0.00	0.00	0.00	2.931.579.79
09	3860000	Other Prop. On Customer Prem.	5,692.66	0.00	0.00	0.00	5,692.66
		Total Distribution Plant	151,169,065.81	694,764.62	0.00	0.00	151,863,830.43
	GENERAL	PLANT					
09	3890010	Land	44,727.62	0.00	0.00	0.00	44,727.62
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.56
09	3900003	Improvements	64,110.56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	3900005	Total Energy	0.00	0.00	0.00	0.00	0.00
09	3900009	Inprov. to Leased Premises	1,377,285.70	0.00	0.00	0.00	1,377,285.70
09	3910000	General Office Equipment	1,576,086.34	0.00	0.00	0.00	1,576,086.34
09	3918300	Office Furn-Coplers & Typewritters	200,478.91	0.00	0.00	0.00	200,478.91
09	3920000	Transportation Equipment	6,044,073.81	0.00	0.00	0.00	6,044,073.81
09	3920020	Irailers	165,969.90	0.00	0.00	0.00	103,909.90
09	3940077	Ditabase	3,064,138.84	0.00	0.00	0.00	3,004,138.84
09	1060477	Dichers	206 012 84		0.00	0.00	706 077 84
09	3969477	Waldars	00,022.04	0.00	0.00	0.00	97 412 75
09	1070000	Communication Equipment	750.059.60	0.00	0.00	0.00	750 059 60
09	3970000	Comm Equip - Fixed Redice	750,055,00	0.00	0.00	0.00	21 697.23
09	3970020	Comm Equip - Mobile Redios	58 073 11	0.00	0.00	0.00	58.023.11
69	3970022	Comm. Equip Telemetering	114 695.01	0.00	0.00	0.00	114.695.01
09	3980000	Miscellaneous Equipment	37.073.03	0.00	0.00	0.00	37,073.03
09	3990000	Oth Tang Prop	0.00	0.00	0.00	0.00	0.00
09	3998500	Oth Tang Prop - MF Hardware	397.277.53	0.00	0.00	0.00	397,277.53
09	3998600	Oth Tang Prop - PC Hardware	504,426.33	0.00	0.00	0.00	504,426.33
09	3998700	Oth Tang Prop - PC Software	185,510.98	0.00	0.00	0.00	185,510.98
09	3998800	Oth Tang Prop - Application software	206,993.96	0.00	0.00	0.00	206,993.96
09	9999117	Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	0.00	0.00	0.00	18,351,444.82
	TOTAL WI	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198,172,611.71	\$689,336.75	\$0.00	\$0.00	\$198,861,948.46

#### DR 35b p. 3

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Nov'98-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF NOVEMBER 30, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 11/30/98
	WESTERN	KENTUCKY DISTRIBUTION DIVISION					
00	1010000		8 170 77	0.00	0.00	0.00	8 379 77
09	3020000	Franchizes & Contents	119 857 69	0.00	0.00	0.00	119 852.69
•,							
		Total Intangible Plant	128,182.41	0.00	0.00	0.00 `	128,182.41
	L-P PROD	UCTION PLANT					
09	3040010	Land	0.00	0.00	0.00	0.00	0.00
09	3050000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110005	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges & Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Mixing Equipment	0.00	0.00	0.00	0.00	0.00
09	3110016	Piping	0.00	0.00	0.00	0.00	0.00
09	3110022	Storage Equipment	0.00	0.00	0.00	0.00	0.00
09	3110026	Vaporizing Equipment	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0.00	0.00	0.00	0.00	0.00
		Total L-P Production Plant	0.00	0.00	0.00	0.00	0.00
	NATURAL	GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Field Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3,492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	47,162.67
09	3320020	Tributary Lines	528,218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sts. Equip.	0.00	0.00	0.00	0.00	0.00
09	3340000	Field Meas. & Reg. Sta. Equip.	198,468.81	0.00	0.00	0.00	198,468.81
09	3360000	Purification Equipment	44,369.30	0.00	0.00	0.00	44,369.30
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261,126.69	0.00	0.00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09	3510020	Compressor Station Equipment	121,264.59	0.00	0.00	0.00	121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144,554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2,172,799.96	0.00	0.00	0.00	2,172,799.96
09	3520002	Well Equipment	535,976.49	0.00	0.00	0.00	535,976.49
09	3520010	Leaseholds	178,530.09	0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	54,614.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	Tributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470,685.43	0.00	0.00	0.00	470,685.43
09	3550000	Meas. & Reg. Equipment	288,850.55	0.00	0.00	0.00	288,850.55
09	3560000	Purification Equipment	239,929.66	0.00	0.00	0.00	239,929.66
		Total Storage Plant	4,884,110.51	0.00	0.00	0.00	4,884,110.51

#### DR 355 p. 4

ļ

## V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Nov'98-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF NOVEMBER 30, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 11/30/98
	TRANSM	ISSION PLANT					
09	3650010	Land	26.951.25	0.00	0.00	0.00	26.951.25
09	3650020	Rights of Ways	403.419.34	0.00	0.00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	0.00	0.00	0.00	32,921.36
09	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(17,387.32)	0.00	0.00	19,283,668.53
0 <b>9</b>	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas. & Reg. Equipment	2,976,155.16	0.00	0.00	(27,864.23)	2,948,290.93
		Total Transmission Plant	22,809,675.37	(17,387.32)	0.00	(27,864.23)	22,764,423.82
	DISTRIBU	TION PLANT					
09	3740010	Land Town Border	61,709.61	180.00	0.00	0.00	61,889.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783,89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,871.73
09	3750010	Structures & Improvements 1.B.	106,375.76	0.00	0.00	0.00	100,375.70
09	3750002	Improvements	7 517 59	0.00	0.00	0.00	7 51 7 58
09	3750003	Improvements	7,517.50 46 <b>6</b> 01.01	0.00	0.00	0.00	45 591 01
09	3750020	Lano Rights Maine	68 571 269 61	285 51 5 00	0.00	27 864 23	68 884 648 84
09	3780010	Meas & Reg. Sta. Koninment General	1.881 560 42	255,515.00	0.00	0.00	1,907,060,08
09	3790030	Meas. & Reg. Sta. Equipment T.B.	1.650.884.25	0.00	0.00	0.00	1,650,884.25
09	3800000	Services	40.476.571.75	638,509,69	0.00	0.00	41,115,081.44
09	3810000	Meters	18.009.962.39	0.00	0.00	0.00	18,009,962.39
09	3810020	V & P Gauges	109,523.93	0.00	0.00	0.00	109,523.93
09	3820000	Meter Installations	13,197,358.85	199,236.30	0.00	0.00	13,396,595.15
09	3830000	Regulators Service	3,428,991.68	0.00	0.00	0.00	3,428,991.68
09	3830020	Regulators Relief	481,544.54	0.00	0.00	0.00	481,544.54
09	3840000	House Reg. Installations	154,276.36	0.00	0.00	0.00	154,276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2,931,579.79	(12,856.00)	0.00	0.00	2,918,723.79
09	3860000	Other Prop. On Customer Prem.	5,692.66	0.00	0.00	0.00	5,692.66
		Total Distribution Plant	151,169,065.81	1,136,084.65	0.00	27,864.23	152,333,014.69
	GENERAL	PLANT					
09	3890010	Land	44,727.62	. 0.00	0.00	0.00	44,727.62
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.56
09	3900003	Improvements	64,110,56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	1000000	Iotal Energy	1 177 295 70	0.00	0.00	0.00	1 377 285 70
09	3910000	Cenaral Office Fourinment	1,576,086,34	0.00	0.00	0.00	1 576.086.34
09	3918300	Office Furn-Conjers & Typewritters	200.478.91	0.00	0.00	0.00	200,478.91
09	3920000	Transportation Equipment	6.044.073.81	0.00	0.00	0.00	6,044,073.81
09	3920020	Trailers	165,969.90	0.00	0.00	0.00	165,969.90
09	3940077	Tools & Work Equipment	3,064,138.84	0.00	0.00	35.37	3,064,174.21
09	3969377	Ditchers	853,614.76	0.00	0.00	(35.37)	853,579.39
09	3969477	Backhoes	706,022.84	0.00	0.00	0.00	706,022.84
09	3969577	Welders	92,412.75	0.00	0.00	0.00	92,412.75
09	3970000	Communication Equipment	750,059.60	0.00	0.00	0.00	750,059.60
09	3970020	Comm. Equip Fixed Radios	21,697.23	0.00	0.00	0.00	21,697.23
09	3970021	Comm. Equip Mobile Radios	58,023.11	0.00	0.00	0.00	58,023.11
09	3970022	Comm. Equip Telemetering	114,695.01	0.00	0.00	0.00	114,695.01
09	3980000	Miscellaneous Equipment	37,073.03	0.00	0.00	0.00	37,073.03
09	3990000	Oth Tang Prop	0.00	0.00	0.00	0.00	U.UU 404 677 14
09	3998500	Oth Tang Prop - Mr Hardware	397,277.33	0.00	(10.666)	0.00	404,077.14 508 876 11
09 00	3330000	Oth Tang Prop - PC Schware	105 510 00	U.UU 7 647 01	0.00	0.00	101 149 01
09 00	1008900	Oth Tang Pron - Application software	202,210.70	7,047.03	0.00	0.00	222,120.01 20K 003 0K
09	9999117	Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	7,647.03	(7,399.61)	0.00	18,366,491.46
	TOTAL WI	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198,172,611.71	\$1,126,344.36	(\$7,399.61)	\$0.00	\$199,306,355.68



#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Dec'98-A

Į

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF DECEMBER 31, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 12/31/98
	WESTERM	KENTUCKY DISTRIBUTION DIVISION					
09	3010000	Organization	8.329.72	0.00	0.00	0.00	8.329.72
09	3020000	Franchises & Consents	119,852.69	0.00	0.00	0.00	119,852.69
		Total Intangible Plant	128,182.41	0.00	0.00	0.00	128,182.41
	L-P PROD	UCTION PLANT					
09	3040010	Land	0.00	0.00	0.00	0.00	0.00
09	3050000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110005	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges & Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Mixing Equipment	0.00	0.00	0.00	0.00	0.00
09	3110016	Piping	0.00	0.00	0.00	0.00	0.00
09	3110022	Storage Equipment	0.00	0.00	0.00	0.00	0.00
09	3110026	Vaporizing Equipment	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0.00	0.00	0.00	0.00	0.00
		Total L-P Production Plant	0.00	0.00	0.00	0.00	0.00
	NATURAL	, GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Field Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3,492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	47,162.67
09	3320020	Tributary Lines	528,218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sta. Equip.	0.00	0.00	0.00	0.00	0.00
09	3340000	Field Meas. & Reg. Sta. Equip.	198,468.81	0.00	0.00	0.00	198,468.81
09	3360000	Purification Equipment	44,369.30	0.00	0.00	0.00	44,369.30
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261,126.69	0.00	0,00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09	3510020	Compressor Station Equipment	121,264.59	0.00	0.00	0,00	121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144,554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2,172,799.96	0.00	0.00	0.00	2,172,799.96
09	3520002	Weli Equipment	535,976.49	0.00	0.00	0.00	535,976.49
09	3520010	Leaseholds	178,530.09	0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	54,614.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	Tributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470,685.43	0.00	0.00	0.00	470,685.43
09	3550000	Meas. & Reg. Equipment	288,850.55	0.00	0.00	0.00	288,850.55
09	3560000	Purification Equipment	239,929.66	0.00	0.00	0.00	239,929.66
		Total Storage Plant	4,884,110.51	0.00	0.00	0.00	4,884,110.51

,

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF DECEMBER 31, 1998

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 12/31/98
	TRANSMI	SSION PLANT					
09	3650010	Land	26,951.25	0.00	0.00	0.00	26,951.25
09	3650020	Rights of Ways	403,419.34	0.00	0,00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	(18,124.03)	0.00	0.00	14,797.33
69	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(17,387.32)	0.00	0.00	19,283,668.53
09	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas. & Reg. Equipment	2,976,155.16	0.00	0.00	(27,864.23)	2,948,290.93
		Total Transmission Plant	22,809,675.37	(35,511.35)	0.00	(27,864.23)	22,746,299.79
	DISTRIBU	TION PLANT					
09	3740010	Land Town Border	61,709.61	180.00	0.00	0.00	61,889.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783.89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,871.73
09	3750010	Structures & Improvements T.B.	106,375.76	0.00	0.00	0.00	106,375.76
09	3750002	Structures & Improvements Other	0.00	0.00	0.00	0.00	0.00
09	3750003	Improvements	7,517.58	0.00	0.00	0.00	7,517.58
09	3750020	Land Rights	46,591.01	0.00	0.00	0.00	46,591.01
09	3760000	Malus	68,571,269.61	270,409.34	0.00	27,864.23	68,869,543.18
09	3780010	Meas. & Reg. Sta. Equipment General	1,881,560.42	25,499.66	0.00	0.00	1,907,060.08
09	3790030	Meas. & Reg. Sta. Equipment T.B.	1,650,884.25	0.00	0.00	0.00	1,650,884.25
09	3800000	Services	40,470,571.75	984,501.81	0.00	0.00	41,401,073.30
09	3810000	Meters	10,009,902.39	0.00	0.00	0.00	10,009,902.39
09	3810020	V & F Gauges	107,525.55	306 870 37	0.00	0.00	13 504 179 17
09	3830000	Pagulatora Service	3 478 901 68	0.00	0.00	0.00	3 428 991.68
09	3830020	Regulators Belief	481.544.54	0.00	0.00	0.00	481.544.54
09	3840000	House Reg. Installations	154,276,36	0.00	0.00	0.00	154.276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2.931.579.79	(12.856.00)	0.00	0.00	2,918,723.79
09	3860000	Other Prop. On Customer Prem.	5,692.66	0.00	0.00	0.00	5,692.66
		Total Distribution Plant	151,169,065.81	1,574,555.13	0.00	27,864.23	152,771,485.17
	GENERAL	PLANT					
09	3890010	Land	44,727.62	0.00	0.00	0.00	44,727.62
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.56
09	3900003	Improvements	64,110.56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	3900005	Total Energy	0.00	0.00	0.00	0.00	0.00
09	3900009	Inprov. to Leased Premises	1,377,285.70	0.00	0.00	0.00	1,377,285.70
09	3910000	General Office Equipment	1,576,086.34	. 0.00	0.00	0.00	1,576,086.34
09	3918300	Office Furn-Coplers & Typewritters	200,478.91	0.00	0.00	0.00	200,478.91
09	3920000	Transportation Equipment	6,044,073.81	0.00	0.00	0.00	6,044,073.81
09	3920020	Trailers	165,969.90	0.00	0.00	0.00	165,969.90
09	3940077	loois & work Equipment	3,004,138.84	0.00	0.00	33.37	3,004,174.21
09	3969377	Ditchers	833,014.70	0.00	0.00	(33.37)	706 077 84
09	39094//	Dacknoes Waldam	/00,022.04	0.00	0.00	0.00	00,022.84
09	1070000	Communication Equipment	750.050.60	0.00	0.00	0.00	750 059 60
09	3970000	Communication Equipatent	750,055.00	0.00	0.00	0.00	21 697.23
00	3970020	Comm Equip - Mobile Padios	58 023 11	0.00	0.00	0.00	58.023.11
09	3970022	Comm Equip Telemetering	114,695,01	0.00	0.00	0.00	114.695.01
09	3980000	Miscellaneous Equipment	37.073.03	0.00	0.00	0.00	37.073.03
09	3990000	Oth Tang Prop	0.00	0.00	0.00	0.00	0.00
09	3998500	Oth Tang Prop - MF Hardware	397,277.53	0.00	(7,399.61)	0.00	404,677.14
09	3998600	Oth Tang Prop - PC Hardware	504,426.33	.000	0.00	0.00	504,426.33
09	3998700	Oth Tang Prop - PC Software	185,510.98	7,647.03	0.00	0.00	193,158.01
09	3998800	Oth Tang Prop - Application software	206,993.96	0.00	0.00	0.00	206,993.96
09	9999117	Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	7,647.03	(7,399.61)	0.00	18,366,491.46
	TOTAL WI	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198,172,611.71	\$1,546,690.81	(\$7,399.61)	\$0.00	\$199,726,702.13



#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Jan'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF JANUARY 31, 1999

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 01/31/99
	WESTERN	KENTUCKY DISTRIBUTION DIVISION					
	INTANGI	BLE PLANT					
09	3010000	Organization	8,329.72	0.00	0.00	0.00	8,329.72
09	3020000	Franchises & Consents	119,852.69	0.00	0.00	0.00	119,852.69
		Total Intangible Plant	128,182.41	0.00	0.00	0.00 ``	128,182.41
	L-P PROD	UCTION PLANT					
09	3040010	Land '	0.00	0.00	0.00	0.00	0.00
09	3050000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110005	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges & Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Mixing Equipment	0.00	0.00	0.00	0.00	0.00
09	3110016	Piping	0.00	0.00	0.00	0.00	0.00
09	3110022	Storage Equipment	0.00	0.00	0.00	0.00	0.00
09	3110026	Vaporizing Equipment	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0.00	0.00	0.00	0.00	0.00
		Total L-P Production Plant	0.00	0.00	0.00	0.00	0.00
	NATURAL	, GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Field Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3.492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	47,162.67
09	3320020	Tributary Lines	528.218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sta. Equip.	0.00	0.00	0.00	0.00	0.00
09	3340000	Field Meas, & Reg. Sta. Equip.	198.468.81	0.00	0.00	0.00	198,468.81
09	3360000	Purification Equipment	44,369.30	0.00	0.00	0.00	44,369.30
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261.126.69	0.00	0.00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09	3510020	Compressor Station Equipment	- 121,264.59	0.00	0.00	0.00	. 121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144,554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2,172,799.96	0.00	0.00	0.00	2,172,799.96
09	3520002	Well Equipment	535,976.49	0.00	0.00	0.00	535,976.49
09	3520010	Leaseholds	178,530.09	0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	54,614.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	Tributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470,685.43	0.00	0.00	0.00	470,685.43
09	3550000	Meas. & Reg. Equipment	288,850.55	0.00	0.00	0.00	288,850.55
09	3560000	Purification Equipment	239,929.66	0.00	0.00	0.00	239,929.66
		Total Storage Plant	4,884,110.51	0.00	0.00	0.00	4,884,110.51

# V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Jan'99-A

# ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF JANUARY 31, 1999

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 01/31/99
	TRANSMI	SSION PLANT					
09	3650010	Land	26,951,25	0.00	0.00	0.00	26,951,25
09	3650020	Rights of Ways	403,419.34	0.00	0.00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	(18,124.03)	0.00	0.00	14,797.33
09	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(17,387.32)	0.00	0.00	19,283,668.53
09	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas, & Reg. Equipment	2,976,155.16	0.00	0.00	(27,864.23)	2,948,290.93
		Total Transmission Plant	22,809,675.37	(35,511.35)	0.00	(27,864.23)	22,746,299.79
	DISTRIBU	TION PLANT					
09	3740010	Land Town Border	61,709.61	180.00	0.00	0.00	61,889.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783.89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,871.73
09	3750010	Structures & Improvements T.B.	106,375.76	0.00	0.00	0.00	106,375.76
09	3750002	Structures & Improvements Other	0.00	0.00	0.00	0.00	0.00
09	3750003	Improvements	7,51 7.58	0.00	0.00	0.00	7,517.58 46 591 01
09	3750020	Land Rights	40,391.01	310 284 66	0.00	27 864 23	68 979 418 50
09	3780010	Mens & Reg Sta Fouinment General	1 881 560 42	25 499.66	0.00	0.00	1,907,060.08
09	3790030	Meas & Reg. Sta. Equipment T.B.	1 650 884 25	0.00	0.00	0.00	1,650,884,25
09	3800000	Services	40.476.571.75	1.247.530.55	0.00	0.00	41,724,102.30
09	3810000	Meters	18.009.962.39	0.00	0.00	0.00	18,009,962.39
09	3810020	V & P Gauges	109,523.93	0.00	0.00	0.00	109,523.93
09	3820000	Meter Installations	13,197,358.85	366,045.93	0.00	0.00	13,563,404.78
09	3830000	Regulators Service	3,428,991.68	0.00	0.00	0.00	3,428,991.68
09	3830020	Regulators Relief	481,544.54	0.00	0.00	0.00	481,544.54
09	3840000	House Reg. Installations	154,276.36	0.00	0.00	0.00	154,276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2,931,579.79	(12,856.00)	0.00	0.00	2,918,723.79
09	3860000	Other Prop. On Customer Prem.	5,692.66	0.00	0.00	0.00	5,692.66
		Total Distribution Plant	151,169,065.81	1,956,684.80	0.00	27,864.23	153,153,614.84 (153,153,614.84)
	GENERAL	PLANT					0.00
09	3890010	Land	44,727.62	0.00	0.00	0.00	44,727.62
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.56
09	3900003	Improvements	64,110.56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	3900005	I otal Energy	0.00	0.00	0.00	0.00	1 277 296 70
09	3900009	Inprov. to Leased Premises	1,377,285.70	0.00	0.00	0.00	1,577,285.70
09	3910000	Office Russ. Conject & Typewritters	200 478 91	0.00	0.00	0.00	200 478 91
09	3920000	Transportation Registrant	6 044 073 81	0.00	0.00	0.00	6 044 073 81
09	3920020	Trailers	165,969,90	0.00	0.00	0.00	165,969,90
09	3940077	Tools & Work Equipment	3.064.138.84	0.00	0.00	35.37	3.064.174.21
09	3969377	Ditchers	853,614.76	0.00	0.00	(35.37)	853,579.39
09	3969477	Backhoes	706,022.84	0.00	0.00	0.00	706,022.84
09	3969577	Welders	92,412.75	0.00	0.00	0.00	92,412.75
09	3970000	Communication Equipment	750,059.60	0.00	0.00	0.00	750,059.60
09	3970020	Comm. Equip Fixed Radios	21,697.23	0.00	0.00	0.00	21,697.23
09	3970021	Comm. Equip Mobile Radios	58,023.11	0.00	0.00	0.00	58,023.11
09	3970022	Comm. Equip Telemetering	114,695.01	0.00	0.00	0.00	114,695.01
09	3980000	Miscellaneous Equipment	37,073.03	0.00	0.00	0.00	37,073.03
09	3990000	Oth Tang Prop	0.00	0.00	0.00	0.00	404 677 4 4
09	J998500	Oth Tang Prop - MF Hardware	397,277.53	0.00	(10.666')	0.00	404,0//.14
09 00	1009700	Oth Tang Prop - PC Hardware	304,440,33	0.00 7 647 03	0.00	0.00	103 152 01
4V 00	3998/00	Oth Tang Prop - Application software	100,010,90	7,047.03	0.00	0.00	206.993 96
09	9999117	Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	7,647.03	(7,399.61)	0.00	18,366,491.46
	TOTAL WI	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198,172,611.71	\$1,928,820.48	(\$7,399.61)	\$0.00	\$200,108,831.80

.







#### DR 355 p. 9

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Feb'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF FEBRUARY 28, 1999

				Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 02/28/99
	WESTERN	KENTUCKY DISTRIBUTION DIVISION	·				
	INTANGI	ILE PLANT				0.00	0 340 77
09	3010000	Organization	8,329.72	0.00	0.00	0.00	8,329.72
09	3020000	Franchises & Consents	119,852.69	0.00	0.00	0.00	119,852.09
		Total Intangible Plant	128,182.41	0.00	0.00	0.00 `	128,182.41
	L-P PROD	UCTION PLANT					
09	3040010	Land	0.00	0.00	0.00	0.00	0.00
09	3050000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110005	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges & Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Mixing Equipment	0.00	0.00	0.00	0.00	0.00
09	3110016	Piping	0.00	0.00	0.00	0.00	0.00
09	3110022	Storage Equipment	0.00	0.00	0.00	0.00	0.00
09	3110026	Vaporizing Equipment	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0,00	0.00	0.00	0.00	0.00
		Total L-P Production Plant	0.00	0.00	0.00	0.00	0.00
	NATURAL	GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Field Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3,492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	47,162.67
09	3320020	Tributary Lines	528,218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sta. Equip.	0.00	0.00	0.00	0.00	0.00
09	3340000	Field Meas. & Reg. Sta. Equip.	198,468.81	0.00	0.00	0.00	198,468.81
09	3360000	Purification Equipment	44,369.30	0.00	0.00	0.00	44,369.30
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261,126.69	0.00	0.00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09	3510020	Compressor Station Equipment	121,264.59	0.00	0.00	0.00	121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144.554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2.172.799.96	23,675.63	0.00	0.00	2,196,475.59
09	3520002	Well Equipment	535,976,49	0.00	0.00	0.00	535,976.49
09	3520010	Leascholds	178,530.09	0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	54,614.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	Tributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470.685.43	0.00	0.00	0.00	470,685.43
09	3550000	Meas. & Reg. Equipment	288,850.55	0.00	0.00	0.00	288,850.55
09	3560000	Purification Equipment	239,929.66	0.00	0.00	0.00	239,929.66
		Total Storage Plant	4,884,110.51	23,675.63	0.00	0.00	4,907,786.14



.

## V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Feb'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF FEBRUARY 28, 1999

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 02/28/99
	TRANSM	ISSION PLANT					
09	3650010	Land	26,951.25	0.00	0.00	0.00	26,951.25
09	3650020	Rights of Ways	403,419.34	0.00	0.00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	(18,124.03)	0.00	0.00	14,797.33
09	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(17,387.32)	0.00	0.00 '	19,283,668.53
09	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas. & Reg. Equipment	2,976,155.16	0.00	0.00	(27,864.23)	2,948,290.93
		Total Transmission Plant	22,809,675.37	(35,511.35)	0.00	(27,864.23)	22,746,299.79
	DISTRIBU	JTION PLANT					
09	3740010	Land Town Border	61,709.61	204.00	0.00	0.00	61,913.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783.89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,8/1./3
09	3750000	Structures & Improvements 1.B.	100,375.70	0.00	0.00	0.00	100,373.70
09	3750002	Improvements	7 51 7 58	0.00	0.00	0.00	7 517 58
09	3750020	Land Rights	46 591 01	0.00	0.00	0.00	46 591.01
09	3760000	Mains	68.571.269.61	742.110.90	92.958.31	27.864.23	69.248.286.43
09	3780010	Meas, & Reg. Sta, Equipment General	1.881,560.42	27,007.78	0.00	0.00	1,908,568,20
09	3790030	Meas. & Reg. Sta. Equipment T.B.	1,650,884.25	0.00	1,546.92	0.00	1,649,337.33
09	3800000	Services	40,476,571.75	1,397,411.48	71,570.10	0.00	41,802,413.13
09	3810000	Meters	18,009,962.39	0.00	292,116.15	0.00	17,717,846.24
09	3810020	V & P Gauges	109,523.93	0.00	0.00	0.00	109,523.93
09	3820000	Meter Installations	13,197,358.85	387,996.83	3,087.11	0.00	13,582,268.57
09	3830000	Regulators Service	3,428,991.68	0.00	0.00	0.00	3,428,991.68
09	3830020	Regulators Relief	481,544.54	0.00	0.00	0.00	481,544.54
09	3840000	House Reg. Installations	154,276.36	0.00	0.00	0.00	154,276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2,931,579.79	22,129.99	0.00	0.00	2,953,709.78
09	3860000	Other Prop. On Customer Prem.	3,692.00	0.00	0.00	0.00	5,092.00
		Total Distribution Plant	151,169,065.81	2,576,860.98	461,278.59	27,864.23	153,312,512.43 (153,153,614.84)
	GENERAL	, PLANT					158,897.59
09	3890010	Land	44,727.62	. 0.00	0.00	0.00	44,727.62
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.56
09	3900003	Improvements	64,110.56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	3900003	Lotal Energy	1 177 446 70	0.00	0.00	0.00	1 377 295 70
09	3010000	Conorol Office Fouriement	1,576,096,34	0.00	1 464 75	0.00	1 574 621 59
09	3918300	Office Furn-Conjers & Typewritters	200 478 91	0.00	1,404.75	0.00	200 478.91
09	3920000	Transportation Equipment	6.044.073.81	0.00	15.217.44	0.00	6.028.856.37
09	3920020	Trailers	165,969,90	0.00	0.00	0.00	165,969.90
09	3940077	Tools & Work Equipment	3,064,138.84	1,181.58	4,300.04	35.37	3,061,055.75
09	3969377	Ditchers	853,614.76	0.00	22,556.15	(35.37)	831,023.24
09	3969477	Backhoes	706,022.84	0.00	0.00	0.00	706,022.84
09	3969577	Welders	92,412.75	0.00	0.00	0.00	92,412.75
09	3970000	Communication Equipment	750,059.60	0.00	2,345.28	0.00	747,714.32
09	3970020	Comm. Equip Fixed Radios	21,697.23	0.00	0.00	0.00	21,697.23
09	3970021	Comm. Equip Mobile Radios	58,023.11	0.00	0.00	0.00	58,023.11
09	3970022	Comm. Equip Telemetering	114,695.01	0.00	0.00	0.00	114,695.01
09	3980000	Miscellaneous Equipment	37,073.03	0.00	0.00	0.00	37,073.03
09	3990000	Oth Tang Prop	0.00	0.00	U.UU (** 100 41)	0.00	0.00 40.4 677 1 4
09	3008400	Oth Tang Prop - MF Hardware	597,277.53	0.00	(10.666'1)	0.00	909,077.14
09	3778000	Oth Tang Prop - PC Hardware	304,420.33	0.00	0.00	0.00	304,420.33 103 400 FF
09	1008900	Oth Tang Prop - re software	202 001 02	0 0 0	0.00	0.00 A AA	205,470.30
09	9999117	Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	9,161.16	38,484.05	0.00	18,322,121.93
	TOTAL W	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198,172,611.71	\$2,574,186.42	\$499,762.64	\$0.00	\$200,247,035.49



,

1



#### DR 35b p. 11

•

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Mar'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF MARCH 31, 1999

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 03/31/99
	WESTER	KENTUCKY DISTRIBUTION DIVISION					
00	1010000	Organization	8 120 77	0.00	0.00	0.00	8 179 77
09	3020000	Franchises & Consents	119,852.69	0.00	0.00	0.00	119,852.69
		Total Intangible Plant	128,182.41	0.00	0.00	0.00	128,182.41
	L-P PROD	UCTION PLANT					
09	3040010	Land	0.00	0.00	0.00	0.00	0.00
09	3050000	Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3110005	Compression Equipment	0.00	0.00	0.00	0.00	0.00
09	3110011	Gauges & Instruments	0.00	0.00	0.00	0.00	0.00
09	3110012	Mixing Equipment	0.00	0.00	0.00	0.00	0.00
09	3110016	Piping	0.00	0.00	0.00	0.00	0.00
09	3110022	Storage Equipment	0.00	0.00	0.00	0.00	0.00
09	3110026	Vaporizing Equipment	0.00	0.00	0.00	0.00	0.00
09	3190000	Gas Mixing Equipment	0.00	0.00	0.00	0.00	. 0,00
		Total L-P Production Plant	0.00	0.00	0.00	0.00	0.00
	NATURAL	GAS PRODUCTION PLANT					
09	3250020	Producing Leaseholds	2,352.50	0.00	0.00	0.00	2,352.50
09	3250040	Rights of Ways	6,069.04	0.00	0.00	0.00	6,069.04
09	3250050	Land	0.00	0.00	0.00	0.00	0.00
09	3250060	Mineral Deeds	0.00	0.00	0.00	0.00	0.00
09	3280000	Fleid Meas. & Reg. Structures	0.00	0.00	0.00	0.00	0.00
09	3290000	Other Structures & Improvements	0.00	0.00	0.00	0.00	0.00
09	3300000	Gas Well Construction	0.00	0.00	0.00	0.00	0.00
09	3310000	Production Gas Wells Equipment	3,492.47	0.00	0.00	0.00	3,492.47
09	3320010	Field Lines	47,162.67	0.00	0.00	0.00	47,162.67
09	3320020	Tributary Lines	528,218.00	0.00	0.00	0.00	528,218.00
09	3330000	Prod. Field Comp. Sta. Equip.	0.00	0.00	0.00	0.00	0.00
09	3340000	Field Meas. & Reg. Sta. Equip.	198,468.81	0.00	0.00	0.00	198,468.81
09	3360000	Purification Equipment	44,369.30	0.00	0.00	0.00	44,369.30
		Total Natural Gas Prod. Plant	830,132.79	0.00	0.00	0.00	830,132.79
	STORAGE	PLANT					
09	3500010	Land	261,126.69	0.00	0.00	0.00	261,126.69
09	3500020	Rights of Ways	4,681.58	0.00	0.00	0.00	4,681.58
09·	3510020	Compressor Station Equipment	121,264.59	0.00	0.00	0.00	121,264.59
09	3510030	Meas. & Reg. Sta. Structures	23,138.38	0.00	0.00	0.00	23,138.38
09	3510040	Other Structures	144,554.11	0.00	0.00	0.00	144,554.11
09	3520001	Well Construction	2,172,799.96	23,675.63	0.00	0.00	2,196,475.59
09	3520002	Well Equipment	535,976.49	0.00	0.00	0.00	535,976.49
09	3520010	Leaseholds	178,530.09	. 0.00	0.00	0.00	178,530.09
09	3520011	Storage Rights	54,614.27	0.00	0.00	0.00	54,614.27
09	3530010	Field Lines	178,500.50	0.00	0.00	0.00	178,500.50
09	3530020	I ributary Lines	209,458.21	0.00	0.00	0.00	209,458.21
09	3540000	Compressor Station Equipment	470,685.43	0.00	0.00	0.00	470,085.43
09 09	3560000	Meas. & Reg. Equipment Purification Equipment	288,850.55	0.00	0.00	0.00	239,929.66
		Total Storage Plant	4,884,110.51	23,675.63	0.00	0.00	4,907,786.14

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Mar'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF MARCH 31, 1999

					Year to Date		
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 03/31/99
	WESTERI TRANSMI	N KENTUCKY DISTRIBUTION DIVISION ISSION PLANT					
09	3650010	Land	26,951.25	0.00	0.00	0.00	26,951.25
09	3650020	Rights of Ways	403,419.34	0.00	0.00	0.00	403,419.34
09	3660020	Structures & Improvements	32,921.36	(18,124.03)	0.00	0.00	14,797.33
09	3660030	Other Structures	69,172.41	0.00	0.00	0.00	69,172.41
09	3670000	Mains	19,301,055.85	(17,387.32)	0.00	0.00	19,283,668.53
09	3680000	Compressor Station Equipment	0.00	0.00	0.00	0.00	0.00
09	3690010	Meas. & Reg. Equipment	2,976,155.16	0.00	0.00	(27,864.23)	2,948,290.93
		Total Transmission Plant	22,809,675.37	(35,511.35)	0.00	(27,864.23)	22,746,299.79
	DISTRIBU	JTION PLANT					
09	3740010	Land Town Border	61,709.61	204.00	0.00	0.00	61,913.61
09	3740030	Land Other	2,783.89	0.00	0.00	0.00	2,783.89
09	3740020	Right of Way	44,871.73	0.00	0.00	0.00	44,871.73
09	3750010	Structures & Improvements T.B.	106,375.76	0.00	0.00	0.00	106,375.76
09	3750002	Structures & Improvements Other	0.00	0.00	0.00	0.00	0.00
09	3750003	Improvements	7,517.58	0.00	0.00	0.00	7,517.58
09	3750020	Land Rights	46,591.01	0.00	0.00	0.00	46,591.01
09	3760000	Mains	68,571,269.61	794,573.50	96,202.52	27,864.23	69,297,504.82
09	3780010	Meas. & Reg. Sta. Equipment General	1,881,560.42	27,007.78	0.00	0.00	1,908,568.20
09	3790030	Meas. & Reg. Sta. Equipment T.B.	1,650,884.25	0.00	1,546.92	0.00	1,649,337.33
09	3800000	Services	40,476,571.75	1,543,506.66	139,581.78	0.00	41,880,496.63
09	3810000	Meters	18,009,962.39	0.00	292,116.15	0.00	17,717,846.24
09	3810020	V & P Gauges	109,523.93	0.00	0.00	0.00	109,523.93
09	3820000	Meter Installations	13,197,358.85	387,996.83	4,512.78	0.00	13,580,842.90
09	3830000	Regulators Service	3,428,991.68	0.00	0.00	0.00	3,428,991.68
09	3830020	Regulators Relief	481,544.54	0.00	0.00	0.00	481,544.54
09	3840000	House Reg. Installations	154,276.36	0.00	0.00	0.00	154,276.36
09	3850010	Ind. Meas. & Reg. Sta. Equipment	2,931,579.79	22,129.99	0.00	0.00	2,953,709.78
09	3860000	Other Prop. On Customer Prem.	5,692.66	0.00	0,00	0.00	5,692.66
		Total Distribution Plant	151,169,065.81	2,775,418.76	533,960.15	27,864.23	153,438,388.65
•••	GENERAL	_ PLANT		<b>A A A</b>			
09	3890010	Land	44,727.62	0.00	0.00	0.00	44,717.02
09	3900002	Structures & Improvements	182,161.56	0.00	0.00	0.00	182,161.50
09	3900003	Improvements	64,110.56	0.00	0.00	0.00	64,110.56
09	3900004	Air Conditioning Equipment	9,771.49	0.00	0.00	0.00	9,771.49
09	3900003	Iotal Energy	0.00	0.00	0.00	0.00	0.00
09	3900009	Inprov. to Leased Premises	1,377,285.70	. 0.00	0.00	0.00	1,377,285.70
09	3910000	General Office Equipment	1,576,086.34	0.00	1,464.75	0.00	1,574,621.59
09	3918300	Office Furn-Copiers & Typewritters	200,478.91	0.00	0.00	0.00	200,478.91
09	3920000	Iransportation Equipment	6,044,073.81	0.00	15,217.44	0.00	0,028,850.37
09	3920020	I railers	165,969.90	0.00	0.00	0.00	103,909.90
09	3940077	loois & work Equipment	3,064,138.84	1,181.38	4,300.04	35.37	3,001,033.73
09	3909377	Dichers	853,014.70	0.00	22,330.13	(35.37)	706 022 84
09	3969477	Backhoes	706,022.84	0.00	0.00	0.00	700,022.84
09	3969377	Welders	92,412.75	0.00	0.00	0.00	92,412.75
09	3970000	Communication Equipment	750,059.60	0.00	2,345.28	0.00	141,114.32
09	3970020	Comm. Equip Fixed Radios	21,697.23	0.00	0.00	0.00	21,097.23
09	3970021	Comm. Equip Mobile Kadlos	58,025.11	0.00	0.00	0.00	56,025.11
09	3970022	Comm. Equip Telemetering	114,695.01	0.00	0.00	0.00	114,095.01
09	3980000	Nilscellaneous Equipment	37,073.03	0.00	0.00	0.00	3/,0/3.03
09	3990000	Oth Tang Prop	0.00	· 0.00	U.UU (7 300 (1)	0.00	U.UU 404 497 44
09	1008600	Oth Tang Prop - MF Hardware	397,277.53	0.00	(1,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	0.00	404,077.14
09	3998600	Ota lang Prop - PC Hardware	504,426.33	0.00	0.00	0.00	304,440.33
09	3998700	Oth Tang Prop - PC Software	185,510.98	7,979.58	0.00	0.00	193,490.56
09 09	99999117	Our rang Frop - Application software Cushion Gas	1,694,832.96	0.00	0.00	0.00	1,694,832.96
		Total General Plant	18,351,444.82	9,161.16	38,484.05	0.00	18,322,121.93
	TOTAL W	ESTERN KENTUCKY DISTRIBUTION DIV.	\$198.172.611.71	\$2,772,744.20	\$572,444,20	\$0.00	\$200,372,911.71



I.





#### R:\0113600 Plant Accounting\@Plant Accounting\TOPPER\Monthly Reports\Pg4 -Plant\[PG4 - 1999.xls]Oct'98-A

# ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF OCTOBER 31, 1998

				Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 10/31/98
	ATMOS G	ENERAL OFFICE					
	GENERAL	, PLANT					
02	3890010	Land	0.00	0.00	0.00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00	0.00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00 `	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn. & Equipment	3,573,799.65	0.00	0.00	0.00	3,573,799.65
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	0.00	39,996,823.23
	TOTAL AT	IMOS GENERAL OFFICE DIV.	\$39,996,823.23	\$0.00	\$0.00	\$0.00	\$39,996,823.23



i T







#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Nov'98-A

I.

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF NOVEMBER 30, 1998

				Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 11/30/98
	ATMOS G	ENERAL OFFICE					
	GENERAL	, PLANT					
02	3890010	Land	0.00	0.00	0.00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00	0.00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00 `	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn. & Equipment	3,573,799.65	0.00	0.00	0.00	3,573,799.65
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	0.00	39,996,823.23
	TOTAL AT	MOS GENERAL OFFICE DIV.	\$39,996,823.23	\$0.00	\$0.00	\$0.00	\$39,996,823.23

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Dec'98-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF DECEMBER 31, 1998

	_			Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 12/31/98
	ATMOS G	ENERAL OFFICE					
	GENERAL	PLANT					
02	3890010	Land	0.00	0.00	0.00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00 `	0,00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn, & Equipment	3,573,799.65	0.00	0.00	0.00	3,573,799.65
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	0.00	39,996,823.23
	TOTAL AT	MOS GENERAL OFFICE DIV.	\$39,996,823.23	\$0.00	\$0.00	\$0.00	\$39,996,823.23

#### DR 355 p. 16

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Jan'99-A

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF JANUARY 31, 1999

				Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 01/31/99
	ATMOS GI	ENERAL OFFICE					
	GENERAL	PLANT					
02	3890010	Land	0.00	0.00	0.00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00 🔪	0.00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn. & Equipment	3,573,799.65	0.00	0.00	0.00	3,573,799.65
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	0.00	39,996,823.23
	TOTAL AT	MOS GENERAL OFFICE DIV.	\$39,996,823.23	\$0.00	\$0.00	\$0.00	\$39,996,823.23

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xis]Feb'99-A

| :

,

#### ATMOS ENERGY CORPORATION PROPERTY, PLANT AND EQUIPMENT AS OF FEBRUARY 28, 1999

.

			-	Year to Date			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 02/28/99
	ATMOS G	ENERAL OFFICE					
	GENERAL	, PLANT					
02	3890010	Land	0.00	0.00	0,00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00 🔪	0.00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn. & Equipment	3,573,799.65	0.00	0.00	0.00	3,573,799.65
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	0.00	39,996,823.23
	TOTAL AT	IMOS GENERAL OFFICE DIV.	\$39,996,823.23	\$0.00	<u>\$0.00</u>	\$0.00	\$39,996,823.23

#### DR 355 p. 18

#### V:\TOPPER-PLANT\Monthly Reports\PG4\[PG4 - 1999.xls]Mar'99-A

1	
	_
<b>N</b>	

ATMOS ENERGY CORPORATION
PROPERTY, PLANT AND EQUIPMENT
AS OF MARCH 31, 1999

				·			
			Balance 9/30/98	Additions	Retirements	Transfers	Balance 03/31/99
	ATMOS G	ENERAL OFFICE				······	
	GENERAL	PLANT					
02	3890010	Land	0.00	0.00	0.00	0.00	0.00
02	3900002	Structures - Brick	0.00	0.00	0.00	0.00	0.00
02	3900003	Improvements	0.00	0.00	0.00	0.00 🝾	0.00
02	3900004	Air Conditioning Equipment	0.00	0.00	0.00	0.00 `	0.00
02	3900009	Improv. to Leased Premises	6,030,709.19	0.00	0.00	0.00	6,030,709.19
02	3910000	Office Furn. & Equipment	3,573,799.65	0.00	0.00	(514.68)	3,573,284.97
02	3918300	Office Furn Copiers & Typewriters	1,132,451.04	0.00	0.00	0.00	1,132,451.04
02	3920000	Transportation Equipment	55,387.90	0.00	0.00	0.00	55,387.90
02	3930000	Stores Equipment	6,063.14	0.00	0.00	0.00	6,063.14
02	3940077	Tools & Work Equipment	33,041.97	0.00	0.00	0.00	33,041.97
02	3970000	Comm. Equipment Telephones	805,561.16	0.00	0.00	0.00	805,561.16
02	3980000	Miscellaneous Equipment	647,851.53	0.00	0.00	0.00	647,851.53
02	3990000	Oth Tang Prop	61,661.77	0.00	0.00	0.00	61,661.77
02	3998400	Oth Tang Prop - CPU	1,095,465.10	0.00	0.00	0.00	1,095,465.10
02	3998500	Oth Tang Prop - MF Hardware	1,171,886.44	0.00	0.00	0.00	1,171,886.44
02	3998600	Oth Tang Prop - PC Hardware	3,719,347.61	0.00	0.00	0.00	3,719,347.61
02	3998700	Oth Tang Prop - PC Software	990,547.27	0.00	0.00	0.00	990,547.27
02	3998800	Oth Tang Prop - Application software	17,871,655.74	0.00	0.00	0.00	17,871,655.74
02	3998900	Oth Tang Prop - System software	2,801,393.72	0.00	0.00	0.00	2,801,393.72
		Total General Plant	39,996,823.23	0.00	0.00	(514.68)	39,996,308.55
TOTAL ATMOS GENERAL OFFICE DIV.			\$39,996,823.23	\$0.00	\$0.00	(\$514.68)	\$39,996,308.55

# WKG RETIREMENTS FOR THE PERIOD OCT 1998 - MAR 1999

BU	Div	Loc	Approp	Description	FERC	Amount
40	09	221	918144	530-EAST LOCUST MAIN RETIREMENT	37601	166.89
40	09	311	918274	311 - NORTHVIEW ST. MAIN RELOCATION	37601	2,505.01
40	09	500	432	2" STEEL MAIN RETIRE		7.55
40	09	500	918163	WEST PARRISH & FREDERICA STREET	37601	2,297.55
40	09	500	918424	CARTER RD. MIDDLEGROUND TO PARRISH	37601	18,710.54
40	09	505	918166	OLD U.S.60 HARDINSBURG 2" STL. REPL.	37601	6,778.74
40	09	515	416	1" STEEL MAIN RETIRE	37601	22.00
40	09	515	432	2" STEEL MAIN RETIRE	37601	623.42
40	09	515	438	4" STEEL MAIN RETIRE	37601	784.32
40	09	515	918551	515 - S. SEMINARY ST. RELOCATION	37601	370.29
40	09	519	918541	519 - WALNUT ST. RELOCATION	37601	278.14
40	09	530	918563	546-LEAK REPAIR CRAYNE,	37601	1,019.47
40	09	533	918492	EDDYVILLE TRANS. RELOCATION	37601	512.73
40	09	550	918489	311 - MORGAN/CORNELL RELOCATION III	37601	248.91
40	09	555	918057	321 - SOUTH STREET MAIN RETIREMENT	37601	2,140.59
40	09	560	432	2" STEEL MAIN RETIRE	37601	1,428.13
40	09	560	436	3" STEEL MAIN RETIRE	37601	76.50
40	09	560	918467	560 - STONEHENGE 8" RELOCATION	37601	415.94
40	09	565	918587	565- SOUTH HIGH STREET REPLACEMENT	37601	2,112.80
40	09	580	432	2" STEEL MAIN RETIRE	37601	941.79
40	09	580	438	4" STEEL MAIN RETIRE	37601	481.60
40	09	580	918130	611 BONTA TRAILER COURT RETIREMENT	37601	71.99
40	09	590	918559	HARRODSBURG TOWN BORDER RELOCATION	37601	1,073.15
40	09	595	918279	651 ISAAC SHELBY DRIVE RELOCATION	37601	144.42
40	09	622	916962	622 WALNUT STREET REPLACEMENT	37601	2,448.30
40	09	500	632	2" PLASTIC MAIN RETI	37602	135.35
40	09	500	638	4" PLASTIC MAIN RETI	37602	168.60
40	09	515	632	2" PLASTIC MAIN RETI	37602	80.95
40	09	515	918336	515-HWY-41 HWY RELOCATION	37602	40,836.20
40	09	549	918112	655 EVERGREEN RELOCATION	37602	7,045.41
40	09	550	632	2" PLASTIC MAIN RETI	37602	24.55
40	09	550	918327	COLDSPRING ROAD RELOCATION	37602	669.58
40	09	580	632	2" PLASTIC MAIN RETI	37602	81.01
40	09	588	918427	622 INDUSTRY DR RELOCATION	37602	1,500.10
				TOTAL MAINS		96,202.52
						· · · · · · · · · · · · · · · · · · ·
40	09	500	918228	ODORIZER REPLACEMENT	37905	189.82
40	09	515	918479	201-REPLACE WICK TYPE ODORANT TANKS	37905	473.52
40	09	515	918481	201-REPLACE WICK TYPE ODORANT TANKS	37905	883.58
				TOTAL M & R STATION EQUIPMENT		1,546.92
40	09	500	212	SERVICE RETIRE	38000	44,274.42
40	09	550	212	SERVICE RETIRE	38000	47,742.99
40	09	580	212	SERVICE RETIRE	38000	47,564.37
				TOTAL SERVICES		139,581.78
40	09	500	917926	WKG-METERS FOR 1998(CLASS 1-9)	38100	40,586.30
40	09	500	917928	WKG-METERS FOR 1998(CLASS 1-9)	38100	7,539.57
40	09	500	917956	HITACHI AUTOMOTIVE HARRODSBURG	38100	152.54
40	09	500	917958	EFM FOR TRIAD INDUSTRIES B.G.	38100	48.83

# WKG RETIREMENTS FOR THE PERIOD OCT 1998 - MAR 1999

BU	Div	Loc	Approp	Description	FERC	Amount
40	09	500	917974	98EFM-FIELD PACKING (OWB.)	38100	316.29
40	09	500	917977	98EFM-PRO-WASH (FRANKLIN)	38100	75.00
40	09	500	918010	1998EFM-SKF BEARING (GLASGOW)	38100	473.68
40	09	500	918143	WKG-METERS FOR 1998(CLASS 1-9)	38100	238,367.95
40	09	500	918218	1998 EFM ELKTON DIE CASTING	38100	1,097.54
40	09	500	918265	1998 EFM-ROGERS GROUP(HOPKINSVILLE)	38100	947.35
40	09	500	918417	EFM EQUIPMENT FOR G.EOBORO	38100	316.29
40	09	500	918419	EFM EQUIPMENT FOR DANA OBORO	38100	473.67
40	09	500	918421	EFM EQUIPMENT FOR WHITE HYDR.HOPK.	38100	1,095.44
40	09	500	918447	1998-EFM FOR HOPK. COATING	38100	625.70
				TOTAL METERS		292,116.15
40	09	515	200	MTR-REG INST.RETIRE	38200	530.29
40	09	550	200	MTR-REG INST.RETIRE	38200	704.94
40	09	580	200	MTR-REG INST.RETIRE	38200	3,277.55
				TOTAL METER INSTALLATIONS		4,512.78
40	09	592	918299	592 COPIER RETIREMENT	39100	1,464.75
				TOTAL GENERAL OFFICE EQUIPMENT		1,464.75
40	09	537	918743	RETIRE VEHICLE 5246 (1990 Chev 1/2 ton Pickup)	39200	15,217.44
				TOTAL TRANSPORTATION EQUIPMENT		15,506.15
40	09	500	918284	TOOLS - OTHER EQUIPMENT OWENSBORO	39400	743.49
40	09	515	918157	201-WMSON MACHINE REPLACEMENT	39400	434.93
40	09	515	918475	201-M-SCOPE-MADSERVICE	39400	810.85
40	09	537	918238	537-BACKFILL TAMPER	39400	200.00
40	09	537	918240	537-ROCKDRILL	39400	335.62
40	09	546	918151	293-WEED TRIMMER-WILLIS	39400	551.25
40	09	550	918180	OTHER EQUIP - PIPE LOCATORS- PADUCAH	39400	387.45
40	09	580	918174	611 COMBUSTIBLE GAS/C.O. DETECTOR	39400	96.86
40	09	592	918508	641-REPLACE OXYGEN DEPLETION METER	39400	739.59
				TOTAL TOOLS & WORK EQUIPMENT		4,300.04
40	09	311	917279	GROUND BED REPLACEMENT - PADUCAH	39603	22,556.15
				TOTAL DITCHERS		22,556.15
40	09	560	918170	560 - BASE UNIT - BOWLING GREEN	39700	2,345.28
				TOTAL COMMUNICATION EQUIPMENT		2,345.28
40	09	565	916257	MAINFRAME PRINTER - FRANKLIN OFFICE	39905	(7,399.61)
				TOTAL MAINFRAME HARDWARE		(7,399.61)
					•	
				TOTAL RETIREMENTS DIV 09 WKG		572,732.91

.

# WKG TRANSFERS OCTOBER 1998 - MARCH 1999

Amount	Approp	Description	Date
(\$27,864.23)	916575	1996 Supplement for AFE 209923-001 Measuring & Regul Station - System Imp (T.B. #4) Reclass out of Meas & Reg Station Equpment to Mains	Nov-98
\$27,864.23	916575	1996 Supplement for AFE 209923-001 Measuring & Regul Station - System Imp (T.B. #4) <i>Reclass to Mains from Meas &amp; Reg Station</i> <i>Equipment</i>	Nov-98
\$35.37	917269	Replace #5140 Trencher Reclass to Tools & Work Equipment from Ditchers	Nov-98
(\$35.37)	917269	Replace #5140 Trencher <i>Reclass from Ditchers to Tools &amp; Work</i> Equipment	Nov-98
\$0.00		otal	

# SHARED SERVICES DIV 02 TRANSFERS

(\$278.69) (\$119.93)	215390 215402	Fire Proof File Cabinet Hardware/Software/Furniture	Feb-99 Correction of investment amt. Feb-99 Correction of investment amt.
 (\$116.06) ( <b>\$514.68</b> )	215365	New Std PC & 1 Std Chair Total Office Furniture & Equipment	Feb-99 Correction of investment amt.
 (+++++)		••••••••••	

Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 35 c Witness: David H. Doggette

# Data Request:

Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-2.2

c. For the forecasted period, explain why Western assumed there would be no retirements. Include in the explanation a discussion of why such an assumption is reasonable.

# Response:

For budget administration, Western's budgeted additions are projected as a net amount less retirements. Western's retirements mostly occur as part of a maintenance project. Since pipe is usually abandoned in place it is rare to have a net salvage credit. The costs associated with the abandonment are shown on DHD-1, line 41, pages 1-6. Western does not budget for plant retirements since they are not known at the time of budget preparation.


RECYCLED 10% P.C.W.

.

36

Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 36 a & b Witness: Rebecca M. Buchanan

## Data Request:

36. Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-2.3, for both the base and forecasted periods.

a. Identify the business unit of Atmos referred to in the schedule as "Division 02 General Office."

b. Indicate whether the assignment of the Division 02 General Office plant is the result of a general allocation of plant by Atmos or whether Western contracted for these assets.

## Response:

36a.) Division 02 General Office is not a separate business unit. It is the part of Atmos that performs the Shared Service General Office functions on behalf of Western and the other Atmos business units. A list of shared services functions is found in FR 10(9)(u) of the filing, Volume 9, tab 2.

36b.) In relation to Schedule B-2.3, pages 4 and 8, the Division 02 General Office plant is allocated to Western based on the residual factor of 16.657%, and carried forward to Schedule B-2.2. This is set out in the footnotes on pages 3 and 6 of Schedule B-2.2. Western and the other business units do not contract for Shared Services General Office plant.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 37 Witness: Donald P. Burman

## Data Request:

Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-3, for both the base and forecasted periods. For each of the plant accounts listed below, provide a detailed explanation as to why the accumulated depreciation exceeds the plant investment. Also explain why accumulated depreciation in excess of plant investment should be included in the calculation of Western's rate base.

- a. Account No. 331.00 Production Gas Wells Equipment.
- b. Account No. 332.10 Field Lines.
- c. Account No. 332.20 Tributary Lines
- d. Account No. 334.00 Field Meas. & Reg. Station Equipment.
- e. Account No. 351.20 Compression Station Equipment
- f. Account No. 366.30 Other Structures (Transmission Plant).
- g. Account No. 375.10 Meas. & Reg. Station Equipment General.
- h. Account No. 399.00 Other Tangible Property (base period only).
- i. Account No. 399.89 -- Other Tangible Property System Software (forecasted period only).

## Response:

Western does not maintain accumulated depreciation accounts for each individual 300 asset account, but rather maintains the accumulated depreciation accounts for classes of assets, i.e. transmission, distribution, etc. The accumulated depreciation accounts referred to above were interpolated by the Company. Unfortunately, in doing so, certain errors were made in the allocation of the amounts. A revised Schedule B-3 has been prepared, allocating the accumulated depreciation accounts to the various 300 asset accounts, as of September 30, 1998.

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Base Period ended September 30, 1999

Data: Type o Workp	_X_ Bas of Filing: paper Re	e Period Forecasted Period Original Updated _X_ Revised ference No(s).: Sched. B-3.1 & wp B-3					FR 10(10)(b)3 Schedule B-3 Page 1 of 6 Witness:	
			Base Period	<		Reserve Balances		>
Line	Acct.		Total Company	Total	Jurisdictional	Jurisdictional		Adjusted
No.	No.	Account Titles	Investment	Company	Percent	Total	Adjustments	Jurisdiction
1		Intangible Plant			100%			
2	301.00	Organization	8,330	0		0	•.	0
3	302.00	Franchises & Consents	119,853	119,853		119,853		119,853
4 5 6		Total Intangible Plant	128,182	119,853		119,853	0	119,853
7		Natural Gas Production Plant						
8	325.20	Producing Leaseholds	2.353	0		0		0
9	325.40	Rights of Ways	6,069	5,744		5,744		5,744
10	331.00	Production Gas Wells Equipment	3,492	3,331		3,331		3,331
11	332.10	Field Lines	47,163	44,969		44,969		44,969
12	332.20	Tributary Lines	528,218	503,644		503,644		503,644
13	334.00	Field Meas. & Reg. Sta. Equip	198,469	189,289		189,289		189,289
14	336.00	Purification Equipment	44,369	42,133		42,133		42,133
15					-			
16 17		Total Natural Gas Production Plant	830,133	789,110		789,110	0	789,110
18		Storage Plant						0
19	350.10	Land	261.127	0		0		Ó
20	350.20	Rights of Way	4.682	3.524		3,524		3,524
21	351.20	Compression Station Equipment	121,402	92.856		92.856		92,856
22	351.30	Meas. & Reg. Sta. Structures	23,138	17.718		17,718		17,718
23	351.40	Other Structures	144.554	110,688		110,688		110,688
24	352.01	Well Construction	2,180,995	1.675,569		1,675,569		1,675,569
25	352.20	Well Equipment	547.827	413,450		413,450		413,450
26	352.10	Leaseholds	178,530	136,738		136,738		136,738
27	352.11	Storage Rights	54,614	41,830		41,830		41,830
28	353.10	Field Lines	190,173	137,146		137,146		137,146
29	353.20	Tributary Lines	212,186	160,825		160,825		160,825
30	354.00	Compressor Station Equipment	470,685	362,087		362,087		362,087
31	355.00	Meas & Reg. Equipment	288,851	222,097		222,097		222,097
32	356.00	Purification Equipment	239,930	184,301	-	184,301		184,301
33 34			4,918,694	3,558,829		3,558,829	0	3,558,829

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Base Period ended September 30, 1999

Data: Type Work	_X_ Bas of Filing: paper Re	e Period Forecasted Period Original Updated _X_ Revised ference No(s).: Sched. B-3.1 & wp B-3			·		FR 10(10)(b)3 Schedule B-3 Page 2 of 6 Witness:	
			Base Period	<		Reserve Balances		>
Líne			Total Company	Total	Jurisdictional	Jurisdictional		Adjusted
No.	Acct.		Investment	Company	Percent	Total	Adjustments	Jurisdiction
<u>(A)</u>	No.	Account Title	(D)	(E)	(F)	(G)	(H)	(i)
		Transmission Plant						
1	365 10	I and	26 051	٥		0	1	٥
2	365.20	Rights of Way	20,331 A03 A10	261 725		261 725		261 725
2	. 266 20	Structures & Improvements	100,410	201,720		201,725		201,120
3	366 30	Other Structures	23,033	Z1,430 A5 246		45 246		45 246
4	267.00	Maine	10 210 520	40,240		40,240		40,240
2	260.10	Mans & Des Equipment	19,310,530	1 056 457		1 056 457		1 056 457
0	309.10	Meas. & Rey. Equipment	2,902,000	1,900,407		1,930,437		1,500,407
8		Total Production Plant - LPG	22,796,738	14,955,009		14,955,009	0	14,955,009
9								
10		Distribution Plant						
11	374.10	Land Town Border	61.836	0		0		0
12	374.30	Land Other	2.784	0		0		0
13	374.20	Right of Way	44.872	19.415		19.415		19,415
14	375.10	Structures & Improvements T.B.	106.376	46.541		46,541		46.541
15	375.02	Structures & Improvements Other	0	0		0		0
16	375.03	Improvements	7.518	3,339		3.339		3.339
17	375.20	Land Rights	46.591	20,694		20.694		20.694
18	376.00	Mains	69.803.621	30.525.555		30.525.555		30.525.555
19	375.10	Meas, & Reg. Sta. Equipment General	1.967.476	839,459		839.459		839,459
20	379.30	Meas & Reg. Sta. Equipment T.b.	1.691.528	735.392		735.392		735.392
21	380.00	Services	42.077.687	17.887.837		17.887.837		17.887.837
22	381.00	meters	18 132 215	7,807,827		7.807.827		7.807.827
23	381.02	V & P Guaques	109.524	48,793		48,793		48,793
24	382.00	Meter Installations	13 621,909	5 865 972		5 865 972		5.865.972
25	383.00	Regulators Service	3 477 781	1 527 007		1 527 007		1 527 007
26	383.02	Regulators Relief	481 545	214 385		214.385		214.385
27	384.00	House Reg. Installations	160 998	71 171		71,171		71,171
28	385.01	Ind Meas & Reg. Sta Fouinment	2 970 314	1 307 712		1.307.712		1.307.712
29	000.01	and more a rogi our equipment		.,	-			
30			154,764,573	66,921,098		66,921,098	0	66,921,098

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Base Period ended September 30, 1999

.-

· . .

Data: , Type o Workp	_X_ Base of Filing: aper Rel	e Period Forecasted Period Original Updated _X_ Revised ference No(s).: Sched. B-3.1 & wp B-3					FR 10(10)(b)3 Schedule B-3 Page 3 of 6 Witness:	
	_		Base Period	<		Reserve Balances		>
Line			Total Company	Total	Jurisdictional	Jurisdictional		Adjusted
No.	Acct.		Investment	Company	Percent	Total	Adjustments	Junsdiction
(A)	No.	Account Title	(D)	(E)	(F)	(G)	(H)	()
		Conoral Plant **					X	
1	380 10	Land	44 728	٥		0		0
2.	390.02	Structures & Improvements	244 220	92 671		92 671		92.671
2	390.02	Improvements	64 111	32 958		32,958		32,958
4	390.00	Air Conditioning Equipment	9 771	5.039		5.039		5.039
5	390.05	Total Energy	0,111	0,000		0		0
6	390.09	Improvement to leased Premises	2.389.182	1.008.493		1.008.493		1,008,493
7	391.00	General Office Equipment	2,296,800	1.063.548		1.063.548		1,063,548
8	391.82	General Office Equipment	15.691	12.847		12,847		12,847
9	391.83	Office Furn - Copiers & Typewriters	389.359	262.804		262.804		262,804
10	392.00	Transportation Equipment	6.048.032	3,241,760		3,241,760		3,241,760
11	382.02	Trailers	165,970	89,107		89,107		89,107
12	392.00	n/a	0	0		0		0
13	394.77	Tools & Work Equipment	3,071,424	1,593,002		1,593,002		1,593,002
14	396.93	Ditchers	845,786	426,505		426,505		426,505
15	396.94	Backhoes	706,023	361,976		361,976		361,976
16	396.95	Welders	92,413	47,403		47,403		47,403
17	397.00	Communication Equipment - Phones	1,032,385	509,103		509,103		509,103
18	397.20	Communication Equip Fixed Radios	21,697	11,529		11,529		11,529
19	397.21	Communication Equipment - Mobile Radios	60,768	30,971		30,971		30,971
20	397.22	Communication Equip Tetemetering	114,695	60,485		60,485		60,485
21	398.00	Miscellaneous Equipment	145,042	49,584		49,584		49,584
22	399.00	Other Tangible Property	10,271	62,799		62,799		62,799
23	399.84	Other Tangible Property - CPU	182,472	128,503		128,503		128,503
24	399.85	Other Tangible Property - MF Hardware	596,748	266,176		266,176		266,176
25	399.86	Other Tangible Property - PC Hardware	2,194,113	810,384		810,384		810,384
26	399.87	Other Tang. Property - P.C. Software	410,794	244,532		244,532		244,532
27	399.88	Other Tang. Property - Application Software	9,044,627	1,868,224		1,868,224		1,868,224
28	399.89	Other Tang. Property - System Software	466,628	404,490		404,490		404,490
29	39x.xx1	Server Hardware	321,217	12,358		12,358		12,358
30	39x.xx2	Server Software	105,374	4,054		4,054		4,054
31	39x.xx3	Network Cost	153,339	5,899		5,899		5,899
32	39x.xx4	Start Up Cost	2,629,307	58,967		58,967		58,967
33 34	999.00	Cushion Gas	1,694,833	0	· -	0		0
35		Total General Plant	35,567,821	12,766,173		12,766,173	0	12,766,173
36 37		Total Plant	219,006,142	99,110,071		99,110,071	0	99,110,071

\*\* All General Plant amounts include Western Kentucky Gas general plant 100%, plus an allocation of General Office general plant at 16.657% KY residual factor.

# Western Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Forecasted Period ended December 31, 2000

L

Data: _ Type o Workp	Base f Filing: _ aper Ref	Period _X_ Forecasted Period Original Updated _X_ Revised erence No(s).: Sched, B-3.1 & wp B-3	•				FR 10(10)(b)3 Schedule B-3 Page 4 of 6	
							Witness:	
			Forecasted	<		Reserve Balances		>
Line	Acct.		Total Company	Total	Juriso.	Jurisdictional		Adjusted
No.	No.	Account Titles	Investment	Company	Percent	Total	Adjustments	Jurisdiction
(A)	(B)	(C)	(D)	(E)	(F)	(G)	(H)	(I)
					40004			
1				•	100%		Υ.	•
2	301.00	Organization	8,330	0		0		0
3	302.00	Franchises & Consents	119,853	119,853		119,853		119,853
4								
5		Total Intangible Plant	128,182	119,853		119,853	0	119,853
5		Network Case Readination Plant						
1	205 00	Natural Gas Production Plant	0.050	•		•		0
8	325.20	Producing Leasenoids	2,303	5 744		5 744		U 5 744
9	325.40	Rights of ways	0,009	5,744		5,744		3,744
10	331.00	Production Gas wells Equipment	3,492	3,3/2		3,372		3,3/2
11	332.10	Field Lines	47,163	45,484		45,484		45,464
12	332.20	Indutary Lines	528,218	509,413		509,413		509,413
13	334.00	Field Meas. & Reg. Sta. Equip	198,469	191,651		191,651		191,651
14	336.00	Purification Equipment	44,369	41,991		41,991		41,991
15								303.055
16		Total Natural Gas Production Plant	830,133	797,655		797,655	0	797,655
1/		Starage Dignt						٥
18	050 40	Storage Plant	004 407	•		0		0
19	350.10		201,127	0 577		0		2 5 7 7
20	350.20	Rights of way	4,082	3,577		3,577		3,377
21	351.20	Compression Station Equipment	121,774	96,212		95,212		90,212
22	351.30	Meas. & Reg. Sta. Structues	23,138	18,356		18,355		18,330
23	351.40	Other Structures	144,554	114,6/5		114,675		114,675
24	352.01	Weil Construction	2,172,800	1,786,876		1,786,876		1,785,875
25	352.20	Well Equipment	579,991	442,387		442,387		442,387
26	352.10	Leaseholds	178,530	140,329		140,329		140,329
27	352.11	Storage Rights	54,614	43,346		43,346		43,346
28	353.10	Field Lines	261,841	144,166		144,166		144,166
29	353.20	Tributary Lines	228,934	167,806		167,806		167,806
30	354.00	Compressor Station Equipment	470,685	380,173		380,173		380,173
31	355.00	Meas & Reg. Equipment	288,851	233,594		233,594		233,594
32 33	356.00	Purification Equipment	239,930	192,280		192,280		192,280
34		Total Storage Plant	5,031,452	3,763,778	_	3,763,778	0	3,763,778

.

#### Westem Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Forecasted Period ended December 31, 2000

Data: Type c Workp	Base of Filing: _ aper Ref	Period _X_ Forecasted Period Original Updated _X_ Revised erence No(s).: Sched. B-3.1 & wp B-3					FR 10(10)(b)3 Schedule B-3 Page 5 of 6 Witness:	
			Forecasted	<	F	Reserve Balances		>
Line			Total Company	Total	Jurisd.	Jurisdictional		Adjusted
No.	Acct.		Investment	Company	Percent	Totał	Adjustments	Jurisdiction
(A)	No.	Account Title	(D)	(E)	(F)	(G)	<u>(H)</u>	(1)
		Transmission Plant					X.	
1	365.10	Land	26 951	0		0		0
2.	365.20	Rights of Way	403,419	266.305		266.305		266.305
3	366.20	Structures & Improvements	32,921	21,936		21,936		21,936
4	366.30	Other Structues	69,172	46,388		46,388		46,388
5	367.00	Mains	19,441,293	13,143,620		13.143.620		13,143,620
6	369.10	Meas. & Reg. Equipment	2,995,622	2.054,801		2,054,801		2,054,801
7						, ,		
8		Total Production Plant - LPG	22,969,379	15,533,051		15,533,051	0	15,533,051
9								
10		Distribution Plant						
11	374.10	Land Town Border	61,710	0		0		0
12	374.30	Land Other	2,784	0		0		0
13	374.20	Right of Way	44,872	20,077		20,077		20,077
14	375.10	Structures & Improvements T.B.	106,376	49,106		49,106		49,106
15	375.02	Structures & Improvements Other	0	0		0		0
16	375.03	Improvements	7,518	3,551		3,551		3,551
17	375.20	Land Rights	46,591	22,010		22,010		22,010
18	376.00	Mains	75,047,099	33,215,849		33,215,849		33,215,849
19	375.10	Meas. & Reg. Sta. Equipment General	2,363,549	919,245		919,245		919,245
20	379.30	Meas & Reg. Sta. Equipment T.b.	1,917,181	802,795		802,795		802,795
21	380.00	Services	45,854,769	20,457,751		20,457,751		20,457,751
22	381.00	meters	19,396,585	8,429,666		8,429,666		8,429,666
23	381.02	V & P Gauges	109,524	53,192		53,192		53,192
24	382.00	Meter Installations	14,560,567	6,479,383		6,479,383		6,479,383
25	383.00	Regulators Service	3,733,713	1,659,558		1,659,558		1,659,558
26	383.02	Regulators Relief	481,545	232,004		232,004		232,004
27	384.00	House Reg. Installations	166,402	77,400		77,400		77,400
28	385.01	Ind. Meas. & Reg. Sta. Equipment	3,211,613	1,427,421		1,427,421		1,427,421
29 30		Total Distribution Plant	167,112,395	73.849.009	-	73.849.009	0	73.849.009
							•	

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Accumulated Depreciation & Amortization 13 Month Avg. - Forecasted Period ended December 31, 2000

FR 10(10)(b)3 Base Period X\_ Forecasted Period Data: Type of Filing: \_\_\_\_ Original \_\_\_ Updated \_X\_ Revised Schedule B-3 Workpaper Reference No(s) .: Sched. B-3.1 & wp B-3 Page 6 of 6 Witness: Reserve Balances Forecasted Total Company Line Total Jurisd. Jurisdictional Adjusted Investment Company Percent Total Adjustments Jurisdiction No. Acct. (A) No. Account Title (D) (E) (F) (G) (H) (l)١Ļ General Plant \*\* 44,728 0 0 Ð 389.10 Land 1 316,621 100,597 100,597 100,597 390.02 Structures & Improvements 2 34,189 3 390.03 Improvements 64,111 34,189 34,189 5,425 9,771 5,425 5,425 4 390.04 Air Conditioning Equipment 0 0 0 ٥ 5 390.05 Total Energy 6 390.09 Improvement to leased Premises 2,504,775 1,220,766 1,220,766 1,220,766 1,219,510 2,550,590 1,219,510 1,219,510 391.00 General Office Equipment 7 15,146 15,146 General Office Equipment 16,898 15,146 8 391.82 405,141 281,098 281,098 281,098 9 391.83 Office Furn - Copiers & Typewriters 3,933,006 6,054,009 3,933,006 3,933,006 10 392.00 Transportation Equipment 165,970 108,135 108,135 108.135 382.02 Trailers 11 0 12 392.00 Stores Equipment 0 0 0 1,745,524 13 394.77 **Tools & Work Equipment** 3,082,589 1,745,524 1,745,524 461,137 853,615 461,137 461,137 14 396.93 Ditchers 706,023 399,615 399,615 399,615 15 396.94 Backhoes 52,306 52,306 92,413 52,306 16 396.95 Welders 603,212 603,212 397.00 Communication Equipment - Phones 1.293,379 603,212 17 397.20 Communication Equip. - Fixed Radios 28,653 13,452 13,452 13,452 18 36,054 36,054 68,220 36,054 19 397.21 Communication Equipment - Mobile Radios 114,695 70.005 70.005 70,005 20 397.22 Communication Equip. - Telemetering 62,394 153,632 62,394 62,394 21 398.00 Miscellaneous Equipment 3,824 3,824 3,824 22 399.00 Other Tangible Property 11,061 190,242 190,242 23 399.84 Other Tangible Property - CPU 196.508 190.242 338,053 24 399.85 Other Tangible Property - MF Hardware 607,494 338,053 338,053 1,527,719 399.86 Other Tangible Property - PC Hardware 3,551,824 1,527,719 1,527,719 25 365,551 546,060 365,551 365,551 26 399.87 Other Tang. Property - P.C. Software 3,714,353 399.88 Other Tang. Property - Application Software 20,278,490 3,714,353 3,714,353 27 553,715 399.89 Other Tang. Property - System Software 502,523 553,715 553,715 28 120,493 29 39x.xx1 Server Hardware 695.971 120.493 120,493 30 39x.xx2 Server Software 228,311 39,527 39,527 39,527 31 39x.xx3 Network Cost 332,234 57,519 57,519 57,519 574,931 574,931 574,931 39x.xx4 Start Up Cost 5,696,831 32 33 999.00 Cushion Gas 1,694,833 0 0 0 34 35 17,847,497 0 17,847,497 **Total General Plant** 52,867,971 17,847,497 36 248,939,511 111,910,842 111,910,842 0 111,910,842 37 Total Plant

 All General Plant amounts include Western Kentucky Gas general plant 100%, plus an allocation of General Office general plant at 16.657% KY residual factor.

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 38 Witness: Donald P. Burman

# Data Request:

Refer to the Application, Volume 10 of 10, Tab 2, Schedule B-3.2. Each of the accounts listed below is shown on Schedule B-3.2 as being fully depreciated. Explain why Western has included a 12-month depreciation expense for each of these accounts, and why it is reasonable to include depreciation expense for a plant account that is fully depreciated per the company's books.

- a. Account No. 331.00 Production Gas Wells Equipment (base period only).
- b. Account No. 332.10 Field Lines (base period only).
- c. Account No. 332.20 Tributary Lines (base period only).
- d. Account No. 334.00 Field Meas. & Reg. Station Equipment (base period only).
- e. Account No. 351.20 Compression Station Equipment (base and forecasted periods).
- f. Account No. 366.30 Other Structures (base and forecasted periods).
- g. Account No. 375.10 Meas. & Reg. Station Equipment General. (base and forecasted periods).

#### Response:

As noted in the response to DR Item 37, Western does not maintain accumulated depreciation accounts for each individual 300 asset account, but rather maintains the accumulated depreciation accounts for classes of assets, i.e. transmission, distribution, etc. The accumulated depreciation accounts referred to above were interpolated by the Company. Unfortunately, in doing so, certain errors were made in the allocation of the amounts. A revised Schedule B-3.2 has been prepared, allocating the accumulated depreciation accounts to the various 300 asset accounts.

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Base Period Ended September 30, 1999

Data: \_X\_ Base Period \_\_\_\_ Forecasted Period Type of Filing: \_\_\_ Original \_\_\_ Updated \_X\_ Revised Workpaper Reference No(s).: Sched. B-3 & wp B-3.2 FR 10(10)(b)3.2 Schedule B-3.2 Page 1 of 6 Witness:

•			Total Comp	any Adjusted J	urisdiction	Annual
Line	Acct.		13 month	average	12 Month	Accrual
No.	No.	Account Titles	Investment	Reserve	Expense	Rate
(A)	(B)	(C)	(D)	(E)	<u>(F)</u>	(G)
1		Intangible Plant				
2	301.00	Organization	8 330	0	0	Amort
2	302.00	Franchises & Consents	119,853	119.853	0	Amort
4	002.00				•	
5		Total Intangible Plant	128,182	119.853	0	
6			120,102		-	
7		Natural Gas Production Plant				
8	325.20	Producing Leaseholds	2,353	0	0	0.00%
9	325.40	Rights of Ways	6,069	5,744	0	0.00%
10	331.00	Production Gas Wells Equipment	3,492	3,331	51	1.53%
11	332.10	Field Lines	47,163	44,969	648	1.44%
12	332.20	Tributary Lines	528,218	503,644	7,260	1.44%
13	334.00	Field Meas. & Reg. Sta. Equip	198,469	189,289	2,917	. 1.54%
14	336.00	Purification Equipment	44,369	42,133	00	0.00%
15						
16		Total Natural Gas Production Plant	830,133	789,110	10,877	
17						
18		Storage Plant				
19	350.10	Land	261,127	0	0	0.00%
20	350.20	Rights of Way	4,682	3,524	41	0.92%
21	351.20	Compression Station Equipment	121,402	92,856	3,314	2.86%
22	351.30	Meas. & Reg. Sta. Structues	23,138	17,718	632	2.86%
23	351.40	Other Structures	144,554	110,688	3,946	2.86%
24	352.01	Well Construction	2,180,995	1,675,569	101,174	4.86%
25	352.20	Well Equipment	547,827	413,450	25,413	4.86%
26	352.10	Leaseholds	178,530	136,738	4,993	2.93%
27	352.11	Storage Rights	54,614	41,830	1,527	2.93%
28	353.10	Field Lines	190,173	137,146	6,517	3.59%
29	353.20	Tributary Lines	212,186	160,825	7,271	3.59%
30	354.00	Compressor Station Equipment	470,685	362,087	18,779	4.18%
31	355.00	Meas & Reg. Equipment	288,851	222,097	11,139	4.04%
32	356.00	Purification Equipment	239,930	184,301	8,611	3.76%
33						
34			4,918,694	3,558,829	193,356	





#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Base Period Ended September 30, 1999

Data: \_X\_ Base Period \_\_\_\_ Forecasted Period Type of Filing: \_\_\_\_ Original \_\_\_\_ Updated \_X\_ Revised Workpaper Reference No(s).: Sched. B-3 & wp B-3.2 FR 10(10)(b)3.2 Schedule B-3.2 Page 2 of 6 Witness:

			Total Com	oany Adjusted J	urisdiction	Annual
Line	Acct.		13 month	average	12 Month	Accrual
No.	No.	Account Titles	Investment	Reserve	Expense	Rate
(A)	(B)	(C)	(D)	(E)	(F)	(G)
		Transmission Plant				
1	365.10	Land /	26,951	0	0	0.00%
2	365.20	Rights of Way	403,419	261,725	3,543	0.92%
3	366.20	Structures & Improvements	23,859	21,496	355	1.56%
4	366.30	Other Structues	69,172	45,246	1,030	1.56%
5	367.00	Mains	19,310,530	12,670,085	447,895	2.43%
6	369.10	Meas. & Reg. Equipment	2,962,806	1,956,457	78,901	2.79%
7		<b>2</b>				
8		Total Production Plant - LPG	22,796,738	14,955,009	531,724	
9						
10		Distribution Plant				
11	374.10	Land Town Border	61,836	0	0	0.00%
12	374.30	Land Other	2,784	0	0	0.00%
13	374.20	Right of Way	44,872	19,415	364	0.85%
14	375.10	Structures & Improvements T.B.	106,376	46,541	2,782	2.74%
15	375.02	Structures & Improvements Other	0	0	0	2.74%
16	375.03	Improvements	7,518	3,339	197	2.74%
17	375.20	Land Rights	46,591	20,694	1,219	2.74%
18	376.00	Mains	69,803,621	30,525,555	2,331,964	3.50%
19	375.10	Meas. & Reg. Sta. Equipment General	1,967,476	839,459	62,536	3.33%
20	379.30	Meas & Reg. Sta. Equipment T.b.	1,691,528	735,392	54,572	3.38%
21	380.00	Services	42,077,687	17,887,837	1,405,710	3.50%
22	381.00	meters	18,132,215	7,807,827	560,753	3.24%
23	381.02	V & P Gauges	109,524	48,793	3,387	3.24%
24	382.00	Meter Installations	13,621,909	5,865,972	508,383	3.91%
25	383.00	Regulators Service	3,477,781	1,527,007	103,902	3.13%
26	383.02	Regulators Relief	481,545	214,385	14,387	3.13%
27	384.00	House Reg. Installations	160,998	71,171	4,610	3.00%
28	385.01	Ind. Meas. & Reg. Sta. Equipment	2,970,314	1,307,712	96,679	3.41%
29		<u> </u>				
30			154,764,573	66,921,098	5,151,445	

### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Base Period Ended September 30, 1999

Data Type Work	: _X_ Bas	se Period Forecasted Period : Original Updated _X_ Revised			FR 10(10)(b)3. Schedule B-3.2 Page 3 of 6 Schedule B-3.2	2 2
V1011	paperine	serence holds. Gened. D-0 a wp D-0.2	Total Com	any Adjusted.	lurisdiction	Annual
l ine	Acct		13 month	average	12 Month	Accrual
No	No.	Account Titles	Investment	Reserve	Expense	Rate
(4)	(B)		(D)	(F)	(F)	(G)
-101	(9)					<u>, (0)</u>
		General Plant				
1	389 10	Land	44 728	0	0	0.00%
2	300.02	Structures & Improvements	244,720	92 671	4 942	2 12%
2	300.02	Improvements	64 111	32,071	1 297	2 12%
3	200.03	Air Conditioning Equipment	0,111	5 030	361	3 87%
4	390.04	Air Conditioning Equipment	5,771	2,039	501	0.00%
5	390.05	Total Energy	0 200 402	1 009 403	65.049	6.00%
6	390.09	Improvement to leased Premises	2,309,102	1,000,493	00,940	0.00%
7	391.00	General Office Equipment	2,296,800	1,063,548	63,055	3.87%
8	391.82	General Office Equipment	15,691	12,847	0	0.00%
9	391.83	Office Furn - Copiers & Typewriters	389,359	262,804	7,406	3.87%
10	392.00	Transportation Equipment	6,048,032	3,241,760	510,694	8.86%
11	392.20	Trailers	165,970	89,107	14,036	8.86%
12	393.00	n/a	0	0	0	0.00%
13	394.77	Tools & Work Equipment	3,071,424	1,593,002	130,768	4.47%
14	396.93	Ditchers	845.786	426,505	36,086	4.47%
15	396.94	Backhoes	706.023	361,976	30,123	4.47%
16	306.05	Welders	92 413	47 403	3,943	4.47%
17	307.00	Communication Equipment - Phones	1 032 385	509 103	60 419	7 05%
10	207.00	Communication Equipment - Thomes	21 607	11 529	1 460	7.05%
10	397.20	Communication Equip Fixed Radios	21,037	20.071	1,400	7.05%
19	397.21	Communication Equipment - Mobile Radios	00,700	50,971	4,005	7.05%
20	397.22	Communication Equip Telemetering	114,090	00,400	/,/10	1.00%
21	398.00	Miscellaneous Equipment	145,042	49,584	4,2/8	12.09%
22	399.00	Other Tangible Property	10,271	62,799	0	0.00%
23	399.84	Other Tangible Property - CPU	182,472	128,503	0	0.00%
24	399.85	Other Tangible Property - MF Hardware	596,748	266,176	38,481	10.04%
25	399.86	Other Tangible Property - PC Hardware	2,194,113	810,384	307,819	20.60%
26	399.87	Other Tang. Property - P.C. Software	410,794	244,532	47,674	20.60%
27	399.88	Other Tang. Property - Application Software	9,044,627	1,868,224	405,562	8.22%
28	399.89	Other Tang. Property - System Software	466,628	404,490	0	0.00%
29	39x.xx1	Server Hardware	321,217	12,358	43,813	14.29%
30	39x.xx2	Server Software	105.374	4,054	14,373	14.29%
31	39x xx3	Network Cost	153,339	5,899	20.915	14.29%
32	394 444	Start Lin Cost	2.629.307	58.967	209.056	8.33%
22	000.004	Cuchion Gas	1 694 833	0	0	0.00%
24	333.00	ousnion das	1,004,000	<u>`</u>		0.0070
04 25		Total Conoral Plant	35 567 901	12 766 172	2 03/ 315	
30		I Olai Generai Fidill	33,307,021	12,700,173	2,004,010	
30		Total Diant	210 006 142	00 110 074	7 021 717	
37		i otal Plant	219,000,142	99,110,071	1,921,111	





.

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Forecasted Period ended December 31, 2000

Data: \_\_\_\_ Base Period \_X\_ Forecasted Period Type of Filing: \_\_\_ Original \_\_\_ Updated \_X\_ Revised Workpaper Reference No(s).: Sched. B-3 & wp B-3.2 FR 10(10)(b)3.2 Schedule B-3.2 Page 4 of 6 Witness:

			Total Comp	any Adjusted Ju	urisdiction	Annual
Line	Acct.		13 Mon	th Avg.	12 Month	Accrual
No.	No.	Account Titles	Investment	Reserve	Expense	Rate
<u>(A)</u>	(B)	(C)	(D)	(E)	(F)	<u>(</u> G)
1		Intancible Plant				
2	301.00		8 330	0	٥	Amort
2	307.00	Eranchicos & Consonts	110,853	110 853	0	Amort
1	302.00	Franchises & Consents	119,000	113,000	U	Amon
5		Total Intangible Plant	128 182	119 853	0	
6			120,102	110,000	Ũ	
7		Natural Gas Production Plant				
8	325.20	Producing Leaseholds	2.353	0	0	0.00%
9	325.40	Rights of Ways	6.069	5,744	0	0.00%
10	331.00	Production Gas Wells Equipment	3,492	3,372	0	0.00%
11	332.10	Field Lines	47,163	45,484	0	0.00%
12	332.20	Tributary Lines	528,218	509,413	0	0.00%
13	334.00	Field Meas. & Reg. Sta. Equip	198,469	191,651	0	0.00%
14	336.00	Purification Equipment	44,369	41,991	0	0.00%
15						
16		Total Natural Gas Production Plant	830,133	797,655	0	
17						
18		Storage Plant				
19	350.10	Land	261,127	0	0	0.00%
20	350.20	Rights of Way	4,682	3,577	41	0.92%
21	351.20	Compression Station Equipment	121,774	96,212	2,243	1.93%
22	351.30	Meas. & Reg. Sta. Structues	23,138	18,356	426	1.93%
23	351.40	Other Structures	144,554	114,675	2,663	1.93%
24	352.01	Well Construction	2,172,800	1,786,876	56,204	2.71%
25	352.20	Well Equipment	579,991	442,387	15,003	2.71%
26	352.10	Leaseholds	178,530	140,329	511	0.30%
27	352.11	Storage Rights	54,614	43,346	954	1.83%
28	353.10	Field Lines	261,841	144,166	3,374	1.35%
29	353.20	Tributary Lines	228,934	167,806	2,950	1.35%
30	354.00	Compressor Station Equipment	470,685	380,173	6,784	1.51%
31	355.00	Meas & Reg. Equipment	288,851	233,594	5,680	2.06%
32	356.00	Purification Equipment	239,930	192,280	2,977	1.30%
33 34		Total Storage Plant	5.031.452	3.763.778	99.810	
			0,000., IOC	-,,. · ·	,	

l i



#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Forecasted Period ended December 31, 2000

Data: \_\_\_\_ Base Period \_X\_ Forecasted Period Type of Filing: \_\_\_ Original \_\_\_ Updated \_X\_ Revised Workpaper Reference No(s).: Sched. B-3 & wp B-3.2 FR 10(10)(b)3.2 Schedule B-3.2 Page 5 of 6 Witness:

			Total Com	Total Company Adjusted Jurisdiction						
Line	Acct.		13 Mor	th Avg.	12 Month	Accrual				
No.	No.	Account Titles	Investment	Reserve	Expense	Rate				
(A)	(B)	(C)	(D)	(E)	(F)	<sup>`,</sup> (G)				
		Transmission Plant								
1	365.10	Land	26,951	0	0	0.00%				
2	365.20	Rights of Way	403,419	266,305	3,427	0.89%				
3	366.20	Structures & Improvements	32,921	21,936	437	1.39%				
4	366.30	Other Structues	69,172	46,388	918	1.39%				
5	367.00	Mains	19,441,293	13,143,620	235,670	1.27%				
6	369.10	Meas. & Reg. Equipment	2,995,622	2,054,801	65,193	2.28%				
7		•								
8		Total Production Plant - LPG	22,969,379	15,533,051	305,644					
9										
10		Distribution Plant								
11	374.10	Land Town Border	61,710	0	0	0.00%				
12	374.30	Land Other	2,784	0	0	0.00%				
13	374.20	Right of Way	44,872	20,077	720	1.68%				
14	375.10	Structures & Improvements T.B.	106,376	49,106	1,980	1.95%				
15	375.02	Structures & Improvements Other	0	0	0	1.95%				
16	375.03	Improvements	7,518	3,551	140	1.95%				
17	375.20	Land Rights	46,591	22,010	867	1.95%				
18	376.00	Mains	75,047,099	33,215,849	1,712,016	2.39%				
19	375.10	Meas. & Reg. Sta. Equipment General	2,363,549	919,245	56,175	2.49%				
20	379.30	Meas & Reg. Sta. Equipment T.b.	1,917,181	802,795	47,030	2.57%				
21	380.00	Services	45,854,769	20,457,751	3,002,511	6.86%				
22	381.00	meters	19,396,585	8,429,666	620,220	3.35%				
23	381.02	V & P Gauges	109,524	53,192	3,502	3.35%				
24	382.00	Meter Installations	14,560,567	6,479,383	425,281	3.06%				
25	383.00	Regulators Service	3,733,713	1,659,558	101,569	2.85%				
26	383.02	Regulators Relief	481,545	232,004	13,100	2.85%				
27	384.00	House Reg. Installations	166,402	77,400	5,353	3.37%				
28	385.01	Ind. Meas. & Reg. Sta. Equipment	3,211,613	1,427,421	83,688	2.73%				
29										
30		Distribution Plant	167,112,395	73,849,009	6,074,149					



T

DR 38 B.3.2 F

#### Western Kentucky Gas Company Case No. 99-070 Jurisdictional Depreciation Expense, Accum. Reserve & Accrual Rates by Account Forecasted Period ended December 31, 2000

Į.

Data: _ Type o Workpa	Base P f Filing: aper Refere	eriod _X_ Forecasted Period _ Original Updated _X_ Revised ence No(s).: Sched. B-3 & wp B-3.2	,		FR 10(10)(b)3. Schedule B-3.2 Page 6 of 6 Witness:	2
			Total Com	pany Adjusted J	urisdiction	Annual
Line	Acct.	4	13 Mo	nth Avg	12 Month	Accrual
NO.	NO.	Account lities	Investment	Reserve	Expense	Rate
<u>(A)</u>	(B)	(0)	(U)	(E)	(F)	<u>_(G)</u>
1		General Plant				
2	380 10	L and	11 728	٥	0	0.00%
2	390.10	Structures & Improvements	316 621	100 597	6407	2 12%
4	300.02	Improvements	64 111	34 189	1 297	2 12%
5	300.00	Air Conditioning Equipment	9,771	5 4 2 5	198	2.12%
6	300.04	Total Eperav	5,711	0,420	0	0.00%
7	390.09	Improvement to leased Premises	2 504 775	1 220 766	67.095	5.00%
, 8	391.00	General Office Equipment	2,550,590	1 219 510	125 407	7.05%
ğ	391.82	General Office Equipment	16 898	15 146	120,101	1.0070
10	301.02	Office Furn - Conjers & Typewriters	405 141	281 098	13 491	7 05%
11	392.00	Transportation Equipment	6 054 009	3 933 006	514 601	8.92%
12	382.00	Trailers	165 970	108 135	14 131	8.92%
13	303.00	Stores Equipment	100,070	100,100	0	0.00%
14	394 77	Tools & Work Equipment	3 082 589	1 745 524	96 289	3 28%
15	306.03	Ditchers	853 615	461 137	22 732	2 79%
16	396.94	Backhoes	706 023	399 615	18 802	2.79%
17	306.04	Welders	92 413	52,306	2 461	2 79%
18	397.00	Communication Equipment - Phones	1 293 379	603 212	57 027	5 21%
19	397.00	Communication Equipment - Fixed Radios	28 653	13 452	1 425	5 21%
20	397.20	Communication Equipment - Mobile Badios	68 220	36 054	3 393	5.21%
21	397.22	Communication Equipment Mobile Natios	114 695	70 005	5 704	5.21%
22	398.00	Miscellaneous Equipment	153 632	62 394	3 871	10 94%
23	399.00	Other Tannible Property	11 061	3 824	0,07	0.00%
20	399 RA	Other Tangible Property - CPU	196 508	190 242	0	0.00%
25	300 85	Other Tangible Property - ME Hardware	607 494	338 053	4 512	1 19%
26	399.86	Other Tangible Property - PC Hardware	3 551 824	1 527 719	499,802	18.51%
27	399 87	Other Tang, Property - P.C. Softward	546.060	365 551	52,631	15.85%
28	399.88	Other Tang, Property - Application Software	20 278 490	3 714 353	1.439.687	12.50%
29	399.89	Other Tang. Property - System Software	502 523	553,715	0	0.00%
30	39x xx1	Server Hardware	695 971	120,493	94,929	14.29%
31	39x xx2	Server Software	228.311	39.527	31.141	14.29%
32	39x xx3	Network Cost	332 234	57,519	45.316	14.29%
33	39x xx4	Start Up Cost	5 696 831	574.931	452.954	8.33%
34	999 00	Cushion Gas	1 694 833	0, 1,001	0	0.00%
35	000.00		100-1000	•	3	
36		Total General Plant	52,867,971	17.847.497	3.575.303	
37						
38		Total Plant	248,939,511	111,910,842	10,054,907	

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 39 a Witness: Adams

## Data Request:

Refer to the Application, Volume 10 of 10, Tab 6, Schedule F-6. Provide the following information concerning the costs for the preparation of this case:

a. A detailed schedule of costs incurred to date. Include the date of the transaction, check number or other document reference, the vendor, amount, a description of the services performed, and the account number in which the expenditure was recorded. Indicate any costs incurred for this case during the base year. Include copies of invoices received from the vendors.

## Response:

a. See attached schedule entitled "Rate Application Costs Incurred By Western Kentucky Gas Company Case No. 99-070 Account WKG10143". Vendor invoices are enclosed.

	/sou	c. Labor																												
	e Meeti	Mis																							`.					
	Emplove	Expense						-																				-		
		Printing	)							-																				
0	Public	Notice																												
		Legal																												
NDDDDD		Consultants		12,560.00	1,440.00	540.00	10,410.00	1,868.03	2,100.00	2,310.00	31,865.00	4,005.00	16,064.05	150.00	2,700.72	7,138.09	1,726.04	8,257.93	33,997.17	18,997.39	15,033.79	6,896.06	247.50	1,740.50	2,016.50	625.00	3,306.33	1,016.50	1,016.50	4,300.55
		Vendor	Consultants	Applied Energy Groups Inc.	Applied Energy Groups Inc.	Applied Energy Groups Inc.	Deloitte & Touche	CH Guernsey & Company	Lukens Consulting Group	Utility & Economic Consulting, Inc.																				
	Check	Number		189357	195638	203677	191808	307492	188576	117347	187701	195935	201832	169010	176324	195668	200378	305577	189367	197302	203723	305697	307625	165301	168815	175914	189341	197895	201624	305805
		Date		3/3/99	4/1/99	5/1/99	3/4/99	5/27/99	1/21/98	4/22/98	2/18/99	4/1/99	4/29/99	11/30/98	12/29/98	3/31/99	4/21/99	5/19/99	2/28/99	3/31/99	4/30/99	5/31/99	6/30/99	11/2/98	11/30/98	1/4/99	3/1/99	4/3/99	4/30/99	6/1/99
		₽		*	*	*	*	*			*	*	*	*	*	*	*	*	*	*	•	*	*	*	*	*	*	*	*	*

RATE APPLICATION COSTS INCURRED BY WESTERN KENTUCKY GAS COMPANY CASE NO. 99-070

ACCOUNT WKG10143

-

,

		Labor			
Montines/	INIGEUINGS/ Alice	MISC			
Employee	Evnance				2,237.63 218.14 484.20 1,021.39 1,189.55 2,381.14 1,510.61 323.49 1,515.87
	Printing	מ 		4,042.61 808.53	
Public	Notice		1,137.78 1,544.13 1,544.13 2,171.61 2,171.61 2,126.25 3,498.39 970.20 1,173.69 4,411.26 4,231.71 1,265.62		
	Legal	1,512.70 8,430.55 5,676.78 3.757 83			
	Consultants				
	Vendor	Legal Sheffer, Hutchinson, Kinney Sheffer, Hutchinson, Kinney Sheffer, Hutchinson, Kinney Ward & Anderson, P.C.	Public Notice The Anderson News The Anderson News The Advocate Messenger Central KY News-Journal Daily News Glasgow Daily Times Glasgow Daily Times Glasgow Daily Times Glasgow Daily Times Glasgow Daily Times Glasgow Daily Times Central News News-Democrat & Leader The Sentinel-News Owensboro Messenger Inquirer The Paducah Sun The Times Leader	<u>Printing</u> Progress Printing Progress Printing	Employee Expense Betty L. Adams Rebecca M. Buchanan Donald P. Burman Donald P. Burman Tom S. Hawkins, Jr. Tom S. Hawkins, Jr. Tom S. Hawkins, Jr. Sid Hudson Thomas H. Petersen
Check	Number	205519 205519 205519 205519 201812	306879 307522 307522 306911 306935 307589 307589 306988 307589 306988 307682 306446 306446 306517	305742 306464	3703 3703 205903 3732 300704 3793 3954 304564 3697 3697 198600
1	Date	2/28/99 3/31/99 4/30/99 3/3/99	6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99 6/30/99	5/28/99 6/15/99	4/5/99 5/15/99 4/10/99 5/13/99 5/13/99 6/18/99 6/18/99 6/118/99 4/1/99
	•	* * * *	* * * * * * * * * * *	* *	* * * * * * * * *

-	Labor	\$452.05	\$452.05	
Meetings/	WISC.	65.00 35.00 65.00 135.38 185.12 185.12 18.19 18.19 13.77 9.95	\$1,032.35	
Employee	1,162.36		\$13,273.72	
Printing	<u>מ</u> 		\$4,851.14	
Public Notice			\$26,118.43	
Legal	2		\$19,377.86	
Consultants			\$192,328.65	\$257,434.20
Vendor	Employee Expense - cont'd Thomas H. Petersen Thomas H. Petersen	<u>Meetings/Misc cont'd.</u> Midwestern Climate Center - Data Midwestern Climate Center - Data Midwestern Climate Center - Data RiverPark Center Fairfield Inn Moonlite Ladderbacks Subway Subway Wetzels Wetzels Wetzels 19.5 hrs.	Totals	Grand Total to Date
Check No.	300711 201347	188134 188134 188134 188929 305756 305730 309390 193030 197028 197013 199351		
Date	6/7/99 5/1/99	2/3/99 2/23/99 3/2/99 5/18/99 5/18/99 5/18/99 5/18/99 5/18/99 6/22/99 3/23/99 4/12/99 4/12/99 4/12/99		
1	* *	* * * * * * * * * *		

\* Note: All costs are base year (BY) unless otherwise indicated

. .

•

Ņ

ო

APPLIED ENERGY GROUP, INC.

CONSULTING TECHNOLOG

490 WHEELER ROAD, SUITE 100 🛢 HAUPPAUGE, NEW YORK 11788 🔳 (516) 434-1414 🛢 FAX (516) 434-1212 🔳 E-MAIL: AEGNEWYORK@aoi.com

M-T.

5/14

March 3, 1999

Mr. Bill Senter VP Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, Kentucky 42303

RECEIVED MAR 1 0 1999 ATMOS ENERGY ACCOUNTS PAYABLE

Invoice:

Professional time for the month of February, 1999 in support of Western Kentucky Gas' upcoming Rate Case. (This invoice represents the completion of Tasks 1&2).

CONSULTING PROJECTS: Senior Partners Managing Consultants Support Subtotal	Hours 56.00 24.00 <u>8.00</u> 88.00	<u>Rate</u> \$180.00 \$90.00 \$40.00	<u>Total</u> \$10,080.00 \$2,160.00 <u>\$320.00</u> \$12,560.00
TOTAL CONSULTANT FEES:			\$12,560.00

INVOICE TOTAL (Net 30 Days):

\$12,560.00

If you have any questions or require additional information, please contact the undersigned.

Sincerely,

M

Michael Marks Senior Partner

MM/rz 3/10/99 CR#189357 #12560.00

and Serte 18607 6021

APPLIED ENERGY GROUP, INC. SYSTEMS CONSULTING TECHNOLOGY

490 WHEELER ROAD, SUITE 100 ■ HAUPPAUGE, NEW YORK 11788 ■ (516) 434-1414 ■ FAX (516) 434-1212 ■ WWW.APPLIEDENERGYGROUP.COM

April 1, 1999

Mr. Bill Senter VP Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, Kentucky 42303

## Invoice:

Professional time for the month of March, 1999 in support of Western Kentucky Gas' upcoming Rate Case. (This invoice represents the completion of Task 3.)

CONSULTING PROJECTS: Senior Partner Subtotal	<u>Hours</u> <u>8.00</u> 8.00	<u>Rate</u> \$180.00	<u>Total</u> <u>\$1,440.00</u> \$1,440.00
TOTAL CONSULTANT FEES:			\$1,440.00
INVOICE TOTAL (Net 30 Days):			\$1,440.00

If you have any questions or require additional information, please contact the undersigned.

Sincerely,

michael Markel Rz

Michael Marks Senior Partner

MM/Z

4/08/99 2 p.41/95638 #144000

Rec'd 4-5.99 #

RECEIVED

APR 07 1999

ATMOS ENERGY ACCOUNTS PAYABLE AEG APPLIED ENERGY GROUP, INC.

490 WHEELER ROAD, SUITE 100 🔳 HAUPPAUGE. NEW YORK 11788 🛢 (516) 434-1414 🔳 FAX (516) 434-1212 📕 E-MAIL: AEGNEWYORK@aol.com

May 4, 1999

Mr. Bill Senter VP Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, Kentucky 42303

## Invoice:

Professional time for the month of April, 1999 in support of Western Kentucky Gas' upcoming Rate Case.

CONSULTING PROJECTS: Senior Partner Manager Subtotal	<u>Hours</u> 2.00 <u>2.00</u> 4.00	<u>Rate</u> \$180.00 \$90.00	<u>Total</u> \$360.00 <u>\$180.00</u> \$540.00
TOTAL CONSULTANT FEES:			\$540.00
INVOICE TOTAL (Net 30 Days):			\$540.00

If you have any questions or require additional information, please contact the undersigned.

Sincerely,

michael marks (R3

Michael Marks Senior Partner

MM

1860

RECEIVED MAY 1 4 1999 ATMOS ENERGY ACCOUNTS PAYABLE

D ... ' -1 1

FROM: ATMOS ENERGY CORP FAX NO.: 972 855 3095 07-29-99 13:15 P.06 **Deloitte &** 49932 Touche Deloitte & Touche LLP Telephone: (214) 777-7000 Suite 1600 Chase Tov. er 2200 Ross Avenue Dallas, Texas 75201-6778 \$5985.00 \$1,000,000 W January 21, 1998 Mr. David Bickerstaff Vice President and Controller Atmos Energy Corp. P.O. Box 650205 Dailas, TX 75265 Dear David: Attached is our invoice for services associated with the Energas salvage rerous with proper grouping, review of answers provided by Energas personnel, reformatting data (or WKG), running edit and preparing reconciliation for the period November 29, 1998 through January 9, 1999. Fees: \$8,085.00 \$210 38.5 hrs. Q. Rhonda Watts JERNE QUIT \$8,085,00 Total RECEIVED MAR 0 4 1999 Very truly yours, Carlow Mar - Solarish TMOS ENERG) **ICCOO** Contraction of the second Donald S. Roff 1860 7 0014 02 DSR/all approved Enclosure CK# 188576 **Deloitte Touche** Tohmatsu

FROM:	ATMOS ENERGY CORP	FAX NO.:	972 855	3095	07-29-99	13:15	P.04
~	Touche LLP						
				Suite 1600 Texas Commerce Tilwer 2200 Ross Avenue Dailas, Texas T5201-6778	Telechone, (214	u 777-7000	

Aprii 23, 1998

Mr. David Bickerstaff Vice President and Controller Atmos Energy Corp. P.O. Box 650205 Dallas, TX 75265

Dear David:

Attached is our invoice for services associated with initiating study and setting up account sheets with pertinent data for the period March 22, 1998 through April 4, 1998.

Ń

with per thousa wette

**C** \* •

Fees:

Rhonda Watts	11 hrs.	Ù	\$210	<u>\$2,310.00</u>	
			Total	<u>\$2,310.00</u>	(2

3

Very truly yours,

WAZ

Donald S. Roff

DSR/agb

Enclosure:

CK#/17347

Oeloitte Touche Tohmatsu International

DELIVERY SYSTEMS, INC.

FROM: ATMOS ENERGY CORP.

FAX NO.: 972 855 3095

07-29-99 13:15 P.05

49932

Deloitte & Touche LLP Suite 1600 Chase Tower 2200 Ross Avenue Dallas, Texas 75201-6778

electione (114) 777-7000

February 18, 1999

**Neinitte** &

Touche

Mr. David Bickerstaff Vice President and Controller Atmos Energy Corp. P.O. Box 650205 Dallas, TX 75265

RECEIVED FEB 2 6 1999 ATMOS ENERGY ACCOUNTS PAYABLE

Dear David:

Fees: Energas

Attached is our invoice for services provided for the period January 10, 1999 through February 6, 1999. Services included finalizing Energas rates and presentation for management. Services for WKG included preliminary discussion with operations, simulation analysis and evaluation as well as preliminary rate calculations.

Don Roff	4 hrs @ \$325	\$1,300.00	
Rhonda Watts	23.5 hrs (a) \$210	54 <b>.9</b> 35.00	
John Johnson	22 hrs @ \$150	<u>\$3,300.00</u>	
10/01 10/04/00/1		\$9,535.00	V
Fees: WKG			
Don Roff	17 hrs @ \$325	\$5,525.00	
Rhonda Watts	106.5 hrs @ \$210	\$22,365.00	
John Johnson	26.5 hrs @ \$150	<u>\$3,975.00</u>	
JOHN JOHNSON		\$31,865.00	•

*,* •.

Total

Out of nocket expenses

A WAR AND A WERE THE AND A PARTY OF A PARTY

Very truly yours.

Donald S. Rotf

DSR/all

Enclosure Deloitto Touche Tehmatsu

1860 7 0014 10 vEnuer St 45. 第らに 1160 2 0014 NK(. ï 9 3[.872 CK# 187701

\$<u>9.75</u>

41.409.75

**Deloitte & Touche LLP** Suite 1600 Chase Tower 2200 Ross Avenue Dallas, Texas 75201-6778

Telephone: (214) 777-7000

March 4, 1999

**Deloitte &** 

Touche

Mr. David Bickerstaff Vice President and Controller Atmos Energy Corp. P.O. Box 650205 Dallas, TX 75265

Dear David:

Attached is our invoice for services associated with the preparation of supporting schedules, discussion of results and strategies, and preparation of draft report for WKG for the period February 7, 1999 through February 20,1999. A S/B dividion

RECEIVED

MAR 1 0 1999

ATMOS ENERGY

ACCOUNTS PAYABLE

Fees:

Don Roff Rhonda Watts John Johnson Administrative staff

8 hrs @ \$325 31.5 hrs @ \$210 7.5 hrs @ \$150 1.4 hrs @ \$50 Total

\$2,600	
6,615.00	
1,125.00	
<u>70.00</u>	
\$ 510.410.00	

Very truly yours,

Donald S. Roff

DSR/all

Enclosure

n00 Deloitte Touche Tohmatsu

1860 7 0014 07 Caperon plances bu 11/99



í



**Deloitte & Touche LLP** P.O. Box 840503 Dallas, Texas 75284-0503 Telephone: (214) 777-7000 Facsimile: (214) 777-7050

Facsimile: (214) 777-7050 Taxpayer I.D. No. 13-3891517

Atmos Energy P.O. Box 650205 Dallas, TX 75265

INVOICE NUMBER: 06088543

DATE: 3/4/99

For services and expenses rendered through February 20, 1999.

<u>\$10,410.00</u>

**Due and Payable Upon Receipt** 

Atmos Energy P.O. Box 650205 Dallas, TX 75265

Deloitte & Touche

**Deloitte & Touche LLP** 

P.O. Box 840503 Dallas, Texas 75284-0503 Telephone: (214) 777-7000 Facsimile: (214) 777-7050 Taxpayer I.D. No. 13-3891517.

INVOICE NUMBER: 06088543

DATE: 3/4/99

For services and expenses rendered through February 20, 1999.

<u>\$10,410.00</u>

Due and Payable Upon Receipt
PLEASE RETURN THIS COPY WITH PAYMENT

07-29-99 13:15 P.03



Ocioitte & Touche LLP Suite 1600

Chase Tower 2200 Ross Avenue Dallas, Texas 75201-6778 Telephone (214) 777-7000

April 1, 1999

Mr. Don Burman Assistant Controller Atmos Energy Corp. P.O. Box 650205 Dallas, TX 75265

Dear Don:

Attached is our invoice for services associated with the workpaper compilation and review, and numerous discussions with WKG personnel.

**:** -

Fees:

Rhonda Watts John Johnson 15.5 hrs @ \$210 5 hrs @ \$150 Total \$3,255.00 <u>750.00</u> <u>\$4,005.00</u>

Very truly yours,

Donald S. Roff

DSR/all

Enclosure

Deloitte Touche Tohmatsu

VL# 195935 4.199 FROM: ATMOS ENERGY CORP

FAX NO.: 972 855 3095



Dejoitte & Touche LLP Suite 1600 Chase Tower 2200 Ross Avenue Dallas, Texas 75201-6778 Telephone, (214) 777-7000

13:15

P.02

07-29-99

Place Cette Visk to Visk to Vision

April 29, 1999

Mr. Don Burman Assistant Controller Atmos Energy Corp. P.O. Box 650205 Dailas, IX 75265

RECEIVED

MAY 0 7 1999 ATMOS ENERGY ACCOUNTS PAYABLE

Dear Don:

Attached is our invoice for professional services rendered and expenses incurred for the period March 21 through April 17, 1999. Services included completion of depreciation rate calculations, supporting schedules, workpapers, and depreciation study report for WKG/

Fees:		·
Don Roff	13 hrs @ \$325	\$4,225.00
Rhonda Watts	40 hrs @ \$210	8,400.00
John Johnson	16.5 hrs @ \$150	2,475.00
Secretary	14.28 hrs @ \$50	<u>714.00</u>
,	Total	<u>\$15,814.00</u>
Francas	-	
Expenses		\$240.30
Meals		<u>9.75</u>
	Total	\$250.05
	TOTAL	<u>\$16.064.05</u>

.....

Very truly yours,

Donald S. Rotf

DSR/all

Enclosure

Deloitte Touche Tolunatsu

(K# 201832

1860 Anales Burn 5/84 approved

....

114964

Deloitte & Touche LLP Tel Suite 1600 Chase Tower 2200 Ross Avenue Dallas, Texas 75201-6778

Telephone: (214) 777-7000

May 27, 1999

Mr. Don Burman Assistant Controller Atmos Energy Corp. P.O. Box 650205 Dallas, TX 75265

**Deloitte &** 

Touche

RECEIVED JUN 0.3-1999 ATMUS ENERGY ACCOUNTS PAYABLE

RECEIVED

Dear Don:

JUN 1 1 1999

Attached is our invoice for professional services rendered and expenses incurred for the period April RGY 18 through May 15, 1999. Services included final report and supporting workpapers for WKC and PAYABLE re-evaluation of Energas net salvage.

Fees: Don Roff 1 hr @ \$325 Ξ \$325.00 Rhonda Watts (WKG) 8.5 hrs @ \$210 1,785.00  $\omega$ (Energas) 6 hrs @ \$210 1,260.00 ٤. Total \$3,370.00 Expenses:

Postage/Courier \$83.03 × ERPENDITUR DIG SHARED SERVICES 453.03 Very truly yours ATTN 10165 \$ 1,585. Donald S. Roff WK610170 was Western Kentrich 0144 DSR/all 13141516 91011 proved Enclosure 1234563 **Deloitte Touche** Tohmatsu

C. H. GUERNSEY & COMPANY Engineers • Architects • Consultants

Remittance Address RO Box \$50017 Oklahoma Dily, OK 73185-0017 Federa-I.D # 73-0590316



B 1860 1 0021 09 2/10/99 fg 8700 0 6111 09 000 5004000

CR#169010 \$150.00

C. H. GUERNSEY & COMPANY Engineers • Architects • Consultants

Femiliance Address. - P.O. Sox 550017 Oklahoima City: OK-73185-0017 Federal I D. # 73-0590816

WESTERN KENTUCKY GAS CO. BILL SENTOR 2401 N. HARTFORD ROAD OWENSBORO, KY 42303 INV. DATE : 11/30/98 INVOICE # : 074207 PROJECT # : 0K-40052-000 CLIENT # : C2684 CLIENT REF:

1

#### LABOR & EXPENSE DETAIL

LABOR

2.00	Ducient	07 40050 000	150.00
** Total	Project	ок-40052-000	150.00



. .

GUERNSEY	35940	)	Remittance Address Oklahoma City Federai	PO Box 85001 v. OK 73195-001. LD. # 73-059031
		· · · · · · · · · · · · · · · · · · ·		
	RECEIVED			
	JAN 1 2 1999			
WESTERN KENTUCKY GAS CO BILL SENTOR 2401 N. HARTFORD ROAD OWENSBORO, KY 42303	ATMUS ENERGY ACCOUNTS PAYABLE	INV. DATE : 12 INVOICE # : 07 PROJECT # : OK CLIENT # : C2 CLIENT REF:	/29/98 4304 -40052-000 684	
FOR PROFESSIONAL SERVIC EXPERT WITNESS	ES RENDERED THROUGH	11/30/98		
LABOR			2,700.00	
OUT-OF-POCKET EX	PENSES		.72	
	*PLEASE PAY THIS	AMOUNT*	2,700.72	

1860 7 0021 09 8700 0 4111 09 000 5004000 5/B

allian Sertige.

CR.H. 176324 # 2700.72

C. H. GUERNSEY & COMPANY Engineers • Architects • Consultants

Remittance Address: PO Box 850017 Oklahoma City, OK 73185-0017 Federal I D. # 73-0590516

				λ.
UPSTERN KENTICKY CAS CO		T	TT DAME - 10/0	0 /00
BILL SENTOR		INVOICE # : 074304		
2401 N. HARTFORD ROAD	PROJECT # : OK-40052-000			
OWENSBORO, KY 42303	CLIENT # : C2684			34
		CI	LIENT REF:	
LABOR & EXPENSE DETAIL				
•				
LABOR				
	HOURS	RATE		AMOUNT
CONSULTING ECONOMIST	9.00	175.00	)	1,575.00
1 ENGINEER/CONSULTANT/ANALYST	15.00	75.00	)	1,125.00
	24.00			2,700.00
				·
PENSES				
COPY CHARGES - IN HOUSE				AMOUNT
COPY CHARGES - INHOUSE				. 72
JOB CHARGEABLE				
				./2
	** Total	Project	OK-40052-000	2,700.72
				=================

.

GUERNSEY


N. 11/1--- 11

C. H. GUERNSEY & COMPANY Engineers • Architects • Consultants

Romittanov Adareas – F.C. Box 851117 Orianoma Div. Cr. 73185-1117 Feaural C. # 73-0531718

WESTERN KENTUCKY GAS CO. BILL SENTOR 2401 N. HARTFORD ROAD OWENSBORO, KY 42303

LABOR & EXPENSE DETAIL

, LABOR

	HOURS	RATE	AMOUNT
CONSULTING ECONOMIST	29.00	175.00	5,075.00
SR2 ENGINEER/CONSULTANT/ANAL	24.25	85.00	2,061.25
	53.25		7,136.25
ENSES			
			AMOUNT
COPY CHARGES - IN HOUSE			
COPY CHARGES - INHOUSE			1.84
JOB CHARGEABLE			

1.84

\*\* Total Project OK-40052-000 7,138.09 

RECEIVED

APR 07 1999 ATMOS ENERGY ACCOUNTS PAYABLE



INV. DATE : 03/31/99 INVOICE # : 074869 PROJECT # : OK-40052-000 CLIENT # : C2684 CLIENT REF:

.



C. H. GUERNSEY & COMPANY conference + Architects + Conductant,

ೆ ಗ್ರಾಪಕ್ಷಿಗಳ ಸಂಘಟನೆಂದು ಕಾರ್ಯಕ್ರಮ ಸ್ಥಾನ ಸಂಶ ವಿಶೇಷ ನಡೆ ನಿರ್ಧಾರ್ಷ ನಿರ್ಧಾರ್ಧ ಕೊಡುವುದು ಕಾರ್ಯವರ್ಷ

25940

WESTERN KENTUCKY GAS CO. BILL SENTOR 2401 N. HARTFORD ROAD OWENSBORO, KY 42303 INV. DATE : 04/21/99 INVOICE # : 075052 PROJECT # : 0K-40052-000 CLIENT # : C2684 CLIENT REF:

FOR PROFESSIONAL SERVICES RENDERED THROUGH 03/31/99 EXPERT WITNESS

> LABOR 1,725.00 OUT-OF-POCKET EXPENSES 1.04 \*PLEASE PAY THIS AMOUNT\* 1,726.04

1840 7 0021 09

RECEIVED

APR 2 8 1999 ATMOS ENERGY ACCOUNTS PAYABLE

1 101€. ●130/99 Ck # 200318 #1726.04

C. H. GUERNSEY & COMPANY End ricers • Architects • Consultants

Hommanco Aduress — P.O. Box SECORT K. Frema City, C.K. Totes-optic Federal (D. # 19-95905 fr

-----

X.

WESTERN KENTUCKY GAS CO.	INV. DATE : 04/21/99
BILL SENTOR	INVOICE # : 075052
2401 N. HARTFORD ROAD	PROJECT 🖸 : OK-40052-000
OWENSBORO, KY 42303	CLIENT # : C2684
	CLIENT REF:

LABOR & EXPENSE DETAIL

.

LABOR

,

GUERNSEY

CONSULTING ECONOMIST SR2 ENGINEER/CONSULTANT/ANAL	HOU 5. 10.	RS 00 00	RATE 175.00 85.00	)	AMOUNT 875.00 850.00
	15.	00			1,725.00
PENSES					
COPY CHARGES - IN HOUSE					AMOUNT
COPY CHARGES - INHOUSE JOB CHARGEABLE					1.04
					1.04
	** Tota	al Pa	roject	OK-40052-000	1,726.04

RECEIVED APR 2 8 1999

ATMOS ENERGY ACCOUNTS PAYABLE

C. H. GUERNSEY & COMPANY Engineers • Architects • Consultants

Remittance Address: P.O. Box 850017 Oklahoma City, OK: 73185-0017 Federal I.D. # 73-0590816





3/10/99 CR#189367 \$ 33,997.17 Schedule of Services Rendered and Compensation Due

February 1, 1999 - February 28, 1999

### WESTERN KENTUCKY

RECEIVED

			MAR		
Case:	Rate Case Testimony	4	MAR 1 0	1999	
		ETA	ATMOS EN	ATMOS ENEDA	
Please remit to:	Lukens Consulting Group, Inc. 1100 Louisiana, #2750	ulting Group, Inc. a, #2750		COUNTS PAYABLE	
Invoice# 1001	Houston, Texas 77002		N		
TOTAL: PRO	DFESSIONAL SERVICES AND EXPEN	NSES:		\$33,997.17	
Dillion Datail Calendar					
Billing Detail February	:				
Consultants					
Protessional Services		Houro	Data/Us	(* A mount	
	lov Lukono		Rale/Hr.		
	Jay Lukens	17.7	\$325.00 \$325.00	· \$0,752.50	
	Daniel ives	12.15	\$225.00	\$10,308.70	
	Research Assistance			\$10,065.00	
TOTAL PROFESSION	NAL SERVICES:			\$32,186.25	
Expenses:					
	Computer Services			\$0.00	
	Mail			\$0.00	
	Phone			\$0.00	
	Travel, Meals			\$1,796.08	
	Reproductions			\$0.00	
	Purchased Data			\$14.84	
	Total:			\$1,810.92	
TOTAL · PROFESSIO				\$32,186.25	
TOTAL FYPENSES	HAL JERVICES.			\$1 810 02	
TOTAL PROFESSIO	NAL SERVICES & EXPENSES			\$32 007 47	

Millian Serting 1840 7 6021 09

Yay P. Lukens

March 4, 1999 Tax ID #76-0588625

Par's stolm



Consulting Group, Inc.

March 3, 1999

Mr. William Senter Vice President – Rates And Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, KY 42303

Dear Bill:

Enclosed is our invoice for services through February 28, 1999. The invoice includes time and expenses for consultants and support staff. As discussed in the project estimate I sent you on February 25<sup>th</sup>, the activities to date include:

- Lukens, Ives client meeting in Nashville, TN
- Ives/Osterholm client meeting in Owensboro, KY
- Research assistance on area expansion tariffs, KY regulations, WKY tariff
- Meeting agenda preparation, client follow-up discussions
- Initial incremental cost study preparation
- Initial testimony preparation

Thank you very much for your business. Please call if you have any questions regarding this invoice.

Sincerely,

Daniel M. Ives Vice President

1100 Louisiana Suite 2750 Houston, Texas 77002 713.659.1900 713.659.1914 fax

	5	5275		
[]	Schedule of Services Re	endered and Compensation Due		
	March 1, 199	99 - March 31, 1999		
Lukens Consuiting Group, Inc.	۱۸.E	STERN KENTUCKY	·	
	Dub Occa Testimory			
Case:	Rate Case Testimony			
Please remit to:	Lukens Consulting Group,	Inc.		
	Houston, Texas 77002		X	
Invoice# 1002				
TOTAL: PROFES	SIONAL SERVICES AND E	XPENSES:		\$18,997.39
Billing Detail March:				
Consultants				
Professional Services		Hours	Rate/Hr.	\$Amount
	Jay Lukens	0.6	\$325.00	\$195.00
	Daniel Ives	53.7	\$225.00	\$12,082.50
	Ryan Osterholm	47.3	\$125.00	\$5,912.50
TOTAL PROFESSIONAL	SERVICES:			\$18,190.00
Expenses:				
	Computer Services			\$234.02
	Mail			\$12.75
	Phone	DEACH		\$15.86
	Travel, Meals	RECEIVED		\$544.76
	Reproductions			\$0.00
	Purchased Data	APR 1,4 1999		\$0.00
		ATMOS ENERGY ACCOUNTS PAVADLE	;	▲ 10.2 ± 22 = 44.0 ± 1 ± 440.0
	Total:	S S S S S S S S S S S S S S S S S S S		\$807.39
	SERVICES		<del></del>	\$18 190 00
TOTAL: EXPENSES:				\$807.3
TOTAL: PROFESSIONA	L SERVICES & EXPENSES:			\$18,997.39
Actor		,	nn	
all			mel	_
41,5199		$\forall$	Jay P. Lukens	-
101202			April 7, 1999	
(RH 1410000	000100	Та	x ID #76-0588625	5
29 1840	TODATUT			
yaan /	$\int \mathcal{Q}_{-1}$			
)'' / 1, <b>//</b> .	Henlyt			
r (NUUP	my ser - 1 p c		fs.	
2	$\sim$ $\sim$			



April 7, 1999

Mr. William Senter Vice President – Rates And Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, KY 42303

### Dear Bill:

Enclosed is our invoice for services in March 1999. The invoice includes time and expenses for consultants and support staff. Activities include:

- Deliver 1<sup>st</sup> draft testimony to client, review with client
- Deliver 2<sup>nd</sup> draft testimony to client
- Attend rate case meeting, Owensboro, March 23<sup>rd</sup>
- Prepare 10,15,20 year Premise Charge and interest method variations
- Make additional testimony and exhibit revisions following 3/23 client meeting
- Obtain new customer hook-up benchmark data for WGL, Piedmont. Lexis search for other benchmarks
- Multiple data collection contacts with Messrs. Senter, Doggette, Petersen, Hudson, Adams and Buchanen

Thank you very much for your business. Please call if you have any questions regarding this invoice.

Sincerely,

Daniel M. Ives Vice President

> 1100 Louisiana Suite 2750 Houston, Texas 77002 713.659.1900 713.659.1914 fax

Understanding Energy Market Economics

Schedule of Services Rendered and Compensation Due

April 1, 1999 - April 30, 1999



WESTERN KENTUCKY

Case:	Rate Case Testimony
Please remit to:	Lukens Consulting Group, Inc. 1100 Louisiana, #2750
	Houston, Texas 77002

Invoice# 1003

## TOTAL: PROFESSIONAL SERVICES AND EXPENSES:

Billing Detail April:

- Consultants
- Professional Services

	Hours	Rate/Hr.	\$Amount
Jay Lukens	1.5	\$325.00	\$487.50
Daniel Ives	43.25	\$225.00	\$9,731.25
Ryan Osterholm	25.9	\$125.00	\$3,237.50
Greg Murphy	7.5	\$130.00	\$975.00

### TOTAL PROFESSIONAL SERVICES:

### Expenses:

Computer Services		\$0.00
Mail	DECENTED	\$0.00
Phone	RECEIVED	\$11.26
Travel, Meals	1 1 1999	\$591.28
Reproductions	MAY 1 4 1999	\$0.00
Purchased Data	ATMUS LIVERGY ACCOUNTS PAYABLE	\$0.00
	1.00-	

Total:

\$602.54

\$14,431.25

\$15,033.79

TOTAL: PROFESSIONAL SERVICES: TOTAL: EXPENSES:	\$14,431.25 \$602.54
TOTAL: PROFESSIONAL SERVICES & EXPENSES:	\$15,033.79
1 lote.	a all

5/17/99 CR# 203728 503.3.79 1860 70021 09

Jay P. Lukens

١,

May 5, 1999 Tax ID #76-0588625

Ren'A Sholog\_10

			RECEIN	/ED	
	Schedule of Services Rende	ered and Compensation Due	JUN 1 1	1999	
	May 1, 1999 -	May 1, 1999 -May 31, 1999 ATM		MUS ENERGY	
Consetting Group, Inc.	WESTE	WESTERN KENTUCKY			
Case:	Rate Case Testimony				
Please remit to:	Lukens Consulting Group, Inc.				
	1100 Louisiana, #2750				
Invoice# 1004	Houston, Texas 77002		N		
TOTAL: PROFES	SSIONAL SERVICES AND EXPE	NSES:		\$6,896.06	
				<u></u>	
Billing Detail May:					
Consultants					
Protessional Services		Hours	Rate/Hr.	\$Amount	
	Daniel Ives	19.6	\$225 AD	<b>*</b> 4 4 0 5 0 4	
	Ryan Osterholm	15.3	\$225.00 \$125.00	\$4,185.00 \$1,912.50	
	Christy Spivey	0.3	\$100.00	\$30.00	
TOTAL PROFESSIONAL	SERVICES:			\$6,127.50	
Expenses:					
	Computer Services			\$171.29	
	Mail Bhana			\$0.00	
	Travel Meals			\$0.00	
	Reproductions			\$558.27	
	Purchased Data			\$0.00	
				\$39.0C	
	Total:			\$768.56	
TOTAL: PROFESSIONAL	SERVICES:			00 407 FO	
TOTAL: EXPENSES:				90,727.50 \$769 56	
TOTAL: PROFESSIONAL	SERVICES & EXPENSES:				
1KG 10143	Defer Consul	EWKG263A	ay P. Lukens		
ite:	1860 14068	109000 Tax I	D #76-0588625		
18/99 bir 2056-97	William Senti	FA-			
#305697	. (/ (				

16. 24. Schedule of Services Rendered and Compensation Due June1, 1999 - June 30, 1999 Lukens WESTERN KENTUCKY Rate Case Testimony Please remit to: Lukens Consulting Group, Inc. 1100 Louisiana, #2750 Houston, Texas 77002 Invoice# 1005 TOTAL: **PROFESSIONAL SERVICES AND EXPENSES:** \$247.50 **Billing Detail June:** Consultants **Professional Services** Hours Rate/Hr. \$Amount **Daniel Ives** 1.1 \$225.00 \$247.50 TOTAL PROFESSIONAL SERVICES: \$247.50 . Expenses: **Computer Services** \$0.00 Mail \$0.00 Phone \$0.00 Travel, Meals \$0.00 RECEIVED Reproductions \$0.00 **Purchased Data** \$0.00 JUL 1 4 1999 ATMOS ENERGY Total: \$0.00 ACCOUNTS PAYABLE TOTAL: PROFESSIONAL SERVICES: \$247.50 **TOTAL: EXPENSES:** \$0.00 TOTAL: PROFESSIONAL SERVICES & EXPENSES: \$247.50 lete: 1/16/99 Jay P. Lukens CK# 307625

13 DEFER MUDIC 2636

Case:

\$ 241.50

INKG JOI4

July 9, 1999 Tax ID #76-0588625

Dan't Thislag 4

539185

# Utility and Economic Consulting, Inc.

£1

November 2, 1998

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road P.O. Box 866 Owensboro, Kentucky 42302

INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period October 1 through October 31, 1998.

Total Total Hourly Hours Rate Fees 13.0 \$125 \$ 1,625.00 RECEIVED 115.50 Expenses NOV 1 6 1998 ATMOS ENERGY ACCOUNTS PAYABLE TOTAL AMOUNT DUE \$ 1,740.50 Payment is due by November 30, 1998. Please make check payable to Utility and Economic Consulting, Inc. and remit to: Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507 2/10/99 fg. 1860 7 0021 09 81000 6111 09 000 500 4000

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

October 1 through October 31, 1998

## Summary Statement

### CONSULTING SERVICES

Name	Hours	Rate	То	tal
Chuck Buechel (CDB)	0.0	\$150	\$	0
Jim Sharpe (JKS)	13.0	\$125	<b>\$</b> 1	,625.00
Total Consulting Services	13.0		<b>\$</b> 1	,625.00

### EXPENSES

٤

Mileag	e				
	CDB JKS	0 miles 350 miles	33¢/mile 33¢/mile	\$	0 115.50
Lodgin	g (receip	ts attached)			0
Meals	(receipts	attached)			0
Federa	l Express	(receipt at	tached)		0
Parkin	g (receip	t attached)		-	0
Total Expen	ses			\$	115.50
TOTAL AMOUN	T DUE			\$	1,740.50

Billing Statement for Jim Sharpe

October 1 through October 31, 1998

Consulting Services

Oct 20 Met with West Kentucky Gas to discuss potential rate case and other PSC related activities. (8 hours)

Oct 22 Trip to Commission to retrive tariffs and other information. (5 hours)

Total Consulting Hours 13.0

### Out of Pocket Expense

Oct 20	Trip to Owensboro and return. (350 miles @ \$.33 per mile)	\$ 115.50
	Total Expense	\$ 115.50

## Gonzales, Frances (temp)

	_ •.
	Frn:
	Sent:
-	To:
	Subject:
	•

.

• •

Purcell, Jackie S. Tuesday, February 09, 1999 4:32 PM Gonzales, Frances (temp) Rate case invoice coding

Hi, Frances,

s,

I was just wondering if you have had a chance to take care of this yet!

Thanks, Jackie

### Frances,

I have some invoices that the coding needs to be changed on when you get a chance, please. All of these were coded to 8700 0 6111 09 000 5004000. They should be changed to 1860 7 0021 09.

١,

<u>Vendor</u> Utility & Economic Consulting, Inc. "	<u>Inv. Date</u> 01/04/99 11/30/98 11/02/98	Amount \$ 625.00 VH 53985 \$ 2016.50 - 9028 - 11/20/98
C.H. Guernsey & Company	12/29/98 11/30/98	\$2700.72 \$ 150.00

I tried to get on the payables system and look up the information to try and make it easy for you, but the system is down.

Please advise me when you have made these changes.

Thanks for your help,

Jackie



Utility and Economic Consulting, Inc.

November 30, 1998

RECEIVED DEC 07 1998 ATMOS ENERGY ACCOUNTS PAYABLE

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road P.O. Box 866 Owensboro, Kentucky 42302

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period November 1 through November 30, 1998.

	Total Hours	Hourly Rate	Total
Fees	16.0	\$125	\$ 2,000.00
Expenses			16.50

TOTAL AMOUNT DUE

\$ 2,016.50

Payment is due by December 31, 1998. Please make check payable to Utility and Economic Consulting, Inc. and remit to:

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507

1860 7 0021 09 2/10/99 /9-1 87000 6111 09 000 500 4000 Will

tility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

November 1 through November 30, 1998

## Summary Statement

## CONSULTING SERVICES

Name	Hours	Rate	Tot	tal
Chuck Buechel (CDB)	0.0	\$150	\$	0
Jim Sharpe (JKS)	16.0	\$125	\$2	,000.00
Total Consulting Services	16.0		\$2	,000.00

## EXPENSES

Mileage	CDB JKS	0 miles 50 miles	s 33¢/mile s 33¢/mile	\$	0 16.50
Lodging	(receipt	s attached	i)		0
Meals ()	receipts	attached)			0
Federal	Express	(receipt a	attached)		0
Parking	(receipt	attached	)	-	0
Total Expense	es			\$	16.50
TOTAL AMOUNT	DUE			\$	2,016.50

Billing Statement for Jim Sharpe

November 1 through November 30, 1998

Consulting Services

Nov 20 Reviewed Court decision and agreement with Hopkinsville concerning franchise tax. (2 hours)

- Nov 23 Trip to Commission to gather information and work on putting together a comparison with other companies. (8 hours)
- Nov 24 Worked on tariff comparisons. (6 hours)

TOTAL CONSULTING HOURS 16.0

### Out of Pocket Expense

Nov 23	Trip to Frankfort and return. (50 miles @ \$.33 per mile)	\$ 16.50
TOTAL EX	PENSE	\$ 16.50

## Gonzales, Frances (temp)

	Fra.n:
	Sent:
	To:
ì	Subject:
	•

• \* ...\*

Purcell, Jackie S. Tuesday, February 09, 1999 4:32 PM Gonzales, Frances (temp) Rate case invoice coding

Hi, Frances,

I was just wondering if you have had a chance to take care of this yet!

### Thanks, Jackie

### Frances,

I have some invoices that the coding needs to be changed on when you get a chance, please. All of these were coded to 8700 0 6111 09 000 5004000. They should be changed to 1860 7 0021 09.

<u>Vendor</u> Utility & Economic Consulting, Inc.	<u>Inv. Date</u> 01/04/99 11/30/98 11/02/98	Amount \$ 625.00 VH 53985 \$ 2016.50 - 9028 - 11/20/98
C.H. Guernsey & Company	12/29/98 11/30/98	\$2700.72 \$ 150.00

I tried to get on the payables system and look up the information to try and make it easy for you, but the system is down.

1

Please advise me when you have made these changes.

Thanks for your help,

Jackie

 $oldsymbol{U}$ tility and **E**conomic **C**onsulting, Inc.

January 4, 1999

RECEIVED

JAN 1 1 1999

ATMOS EIVERGY

ACCOUNTS PAYABLE

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road P.O. Box 866 Owensboro, Kentucky 42302

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period December 1 through December 31, 1998.

Total Hourly Total Hours Rate 5.0 625.00 Fees \$125 \$ 0 Expenses LTOTAL AMOUNT DUE \$ 625.00 Payment is due by January 31, 1999. Please make check payable to Utility and Economic Consulting, Inc. and remit to: Utility and Economic Consulting, Inc. 101 N. MIII St., Suite 7 Lexington, Kentucky 40507 5B 1860 7 0021 09, 2/10/99 89 \$625 87000 6111 09000 5004000 CKtopay, William Sentiga. 1/2/99

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

December 1 through December 31, 1998

## Summary Statement

### CONSULTING SERVICES

Name	Hours	Rate	тс	otal
Chuck Buechel (CDB)	0.0	\$150	\$	0
Jim Sharpe (JKS)	5.0	\$125	\$	625.00
Total Consulting Services	5.0		\$	625.00

## EXPENSES

Mileage	CDB JKS	0 miles 0 miles	33¢/mile 33¢/mile	\$ 0 0	
Lodging	(receipts	attached)		0	
Meals (1	receipts at	ttached)		0	
Federal	Express (	receipt at	tached)	0	
Parking	(receipt a	attached)		 0	•
Fotal Expense	es			\$ 0	
TOTAL AMOUNT	DUE			\$ 625.00	)

Billing Statement for Jim Sharpe

December 1 through December 31, 1998

### Consulting Services

Dec	17	Discussed	regulation	and	alternatives	for
		filing in	Kentucky.			N.
		(2 hours)	-			

- Dec 18 Reviewed information on CIAC and discussed with PSC Staff. (2 hours)
- Dec 23 Continued to work on CIAC. (1 hours)

TOTAL CONSULTING HOURS \_\_\_\_\_5.0

## Gonzales, Frances (temp)

- 2
Erc of
Combi
Sent:
To:
Subject:
Canjoon

Purcell, Jackie S. Tuesday, February 09, 1999 4:32 PM Gonzales, Frances (temp) Rate case invoice coding

Hi, Frances,

I was just wondering if you have had a chance to take care of this yet!

#### Thanks, Jackie

### Frances,

I have some invoices that the coding needs to be changed on when you get a chance, please. All of these were coded to 8700 0 6111 09 000 5004000. They should be changed to 1860 7 0021 09.

<u>Vendor</u> Utility & Economic Consulting, Inc. "	<u>Inv. Date</u> 01/04/99 11/30/98 11/02/98	Amount \$ 625.00 \$ 2016.50 \$ 1740.50 - 9028 - 11/20/0
C.H. Guernsey & Company	12/29/98 11/30/98	\$2700.72 \$ 150.00

I tried to get on the payables system and look up the information to try and make it easy for you, but the system is down.

1

198

Please advise me when you have made these changes.

Thanks for your help,

Jackie

F7985

# Utility and Economic Consulting, Inc.

- 1

RECEIVED

March 1, 1999

MAR 1 0 1999

ATMOS ENERGY ACCOUNTS PAYABLE

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road P.O. Box 866 Owensboro, Kentucky 42302

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period February 1 through February 28, 1999.

	Total Hours	Hourly Rate	Total
Fees	25.0	\$125	\$ 3,125.00
Expenses			181.33

TOTAL AMOUNT DUE

2# 18934

330

\$ 3,306.33

Payment is due by March 31, 1999. Please make check payable to Utility and Economic Consulting, Inc. and remit to:

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507

1860 26

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

## February 1 through February 28, 1999

Summary Statement

CONSULTING SERVICES

Name	Hours	Rate	Total
Chuck Buechel (CDB)	0.0	\$150	<b>\$</b> 0
Jim Sharpe (JKS)	25.0	\$125	\$ 3,125.00
Total Consulting Services	25.0		\$ 3,125.00

## EXPENSES

Mileage						
	CDB JKS	0 350	miles miles	33¢/mile 33¢/mile	\$	0 115.50
Lodging	(receip	ts att	ached)			65.83
Meals (r	receipts	attac	hed)			0
Federal	Express	(rece	eipt at	tached)		0
Parking	(receip	t atta	ched)		-	0
Total Expense	s				\$	181.33
TOTAL AMOUNT	DUE				\$	3,306.33

Billing Statement for Jim Sharpe February 1 through February 28, 1999

CONSULTING SERVICE

- Feb 1 Discuss rate case filing with Mr. Senter. Gathered information. (1.0 hours)
- Feb 2 Trip to Commission to get information on depreciaton and other information to forward to Bill. (2.0 hours)
- Feb 8 Discuss depreciation and rate case. (1.0 hours)
- Feb 25 Work with WKG on rate case and testimony. (8 hours)
- Feb 26 Work with WKG on rate case and testimony. (8.0 hours)
- Feb 27 Work with WKG on rate case and testimony. (5.0 hours)

TOTAL CONSULTING HOURS 25.0

### Out of Pocket Expenses

Feb. 26	Trip to Owensboro and return. (350 miles @ \$.33 per mile)	\$ 1	15.50
Feb. 26	Overnight at Fairfield Inn		55.83
	Total Expenses	1	81.33



OWENSBORD FAIRFIELD INN 800 GALEN DR. OWENSBORD KY. 42303 (502) 688-8887

### TOLL FREE RESERVATIONS (800) 228-2800

#### GUEST CHARGES SUMMARY

٩

Thank you for choosing Fairfield Inn. We hope you enjoyed your stay. If you are not an INNsiders Club or Marriott Rewards<sup>54</sup> member already, fill out an application today. For a \$10 one-time enrollment fee, you can start enjoying the benefits of INNsiders Club membership like Express Check-In, a free USA Today each weekday morning, personal check cashing privileges, and automatic membership in our exciting frequent guest program, Marriott Rewards. Or just join Marriott Rewards for free. Either way, you'll earn valuable points toward free vacations when you stay at any Fairfield Inn, or one of the other participating lodging brands in the Marriott family of hotels. Call 800-443-7396 to join.

		-	F	airfield Inn Staff
GUEST NAME	JIM SHARPE	DEPART	ROUM WIMBER ROOM TYPE NO. OF GUESTS RATE CLERK	RE-50154 402 DD 1 59.00 308
DATE	de en el el el	DESCRIPTION	CHARGES	CREDITS
25-99 25-99 25-99 25-99	LONG ROOM ROOM CITY	DISTANCE #502~606-272 CHARGE - REGULAR TAX OCC.	2~ 1.41 59.00 3.65 1.77	
26-99	VISA			65.83

\*\* BALANCE \*\* .00 .00 .00 uirfield Inn voted "Best Overall Value" in the 1995 Zagat's Hotel Survey

Utility and Economic Consulting, Inc.

.۱,

April 3, 1999

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road P.O. Box 866 Owensboro, Kentucky 42302

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period March 1 through March 31, 1999.

Total Hourly Total Hours Rate Fees 8.0 \$125 \$ 1,000.00 RECEIVED Expenses 16.50 APR 1 6 1999 ATMOS ENERGY ACCOUNTS PAYABLE TOTAL AMOUNT DUE \$ 1,016.50 PRH 197895 \$ 1016 Brayment is due by April 30, 1999. Please make check payable to Utility and Economic Consulting, Inc. and remit to: Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507 \$ 1016.50 .1860 7 0021 09 Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

Rec'd 4/14/99

March 1 through March 31, 1999

## Summary Statement

## CONSULTING SERVICES

Name	Hours	Rate	Τo	tal
Chuck Buechel (CDB)	0.0	\$150	\$	0
Jim Sharpe (JKS)	8.0	\$125	\$ 1	,000.00
Total Consulting Services	8.0		\$ 1	,000.00

## EXPENSES

.

.

Mileage						
	CDB JKS	0 50	miles miles	33¢/mile 33¢/mile	\$	0 16.50
Lodging	(receipts	at	tached)			0
Meals ()	receipts a	itta	ched)			0
Federal	Express	rec	eipt at	tached)		0
Parking	(receipt	att	ached)		-	0
Total Expense	es				\$	16.50
TOTAL AMOUNT	DUE			· .	\$	1,016.50

Billing Statement for Jim Sharpe

March 1 through March 31, 1999

### CONSULTING SERVICE

- Mar 1 Work on testimony, (3.0 hours)
- Mar 9 Trip to Commission to review information on Columbia future test year and provide to Mr. Sentor. (3.0 hours)
- Mar 24 Review regulations and discuss late payment with Mr. Sentor. (1.0 hours)
- Mar 25 Discuss WNA and other proposed rates with Mr. Sentor. (1.0 hours)

TOTAL CONSULTING HOURS 8.0

### Out of Pocket Expenses

Mar 9	Trip to Frankfort.	
	(50 miles @ \$.33 per mile)	\$ 16.50
	Total Expenses	\$ 16.50



April 30, 1999

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, Kentucky 42303

RECEIVED

MAY 0 5 1999 ATMOS ENERGY ACCOUNTS PAYABLE

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period April 1 through April 30, 1999.

	Total Hours	Hourly Rate	Total
Fees	8.0	\$125	\$ 1,000.00
Expens	es		16.50

TOTAL AMOUNT DUE

R# 201624

#1016.50

\$ 1,016.50

Payment is due by May 31, 1999. Please make check payable to Utility and Economic Consulting, Inc. and remit to:

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507

Le 11 11 am Ser 1840 7 002

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

April 1 through April 30, 1999

## Summary Statement

## CONSULTING SERVICES

Name	Hours	Rate	Total	
Chuck Buechel (CDB)	0.0	\$150	<b>\$</b> 0	
Jim Sharpe (JKS)	8.0	\$125	\$ 1,000.	00
Total Consulting Services	8.0		\$ 1,000.	00

## EXPENSES

Milea	ge				
	CDB JKS	0 miles 50 miles	33¢/mile 33¢/mile	\$	0 16.50
Lodgi	ng (receipt	s attached	.)		0
Meals	(receipts	attached)			0
Federa	al Express	(receipt a	ttached)		0
Parki	ng (receipt	attached)		-	0
Total Expe	nses			\$	16.50
TOTAL AMOU	NT DUE			\$	1,016.50

Billing Statement for Jim Sharpe

April 1 through April 30, 1999

٦,

### CONSULTING SERVICE

- Apr 20 Work on testimony, (4.0 hours)
- Apr 21 Trip to Commission to review information for Mr. Senter. (3.0 hours)
- Apr 22 Discussed testimony and made suggestions to Mr. Senter. (1.0 hour)

Total Consulting Hours \_\_\_\_\_8.0

### Out of Pocket Expenses

Apr 21	Trip to Frankfort. (50 miles @ \$.33 per mile)	\$ 16.50
	Total Expenses	\$ 16.50

Utility and Economic Consulting, Inc.

June 1, 1999

Mr. Bill Senter Vice President - Rates & Regulatory Affairs Western Kentucky Gas Company 2401 New Hartford Road Owensboro, Kentucky 42303

### INVOICE

For consulting services provided by Utility and Economic Consulting, Inc. to Western Kentucky Gas Company for regulatory support and advice during the period May 1 through May 31, 1999.

ALEANAY STAUODDA	Total Hours	Hourly Rate	Total
Fees 6661 8 TO THE	33.0	\$125	\$ 4,125.00
Expenses 03N1394			175.55

TOTAL AMOUNT DUE

\$ 4,300.55

Payment is due by June 30, 1999. Please make check payable to Utility and Economic Consulting, Inc. and remit to:

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7 Lexington, Kentucky 40507

WKG 10143 Majer - Consulture Les. WKG 2626 1860

Utility and Economic Consulting, Inc. 181 N. Mill St., Suite 7, Lexington, Kentucky 40507 (606) 255-4850

May 1 through May 31, 1999

Summary Statement

## CONSULTING SERVICES

Name	Hours	Rate	Total	
Chuck Buechel (CDB)	0.0	\$150	\$	0
Jim Sharpe (JKS)	33.0	\$125	\$4	,125.00
Total Consulting Services	33.0		\$4	,125.00

### EXPENSES

Mileage						
	CDB JKS	0 350	miles miles	33¢/mile 33¢/mile	\$	0 115.50
Lodging	(receip	ts at	tached)			60.05
Meals (r	eceipts	atta	ched)			0
Federal	Express	(rec	eipt at	tached)		0
Parking	(receip	t atta	ached)		-	0
Total Expense	es				\$	175.55
TOTAL AMOUNT	DUE				\$	4,300.55
UTILITY AND ECONOMIC CONSULTING, INC.

Billing Statement Jim Sharpe

May 1 through May 31, 1999

## CONSULTING SERVICE

- May 7 Review testimony. (7.0 hours)
- May 8 Review testimony and prepare for meeting. (5.0 hours)
- May 10 Meeting on rate case. (8.0 hours)
- May 11 Meeting on rate case. (8.0 hours)
- May 17 Reviewing and working on testimony. (3.0 hours)
- May 18 Reviewing testimony. (2.0 hours)

Total Consulting Hours 33.0

## Out of Pocket Expenses

Мау	10	Trip to and return to Owensboro. (350 miles @ \$.33 per mile)	\$ 115.50
May	10	Hotel	\$ 60.05
		Total Expenses	\$ 175.55

	Executive- Rivern	Inn 10nt			One Ex Owensbo 1-: Call Toll F 1-:	ecutive B bro, Kentu 502-926-8 ree For Re 300-626-1	oulevard cky 42301 000 eservations 936
			ROOM NU	MBER	GUEST	NUMBER	
			ARRIVAL		DEPAR	TURE DATE	
			NIGHTS	GUEST PLAN	NO. IN PARTY		
CUTIVE	INN - RIVERMONT		RATE		TAX	тот	AL
NSBORO, 2) 926-	, KY 42301 -8000		METHOD	DF:PAYMENTSUE		<b>R</b> ] 960000	
			CLERK		2548		
AY'S DATE				1.071 1.071	9798 2000 - 200	12 In the state	10/20/98
			i. T				1 No.
VANDER	RELT						
TED STA	KENTUCKY 40571		ດງຮຽດຍ	ER CORD	6.0110.064	01500507	n
			¥7.0.070700¥		an a	en en els	2 
				TRACTE		4,31-198 #	ci.cl.(J
	10/20/98	CHECKOUT	``TIME <b>:</b>	06:38	pq	05E <b>:</b> 1	
Ъ	RANSACTION	REFERENCE		CHE	RGES CRE	DITS	BALANCE
19 F 19 S 19 C	OOM CHARGE STATE SALES JAX ICCURANCY NAX SCOVER-GARD	RH 2548 SH RH 2548 SH M 2548 SH SM 2548 SH SM 206991	IARPES IARPES IARPES 502528		5.00 3.40 1.65	0.05	
	K NEX RESERVATI	S GDEST F CUA DIRECT AU		n XXXXXXX CHII	IS. NET	en seren en	<b>\$ (0.00</b>
			1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1				and so the Second Second Se Second Second
THE LESS							
•			GUEST SI	gunts	the	yco	

- Jul-28-99 16:23 F

From

The Law Firm of SHEFFER HUTCHINSON KINNEY 115 East Second Street Owensboro, Kentucky 42303 (502)-684-3700

Tax I.D. No. 61-1023845

Please mail all payments to the above address.

Atmos Energy Corporation P.O. Box 650205 Dallas TX 75265 Page: 1 02/28/99 ACCOUNT NO: 2609-0003K STATEMENT NO: 37

ATTN: Mr. Glen Blanscet

Public Service Commission Matters

Jul-28-99 16	:23	From-						T-026	P.03/	/19 F-924	
Ac Pu	mos E blic	Inergy Servi	Corpo	ration Mission	Matte	rs		ACCOUNI STATEMENI	' NO : ' NO :	Page: 2 02/28/99 2609-0003K 37	
02/02/99								1			
MRH	Exte	nsive ernin	telep) g rate	one cor case up	iferenc Xdate	e with	Bill	Senter	. 75	5 93.75	Α
MRH	Telep poten	bone Itial	confere case is	ence wit saues	ch Lee	Everet	t con	cerning	- 40	50.00 Å	at L
	Jul-28-99 18 Ac Pu 02/02/99 MRH 02/03/99 MRH	Jul-28-99 16:23 Atmos H Public 02/02/99 MRH Exte Conc 02/03/99 MRH Telep	Jul-28-89 18:23 From- Atmos Energy Public Servi 02/02/99 MRH Extensive concernin 02/03/99 MRH Telephone potential	Jul-28-99 18:23 From- Atmos Energy Corpo Public Service Com 02/02/99 MRH Extensive teleph concerning rate 02/03/99 MRH Telephone conference potential case is	Jul-28-00 18:23 From- Atmos Energy Corporation Public Service Commission 02/02/99 MRH Extensive telephone cor concerning rate case up 02/03/99	Jul-28-99 18:23 From- Atmos Energy Corporation Public Service Commission Matter 02/02/99 MRH Extensive telephone conference concerning rate case update 02/03/99	Jul-28-88 18:23 From- Atmos Energy Corporation Public Service Commission Matters 02/02/99 MRH Extensive telephone conference with concerning rate case update 02/03/99	Jul-28-60 18:23 From- Atmos Energy Corporation Public Service Commission Matters 02/02/99 MRH Extensive telephone conference with Bill concerning rate case update 02/03/99	Jul-22-00 18:23 From- Atmos Energy Corporation Public Service Commission Matters 02/02/99 MRH Extensive telephone conference with Bill Senter concerning rate case update 02/03/99 MRH Telephone conference with Lee Everett concerning potential case issues	Jul-28-88 18:23 From- Atmos Energy Corporation Public Service Commission Matters 02/02/99 MRH Extensive telephone conference with Bill Senter concerning rate case update .79 02/03/99 MRH Telephone conference with Lee Everett concerning potential case issues .40	Jul-22-09 18:23 From Atmos Energy Corporation Public Service Commission Matters 02/02/99 MRH Extensive telephone conference with Bill Senter concerning rate case update .75 93.75 02/03/99 MRH Telephone conference with Lee Everett concerning potential case issues .40 50.00 f.

·					
Atmos	Energy Corporation		ACCOUNT	9 02 NO: 2609	age: 3 /28/95 -0003K
Public	Service Commission Mat	ters	STATEMENT	NO :	37
	- · -	• • • • •	H	ours	
			×		
02/09/99					
MRH Tele rate	phone conference with case	Bill Senter conc	cerning	.25	31.25 fe

.....

MRH Conference with Bill Senter concerning rate case ,33 41.25

. . . . . . .

. ....

Jul-28-99	J 18:24 From-	Γ-026	P.05/1	19 F-924
1	Atmos Energy Corporation ACCOU STATEME Public Service Commission Matters	INT 1 INT 1	NO : ; NO :	<b>Page: 4</b> 02/28/99 2609-0003K 37
MF	H Telephone conference with Doug Walther concernin rate case	a HC	)URS .20	25.00 \$
4C 4C	9 H Telephone call with RH re: rate case; review cas schedule	e	- 25	46.25 K
MR	H Telephone conference with Jack Hughes concerning sample meter filing and rate case		.25	31.25 fr
JA	I Review case files re: future test year applications	3.	50	647.50 J
02/24/99 MRH	Telephone conference with Bill Senter concerning rate case	.2	5	31.25 Ant

	tmos Energy Corporation ublic Service Commission Matters	ACCOUNT STATEMENT	NO : : : NO :	Page: 5 02/28/99 2609-0003K 37
02/25/99 J <b>A</b> I	) I Telephone call with BS re: rate case prep	H aration	.17	31.45_V.
MRH 02/26/99 JAH	To review of background material regarding adjustment application using a forcasted to period; review of regs concerning notice of intent; work on first draft of notice of i telephone conference with Bill Senter Research PSC regulations re: rate case not	ice	. 00	375.00-C 46.25-F
MRH	Preparation of revisions of letter to Bob Hazelrigg; telephone conference with Bill & telephone conference with Jack Hughes; preparation of revisions to notice of inter overnight filing with the Public Service Commission	Senter; At for	50	62.50 Jas

T-026 P.06/19 F-924

Jul-28-99 16:24

From-

Jui-28-99 1	6:25 From-			T-026 P.08/1	9 F-924
•					5-0 4-1
)		The Law SHEFFER HUTCH 115 East Seco Owensboro, Ker (502)-684	Firm of INSON KINNEY and Street Stucky 42303 4-3700		
		Tax I.D. No.	61-1023845		
	Please (	mail all payments	to the above add	Ce85.	
Ati P.( Da AT	mos Energy Co: O. Box 650205 llas TX 752 TN: Mr. Glen 1	Slanscet	ACCO STATES	OUNT NO: 2 TENT NO:	Page: 1 03/31/99 2609-0016K 1
19	99 Rate Case	this entire hill goes in	elevedaciont		
03/01/99			- V	HOURS	
JAH	Telephone cal	ll with RH re: not	ice of intent	- 20	30.00 Law
03/02/99 MRH	Telephone con	iference with Doug	Walther	,25	31.25 And
03/08/99 MRH	Telephone con Hughes concer	ferences with Bil ming scheduling c	1 Senter, Jack of conference	. 75	علم 25 . 32 هسر 75 . 32
MRH	Review of exp	erimental rate ca	se filing by Delt	a 1.50	187.50 Aat
03/12/99 JAH	Conference wi preparation 9	th RH; DW; BS re: 99-070	rate case	7.75	1,162.50 ممتل
MRH	To round trip work with Jac concerning ra	) travel to Louisv 1k Hughes, Doug Wa 1ts case	ille and all day lther, Bill Sente	er 9.00	1,125.00
03/19/99 JAH	Review Case 9 prepare for 3 99-070	95-010 testimony, 3/22 conference	exhibits, etc. to	3.50	525.00 Aud
03/22/99 MRH	Review of fil conference an	e in preparation WKG concerning r	for conference; ate case	7.50	937.50
JAH	Conference re 99-070	: rate case prepa	ration	6.50	975.00



1

£

T-026 P.09/19 F-924

At	mos Energy Corpor	ation		ACCOUNT	NO :	Page: 2 03/31/99 2609-00161	z Ə K
19	99 Rate Case		5	STATEMENT	<b>NO :</b>	:	L
03/23/99				1	HOURS	3	
MRH	Continued confer	ence at WKG on	n rate case	x	6.50	812.50	)
JAH	Conference re: r 99-070	ate case prepa	aration	:	10.25	1,537.50	)
03/25/99 Mrh	Telephone confer for PSC	ence with Rich	ard Raft, co	ounsel	. 25	31.25	i
03/27/99 MRH	Telephone confer Jack Hughes conce	ence with Doug erning rate ca	Walther and Se	with	.40	50.00	
MRH	Receipt, review a testimony of Char Louisianna PSC	and initial co rles Goodman a	nsideration nd Tristan W	of est in	1.00	125.00	
99/29/99 Jah	<b>Review Case 95-0</b> 99-070	10 for public	notice forma	E	1.80	<b>270</b> .00	
JAH	Draft filing req 99-070	irement sched	ules		1.25	187.50	
03/30/99 JAH	Staff conference rate case 99-070	re: use of sh	ort form not:	ice for	20	45.00	
03/31/99 MRH	Telephone confere Senter and Smitty review of publica	nces with Jac Taylor conce tion in prior	k Hughes, Bil rning rate ca filing	ll 180;	.75	93.75	
	FOR PROFESSIONAL	SERVICES REND	BRED	- 5	9.45	8,220.00	n. '
ATTORNEY		RECAPITUL Title	Ation Hours H	iourly rat	re	TOTAL	"" Mai
Hutchins Jack Hug	on/Partner hes/ Partner	Partner Partner	27.90 31.55	\$125.0 150.0	D0 D0	\$3,487.50 4,732.50	

03/05/99 Owensboro office - Postage

1.65

Acr	nos Energy Corporation	ACCOUNT	NO:	Page: 3 03/31/99 2609-0016K	
199	99 Rate Case	STATEMENT	NO:	1	
03/25/99	Owensboro office - Copies at \$.20			3.60	
	TOTAL EXPENSES	۰.		5.25	
03/12/99	Mileage from Frankfort to Louisville and for conference; parking fee	return			
	99-070			45.90	
13/23/33	hotel for conference	return,		159.40	
	TOTAL ADVANCES			305 30	
				205.30	
	TOTAL CURRENT WORK			8,430.55	
	BALANCE DUE		$\langle$	\$8,430.55	l

		4C	CO'JN	(18)		CEN	EH (12,	Mu	Cr. + (10)
Amount	ACC	X	Elur.	Detail	0.	Twn	Resp Car	Tun	Approp.
	1-47	¢	77-5-	11-18	1-2	3-5	6-12	1-3	4-10
8430.55	18/0	7	0021		69	000	005200		
-									
		T			1-				
		$\dashv$			4		ومعبدية بمشعدة فالقات		
[									
				A					
Care on	Aprile		A			Auction			
5-10-99		X		Danser					

Statement reflects payments received through March 31, 1999 (Please include account number on payment) Jul-28-99 16:26 From-

(.C 5-12-4

The Law Firm of SHEFFER HUTCHINSON KINNEY 115 East Second Street Owensboro, Kentucky 42303 (502)-684-3700

## Tax I.D. No. 61-1023845

Please mail all payments to the above address.

•		Page: 1
Atmos Energy Corporation		04/30/99
P.O. Box 650205	ACCOUNT NO:	2609-0016K
Dallas TX 75265	STATEMENT NO:	2

ATTN: Doug Walther

1

-19	19 Race case This all gase into the for the of	ate .	Carl
4/05/99 Jah	Review draft abbreviated notice; telephone call with Bill Senter 99-070	.20	30.00
4/07/99 <b>JA</b> H	Draft rate application 99-070	2,75	412.50
9/99 JAH	Telephone call with Bill Senter re rate conference 99-070	.10	15.00
/14/99 MRH	Receipt and review of correspondence from Bill Senter concerning non-recurring charges/premises charge issue; review of regs and consideration of same; telephone conferences with Jack Hughes and with Bill Senter	1.00	125.00
JAH	Telephone conference with Randy Hutchinson re premises charge for rate case; letter to PSC re abbreviated notice/file PSC 99-070	.75	112.50
/16/99 MRH	Telephone conference with Doug Walther concerning rate case filing requirement	- 20	25.00
19/99 MRH	Review of proposed prefiled testimony of Conrad Gruber, Earl Fischer and others	2.50	312.50
Jah	Telephone call with Doug Walther re case preparation 99-070	.10	15.00

T-026 P.14/19 F-924

At	mos Energy Corporation	ACCOUNT	NO ·	Page: 3 04/30/99 2609-00164	
19	99 Rate Case	STATEMENT	NO:	2009-00188	
	TOTAL EXPENSES			4.29	
04/21/99	Mileage to Owensboro @ \$.32 a mile	Υ.		106.24	
	TOTAL ADVANCES			106.24	
	TOTAL CURRENT WORK			5,676.78	h
	PREVIOUS BALANCE			8,430.55	•
	BALANCE DUE			\$14,107.33	
	PAST DUE AMOUNTS				

		~~~~+++			
181+	121-180	91-120	61-90	31-60	0-30
	~~~~~~				***
0.00	0.00	0_00	0.00	0.00	14,107.33

		λC.	COUNT	(18)	I	CEN	TER (12)	Pro	Code (10)
Amount	Acct	X 5	Elem 6-9	Detail 10-18	Dv 1-2	3-6	6-12	1-3	4-10
5676.78	1860	5	00.20		69	000	0052002		
					-	<b></b>			
						<u> </u>	1		
	L	L	[]			┣			
	L	L			+-		<b></b>	+	<b> </b>
		5		+A-		ALE		<b></b>	
5,00	TT:	7	0.5	Drice	5	f			
	<u> </u>	2.2		The second second	-				

Statement reflects payments received through April 30, 1999 (Please include account number on payment)

LAW OFFICES WARD & ANDERSON, P.C. SUITE 503 1000 THOMAS JEFFERSON STREET, N.W. WASHINGTON, D.C. 20007

> (202) 298-6910 TELECOPIER (202) 298-6914 March 3, 1999

RECEIVED

MAY 07 1999

14-152

ATMOS ENERGY ACCOUNTS PAYABLE

Attention: Doug Walther, Attorney

To professional services rendered during February 1999:

### WESTERN KENTUCKY GAS COMPANY

ATMOS Energy Corporation

Dallas, Texas 75265-0205

Post Office Box 650205

Re: CP98-0645

\$ 110.00 0.50 hours at \$220

A.W.Anderson @ \$220 Review data responses and order.

Re: Miscellaneous

300.00 1.25 hours at \$240 \$ \$ 275.00 1.25 hours at \$220

D.B.Ward @ \$240 Conference D.Walther <u>re</u> procedures; review material re LDC unbundling.

Monitor FERC orders and notices and A.W.Anderson @ \$220 transmit relevant orders and notices to D.Walther; phones D.Walther.

<u>Re:</u> PL99-0001

60.00 0.25 hours at \$240 Ś 0.50 hours at \$220 110.00 \$

Obtain and fax material to D.Walther. D.B.Ward @ \$240

A.W.Anderson @ \$220 Review notice; phone D.Walther.

(Continued on next page)

ATMOS Energy Corporation March 3, 1999 Page 2 Re: RM98-0010 1.25 hours at \$240 \$ 300.00 1.50 hours at \$220 330.00 \$ D.B.Ward @ \$240 Review comments re Reliant P.L. issue. A.W.Anderson @ \$220 Review comments; review order re waiver of shipper title; letter J.Hack. Re: RM98-0012 0.50 hours at \$220 \$ 110.00 A.W.Anderson @ \$220 Review comments and report. Re: RP91-0203 1.00 hours at \$220 220.00 Ś A.W.Anderson @ \$220 Obtain and review court decision. Re: RP95-0064 0.50 hours at \$240 \$ 120.00 D.B.Ward @ \$240 Review settlement; conf. A.Anderson. <u>Re:</u> RP95-0112 0.25 hours at \$240 60.00 Ŝ 0.75 hours at \$220 Ŝ 165.00 D.B.Ward 0 \$240 Review reply comments re settlement. A.W.Anderson @ \$220 Review reply comments re settlement. Re: RP97-0344 0.25 hours at \$220 \$ 55.00 A.W.Anderson @ \$220 Review notice. Re: RP98-0014 1.50 hours at \$220 \$ 330.00 A.W.Anderson @ \$220 Review settlement agreement and other docs.; phone D.Walther. (Continued on next page)

ATMOS Energy Corporation March 3, 1999 Page 3 Re: RP98-0140 0.25 hours at \$220 55.00 A.W.Anderson @ \$220 Review compliance filing. <u>Re</u>: RP98-0206 0.25 hours at \$220 \$ 55.00 A.W.Anderson @ \$220 Review draft order. Re: RP99-0167 0.25 hours at \$220 \$ 55.00 A.W.Anderson @ \$220 Review draft order. <u>Re:</u> RP99-0209 0.25 hours at \$220 \$ 55.00 A.W.Anderson @ \$220 Review new cash-out report. Re: RP99-0214 0.50 hours at \$240 \$ 120.00 1.75 hours at \$220 385.00 D.B.Ward @ \$240 Review TXGT cash-out report; conf. A.Anderson. A.W.Anderson @ \$220 Review cash-out report; conf. D.Ward; phone D.Walther; draft and file motion to intervene. Re: Y2K 1.50 hours at \$220 330.00 Attend conference at FERC; memo A.W.Anderson @ \$220 D.Walther.

\$ 3,600.00

(Continued on next page)

ATMOS Energy Corporation March 3, 1999 Page 4

## Expenses:

<u>`</u>,

Telephone	\$ 5.70
Telecopy	2.50
Copying	80.80
Postage	16.28
Local Transp.	20.00
Office Supplies	4.41
Couriers	23.38
FERC Docs.	4.76

\$\_\_\_\_\_157.83 <u>3,757.83</u> fw \$\_\_\_\_

|loto: 5/7/99 CR# 201812 #3,757.83

TOTAL

1			AC	CC: INT	T (18)					/
1	Amount	Acc:	IX	Elom		_	CEN	TER (12)	Proj.	Code (10)
F	2	1-0	6	5-9	10-18	Dv 1-2	Twn 3-5	Resp Ctr 6-12	Twn	Approp.
	3,157.83	1860	7	0021		09	000	005,200		4-10
+										
F			-							
F			$\neg$		· · · · · · · · · · · · · · · · · · ·					
-			-+			+				
D	210:	pprog	<u> </u> :	~/	9/	1	Audito			
	2125199	L	<u>2h</u>	K14	Daver	_	100110	0.		

THE ANDERSON NEWS PO BOX 116 IAWRENCEBURG, KY 40342-011		999 WESTERN KY G	AS CO 15.541
	0 1137.78	PAYMENT DUE BY	7/30
ADVERTISING INVOICE and STATEMENT	1137.78	• 00	• 00 • 00
WESTERN KY GAS CD ATTCS PAYABLE 2401 NEW HARTFORD RD OWENSBORD KY 42909-19	12	FIE AN LANDMARK CO SHELBYVILLE	DERSON NEWS MM. NEWSPAPERS BOX 1118 , KY 40066-1118
AI         PAGE         SI         BILLING DATE         To I         BILLED ACCOUNT NUMBER 171           1         6/30/1999         003-787663         003-7876653         003-7876653         003-7876653         003-7876653         003-7876653         003-7876653         003-7876653         003-7876653         003-7876653	ADVERUSER/CUENT NUMBER	Change of Address?	
PLEASE DETACH AND TD AVOID DELAYS IN CH REMITTANCE STUB WITH OR PRINT THE ACCOUNT	RETURN UPPER PORTION REDITING YOUR AC YOUR PAYMENT IN NUMBER FROM THI	WITH YOUR REMITTANCE COUNT. PLEASE RET THE SUPPLIED EN S BILL ON YOUR CH	TURN THE VELOPE
10       Date       11       REE [12]13]14       DESCRIPTION-OTHER COMMENTS/CHARGES         6/09       PREVIOUS       BAL'ANCE         6/16       1999       RATE       CASE         6/16       1999       RATE       CASE       FILING         6/23       TOTAL!       ANDERSON       NEWS	SX 21.50         5.88           3X 21.50         5.88           3X 21.50         5.88           3X 21.50         5.88           3X 21.50         5.88	19 GROSS AMOUNT 0 00 379.26 379.26 379.26 1137.78	ISC.         20         NET AMOUNT           • 00         379•26           • 00         379•26           • 00         379•26           • 00         379•26           • 00         1137•78
WKC	3 10143 Task DEFEI	E ( 10 7/8/99 V	KG 2636
Note: 7/14/99 0k# 306879 #1137.78		RECEIV JUL 09 19 ATMOS ENER ACCOUNTS PAY	ED 99 IGY ABLE
TOTAL: DUE		1137.78	.00 1137.78
211 CURRENT NET AMOUNT DUE 221 30 DAYS 60 1137.78 .00	OO	GROSS AMOUNT DUE	ANNUAL PERCENTAGE RATE) WILL BE IMPOSED
FOR BILLING QUESTIONS: THE ANDERSON NEWS, F 502-839-6906	PO BOX 116, LAWRE	ENCEBURG, KY 4034	<sup>2</sup> <b>S</b> ∎I™
231 [25] 1 BILLING PERIOD [3.1 BILLED ACCOUNT JUNE 1999 003-7876 YEAR TO DATE DOLLARS CONTRACT EXPERATION	ADVERTISER INFORMATION NUMBER 771 ADVERTISER/CUENT NUMBER 563 CONTRACT REQUIREMENTS	U 21 ADVERTI WESTERN KY GA	SER/CLIENT NAME
1137.78			

LANDMARK COMMUNITY NEWSPAPERS INC . PO BOX 549 . SHELBYVILLE KY 40066-0549 . 502/633-4334

Ehe, Advorale Allessenger)	06/01/99 -	- 06/30/99	WEST	ERN KENT	UCKY GAS	COMF'ANY
	1,544	4.13	APPLIED AND	Net 3	0 days	PANARAT SELECTION AND
330 S. FOURT STREET + DANVILLE KY + (606/238-255)	1,544.13	A 1997 1997 1997 1997 1997 1997 1997 199	00		.00	. OO
1 06/30/99 WESTEEN FO BOX 6 DANVILLE	KENTUCKY GAS 40 KY 40423-0640					
	DETACH AND RETURN UPPE	R PORTION WITH	YOUR RE	MITTANCE	19	
5/31 BALANCI UBLICATION: THE ADVOCATE-MI	E FORWARD ESSENGER					0.00
6/03 LEGAL (	AD	3x2	1.5I 4.5T	1		514.71
6/10 ATTN: 0	GIL MORRISON	3x2	1.5I 4.5I	1 7.98		514.71
6/17 GIL MDI Ad Class Fublica	RRISON-LEGAL ss Totals: ation Totals:	3x2 6 \$1,544.13 \$1,544.13	1.5I 4.5I	1 7.98 193.500	inch	514.71
9 1544.13 WKC 1544.13 WKC William Cent Cto 17/16/99 (k# 307522 # 15444 13	5 10143 Tast DEFE	E Misi 1/12/99	02	737 RECE JUL 1 ATMOS E ACCOUNTS	IVED 1999 NERGY PAYABLE	
STATEMENT OF ACCOUNT AGING OF PAST DUE	AMOUNTS			·	52	<b>U</b>
1,544.13 .00	.00		.00	TO A CONTRACT OF A CONTRACT		1,544.13
P.O. 8 330 S DANV (608) 47490 06/01/99 - 06/30/	30x 149 . FOURTH STREET 'ILLE, KY 40423-0149 236-2551 or 1-800-428-0409		UNAPPLIE	D AMOUNTS ARE	INCLUDED IN T	OTAL AMOUNT DUE

	LEI MDS			BOX	5-JUI 1138	UKNAL		UNE 19	99 H	ESTERN		MENT	15522
$\overline{\nabla}$			5.11					LO40.04	PAYM	ENT DU	EBY	7/30	
			0.1010/0					1040.04		.00		.00	.00
A	ADVE	HTISIN	GINVC		DACCOUNT	MENT NAME AND ADDRESS	ر		9		REMITIAN	CE ADDRESS	
		JEST ATTN 2401 DWEN	ERN MAR NEL SBOR	KY G IAM I HAR J KY	AS TFORI 42	D RD 303-131	.2			CENTR LANDMA SHELBY	AL KY NI RK COMM PO BOX VILLE, I	EWS . NEW 1118 (Y 40	JOURNAL SPAPERS 3066-1118
AT PAGE # 15	L	BAUNG	DATE	61 BILLER	ACCOUNT	NUMBER 71	ADVERTISER/CUENT N	UMBER	Change of J	Address?	• • • •		. • •
1	6/	/30/	1999	085	-910	600			on reve	erse.		ND\$_	
			AVO MITT PRI	PLEA ID D ANCE NT T	ELAY	ACH AND S IN CR B WITH CCOUNT	RETURN UPP EDITING YOUR PAT NUMBER I	ER PORTION YOUR AC MENT IN ROM THI	COUNT THE S BILI	OUR REMI PLEA SUPPLI L DN Y	TTANCE SE RETUI ED ENVEL DUR CHE(	RN TH OPE	fE-
IO DATE	11 REF.	121314	DES	RIPTION-OT	HER COMME	NTS/CHARGES	TEI BRUED UNITS	TEI TIMES RUN TEI RATE	. 19 . GRO		DISC.		20 NET AMOUNT
6/29 6/07 6/14	<b>T</b> .01		MENT AL A CENT	DVER RAL	TISII KY NI	L888 Ng Ng Ews Jou	6X13.50 6X13.50 RNAL	6.73 6.73	1	520.02 520.02 545.13 545.13 090.26	L 2 L 2 5	5.11 .11 .22	520.02 520.02 520.02 520.02 1040.04
													ал 1. с тоба 1. с с с с с с с с.
•							WKG	1014/3	DEF2	ER AV	7/8/99	W	KG. 2636
		0					little	form f.	Sent	logh	•		
		In	L.					4		RE	CEIV	ED	
		TI in	LIQQ							JI	JL 0 9 19	99	
	ć	?RH	:30E	6911						ATI ACCO	IOS ENER UNTS PAY	GY ABI F	
		# [[	040.	04									
	*** TC	TEA TAL	R SHE DUE	ETS	TO FO	DLLOW I	N SEPARA	TE MAIL	ING FR	ROM LOC 90.26	AL PAPE 50	R*** •22-	1040.04
STATEMENT			JNT AG	ING OF PAS			AL PAYMENT IS NOT REC	EIVED BY DUE DATE, A	SERVICE CHARGE	OF 1.5% PER		ANNUAL PERC	CENTAGE RATE) WILL BE IMPOSED
1	040	• 94			.00		•00		.00		1040.04	202 100	1040.04
FOR BILLING Q	UESTIO	NS:	CENT 4271	RAL   9-11	(Y NE 38, 2	WS JOU 70-465	RNAL, PO -8111	BOX 11	38, CA	MPBELL	SVILLE,	ΚY	sal
241 251		JUN	ailung per	100 1999	6	BILLED ACCOUNT	NUMBER 171	DMERIISER INFORMATIC ADMERIISER/CUENT NUM	ABER [2]	ESTERN	ADVERTISER	CUENT NAM	мЕ
YEAR IC	D DATE DO			CO	NTRACT EXPRA	MON	CONTRACT	EQUIREMENTS		ONTRACT CURREN	TMONTH	<u> </u>	
L	LA	NDM	ARK CO	DMMU		WSPAPERS	INC. • P.O. B	OX 549 • SHI	ELBYVILLE	, KY 40066	-0549 • 502	2/633-4	1334

AFFERENCE CODES PLEASE SEE REVERSE SIDE FOR COMPLETE LIST OF REFERENCE CODES.	1012) 1114/199 1114/199 1114/199 1114/199 1114/199	<u>JOUTE</u> 11) AFFINENCE 11 6/02 3050 R BALANG 6/09 3050 R 1999 F 6/16 3050 R 1999 F	2 WESTERN KY GAS C/O JERRY HARMO P.O. BOX 598 BOWLING GREEN	DAILY NEWS P.O. BOX 90012 BOWLING GREEN, KENTUCKY 42102 502 - 781-1700
AGING OF PAST DUE AMOUNTS 90 DAYS 90 D	WKG 10143 Tac Ulliam Seren	CHARGE OF CREDITS DE FORWARD CASE FILING NATE CASE FILING NATE CASE FILING	LEDAGOQUNE COMPANY JN KY 42101	-9012 3 DOCUMENTINO: 4 BILLING
25     TOTAL     21     CURRENT       YS     AMOUNT DUE     CROSS AMOUNT       111     21     CROSS AMOUNT       200     111     21       200     21     21       200     21     21       2100     21	R=DEFER Mac 7/8/99 WK	43.00 11.49 723.8 43.00 11.49 723.8 43.00 11.49 723.8	7 BILLED (ACCOUNT NO. 11 8 4096-04 9 VIAMEOF ADVERTISE	V NEWS PUBLISHING CO FEDERAL I.D. NO. 61
THANK	RECEIVER JUL 0 9 199 ACCOUNTS PAR		ADVENTISER / CUEN	PAYNENT
PLEASE DETACH AND RETURN WITH REMITTANCE DAILY N P.O. BOX 9 BOWLING GREEN, KENT 502 - 781-1	AGY IS O	DATE PREFERENCE BAL FWD 6/02 3050 R 6/09 3050 R 6/16 3050 R	DAILY NEWS P.O. BOX 90 BOWLING GRE	28 4096-1
12,171.61		AMOUNT .00 723.87 723.87 723.87	-9012	DUT NO.



\_\_\_\_\_

•	INCORPORATED FUDISHED Daily Excep P.O. BOX 729 - HOPKINSVI MAIN OFFICE 502/886-4444 • 1	EU ETA. No Sunday LLE, KY. 42241 FAX 502/887-3222			1423	одт: 06/30/99	555 Sanate Mean 14 M	7 NUMBER 764
WEST 2401 P O 1 OWENS	ERN KENTUCKY GAS ( NEW HARTFORD ROA) BOX 866 BBORD, KENTUCKY	20 0 42303					TOTAL EXP DATE	
Date 1	TRANSACTION DES	CRIPTION	INCHE9	LINES 30 30 30	BAT 6 6	CHARGE 170 1166 170 1166 170 1166		ESRS
	wko p Itury	uls pere Interfor		WROS	REC JUL ATMOS CCOUN	EIVED 0.9 1999 IENERIGN TS: PAYABLE		
HAVE ANY AT 270-B	**************************************	ARBANNEL AR IE LOTH DE EA NG YOUR BILL FOR YOUR BU	CH MONTH CONTACT BINESSI	***** IF YON KRIST	i 12 			
PREVIOUS BALANC		. 00	. SERVICE CHARGE . . OO	+ 3498	Harges 1. 39		AMOL DUE N 78, 97	NI A
CURRENT 98. 39 CHARGES AND PAY REFLECTED ON YOU TO BALANCES DUE O	ACCOUNT ST 30 DAYS 00 .00 MENTS MADE AFTER THE LAST DA JR NEXT STATEMENT. A 11/2% FINANC OVER 30 DAYS. AN ANNUAL RATE OF	ATUS 90 DAYS & OVER 1 . 00 Y OF THE STATEMENT M E CHARGE WILL BE ADDE 18% (A MINIMUM CHARGE REMITTAN	MONTH WILL BE ED EACH MONTH OF 50¢).					

P O Box 5	23. Madisonville Ken	iucky 42431	-		199 - 00/30	199	W W	ESTERN	KENTI	JCKY GAS O	<b>o</b> .	14ig
TEC 100	21 al.4-3300 FAX (502) FED (D# 51-0266566	821-6855		2126 2	E	- UNAC	LED MOL	<u></u> 131		TERMS OF T	THENT	
	ADVERTISINO		21 CUBLEN	T NET AMOUNT DUE	2	JO DAYS		<u> </u>		DUE BY 07	31/39	
INVO	HCE and STATEMEN	ī	212	6 25		00						90 DATS
I AGE . ST	BILLING DATE 8		BILLED ACC	OUNT NAME AND ADD	U			0	00		0.0	00
05	20.00	WESTER		KY CLC CO			?	· · · · · · · · · · · · · · · · · · ·	RE/	MITTANCE ADDRESS		
055A GILIE	UNT NUMBER	2401 New	Hartford R	load			The	Marran				
		Atm: Tom	Higdon/Ro	ger Gam			221	S. Main S	jer Street			
001336		OWENSB	ORO, KY 4	123020866			P.O	Box 529				
ADVERTISER CI	LIENT NUMBER						Mad	lisonville	KY 424:	31-0529		
				Payment du	ie by 07/31	199.						
		PLEASE DET										_
DATE 11	NEWSPAPER REFERENCE	12 13 14	DESCRIPTION	OTHER COMMENTS/CH	PORTION	WITH YO	SUZE DI	AITTANC	E			
						16 BILLED		RATE	19	GROSS AMOUNT	20	NET AMOU
06/01/00 04/	520027004	Beginning	Balance									55,55
00101133 04	J29927001	Payment										-55.00
060200 1 04	51327000	Credit										-0.55
00102199 04:	534439001	6×10.5-no	lice			6.00 ×			1		7	0.00
06/00/00 1 0/1						63.00		11.25 <i>1</i> i	n			00.75
00104133	534440001	6×10.5-no	tice			6.00×			1		7	09 75
051500 2 045						63.00		11.25 M	n		•	00.00
00110199 045	034441001	6×10.5-noi	ice			6.00 ×			1		7	08 75
						63.00		11.25 /i	n		,,	00.10
•												
				1 100	V	12	}					
		1//	10.	/ /**		1.0	n/	21				
		WKC	ין0ן ב'	43 pe	FER		At	26				
		//	$\cdot$		1				/-r	<b>`</b>		
			// { ]		^	N.	RE	CEI	VEL	)		
		// t//	1-1/	Xanta	Dh/			ц. 1.				
		VM	lamp	and of	Y   Y		μυ	儿 16	1999			
11070		•	14		/		ATN	AOS EN	EKGY			
	ino		$\mathcal{V}$	$\mathcal{O}$			ACCO	UNTS F	AYAB	LE		
11.21	199 1											
0 h ii	nain											
CR#-	208169											
17 0	INI NE											
1 × 2/	26.20											
IATEMENT OF	ACCOUNT AGING O	F PAST DUE AMO	UNTS							621.*		
CORRENT NET AMOUNT	22]30	DAYS	60	DAYS	OVER	90 DAYS		UNAPPLIED A	MOUNT	23 FOTAL	AMOUN	IT DUE
2126.25	0.0	00	0	00.1	0	00					212	6.25
The												
viessen	Iger TEL 15021	K 529, Madison 824-3300 • FAX - 53	ville, Kentuc	ky 42431								
2	25				NERTICA	*UNAPI	PLIED AM	OUNTS AR	E INCLU	DED IN TOTAL A	MOUI	NT DUE
	BILLING P	ERIOD	6] B	ILLED ACCOUNT NUMBE	R 7	ADVERTISER/O	CLIENT NUM	BER 2		ADVERTISEL/CUEN	TNAME	
	06/01/99 - 06/3	0/99	001336	ò					WEAR			
									WEST	ERN KENTU	скү (	GAS

## News-Democral & LEADER



# News-Democrat & LEADER



11

SHELBYVIL	BDX 399 LE. KY 40066-0399			99 WE	STERN		16836
$\overline{\nabla}$		211 CURRENT NE	73.69	PAYME	NT DUE	E BY 7/30	OVER 90 DAYS
	WOICE and STATEMENT		173.69		.00	•00	•00
Abventioned in	BILLED ACCOUNT NAME AND ADDRESS			9]		REMITTANCE ADDRESS	
WESTER 2401 N OWENSB	N KY GAS CO Ew Hartford RD Ord Ky 42303-1312	2		L	THE ANDMAR SHELBY	SENTINEL NE RK COMM. NE D BOX 1118 VILLE, KY 40	WS NSPAPERS 0066-1118
41 PAGE 7 151 BILLING DATE		ADVERTISER/CLIENT NUA	ABER	Change of A O Note ch	ddress? anges		
6/30/19	PLEASE DETACH AND R	ETURN UPPEI		WITH YC	UR REMI	TTANCE	
TD A REMI DR P	VOID DELAYS IN CRE TTANCE STUB WITH Y RINT THE ACCOUNT N	DITING OUR PAY	OUR AC	COUNT. THE S		SE RETURN TH ED ENVELOPE DUR CHECK	IE
10 DATE 11 REF. 121314	DESCRIPTION-OTHER COMMENTS/CHARGES	TAL BALLED UNITS	TEL TIMES BUN	19 GROS	S AMOUNT	DISC.	20 NET AMOUNT -
6/04 6/09 6/16 Total Sei	DUS BALANCE ADVERTISING ADVERTISING ADVERTISING NTINELHNEWS	6X10.50 6X10.50 6X10.50	5.76 5.76	4 3 3 11	.00 47.93 62.88 62.88 73.69	• 00 • 00 • 00	.00 447.93 362.88 362.88 1173.69
U U U U U U U U U U U U U U U U U U U	1KG 10143 D Wandferby te: 6/99 1 307682 15 \$117269	EFER 1/11/99	Miec	JE36	RE JI ACCC	CEIVED JL 1 4 1999 MOS ENERGY JUNTS PAYABLE	1173-69
STATEMENT OF ACCOUNT	AGING OF PAST DUE AMOUNTS IF TOTAL	PAYMENT IS NOT RECEN	ED BY DUE DATE, A		OF 1.5% PER	R MONTH ( 18% ANNUAL PER	CENTAGE RATE) WILL BE IMPOSED
1172 40	30 DAYS 60	DAVS	CVER 90 (	DAYS	GROSS /	1173 40	NET AMOUNT DUE
FOR BILLING QUESTIONS:	SENTINEL-NEWS. P	00. 0 80x 30	9. SHE		1 F . KY	40066-0399	11/3.09
502	2-633-2526						201
JUNE	1999 019-7608	AUMBER 171 ÃO	VERTISER/CUENT NUA	WEER 21	ESTERN	ADVERTISER/CLIENT N	
1579.49			Y 5 1 0 • SHI			1 MONTH S	

UWEN 	1401 Frederica Street F Owensboro 54 42 Tel: (270 926-0123 FAX: (270) 684	P. O. Box 1480 2302-1480 Ext. 7100 4-3549		06/01/99 23] TOTAL AMOU \$4,411	06/30	199 *UNAPPLIES	WEST KY GAS	14382
	FED ID#: 75-25	88203	21 CURREN	NET AMOUNT DUE	22	O DAYS	00 DAYS	- DVER PO DAYS
	DISPLAY ADVERT INVOICE and STATI	ISING EMENT	\$4,41	1.26	\$	0.00	\$0.00	\$0.00
C	S BILLING DATE	8	BILLED ACC	OUNT NAME AND ADDR	55		REW1773	NCE ATTEL
	UG/30/99	2401 N OWENSE	RN KENTUC IEW HARTF Boro ky	KY GAS ORD RD 42301			OWENSBORO MESS 1401 FREDERICA	ENGER-INQUIREF
WESO	DVERTISEP CLIENT NUMBER						PO BOX 1480	
WESO	06						OWENSBORD KY	42302-1480

DATE	11	EWSPAPER REFERENCE	12 13 14	DESCRIPTION-OTHER COMMENTS/CH	HARGES	15 SAU SIZE	TIMES RUN		
<u>د</u>	<u>+</u>					16 BILLED UNITS	RATE		
	1		BALANCE	FORWARD				.00	
06/04	4064	6201	NOTICE	E PROPOSED CH	ANGES	3 0X 2	1 00"	1470 42	
	RG				MINUE 0	63 00TN	27 74	1410.46	
6/11	4064	6202	NOTICE	E PROPOSED CH	ANCES	3 07 2		1470 42	
	RC					67 DOTN	27 74	1470.46	
6/18	4064	6203	NOTICE	E PROPOSED CH	ANCES	3 08 2		1470 42	
	RG					63 ONTH	27 74	1410.46	
						03.0011	23.34		
							TOTAL	DUE	a all 26
							IVIAL	DUL	
-	1	PREVIO	NUS AMOUN	OWED:	0.0				
	l	NEW CHARG	ES THIS	PERIOD: 4 4	11 26	RF	CFIVE	Ď	
		C/	SH THIS P	PERIOD:	. 00			. –	
Ì	EBIT	ADJUSTMEN	TS THIS	PERIOD:	. 0 0	11	1 08 199	9	
CF	EDIT	ADJUSTMEN	TS THIS	PERIOD:	.00	50			
						ITA I	MOS ENERO	ΞY	
	* *	* * * * *	* * * * *	* * * * * * *	* * * *	* * ACGC	IYNIS PAYA	VBLE *	
			WE APPRE	CIATE YOUR BU	ISINESS.				
	* *	* * * * *	* * * * *	* * * * * *	* * * *	* * * * *	* * * * *	* * *	
A.t									
1 LCIS	U.L.		040 000	n ICIN ulmi	CD 800				
11. 1	-		010,000	0 400 406	8 0070	00			
1121	44		1 4.1	( YV CA	! [\				
il H	201	4416		- 1 Senll	with				
TE LL	202		umo		y IL				
TATELL			<u></u>	(J	<i>[</i>	I	L		м
IAICME		ACCOUNT AGIN	IG OF PAST DUE AMO	UNTS				531	
CURREN	I NET AMOUN	NT DUE 1221	30 DAYS	60 DAYS	OVE	R 90 DAYS	UNAPPLIED AM	OUNT [23]	OTAL AMOUNT DUE
\$4,41	1.26		\$0.00	\$0.00	\$	0.00	. (	00 \$4	4,411.26
		MEGGENIOF		INC			L		· · · · · · · · · · · · · · · · · · ·
				, INC.					
o) 926	-0123 FAX (2	270) 684-3549 FED. ID#7	75-2588203			*UNAPPLIE	D AMOUNTS ARE	INCLUDED IN TOT	AL AMOUNT DU
······		25	ING PERIOD		ADVERTISER INF		TN-MBER 191	ADV67	CLIENT NAME
00005	×454	100/01/99	06/30/99	WES006	W	ES006	1	JEST KY GA	15

6 2F 6/11

NOTICE OF PROPOSED CHARGES IN GAS TARIFFS WHICH WILL RESULT IN INCREASED CHARGES Notice is hereby given that Western Kentucky Gas Company ("Western"), a public utility furnishing natural gas service within the Commonwealth of Kentucky, on/or about the 28th day of May, 1999, pursuant to Kentucky Revised Statute 278.180 and the Rules of the Public Service Commission of Kentucky, respecting tariffs, filed its notice to the Kentucky Public Service Commission ("KPSC"), proposing to change its gas rates effective July 1, 1999. The present rates charged in all territory served by Western are as follows: Present Rates (Effective April 1, 1999)

. . .

	Rate 6 - 1 General Sales Service	· · · · · · · · · · · · · · · · · · ·	Distribution Charge	1
	Monthly Base Charge:		First 300 Mcf or less per month	\$1,2000 per 1 000 cubic feet
•	\$5.10 per meter for residential service		Next 14,700 Mcf per month	\$0.6946 per 1.000 cubic feet
	\$13.60 per meter for non-residential	-	Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Commodity Charge		Gas Charge, all Mcf	\$2.5045 per 1,000 cubic feet
	First 300 Mcf or less per month	\$3.5660 per 1,000 cubic feet	Minimum Charge: The Base Charge	
	Next 14,700 Mcf per month	\$3.0630 per 1,000 cubic feet	Rate G - 1, High Load Factor Firm Sales Se	nvice j
÷ .	Over 15,000 Mcf per month	\$2.9130 per 1,000 cubic feet	Monthly Base Charge:	
	Minimum Charge: The Base Charge	, ,	\$24.00 per meter for non-residential	\$4 2000 per t 000 public (per of
	i Kale 6 - 1, Nign Load Factor Firm Sales Ser Monthly Roce Charge:	vice	Demand Gharge	Daily Contracted Demond
i	\$13.60 per meter for non-residential		Distribution Charne	Daily Constantieu Demand
11	. Demand Charne	\$4 2809 per 1 000 cubic feet of	First 300 Mcf or less ner month	\$1,2000 per 1 000 cubic feet
		Daily Contracted Demand	Next 14,700 Mcf per month	\$0.6946 per 1.000 cubic feet
	Commodity Charge		Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic feet
1:	First 300 Mcf or less per month	\$3.0111 per 1,000 cubic feet	Gas Charge, all Mcf	\$1.9496 per 1,000 cubic feet
	Next 14,700 Mcf per month	\$2.5081 per 1,000 cubic feet	Minimum Charge: The Base Charge	-
ļi	Over 15,000 Mcf per month*	\$2.3581 per 1,000 cubic feet	Rate 6-2, Interruptible Sales Service	
	Minimum Unarge: The Base Unarge	. 1	Montiny Base Unarge:	1
	Nonthly Base Charge:		Distribution Charge	
14	\$150.00 ner meter	с	First 15 000 Mcf or less per month	\$0.5300 per 1.000 cubic feet
	Commodity Charge with sale		Over 15.000 Mct per month*	\$0.3301 per 1.000 cubic feet
44	First 15,000 Mcf or less per month	\$2.4756 per 1,000 cubic feet	Gas Charge, all Mcf	\$1.9820 per 1.000 cubic feet
- <b>i</b> .	Over 15,000 Mcf per month*	\$2.3256 per 1,000 cubic feet	Minimum Charge: The Base Charge	
1	Minimum Charge: The Base Charge		Rate LVS-1, Firm Sales Service	
	Rate LVS-1, Firm Sales Service		Monthly Base Charge:	
ľ	Monuny Base Charge: A Marchard Charge		Distribution Charge	
	Si Sou per meter		First 300 Mcf or less ner month	\$1 2000 per 1 000 cubic feet
	First 300 Mcf or less per month	\$3 5916 per 1 000 cubic feet	Next 14 700 Mcf oar month	\$0.6948 ner 1.000 cubic feet
	Next 14,700 Mcf per month	\$3,0886 per 1,000 cubic feet	Over 15.000 Mcf per month	- \$0,4299 ser 1,000 cubic fest
1	Over 15,000 Mcf per month*	\$2.9386 per 1,000 cubic feet	Gas charge, all Mcf	\$2.5301 per 1,000 cubic feet
4	Minimum Charge: The Base Charge		Minimum Charge: The Base Charge	
	Rate LVS-1, High Load Factor Firm Sales Se	rvice	Rate LVS-1, High Load Factor Firm Sales Se	rvice
	Monthly Base Charge:		Monthly Base Charge:	• <u>•</u> •••
· †·	\$13.60 per meter for non-residential	\$4 2800 per 1 000 cubic feat of	S24.00 per mener for non-residential	\$4 2800 per 1 000 wikin fast of
	Delitatio citatile	Daily Contracted Demand	Ucinalio visa yo	Daily Contracted Demand
	Commodity Charne	Daily Contractor Demand	Distribution Charge	Daily Considered Demand
1	First 300 Mcf or less per month	\$3.0367 per 1.000 cubic feet	First 300 Mcf or less per month	\$1,2000 per 1,000 cubic feet
	Next 14,700 Mcf per month	\$2.5337 per 1,000 cubic feet	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic feet -
1.	Over 15,000 Mcf per month*	\$2.3837 per 1,000 cubic feet	Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Minimum Charge: The Base Charge		Gas Charge, all MCF	\$1.9752 per 1,000 cubic fest
	Rate LVS-2, Interruptible Sales Service		Minimum Unarge: The Base Unarge	
4	C150 00 por motor		Monthly Roce Charge	:
	Commodity Chame		\$250 00 per meter	· · ·
1	First 15,000 Mcf or less per month	\$2,4710 per 1,000 cubic feet	Distribution Charge	· ·. :
	Over 15,000 Mcf per month*	\$2.3210 per 1,000 cubic feet	First 15,000 Mcf or less per month	\$0.5300 per 1,000 cubic feet
<b>.</b>	Minimum Charge: The Base Charge		Over 15,000 Mcf per month*	\$0.3301 per 1,000 cubic feet
	Rato T - 2 General Transportation Service		Gas Charge, all Mcf	\$1.9774 per 1,000 cubic feet
	(includes standby sales service under o	corresponding Sales rates)	Minimum Unarge: The Base Unarge	· * ·
	Monthly Race Charne:	an standby sales service	(includes standby cales service under corres	nonding Sales rates)
8	\$13.60 per meter		Rate T- 2/G-1, Fina Transportation with Fin	a Staring Scies Service
8-	\$45.00 Administration Charge	a de Ar	Monthly Base Charge:	
	Commodity Charge		\$24.00 per meter	
	First 300 Mcf or less per month	\$1.7902 per 1,000 cubic feet	\$50.00 Administration Charge	
	Next 14,700 Mcf per month	\$1.28/2 per 1,000 cubic feet	UISTRIBUTION Charge	1 0000 and 1 000 autoin fact
	Ver 15,000 Mct per month	\$1.1372 per 1,000 cubic teet	Next 14 700 Mot or less per month	\$1.2000 per 1,000 cubic feet
8.	Rate T. 2/C . 1 High Load Factor Firm Servi	ce l	Over 15 000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Monthly Base Charge:		Gas Charge, all Mcf	\$0,7287 per 1,000 cubic feet
	\$13.60 per meter for non-residential		Minimum Charge: The Base Charge	
2	\$45.00 Administration Charge		Rate T-2/G-1, High Load Factor Firm Servic	•
	Demand Charge	\$4.2809 per 1,000 cubic feet of	Monthly Base Charge:	
	Commodia. Charge	Daily Contracted Demand	\$24.00 per meter for non-residential	
	First 200 Met or less per month	\$1 2353 per 1 000 cubic feet	Demand Charge	\$4 2800 nor 1 000 outpic fact of
	Next 14 700 Mcf oer month	\$0 7323 per 1 000 cubic feet	Genand Charge	Daily Contracted Demand
	Over 15 000 Mcf ner month*	\$0.5823 per 1,000 cubic feet	Distribution Charge	wary Generation Demand
	Minimum Charge: The Base Charge		First 300 Mcf or less per month	\$1.2000 per 1,000 cubic feet
	Rate T-2/G-2, Interruptible Transportation	with interruptible Standby Sales	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic feet
	Service		Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Monthly Base Charge:		Gas Charge, all MCT	\$0.1738 per 1,000 cubic feet
	\$150.00 per meter		Minimum Charge: The Base Charge	with Intermetities Otracities Colors
	P45.00 Auministration Unarge		nato 1*2/0*2, interruptione fransportation "	aire internations standon 25102
1	Circl 15 000 Mot a Sace par month	\$0.6009 per 1.000 cubic feet	Monthly Base Charge	

408 KE F PADUC PHON	ENTUCKY AVE P.O. BOX 2300 (AH, KY 42002-2 NE (502) 575-86	07 / 27 / 27 / 27 / 27 / 27 / 27 / 27 /	17:00	) /		<b>6 8 8 1 1</b>	ING DATE N 6/30/99	JERMSO 30 day	877 F:PAYMENT
	VESTERN KY 401 NEW HAR WENSBORO	GAS/OWEN GAS/OWEN RTFORD RE KY 42303-	<b>ecount</b> N D 1312				UED/ACCOL 05100404-0 NAM	UNT NO AGENO 000 E OF AGENCY/CU	
DATE S	REFERENCE,#	CHARGE OR	CREDITS DESCRIPTIO	ON INS	SAU	s BILLED UNIT	SRATE	AMOUNT.	TOTAL
06/05/99 06/09/99 06/16/99	09505382-001 09505734-001 09505735-001	01 - 47411 01 - 47411 01 - 47411	NOTICE OF NOTICE OF NOTICE OF	1	6×10.50 6×10.50 6×10.50	63.00 n 63.00 in 63.00 in	22.39 22.39 22.39	1410.57 1410.57 1410.57	1410.5 2821.1- 4231.7
		oto Utu	0000 1 liam Si	soo nlu	14062	8 <i>00</i> 900	8	EIVED	
Plete: 17/12/99 0/11-300	446 71						JUL ATMO ACCOUR	0 8 1999 S ENERGY ITS PAYABLE	
CURRENT	1-30	DAYS	AGIN 31-60 DAYS	ତ୍ରି 🎨 61-90	D DAYS	OVER 90 DAY	s	TOTAL NET AMO	UNT DUE 4 A
4231.	71	0.00	0.00		0.00	0.0	00		4231.71
PLEASE RETURI	N THIS PORTION	I WITH YOUR .	REMITTANCE If you and re Acct# Signa	ture	charge this a e address abo	amount to your c ove: [ ] VISA	redit card, ple [ ] Maste	ase complete the follow rcard Exp Date	wing informatio
		<b>()</b> ()	ACCOUNT NO + 5100404-000	Ŵ	ESTERN K	ED ACCOUNT (Y G.ASIOWE	NAME N	AMOUN	NT REMITTER
•		itenzebe <del>set</del>	The Pad P.O. B Paducah, K	Minito Jucah Jox 230 Jy. 4200	Sun )0 )2-2300		OUR TERMS Payment in fu <b>REMITTAN</b>	ARE NET 30 DAYS Il is due upon receipt	of the statemer

("Western"), a public utility furnish	ing natural gas service within the	Commodity Chame	•
Commonwealth of Kentucky, on/or	about the 28" day of May 1999,	First 15,000 Mcf or less per month	\$0.6999 per 1,000 cubic f
Public Service Commission of Ke	ntucky, respecting tariffs, filed its	Over 15,000 Mct per month*	\$0.5498 per 1,000 cubic f
notice to the Kentucky Public Servic	e Commission ("KPSC"), proposing	Rate T-3, Interruptible Carriage Ser	vice .
The present rates charged in all t	erritory served by Western are as	Iransportation only service Monthly Base Chame	•
follows:		\$150.00 per meter	
Presen Effective A	t Rates	\$45.00 Administration Charge	
Rate G - 1, General Sales Service	prir 1, 1999)	First 15.000 Mcf or lass per month	\$0.4936 per 1.000 cubic #
Monthly Base Charge:		Over 15,000 Mcf per month*	\$0.3436 per 1,000 cubic fr
\$13.60 per meter for residential ser \$13.60 per meter for non-resident	VICO int	Rate T-4 Firm Carriage Service	
Commodity Charge		Transportation only service	
First 300 Mcf or less per month	\$3.5660 per 1,000 cubic feet	Monthly Base Charge:	
Over 15,000 Mcf per month*	\$2.9130 per 1,000 cubic feet	\$45.00 Administration Charge	•
Minimum Charge: The Base Charge		Commodity Charge	A1 0012 1'000 1' 1
Rate G - 1, High Load Factor Firm	Sales Service	<ul> <li>Next 14,700 Mcf per month</li> </ul>	\$1.0615 per 1,000 cubic fr \$0.5585 per 1,000 cubic fr
Monthly Base Charge:	-	Over 15,000 Mcf per month	\$0.4085 per 1,000 cubic fe
Demand Charge	al \$4.2809 per 1.000 cubic feet of	Special Charges	· ·
	Daily Contracted Demand	Tum on new service with meter set	\$28.00
First 300 Mcf or less per month	\$3.0111 par 1.000 cubic feet	Turn on service, shut in test required	\$18.00
Next 14,700 Mcf per month	\$2.5081 per 1,000 cubic feet	Reconnect delinquent service	No Charge
Over 15,000 Mcf per month* Minimum Charge: The Base Charge	\$2.3581 per 1,000 cubic feet	Reconnect service off temporarity at	60E 00
Charge. The Dase Charge	يستسم بالإلاي المالية المحاصرة بنبه	Termination or field collection charge	⇒∠5.00 \$ 5.00
Rate G-2, Interruptible Sales Servi		Meter test charge	\$20.00
.\$150.00 per meter		Optional facilities charge for Electronic	Flow Measurement
Commodity Charge	\$2 4758 port 1 000 aubin foot	Class 1 E FM	\$105 per mor
Over 15,000 Mcf per month*	\$2.3258 per 1,000 cubic feet		\$210 per mor
Minimum Charge: The Base Charge		Proposed	Rates
Rate LVS-1, Firm Sales Service		Monthly Base Charge:	• • •
Monthly Base Charge:		\$9.00 per meter for residential servi	<b>ce</b>
SIGOU per meter	· · · · · · · · · · · · · · · · · · ·	\$24,00 per meter for non-residentia Distribution Chame	1
First 300 Mcf or less per month	\$3.5916 per 1,000 cubic feet	First 300 Mcf or less per.month	\$1,2000 per 1,000 cubic fe
Next 14,700 Mcf per month	\$3.0886 per 1,000 cubic feet	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic fr
Minimum Charge: The Base Charge	es acco her i hon choic teat	Gas Charge, all Mc	\$0.4239 per 1,000 cubic fi \$2.5045 per 1,000 cubic fi
Data 1000 d. Ellah di sala mata ang		Minimum Charge: The Base Charge	
Monthly Base Charges	Sales Service	Rate Q - 1, High Lond Factor Firm S	alos Sandra
\$13.60 per meter for non-resident	al	Monthly Base Charge: 100	
Demano Charge	54.2809 per 1,000 cubic teet of Daily Contracted Demand	524.00 per meter for non-residentia Demand Charge	\$4 2809 per 1 000 cubic #
Commodity Charge			Daily Contracted Demand
Next 14,700 Mct per month	\$3,0367 per 1,000 cubic feet	First 300 Mcf or less per month	\$1 2000 per 1 000 cubic fe
Over 15,000 Mcf per month	\$2.3837 per 1,000 cubic feet	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic fe
Minimum Charge: The Base Charge		Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic fe
Rate LVS-2, Interruptible Sales Se	vica	Minimum Charge: The Base Charge	a i tavan fisit i 'noon critike ie
Monthly Base Charge: \$150.00 per meter		Rata (L2) Internutible Salas Santa	
Commodity Charge		Monthly Base Charge:	
First 15,000 Mcf or less per month	\$2.4710 per 1,000 cubic feet	\$250.00 per meter	
Minimum Charge: The Base Charge	Actor to per 1,000 cubic teet	First 15,000 Mcf or less per month	\$0.5300 per 1 000 cubic fe
		Over 15,000 Mcl per month*	\$0.3301 per 1,000 cubic fe
Includes standby sales service in	der corresponding Sales rates)	Gas Charge, all Mc	51.9820 per 1,000 cubic fe
		A STATE OF A	
Hate T-2/G-1, Firm Transportation	with Firm Standby Sales Service	Rate LVS-1, Firm Sales Servico	
\$13.60 per meter		\$24.00 per mater	
\$45.00 Administration Charge		Distribution Charge	\$1 000 mar 1 000 million
First 300 Mcf or less per month	\$1.7902 per 1,000 cubic feet	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic fé
Next 14,700 Mcf per month	\$1.2872 per 1,000 cubic feet	Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic fe
"Minimum Charge: The Base Charge	The rest of the cubic test	Minimum Charge: The Base Charge	
Pote T.O.C. I Link I and Frank	n Candaa (Katala) (Ka	Data 11/C 4 Little 6 Address	n engen i gen værdet sinder gen en en gen state i værdet sinder Delege Openstere
Monthly Base Charge:	in Jelvice	Monthly Base Charges Factor Firm	Sales Service
\$13.60 per meter for non-residenti	al <u>a charactar a charactar a charactar</u> a	\$24.00 per meter for non-residential	
Demand Charge	\$4,2809 per 1.000 cubic feet of	Demand Charge	54.2809 per 1,000 cubic fe Daily Contracted Demand
	Daily Contracted Demand	Distribution Charge	wany wane considered
Commodity Charge	\$1 2353 per 1 000 cubin foot	First 300 Mcf or less per month	\$1:2000 per 1,000 cubic fe
Next 14,700 Mcf per month	\$0.7323 per 1,000 cubic feet	Over 15,000 Mcf per month*	\$0.5546 per 1,000 cubic fe \$0.4299 per 1,000 cubic fe
Over 15,000 Mcf per month*	\$0.5823 per 1,000 cubic feet	Gas Charge, all Mcf	\$1.9752 per 1,000 cubic fe
waaanum Charge: Frie Base Charge		Musmum Charge: The Base Charge	
Rate T-2/G-2, Interruptible Tra	insportation with interruptible	Rate LVS-2, Interruptible Sales Serv	ice · · ·
-DALIGUY SAIRS SPLVICE	•	MUNITINY Base Charge:	
Monthly Base Charge:		\$250.00 per meter	

•

1

ł

オートレーフレー

) **!**"

-----

ł

. . . . . . . .

			CYC Res					1415	14
I KADE	K	PACE	SETTER		ACCOUNT	NUMBER	DATE	INVOI	
607	West Washington St. Princeton, Kentuckv	P.O. Box 439 • 42445-0439	Prinsting			2865 0	6/30/99		68
Princeton Phone 502-365-5 Fax 502-365-72	588 99	H Phone Fax S	opkınsville 502-885-7667 502-887-3222		CH	ARG	EOP		
WESTE 2401	RN KENTUCI NEW HARTFO	KY GAS JRD ROAD					E YOU TO CHA		
OWENS	BORD, KEN	TUCKY	42303-	1312			MC CC		
					ACCT#	·		DATE	
RANS/POSTING	TRANS	SACTION DESCR	PTION	INCHES	LINES	BATE	CHARGE	DATE S CF	REDIT
DATE 06/02	JARIFF CH	ANGES-N/C		72	00	. 00	0.	00	
26/02	TARIFF CH	ANGES-N/C		168	00	. 00	0 411	00	
16/07 J	TARIFF CH	ANGES	*** **	168	00	<. 45 2. 45	411.	60	
6/16 1	CHANGE IN	TARIFFS	2 - 4 2 - 2 2 - 2 2	168	00	2.45	411.	60	
						· ·			
			, to a second			. <u>.</u> .			
		1				ي کام در د به سويه معري از در ۲ اليون الاه در آري			
		40 000Q	1860 14	R68 00	1000	RECE			
			112			NLUE			•
	- UL	llan Ax	Sunt	1		JUL O	8 <b>1999</b>		
		1/				ATMOS E	NERGY		• • •
	ار با معدم این این این این این این این این این این	U V			4	CCOUNTS	PAYABLE		
		3.5							
******	*****	*****	***	***	***				
LEASE RE	MIT PAYME	NT BY THE	10TH OF E	ACH MONT	I. IF YO	J ja			
AVE ANY	QUESTIONS	REGARDIN	G YOUR BIL	Li CALL	JENNIFER	· · ·			
1 270-32	> <b>フ−</b> フ3881	nank ybu	UK YUUK B	V21NE821					
1219	517 0156	2. TO	TALS 🌶	744.	00 (	<b>b</b>			
REVIOUS BALAN	CE PAYMENTS		MOUNT SUBJECT	SERVICE CHARG	NEW C	HARGES	TOTA		UN
30.32	2 -	. 00 =	30. 32	+ . 50	)+ 123	4. 80	1,	265. 62	<b>•</b>
	ACC	COUNT STAT	JS				·		
CURRENT	30 DAYS	60 DAYS	90 DAYS & OVER	.,					
1235.30	30. 32	. 00	. 00				a 🕨		
I IARGES AND PAY IFLECTED ON YO ONTH TO BALANC ¢).	MENTS MADE AFTE DUR NEXT STATEME ES DUE OVER 30 D	R THE LAST DAY ( Ent. A 1½ % Final Ays. An <b>Annual R</b>	OF THE STATEMENT NCE CHARGE WILL I ATE OF 18% (A MINI REMITTA)	MONTH WILL BE BE ADDED EACH MUM CHARGE OF NCE COPY				_ 	

-		
		Section D. Pone 7
	LEGAL NOTICES LEGAL NOTICES	LEGAL NOTICES
	Special Charges	
	Turn on new service with meter set	\$28.00
	Turn on service, shut in test required	\$18.00
	Turn on service, meter read only required	\$10.00
معرمين والمام مراجع	Reconnect delinquent service	No Charge
-	Reconnect service off temporarily at customer re	equest \$25.00
	Termination or field collection charge	\$ 5.00
	Meter test charge	\$20.00
	Returned check charge	\$15.00
	Optional facilities charge for Electronic Flow Mea	surement \$105 per month
		\$210 per month
	Class 2 E FM	\$2 TO POLITIONIN
	Proposed Ra	tae
	rioposed ita	163
	Rate G-1 General Sales Service	
	Monthly Base Charge:	
	\$9.00 per meter for residential service	<b>,</b>
	\$24.00 per meter for non-residential	
	Distribution Charge	and the second
	First 300 Mcf or less per month	\$1.2000 per 1,000 cubic feet
	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic feet
	Over 15,000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Gas Charge, all Mcf	\$2.5045 per 1,000 cubic feet
	Minimum Charge: The Base Charge	and management and
	Rate G = 1, High Load Factor Firm Sale	
	Monthly Base Charge:	
	\$24.00 per meter for non-residential	\$4 2809 per 1 000 cubic feet of
		Daily Contracted Demand
	Distribution Chame	
	First 300 Mcf or less per month	\$1,2000 per 1,000 cubic feet
	Next 14 700 Mcf per month	\$0.6946 per 1.000 cubic feet
	Over 15 000 Mcf per month*	\$0.4299 per 1,000 cubic feet
	Gas Charge, all Mcf	\$1.9496 per 1,000 cubic feet
	Minimum Charge: The Base Charge	
	Date C. 2. Intermutible Sales Sandes	
	Rate G-2, Interruptible Sales Service	
	Montiniy base Charge.	
	Distribution Chame	
	First 15 000 Mcf or less per month	\$0.5300 per 1,000 cubic feet
	Over 15 000 Mcf per month*	\$0.3301 per 1,000 cubic feet
	Gas Charge, all Mcf	\$1.9820 per 1,000 cubic feet
	Minimum Charge: The Base Charge	i
	Rate LVS-1, Firm Sales Service	
	Monthly Base Charge:	1
	\$24.00 per meter	
	Distribution Charge	
	First 300 Mcf or less per month	\$1.2000 per 1,000 cubic feet
	Next 14,700 Mcf per month	\$0.6946 per 1,000 cubic feet
	Over 15,000 Mcf per month*	\$0,4299 per 1,000 cubic feet
N.	Gas charge, all Mcf	\$2.5301 per 1,000 cubic feet
5	Minimum Charge: The Base Charge	
• •		- Convice
	Rate LVS-1, High Load Factor Firm Sale	52 DELAICS
	Monthly Base Charge:	
	\$24.00 per meter for non-residential	\$4 2809 per 1 000 cubic feet of
		Dutter Contracted Domand

:

\$4.2809 per 1,000 cubic feet of

\$24.00 per me Demand Charge

•

. · ·

•

.





301 Hale Avenue 
Owensboro, KY 42301

(270) 684-2324

INVOICE NUMBER: V992068-IN INVOICE DATE: 05/28/99

RECEIVED

ORDER NUMBER: ORDER DATE: SALESPERSON: 0010 CUSTOMER NO: 00-00

0010 00-0015880

SOLD TO: WESTERN KENTUCKY GAS ATTN: ACCOUNTING DEPT 2401 NEW HARTFORD ROAD OWENSBORD K

JUN 0 4 1999 ATMOS ENERGY WESTERN KENTUCKY GAS ACCOUNTS PAYABLEATTN: ACCOUNTING DEPT AD KY 42303-1312 OWENSBORO KY 42303-1312

CUSTOMER P.O. TOM HIGDON	SHIP VIA		F.O.B.	TERMS NET 30	DAYS			
ITEM		TY ORDERED	QTY SHIPPED		PRICE	AMOUNT		
/BOOKLETS RATE HIKE	1000 APPLICAT	0.025 ION VOL IX	0.025	0.000	4,097.200	102.43		
/BOOKLETS RATE HIKE	1000 APPLICAT	0.025 ION VOL X	0.025	0.000	8,229.600	205.74		

1/8/99 CR# 305742 # 4042.61

1860 14068 009000

WKG 10143 Delei

ATEMENT MAILED UPON REQUEST.

DO WKG 2636 PAY FROM INVOICE!

NET INVOICE:	3,813.78
FREIGHT: SALES TAX:	.00 228.83
INVOICE TOTAL:	4,042.61

INVUIUE



301 Hale Avenue ● Owensboro, KY 42301 (270) 684-2324

WESTERN KENTUCKY GAS

ATTN: ACCOUNTING DEPT

2401 NEW HARTFORD ROAD

SOLD TO:

OWENSBORD

INVOICE NUMBER: V992068-IN INVOICE DATE: 05/28/99

ORDER NUMBER: ORDER DATE: SALESPERSON: 0010 CUSTOMER NO: 00-0015880

SHIP TO: WESTERN KENTUCKY GAS ATTN: ACCOUNTING DEPT 2401 NEW HARTFORD ROAD KY 42303-1312 OWENSBORO KY 42303-1312

CUSTOMER P.O. TOM HIGDON	SHIP VIA	F.O.B.	TERMS NET 30 D	AYS	ø
ITEM	UNIT QTY ORDERED	QTY SHIPPED	<u></u>	PRICE	AMOUNT
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL I	0.025	0.000 14	,775.600	369.39
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL II	0.025	0.000 13	,715.200	342.88
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL II	0.025 I	0.000 4	, 426. 400	110.66
7BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL IV	0.025	0.000 17	, 920. 400	448.01
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL V	0.025	0.000 19	, 250. 800	481.27
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL VI	0.025	0.000 33	,878.800	846.97
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL VI	0.025 I	0.000 21	, 225. 600	530.64
/BOOKLETS RATE HIKE	1000 0.025 APPLICATION VOL VI	0.025	0.000 15	,031.600	375.79

NET INVOICE:

FREIGHT: SALES TAX:

INVOICE TOTAL:

CONTINUED

LASE PAY FROM INVOICE! TATEMENT MAILED UPON REQUEST. INVOICE

14315



(270) 684-2324

SOLD TO:

INVOICE NUMBER: INVOICE DATE:

V992250-IN 06/15/99

**ORDER NUMBER:** ORDER DATE: SALESPERSON: 0010 **CUSTOMER NO:** 00-0015880

### SHIP TO:

WESTERN KENTUCKY GAS ATTN: ACCOUNTING DEPT 2401 NEW HARTFORD ROAD OWENSBORD KY 42303-1312

WESTERN KENTUCKY	GAS
ATTN: ACCOUNTING	DEPT
2401 NEW HARTFORD	) RUAD
OWENSBORD	KY 42303-1312

CUSTOMER P.O. TOM HIGDON	SHIP VIA F.O.B.		F.O.B.	NET 3	NET 30 DAYS		
ITEM		ORDERED	QTY SHIPPED		PRICE	AMOUNT	
/BOOKLETS RATE INCRE	1000 ASE MANUA	0.005 N VOL IX	0.005	ଡ. ଏଡଡ	4, 098. 000	20.49	
/BOOKLETS	1000	0.005	0.005	ଡ. ଡଡଡ	8,230.000	41.15	

/BOOKLETS 1000 0.005 RATE INCREASE MANUAL VOL X

040 0000 1860 14068 009000 Man Junto J

## RECEIVED

JUL 0 8 1999 ATMOS ENERGY ACCOUNTS PAYABLE

Note:	
7/12/99	Ĵ
CRH 30646	/
$\bigcirc 300$	

LEASE PAY FROM INVOICE! STATEMENT MAILED UPON REQUEST.

NET INVOICE:	762.76
FREIGHT: SALES TAX:	. ᲢᲢ 45. 77
INVOICE TOTAL:	808.53

INVOICE



(270) 684-2324

KY 42303-1312

WESTERN KENTUCKY GAS

ATTN: ACCOUNTING DEPT

2401 NEW HARTFORD ROAD

SOLD TO:

OWENSBORD

INVOICE NUMBER: V992250-IN INVOICE DATE: 06/15/99

ORDER NUMBER: ORDER DATE: SALESPERSON: ଉଦୀଉ CUSTOMER NO: ଉଦ-ଉଦୀ5880

SHIP TO:

WESTERN KENTUCKY GAS ATTN: ACCOUNTING DEPT 2401 NEW HARTFORD ROAD OWENSBORD KY 42303-1312

CUSTOMER P.O. TOM HIGDOŅ	SHIP VIA		F.O.B. TERMS NET 3		S Ø DAYS	
ITEM	UNIT QTY	ORDERED	QTY SHIPPED		PRICE	AMOUNT
BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 Vol 1	0.005	0.000	14,776.000	73.88
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 VOL II	0.005	ଡ. ଡଡଡ	13,716.000	68.58
/BOOKLETS	1000 ASE MANUAL	0.005 VOL III	0.005	Q. QQQ	4,426.000	22.13
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 Vol IV	0.005	0.000	17, 320. 000	89.60
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 Vol V	0.005	0.000	19,250.000	96.25
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 Vol VI	0.005	ଡ. ଡଡଡ	33,878.000	169.39
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 Vol VII	0.005	0.000	21,226.000	106.13
/BOOKLETS RATE INCRE	1000 ASE MANUAL	0.005 VOL VII	0.005 I	ଡ.ଡଡଡ	15,032.000	75.16

NET INVOICE:

FREIGHT: SALES TAX:

INVOICE TOTAL:

CONTINUED

EASE PAY FROM INVOICE! TATEMENT MAILED UPON REQUEST.
Invoice 55/8 Job #:991685

Invoice Date:02/03/99



2204 Griffith Drive Champaign, II 61820 Ph: (217) 244-8226 Fax: (217) 244-0220 Email: requests@mcc.sws.uiuc.edu

Ship to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro, KY 42303 Sold to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro,KY 42303

Salesperson: Karin Gleason Client Phone: 502-685-8072 Client Fax: - -

Quantity	Description	Unit Cost	Total	
1.00	Daily	\$ 50.0	0 <b>\$</b>	50.00
1.00	FTP File	\$ 5.0	<b>\$</b>	5.00
1.00	Standard handling fee	\$ 10.0	D <b>\$</b>	10.00
	*** TOTAL ***		\$	65.00

### \*\*\* Payment due within 30 days of invoice date \*\*\*

#### Invoice notes:

10tex 314199 16.4188134

Daily heating/cooling degree days for Paducah, Louisville, Lexington, Danville and Madisonville, Kentucky during January 1960 - January 1999.

Client Case Number: ref: Kentucky Weather Normalization Tax ID number: 37-6000-511 Deposit in revolving account: 1-3-10943

1860 002

 $\underline{C}$ Make check payable to "University of Illinois/ISWS" and send to:

Illinois State Water Survey 2204 Griffith Drive Champaign, IL 61820-7495

For more information about the Midwestern Climate Center and products, check out our Web page: http://mcc.sws.uiuc.edu/



Return with remittance

Accounting

RECEIVED

MAR 03 1999

ATMOS ENLAGY ACCOUNTS PAYABLE

The Midwestern Climate Center is a cooperative program of the Illinois State Water Survey (Illinois Department of Natural Resources) and the National Climatic Data Center (National Oceanic and Atmospheric Administration, U.S. Department of Commerce)

8 5521



Job #:991888

Invoice Date:02/23/99



2204 Griffith Drive Champaign, II 61820 Ph: (217) 244-8226 Fax: (217) 244-0220 Email: requests@mcc.sws.uiuc.edu

Ship to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro, KY 42303 Sold to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro KY 42303

Salesperson: Karin Gleason Client Phone: 502-685-8072 Client Fax: - -

Quantity	Description	Unit Cost	Total	
1.00	Daily	\$ 20.00	) \$	20.00
1.00	FTP File	<b>\$</b> 5.00	) \$	5.00
1.00	Standard handling fee	\$ 10.00	) \$	10.00
	*** TOTAL ***		S	35.00

\*\*\* Payment due within 30 days of invoice date \*\*\*

Invoice notes:

Daily heating/cooling degree days for Evansville, Indiana and Nashville, Tennessee. Client Case Number:

Tax ID number: 37-6000-511 Deposit in revolving account: 1-3-10943 RECEIVED //llan MAR 0 3 1999

ATHUS ENERGY ACCOUNTS PAYABLE

Make check payable to "University of Illinois/ISWS" and send to:

Illinois State Water Survey 2204 Griffith Drive Champaign, IL 61820-7495

For more information about the Midwestern Climate Center and products, check out our Web page: http://mcc.sws.uiuc.edu/

\_\_\_\_\_ Return with remittance

\_\_\_ Accounting

The Midwestern Climate Center is a cooperative program of the Illinois State Water Survey (Illinois Department of Natural Resources) and the National Climatic Data Center (National Oceanic and Atmospheric Administration, U.S. Department of Commerce)

Invoice

Invoice Date:03/02/99

Job #:991943

MIDWESTERN T

2204 Griffith Drive Champaign, Il 61820 Ph: (217) 244-8226 Fax: (217) 244-0220 Email: mcc@sws.uiuc.edu

Ship to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro, KY 42303 Sold to: William Senter Western Kentucky Gas Co. 2401 New Hartford Rd. Owensboro,KY 42303

RECEIVED

MAR 0 8 1999

ATMUS ENERGY ACCOUNTS PAYABLE

Salesperson: Karin Gleason Client Phone: 502-685-8072 Client Fax: - -

Quantity	Description	Unit Cost	<u> </u>	otal	
1.00	Monthly	\$ 5	0.00	\$	50.00
1.00	FTP File	\$	5.00	\$	5.00
1.00	Standard handling fee	<b>\$</b> 1	0.00	\$	10.00
L	*** TOTAL ***			S	65.00

### \*\*\* Payment due within 30 days of invoice date \*\*\*

Invoice notes:

Monthly heating and cooling degree days during 1961 - 1990 for Evansville, Indiana; Paducah, Kentucky; Louisville, Kentucky; Lexington, Kentucky; and Nashville, Tennessee. Client Case Number:

Tax ID number: 37-6000-511 Deposit in revolving account: 1-3-10943

Make check payable to "University of Illinois/ISWS" and send to:

Illinois State Water Survey 2204 Griffith Drive Champaign, IL 61820-7495

For more information about the Midwestern Climate Center and products, check out our Web page: http://mcc.sws.uiuc.edu/



**Return with remittance** 

\_\_ Accounting

The Midwestern Climate Center is a cooperative program of the Illinois State Water Survey (Illinois Department of Natural Resources) and the National Climatic Data Center (National Oceanic and Atmospheric Administration U.S. Department of Commerce)





P.O. BOX 549, OWENSBORO, KENTUCKY, 42302 TELEPHONE '502' 687-2770

# DATE: MAN INVOICE #

05/28/99 990510F-IN

# RECEIVED

JUN 0 3 1999

BILL TO:

ATMOS ENERGY ACCOUNTS PAYABLE

CUSTOMER NO:

80-WKGJP

WESTERN KENTUCKY GAS 2401 NEW HARTFORD ROAD Owensboro KY 42303

JACKIE PURCELL

MEETING

PONUMBER SIT: S 

NET 30

#### DESCRIPTION RATE AMOUNT

08TABLE CLOTHS/SKIRTING100.009CONCESSIONS - BEVERAGES175.310CONCESSIONS - LABOR30.013CONCESSIONS - SALES TAX10.5
--

1860 14068 009000 William Seuls

Thoject Task Expenditure Type Expenditure WKG 10143 Defev Mealix Externation WKG 2636

RH 305756 afeguard # 324.67



324.67

TO: Western Ky Gas P. O. Box 866 Owensboro KY 42302 Attn: Pearl Simon

Statement Date: May 18, 1999

A Marriott

RECEIVED MAY 2 8 1999 ATMUS ENERGY ACCOUNTS PAYABLE

Date	Description	Folio No.	Debit	Credit	Balance
	•				
5 10 00	Thuches Test	DE 62070	(7 (0 f	1	
5-10-99	Rugnes, Jack	BE53270	07.09 N	-	
5-10-99	Buchanan, Becky	BE23110	67.69		
5-11-99	Webster, Joan	BE53114	67.69 <u>E</u>	31112	
5-14-99	Matthews, Jerry	BE53113	270.76 <sub>(</sub>	h. t	
5-14-99	Owen, Tim	BE53112	270.76		
5-14-99	Vaughn, Sherri	BE53111	284.16		
5-14-99	Kirby, Clay	BE53110	286.97		
5-14-99	Benningfield, Ronnie	BE53109	270.76		
5-14-99	Teater, George	BE53108	270.76		
5-14-99	Mills, Dick	BE53107	270.76		
5-14-99	Clark, Mike	BE53106	270.76		
5-14-99	Wheatley, Pam	BE53105	270.76		
5-14-99	Tucker, Eddie	BE53043	270.76		
5-14-99	Berry, Tom	BE52767	338.45		
5-18-99	Madere, Jenny	BE53620	67.69		· · · · · · · · · · · · · · · · · · ·
			NUC.	INH2 Del	en WKG date
/35	38 1860 14062	8 009000	WKG	10175 000	
21/22	ATTINORD 3990	· Andaina	EXP	ORG	
, <i>5,57 ا</i> و	5 1000 0110	and the			
67.6	9 10117 39908	, H	Total Am	ount Due \$334	46.42
ID			ATM-D	AL-US PROJ	ECT 1202

Account No. E0063BE

Please remit to: Owensboro Fairfield Inn 800 Salem Drive Owensboro, KY 42303

DAL-CIS PROJECT 1202

Thank you for choosing the Owensboro Fairfield Inn. Motes 1/s/99 CKW 305739

DITION MOONLITE ZAR-R-Q

2840 W. Parrish Avenue •Owensboro, KY 42301 Ph. (270) 684-8143 • FAX (270) 684-8105 www.moonlite.com

14888

# **Catering Invoice**

DALE INVOICED 6/22/99 1272

CILLIO:

Potato Salad \* Baked Beans

Banana Pudding 🦯

Small Bottle Hot Sauce **Disposable** Utensils

Pt BBQ Sauce

Western Kentucky Gas

MEMOLE .....

Guests Dinner Delivered

SM PAN Pickle Slices SM PAN Onion Slices

Charge

Delivery

Lbs.

Lbs. Dz

Each Each

Lbs.

Lbs. Lbs.

CUANITY

25

6

DENVER

cky Gas	Off	ice Complex w Hartford Road	· · · ·	Juplicate
			- 10-2 2000000-00-00-00-00-00-00-00-00-00-00-00	Mar
	UNERMPERSON	DEPARTUR		
#7	Kenny	TUES 5	PM	Be There 5:40 PM
la de la desta	CRIPTION		RATE	AMOUNT
Dinner Delivered Delivery & Set Up Charg	e RE(	CEIVĘD	10.00	0.00T 10.00T
Sliced Mutton	ATM	2 2 1 1999 DS ENERGY	8.55	51.30T
Potato Salad Concernent	jies ≠ 2	NTS PAYABLE	2.29 2.29 2.29	47.701 18.32T 13.74T
Buns White Bread	· · · · · /	·	1.89	3.78T
Rye Bread		Leody	1.89	1.89T
Onion Slices		Roch	1.50	1.50T

Sales Tax

Thank You. We appreciate your business.

NO DRINKS NO PNF

CR45 - 30970

\$ 15523

Ate Case

on Cu

6.00%

TOTAL

23.12T

10.48

\$185.12

926-7414

# Ladderbacks Restaurant

INVOICE

118 W. Second St. Owensboro, Ky. 42301 Federal ID # 61-1309831

SOLD TO:

Western Kentucky Gas Co.

INVOICE NUMBER 032399-1 INVOICE DATE 03/23/99 OUR ORDER NUMBER YOUR ORDER NUMBER TERMS 7days SALES REP SHIPPED VIA F.O.B. BILLED yes

SHIPPED TO:

1 (076): 3/26/99 CRH 193186 #162.08

QUANTITY	DES	CRIPTION	UNIT PRICE	AMOUNT
22	Grilled Prok Chops, Fried Catfisl green beans, cole slaw and corr	n, mashed potatoes and gravy, I bread	6.95	\$152.90
	\$ 186.39 1860 Rate & William Ser	7 0021 09 Ase Review Mi	RECE MAR 2 AIMOS E ACCOUNTS	IVED 5 1999 NERGY PAYABLE
	$\mathcal{O}($	//	SUBTOTAL TAX	152.90 9.18
		0	GRATUITY	2431
uestions co all:	oncerning this invoice? John W. Burton Matt P. Burton	MAKE ALL CHECKS P Ladderbacks 118 W. Second St. Owensboro, Ky. 42301	AYABLE TO:	\$162.08 PAY THIS AMOUNT

THANK YOU FOR YOUR BUSINESSI

32646

والمعر والمراجع والمراجع المحاصي الأراجي والمحاصي والمحاصي والمحاصي والمحاصي والمحاصي والمحاصي والمحا	i a constante de la constante d	· [* · · · · ·	er a		<b>4</b>	• *
INVOICE Nº 963	DATE 4.12 99					
CUSTOMER	SALESPERSON <u>- C - J C</u> <u>C - J C</u> <u>C - J C</u> <u>C - J C</u> <u>C - J C</u>	(	GU	:)		D
CITY 1 10 11 118 CUSTOMER'S ORDER # 111N: FILLE	STATEZIP Purchaser's / Exemption #	1-03	P.C Owens	). Box boro,	22473 KY 42304	ł
OUANTITY ORDERED	DESCRIPTION		- UNIT PRI	Ê.	AMOUN	
4 Deced Cut	rs.		4	29	-17	ilo
	There F	ECEIVE	D	•	·	
1860	7 002/ 09 A	APR 1,4 1999 TMOS ENERGY				
NET 10 DAYS. PLEASE PAY FROM THIS INVOIC	migentin ACC	OUNTS PAYAB	C Subto Sales	TAL TAX	11 	14 62
RECEIVED BY	trating and the second se		PLEASE	AY NT		

Acte: 4115/99 CR41197028 #18.19

32646 9-276 A & 200 INVOICE 561 3-22-99 DATE SALESPERSON\_STOCUL 5336 Western Ky Gas CUSTOMER 2401 Alus Mailford ADDRESS WERSBOO P.O. Box 22473 ZIP 12303 CIT Owensboro, KY 42304 CUSTOMER'S ORDER + Alt. JACKIE Purchaser's Exemption # OUANTITY SCRIPTION VITPENCE DOZEN COOKIS assorted WITH 18 ų.... 29 19 TO TAY j 2002 1840 REC EIVFN Keulsin lettra MAR <u>1999</u> ATMOS ENERGY ACCOUNTS PAYAR NET 10 DAYS. PLEASE PAY FROM THIS INVOICE. UBIOTAL RECEIVED BY 

Note: 3/26/99 CRH 193041 # 18.19 25106

Hometown Supermarkets

• •

				2304-2200 •	(502) 684-9952
	жжжж Жужжи	SUR BUSIN	******** ESS IS * TED * ***	/ / ; : 	
	WETZ	EL'S SUPE TREET A CENSBORC THANK YOU	R MARKET T US 231 ) KY	<u></u>	
	MON	4/12/99	9 11:47AM		
			10.00 TF		
		\$2.58	2.99 TF		
	SLS	TAX	.78		1 77
	тс	TAL	\$13.77		
	mı	SCTD	\$13.77		
	CH	IANGE	\$.00		
		5 ITE	15	RECE	
	NAME	RUTHIE <b>‡</b> 30	0004-0004 9195	APR 1	
				ATMOS FI	
	- <u>-</u> .			ACCOUNTS	PAYABLE
Λ		18/00	7002/09		
1 to		1.711-	0 - A A		
100		Ullim	Kinlig L	<u>-</u>	
415/99		+ $ O$	·		
ah 119101	B				
CRUIT			,		
\$ 131	Received	by:			

(	)ha	aro	les)
$\langle \cdot \rangle$		A 1 S	,,

Wetzels

Ļ

ł

Owensboro's Hometown Supermarkets

. .

Office: P.O. Box 22200 • Owensboro, Kentucky 42304-2200 • (502) 684-9952

Date \_\_\_\_\_ 19 \_\_\_\_ Address Eler Aren Harrisonal Res. Bill To: Tax No. CASH 🗖 CHARGE 🔲 Make Duplicate 011 RECEIVED APR 2 6 1999 ATMOS ENERGY ACCOUNTS PAYABLE 1860 7 0021 09 1 YUT9 nni ann Received by:

Note: 4/21/99 CR# 199351 #9.95

## Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 39 b Witness: Adams

### Data Request:

A detailed explanation of how the estimate shown on Schedule F-6 was determined, with all supporting workpapers and calculations.

### Response:

See attached schedule entitled, "WKG Rate Case Expenses – Case. No. 99-070", which summarizes the estimation of rate case expenses.

WKG Rate Case Expenses - Case. No. 99-070	)
Total Expense to be amortized over three years	

DR# 39b FR 10(10)(f) Workpaper

<u>Total \$</u>

Consulting	<u>Ave.\$/hr</u>	<u>Est. hrs</u>				
-Deloitte & Touche	\$250	280	\$ 70,000			
-Lukens	225	349	\$ 78,500		<i>,</i>	
-Guernsey	130	154	\$ 20,000			
-AEG	160	116	\$ 18,500			
-UEC	125	96	<u>\$ 12,000</u>			
					\$	199,000
Legal	<u>Ave.\$/hr</u>	<u>Est. hrs</u>				
-Hutchinson	125	200	\$ 25,000			
-Hughes	185	300	<u>\$ 45,000</u>			
					\$	70,000
Employee Expense	<u>Ave,\$/trip</u>	<u>Est. trips</u>				
- airfare	\$750	36	\$ 27,000			
-car rentals	\$75	20	<u>\$ 1,500</u>			
				\$ 28,500		
Lodging	<u>Ave.\$/stay</u>	<u>Est. stays</u>				
	\$80	75		\$ 6,000		
Meals	<u>Ave.\$/day</u>	<u>Est. days</u>				
	\$50	130		\$ 6,500		
					\$	41,000
Miscellaneous (printing	advertising)					
-printing	, uurornoing,			\$ 5,000		
-advertising (assume	s abbreviated	public notice)		\$ 15,000		
		. ,		 	<u>\$</u>	20,000
Total Rate Case					\$	330,000

# Western Kentucky Gas Company Case No. 99-070 KPSC Data Request Dated July 16, 1999 DR Item 39 c Witness: Adams

#### Data Request:

c. Monthly updates of the actual costs incurred during the course of this proceeding, in the manner prescribed above. Updates will be due on September 3, 1999 (Western's due date for responses to additional data requests), November 1, 1999 (due date for requests to Intervenors), and 30 days after the date of the public hearing or the due date for final briefs, whichever date is later.

### Response:

c. Western will provide updates per the schedule indicated.